

Ocular and systemic vascular dysfunction in neurodegenerative disease

Stephanie Mroczkowska

If you have discovered material in AURA which is unlawful e.g. breaches copyright, (either yours or that of a third party) or any other law, including but not limited to those relating to patent, trademark, confidentiality, data protection, obscenity, defamation, libel, then please read our takedown policy at http://www1.aston.ac.uk/research/aura/aura-take-down-policy/ and contact the service immediately eprints@aston.ac.uk/research/aura/aura-take-down-policy/

OCULAR AND SYSTEMIC VASCULAR DYSFUNCTION IN NEURODEGENERATIVE DISEASE

STEPHANIE ANN MROCZKOWSKA Doctor of Philosophy

ASTON UNIVERSITY

December 2011

Stephanie Ann Mroczkowska, 2011 asserts her moral right to be identified as the author of this thesis

This copy of the thesis has been supplied on condition that anyone who consults it is understood to recognise that its copyright rests with its author and that no quotation from the thesis and no information derived from it may be published without proper acknowledgement.

Thesis Summary

The important role played by vascular factors in the pathogenesis of neurodegenerative disease has been increasingly realised over recent years. The nature and impact of ocular and systemic vascular dysfunction in the pathogenesis of comparable neurodegenerative diseases such as glaucoma and Alzheimer's disease (AD) has however never been fully explored. The aim of this thesis was therefore to investigate the presence of macro- and micro-vascular alterations in both glaucoma and AD and to explore the relationships between these two chronic, slowly progressive neurodegenerative diseases.

The principle sections and findings of this work were:

1. Is the eye a window to the brain? Retinal vascular dysfunction in Alzheimer's disease

• Mild newly diagnosed AD patients demonstrated ocular vascular dysfunction, in the form of an altered retinal vascular response to flicker light, which correlated with their degree of cognitive impairment.

2. Ocular and systemic vascular abnormalities in newly diagnosed normal tension glaucoma (NTG) patients

• NTG patients demonstrated an altered retinal arterial constriction response to flicker light along with an increased systemic arterial stiffness and carotid artery intima-media thickness (IMT). These findings were not replicated by healthy age matched controls.

3. Ocular vascular dysregulation in AD compares to both POAG and NTG

 AD patients demonstrated altered retinal arterial reactivity to flicker light which was comparable to that of POAG patients and altered baseline venous reactivity which was comparable to that of NTG patients. Neither alteration was replicated by healthy controls.

4. POAG and NTG: two separate diseases or one continuous entity? The vascular perspective

- POAG and NTG patients demonstrated comparable alterations in nocturnal systolic blood pressure (SBP) variability, ocular perfusion pressure, retinal vascular reactivity, systemic arterial stiffness and carotid IMT.
- Nocturnal SBP variability was found to correlate with both retinal artery baseline diameter fluctuation and carotid IMT across the glaucoma groups.

Keywords: Neurodegeneration, Glaucoma, Alzheimer's disease, Retinal Vessel Reactivity, Vascular dysregulation

I would like to thank my supervisor, Dr Doina Gherghel, for her expertise and guidance throughout this thesis. I would additionally like to gratefully acknowledge Aston University as the funding body for this research via the clinical demonstrator programme.

I would like to thank all of the staff at Heartlands hospital, Birmingham Midland Eye Centre and The Barberry who supported and assisted me in my patient recruitment, in particular Mr Velota Sung, Mr Anil Negi and Ms Jan Wright. I would additionally like to extend my sincere thanks to all the patients and subjects who kindly gave up their time to make this research possible and to my colleagues, Lu Qin and Sunni Patel, for their assistance with data collection and blood analyses at times when one pair of hands was just not enough

My thanks also go to my friends and colleagues in the Aston University Optometry department for their continued support and encouragement, in particular to Olivia Hunt for her clinic and timetabling support and to my dear office buddy Olivia Howells for keeping me going, plying me with sweets and always being there to either lend an ear or offer a distraction.

Finally my special thanks go to the friends who encouraged me into starting this whole process and to my family for their unfailing support, guidance and encouragement throughout.

Contents

1.	Introduction18		
1.1	Prima	ry Open Angle Glaucoma	20
	1.1.1	Risk factors for glaucomatous damage	22
	1.1.2	POAG vs. NTG	22
1.2	Anato Circula	my and Physiology of the Retina, Optic Nerve and Ocular ation	25
	1.2.1	The Retina	25
	1.2.2	The Optic Nerve	27
	1.2.3	The Ocular Circulation	30
1.3	Glauc	omatous Optic Neuropathy	39
	1.3.1	The Role of Intraocular Pressure	40
	1.3.2	The Vascular Theory	42
1.4	Ocula	r Blood Flow	44
	1.4.1	Ocular perfusion pressure	44
	1.4.2	Vascular resistance	45
	1.4.3	Autoregulation	46
1.5	The E	ndothelium	51
	1.5.1	Background	52
	1.5.2	Nitric oxide (NO)	53
	1.5.3	Endothelin-1 (ET-1)	55
	1.5.4	Endothelial dysfunction	57
	1.5.5	Assessment of endothelial function	58
1.6	Ocula	r Blood Flow and Glaucoma	62
	1.6.1	Autoregulatory Dysfunction	63
	1.6.2	Vascular dysregulation syndrome	65
	1.6.3	Endothelial dysfunction	71
	1.6.4	The Autonomic Nervous System (ANS)	75
1.7	The R	ole of Systemic Blood Pressure	82
	1.7.1	Systemic Blood Pressure - Background	82
	1.7.2	Hypertension and glaucoma	84
	1.7.3	Hypertension and IOP	85
	1.7.4	Hypotension and glaucoma	86
1.8	The R	ole of Cardiovascular Risk Factors	88
	1.8.1	Physiology of the Cardiovascular System	88
	1.8.2	Anatomy of the Cardiovascular System	88
	1.8.3	Cardiovascular risk and Glaucoma	93
1.9	Oxida	tive stress and the concept of reperfusion injury	96
	1.9.1	Background	96
	1.9.2	Oxidative Stress and Endothelial Dysfunction	

	1.9.3	Oxidative stress and glaucoma	100
1.10	Overal	Il summary of the current pathogenetic concept of glaucoma	104
1.11	Cereb	ral Neurodegeneration: Alzheimer's disease	106
	1.11.1	Dementia overview	106
	1.11.2	Clinical presentation, diagnosis and treatment of AD	106
1.12	Anator	my and Physiology of the Cerebral Vasculature	112
	1.12.1	Macrovasculature	112
	1.12.2	Microvasculature	113
	1.12.3	Regulation of cerebral blood flow	115
	1.12.4	The Vascular Theory of Alzheimer's disease	117
	1.12.5	Cerebral and Ocular Microcirculations: Associations	122
	1.12.6	The Concept of using the 'Eye as a Window to the Brain'	122
	1.12.7	Alzheimer's disease and glaucoma	124
2.	Researc	ch Rationale	127
2.1	Aims		128
3.	Subject	s and Methods	129
3.1	Patien	t recruitment	129
	3.1.1	Recruitment of mild newly diagnosed AD patients	129
	3.1.2	Recruitment of newly diagnosed POAG patients	130
	3.1.3	Recruitment of newly diagnosed NTG patients	130
	3.1.4	Recruitment of healthy controls	131
	3.1.5	Exclusion criteria for all groups	132
3.2	Ethica	l approval	133
3.3	Metho	ds	133
	3.3.1	Experimental Protocol	135
	3.3.2	Preliminary Assessments	136
	3.3.3	Assessment of Ocular Vascular Function: Dynamic retinal vertical v	ssel 136
	3.3.4	Assessment of systemic parameters	152
	3.3.5	Assessment of Cardiovascular risk	159
	3.3.6	Evaluation of Oxidative Stress	166
4.	Is the ey	ye a window to the brain? Retinal vascular dysfunction in	168
41	Abstra	ict	168
4.2	Introdu	uction	169
4.3	Aims		
4.4	Hvpotł	nesis	
4.5	Subiec	cts and Methods	
4.6	Statist	ical analysis	
4.7	Power	calculations	171

4.8	Result	ts	172
	4.8.1	Baseline values	172
	4.8.2	Systemic endothelial dysfunction	173
	4.8.3	Dynamic retinal vessel analysis	173
	4.8.4	Correlations	177
4.9	Discus	ssion	178
	4.9.1	Main findings	178
	4.9.2	Systemic endothelial function	178
	4.9.3	Retinal vessel reactivity	179
	4.9.4	Vascular function vs. Cognitive impairment	182
4.10	Concl	usion	182
4.11	Limita	tions	183
5.	Ocular	and Systemic Vascular Abnormalities in Newly-Diagnos	ed
- 4	Normal	Iension Glaucoma Patients	
5.1	Abstra	act	
5.2	Introd	uction	
5.3	Aims		
5.4	Hypot		
5.5	Subje	cts and Methods	
5.6	Statis		
5.7	Power	r calculations	
5.8	Result	ts	
	5.8.1	Baseline values	
	5.8.2	Systemic vascular parameters	
	5.8.3	Dynamic retinal vessel analysis	
5.9	Discus	ssion	
	5.9.1	Main findings	
	5.9.2	Systemic macro-vascular alterations in NTG	
	5.9.3	Retinal microvascular reactivity in NTG	
	5.9.4	Ocular microvasculature vs. Systemic macrovasculature	
5.10	Concl	usion	
5.11	Limita	tions	200
6.	Ocular and NT	Vascular Dysregulation in AD compares to both POAG G	
6.1	Abstra	act	
6.2	Introd	uction	
6.3	Aims		
6.4	Hvpot	hesis	
6.5	Subie	cts and Methods	
6.6	Statist	tical Analysis	
		, = - · · · · · · · · · · · · · · · · · ·	

6.7	Power	calculations	204
6.8	Results		205
	6.8.1	Baseline values	205
	6.8.2	Systemic endothelial function	206
	6.8.3	Dynamic retinal vessel analysis	207
	6.8.4	Correlations	210
6.9	Discu	ssion	211
	6.9.1	Main findings	211
	6.9.2	Systemic endothelial function	211
	6.9.3	Retinal vessel reactivity	212
	6.9.4	Retinal vascular dysfunction in AD compares to both POAG and NTG	214
6.10	Concl	usion	215
6.11	Limita	tions	216
7.	Primary separat	open angle glaucoma and Normal tension glaucoma: two e diseases or one continuous entity? The vascular	217
71	Abstra	oct	217
7.2	Introd	uction	218
7.3	Aims		219
74	Hypot	hesis	220
7.5	Subie	cts and Methods	220
7.6	Statist	tical Analysis	220
7.7	Power	r calculations	221
7.8	Result	ts	221
	7.8.1	Baseline values	222
	7.8.2	Ambulatory blood pressure	223
	7.8.3	Heart rate variability	224
	7.8.4	Pulse wave analysis	225
	7.8.5	Intima-media thickness measurement	225
	7.8.6	Systemic endothelial function	226
	7.8.7	Dynamic retinal vessel analysis	226
	7.8.8	Oxidative stress markers	229
	7.8.9	Correlations	229
7.9	Discus	ssion	231
	7.9.1	Main findings	231
	7.9.2	Systemic vascular parameters	231
	7.9.3	Retinal vascular function	236
	7.9.4	Oxidative stress	238
	7.9.5	POAG vs. NTG	239
7.10	Concl	usion	241

7.11	Limitat	tions	241
8.	Summa	ry and Conclusions	243
8.1	Summ	nary	243
	8.1.1	Is the eye a window to the brain? Ocular and systemic vascular dysfunction in Alzheimer's disease	243
	8.1.2	Ocular and systemic vascular abnormalities in newly diagnosed normal tension glaucoma patients	244
	8.1.3	Ocular Vascular Dysregulation in AD compares to both POAG and NTG	d 246
	8.1.4	Primary open angle glaucoma and Normal tension glaucoma: two separate diseases or one continuous entity? The vascular perspective	247
8.2	Conclu	usions	248
	8.2.1	To investigate the presence and impact of ocular and systemic vascular alterations in AD and to explore the concept of using the 'eye as a window to the brain'	248
	8.2.2	To investigate the presence and impact of ocular and systemic vascular alterations in NTG	249
	8.2.3	To investigate the possibility of a shared vascular aetiology, involving both the ocular and systemic circulations, in AD and both POAG and NTG.	h 249
	8.2.4	To compare and contrast vascular alterations at both the ocular as systemic level in POAG and NTG and to explore their validity as distinct clinical entities	nd 250
8.3	Overa	II Limitations	250
8.4	Clinica	al implications	252
	8.4.1	Implications with regard to AD	252
	8.4.2	Implications with regard to glaucoma	252
	8.4.3	Implications with regard to assessment of ocular vascular function	253
8.5	Future	Directions	253
	8.5.1	Expansion of preliminary data	253
	8.5.2	Assessment of retinal vascular dysfunction in mild cognitive impairment (MCI)	254
	8.5.3	Impact of ocular and systemic vascular dysfunction on the progression of POAG and NTG	254
	8.5.4	Development of dynamic retinal vessel analysis as a tool for the assessment of neurodegenerative disease	255
	8.5.5	Expansion of flow mediated dilation analysis	255
9.	Referen	ces	257
10.	Append	lices	296

List of Figures

Figure 1.1: Classification of glaucoma	21
Figure 1.2: Diagrammatic representation of the human retina	27
Figure 1.3: Diagrammatic representation of the optic nerve head and its	
vasculature	29
Figure 1.4: Diagrammatic representation of the healthy optic nerve head	30
Figure 1.5: Diagrammatic representation of the retrobulbar circulation	32
Figure 1.6: Diagrammatic representation of the optic nerve head in glaucoma	39
Figure 1.7: Summary of the factors which trigger autoregulation of blood flow in the	;
ocular circulation	47
Figure 1.8: Diagrammatic representation of the structure of the vascular wall	
and the location of the vascular endothelium	. 52
Figure 1.9: Mechanism of endothelial nitric oxide (NO) synthesis	54
Figure 1.10: Mechanism of Endothelin-1 (ET-1) synthesis	56
Figure 1.11: Summary of the potential contributing factors to disturbed	
autoregulation and hence unstable ocular blood flow	
in glaucoma	65
Figure 1.12: Diagrammatic representation of the heart	. 90
Figure 1.13: Diagrammatic representation of the arteries of the neck	.91
Figure 1.14: Diagrammatic representation of the artery wall	.92
Figure 1.15: Chemical structure of the tripeptide glutathione	. 99
Figure 1.16: Summary of the current pathogenetic concept of glaucomatous optic	
neuropathy development 1	105
Figure 1.17: Diagrammatic representation of the cerebral macrovasculature 1	113
Figure 1.18: Diagrammatic representation of the structure of the cerebral	
capillaries1	114
Figure 3.1: Diagrammatic representation of the dynamic retinal vessel analyser	
set up1	137
Figure 3.2: Dynamic retinal vessel analyser (DVA)1	139
Figure 3.3: Example of retinal vessel selection prior to dynamic retinal vessel	
analysis1	142
Figure 3.4: Breakdown of the 350 second DVA testing period 1	143
Figure 3.5: Diagrammatic representation of a typical retinal vessel response to	
flicker light on dynamic retinal vessel analysis1	144
Figure 3.6: Diagrammatic representation of the parameters calculated from the	
dynamic retinal vessel analysis (DVA) response profile 1	148

Figure 3.7: Imaging screen in FMD with brachial artery segment highlighted153
Figure 3.8: Breakdown of the 7 minute FMD examination protocol154
Figure 3.9: Diagrammatic representation of the electrode positioning for 24 hour
ECG recording with the Cardiotens-01 device 158
Figure 3.10: SphygmoCor device set up160
Figure 3.11a: forward aortic waveform161
Figure 3.11b: reflected waveform161
Figure 3.11c: complete aortic waveform161
Figure 3.12a: Normal aortic pulse waveform161
Figure 3.12b: Aortic waveform in presence of arterial stiffness
Figure 3.13: Diagrammatic representation of the aortic pressure waveform and
augmentation index162
Figure 3.14: Diagrammatic representation of the carotid artery wall and IMT164
Figure 3.15: Ultrasound image indicating carotid artery intima-media thickness
measurement site. C-IMT: carotid-intima media thickness 164
Figure 4.1: Correlation between arterial reaction time (RT) on the 1st flicker
cycle and MMSE score177
Figure 5.1: Averaged arterial response profile for NTG patients and healthy
controls generated through Matlab191
Figure 5.2: Averaged venous response profile for NTG patients and healthy
controls generated through Matlab192
Figure 7.1: Nocturnal systolic blood pressure variability vs. intima-media
thickness in POAG patients230
Figure 7.2: Nocturnal systolic blood pressure variability vs. retinal arterial
baseline diameter fluctuation in NTG patients

List of Tables

Table 1.1: Comparison between POAG and NTG	24
Table 1.2: Overview of techniques available for assessment of ocular blood flow.	38
Table 1.3: Summary of the vasoactive agents released by the vascular	
endothelium	53
Table 1.4: Methods of assessing endothelial function	61
Table 1.5: Classification of hypertension	83
Table 1.6: Overview of the techniques available for the detection of	
atherosclerosis	93
Table 1.7 Overview of commonly used indices of arterial stiffness	95
Table 1.8: Summary of common reactive oxygen and reactive nitrogen species	96
Table 1.9: Summary of enzymatic and non-enzymatic antioxidants	97
Table 1.10: Overview of the subcategories of dementia	. 106
Table 1.11: NINDS-ADRDA Diagnostic criteria for AD	110
Table 1.12: Clinical stages of Alzheimer's disease	. 111
Table 1.13: Features of Alzheimer's disease vs. Vascular dementia	.121
Table 3.1: An overview of the investigative techniques conducted in this thesis	. 134
Table 3.2: Technical specification of the DVA	.139
Table 3.3 Summary of the parameters which can be calculated and in used for	
analysis of the dynamic retinal vessel response profile	148
Table 3.4: Alx Reference values for healthy individuals provided by	
SphygmoCor	. 162
Table 4.1: Summary of the baseline characteristics of the study groups	173
Table 4.2: Systemic endothelial function	.173
Table 4.3: Arterial vascular function parameters determined using dynamic retinal	I
vessel analysis	.175
Table 4.4: Arterial vascular function parameters by flicker cycle	175
Table 4.5: Dynamic characteristics of the retinal vascular response profiles	
determined using our novel computational model	.175
Table 4.6: Venous vascular function parameters determined using dynamic retina	al
vessel analysis	.176
Table 4.7: Venous vascular function parameters by flicker cycle	176
Table 4.8: Dynamic characteristics of the retinal vascular response profiles	
determined using our novel computational model	176
Table 5.1: Summary of the baseline characteristics of the study groups	. 189
Table 5.2: Systemic macro-vascular parameters	. 190

Table 5.3: Systemic endothelial function	.190
Table 5.4: Arterial vascular function parameters determined using dynamic retinal	l
vessel analysis	.193
Table 5.5: Dynamic characteristics of the retinal vascular response profiles	
determined using our novel computational model	193
Table 5.6: Venous vascular function parameters determined using dynamic retina	ıl
vessel analysis	.194
Table 5.7: Dynamic characteristics of the retinal vascular response profiles	
determined using our novel computational model	194
Table 6.1: Summary of the baseline characteristics of the study groups	206
Table 6.2: Systemic endothelial function	.206
Table 6.3: Arterial vascular function parameters determined using dynamic retinal	l
vessel analysis	208
Table 6.4: Arterial vascular function parameters by flicker cycle	208
Table 6.5: Dynamic characteristics of the retinal vascular response profiles	
determined using our novel computational model	209
Table 6.6: Venous vascular function parameters determined using dynamic retina	ıl
vessel analysis	.210
Table 6.7: Dynamic characteristics of the retinal vascular response profiles	
determined using our novel computational model	210
Table 7.1: Summary of the baseline characteristics of the study groups	223
Table 7.2: Ambulatory blood pressure parameters	224
Table 7.3: Coefficient of variation for systolic blood pressure across 24 hours,	
diurnally and nocturnally	224
Table 7.4: Heart rate variability parameters	225
Table 7.5: Systemic arterial stiffness	225
Table 7.6: Carotid artery intima-media thickness	225
Table 7.7: Systemic endothelial function	226
Table 7.8: Arterial vascular function parameters determined using dynamic retinal	I
vessel analysis	227
Table 7.9: Dynamic characteristics of the retinal vascular response profiles	
determined using our novel computational model	227
Table 7.10: Venous vascular function parameters determined using dynamic retin	al
vessel analysis	228
Table 7.11: Dynamic characteristics of the retinal vascular response profiles	
determined using our novel computational model	229
Table 7.12: Oxidative stress analysis	229

List of Equations

Equation 1.1: Calculation of ocular perfusion pressure	45
Equation 1.2: Calculation of mean arterial blood pressure	45
Equation 1.3: Conversion of GSH to GSSG and vice-versa	99
Equation 3.1: Baseline corrected flicker response	
Equation 3.2: Dilation amplitude	147
Equation 3.3: Dilation and constriction slope	151
Equation 3.4: FMD dilation response	155
Equation 3.5: NMD dilation response	155
Equation 3.5: Nocturnal blood pressure dip	
Equation 3.6: Coefficient of variation	158
Equation 3.7: BMI	166
Equation 3.8: Calculation of total-GSH	167

Abbreviations

ABPM	Ambulatory blood pressure monitoring
ACE-R	Addenbrooke's Cognitive Examination Revised
AD	Alzheimer's disease
Alx	Augmentation index
ANS	Autonomic nervous system
ARMD	Age related macular degeneration
Aß	Amyloid-beta
BBB	Blood-brain barrier
BDF	Baseline diameter fluctuation
BFR	Baseline corrected flicker response
BMI	Body mass index
BP	Blood pressure
CBF	Cerebral blood flow
CCD	Charged coupling device
CNS	Central nervous system
CO ₂	Carbon dioxide
CRA	Central retinal artery
CRV	Central retinal vein
CSF	Cerebral spinal fluid
DA	Dilation amplitude
DBP	Diastolic blood pressure
DSM-IV-TR	Diagnostic and Statistical Manual of Mental Disorders, VIth
	edition
DVA	Dynamic retinal vessel analysis
ECG	Electrocardiogram
EDCF	Endothelial derived constricting factor
EDRF	Endothelial derived relaxing factor
EDVF	Endothelial derived vasoactive factors
ELM	External limiting membrane
eNOS	Endothelial nitric oxide synthase
ET-1	Endothelin-1
ET-2	Endothelin-2
ET-3	Endothelin-3
FMD	Flow mediated dilation
GON	Glaucomatous optic neuropathy

GSH	Reduced glutathione
GSSG	Oxidised glutathione
GTN	Nitroglycerin
H⁺	Hydrogen
HDL-C	High density lipoprotein cholesterol
HF	High frequency
HR	Heart rate
HRV	Heart rate variability
HTG	High tension glaucoma
ICA	Internal carotid artery
ILM	Inner limiting membrane
IMT	Intima-media thickness
iNOS	Inducible nitric oxide synthase
IOP	Intraocular pressure
IQR	Inter-quartile range
K ⁺	Potassium
LDL-C	Low density lipoprotein cholesterol
LF	Low frequency
LF/HF	Low frequency/high frequency ratio
LGN	Lateral geniculate nucleus
LPCAs	Long posterior ciliary arteries
MABP	Mean arterial blood pressure
MC	Maximum constriction
MC%	Percentage constriction
MCI	Mild cognitive impairment
MD	Maximum diameter
MD%	Percentage dilation
MMP-9	Metaalloproteinase-9
MMSE	Mini mental state examination
MRI	Magnetic resonance imaging
NFL	Nerve fibre layer
NINCDS-ADRDA	National Institute of Neurological Disorders and Stroke-
	Alzheimer Disease and Related Disorders working group
NMD	Nitroglycerin mediated dilation
nNOS	Neuronal nitric oxide synthase
NO	Nitric oxide
NO ₂	Nitrite

NOS	Nitric oxide synthase
NRR	Neuro-retinal rim
NTG	Normal tension glaucoma
NVU	Neurovascular unit
O ₂	Oxygen
OA	Ophthalmic artery
OAG	Open angle glaucoma
OBF	Ocular blood flow
OHT	Ocular hypertension
ONH	Optic Nerve Head
OPP	Ocular perfusion pressure
PCAs	Posterior ciliary arteries
pCO ₂	Partial carbon dioxide pressure
PET	Positron emission tomography
PNS	Parasympathetic nervous system
pO ₂	Partial oxygen pressure
POAG	Primary open angle glaucoma
PVD	Primary vascular dysregulation syndrome
PWA	Pulse wave analysis
RBC	Red blood cell
RGC	Retinal ganglion cell
RNFL	Retinal nerve fibre layer
RNS	Reactive nitrogen species
ROS	Reactive oxygen species
RPCs	Radial peripapillary capillaries
RPE	Retinal Pigment Epithelium
RRI	Recurrent reperfusion injury
RT	Reaction time
SA node	Sinoatrial node
SBP	Systolic blood pressure
SDRA	Sequential and diameter response analysis
Slope _{AC}	Arterial constriction slope
Slope _{AD}	Arterial dilation slope
Slope _{VC}	Venous constriction slope
Slope _{VD}	Venous dilation slope
SNS	Sympathetic nervous system
SOV	Superior ophthalmic vein

SPCAs	Short posterior ciliary arteries
SVD	Secondary vascular dysregulation syndrome
TG	Triglycerides
t-GSH	Total glutathione
ТМ	Trabecular meshwork
tMC	time to maximum constriction
UM	Units of measurement
VaD	Vascular dementia
VF	Visual field
vSMCs	Vascular smooth muscle cells
vWf	von Willebrand factor

1. Introduction

The term neurodegeneration refers to a progressive loss of structure and function in the neurons of the central and/or peripheral nervous system and encompasses a large range of different disease conditions. One of the most significant risk factors for neurodegeneration is advancing age and as life expectancies increase so does its prevalence, making it a global issue of increasing concern¹. Discovering the triggers which promote neurodegeneration in the elderly and others and discovering the pathological mechanisms by which neurodegenerative disease develops is therefore currently a research area of intense interest. One of the most recognised locations for neurodegeneration within the central nervous system (CNS) is the brain, where it is commonly affiliated with the development of cognitive impairments such as Alzheimer's disease (AD). AD is a prevalent neurodegenerative disease of poorly understood aetiology which affects over 35 million people worldwide². One of the major factors limiting our aetiological understanding of the neurodegenerative processes in AD and other cerebral diseases is the notorious difficulty associated with assessing and visualising the brain and cerebral neurons directly. As a result researchers are naturally starting to look to other more accessible regions of the CNS in order to gain a potential insight into the alterations which may be occurring at the cerebral level. Of particular interest in this regard is the eye and the neurons of the retina and optic nerve head (ONH), which also form part of the CNS and have been found to exhibit a number of features comparable to that of the cerebral unit ³. Indeed the ONH itself is recognised as an important site of neurodegeneration, whereby the progressive degeneration of ONH neurons is strongly affiliated with the development of the ocular disease glaucoma. Interestingly tentative associations have previously been made between the occurrence of ocular neurodegeneration in the form of glaucoma and cerebral neurodegeneration in the form of AD. This in combination with the multiple similarities previously identified in the structure and functioning of the

ocular and cerebral units has raised the question of whether these two neurodegenerative diseases may share a common underlying aetiology. This thesis aims to explore this possibility and to additionally address the question of whether functional assessment at the ocular level could prove effective as an indirect measure of cerebral function and hence whether the eye could be effectively used as a 'window to the brain' in neurodegenerative disease. Of particular interest to this thesis is the involvement of vascular factors in the aetiology of both AD and glaucoma. Indeed one of the most prominent features shared by the ocular and cerebral units is the nature of their vascular supply, with the ocular and cerebral microcirculations in particular demonstrating a large number of anatomical and physiological similarities⁴. Furthermore, although the involvement of vascular factors in the aetiology of both AD and glaucoma individually has been increasingly realised, the nature of this involvement is still somewhat uncertain and many questions remain. As such this thesis additionally aims to explore the presence and aetiological relevance of vascular disorders at both the ocular and systemic level in glaucoma and AD in order to try and enhance our understanding of the pathological mechanisms involved in these neurodegenerative diseases individually. To provide a basis for the research conducted in this thesis the following sections will outline the background and current aetiological thinking, firstly for the development of glaucoma and secondly for the development of AD, with particular emphasis on the role of vascular factors. The relevant ocular, cerebral and cardiovascular anatomy and physiology will also be discussed, followed by an outline of the current literature associating these two neurodegenerative diseases.

1.1 Primary Open Angle Glaucoma

Primary open angle glaucoma (POAG) is a chronic, slowly progressive optic neuropathy characterised by progressive visual field loss and a distinctive excavation of the ONH ⁵. It is one of the leading causes of blindness in the world, affecting over 66 million people worldwide and accounting for approximately 13% of those on the blind register in England and Wales ⁶. Its origin is strongly linked to the presence of elevated intraocular pressure (IOP) ^{7, 8}, but despite this IOP is no longer included in the definition of glaucoma as its involvement with the disease process has been shown to be inconsistent.

The first description of the term glaucoma (or 'glaucosis' in Greek) is accredited to Hippocrates in approximately 400 B.C⁹. It was thought for many centuries that the 'hardness of the eyeball' first identified as glaucoma resulted from some form of vitreous abnormality and it wasn't until the mid-nineteenth century, following the development of the ophthalmoscope, that the now characteristic features of abnormal cupping of the ONH, increased 'eye tension' and visual loss were first linked ⁹. A differentiation between different forms of glaucoma, namely acute, chronic and secondary, all of which were associated with increased IOP, was first made by von Graefe in 1854⁹. This was followed by the proposal that there may also be IOPindependent causes of glaucomatous optic neuropathy (GON) by Jaeger in 1858¹⁰ and that both mechanical and vascular factors may be involved in the development of glaucoma by Smith in 1886¹¹. The suggestion that atrophy and excavation of the ONH occurring in the absence of elevated IOP should be considered a distinct form of glaucoma was subsequently made by Schnabel in 1892, leading to the categorisation of what is now commonly referred to as normal tension glaucoma (NTG). Finally, following the development of gonioscopic devices in the early 20th century, which allowed the anterior chamber drainage angle to be observed, a further

differentiation between open angle and closed angle glaucoma was made, leading ultimately to the adoption of a simple and broad classification for glaucoma in 1954 ⁹ which still forms the basis for glaucoma classification in the present day (Figure 1.1).



Figure 1.1: Classification of glaucoma

Open angle glaucoma (OAG) is the most prevalent type of glaucoma and forms the focus of this research. It is commonly divided into two subcategories, namely high tension or primary open angle glaucoma (HTG/POAG), in which IOP is elevated (>21 mmHg) and NTG, in which IOP falls within the normal range (10-21 mmHg on diurnal testing). The distinctions between these two subcategories of OAG however have become somewhat blurred over recent years and this is discussed further in section 1.1.2. Nevertheless, as it is still currently common practice in the recent literature to come across the terms POAG and NTG being referred to individually, for the purpose of this thesis, the term POAG will be used to refer to the development of GON in the

presence of raised IOP and NTG, which accounts for approximately 1/3 of all open angle glaucoma cases, will be considered separately. The only exception to this will be when citing previous research in which no distinction between patients with regard to their IOP level was made. In these cases the term 'glaucoma' is used to describe participants and will indicate a non-specific diagnosis of open angle glaucoma.

1.1.1 Risk factors for glaucomatous damage

Both POAG and NTG are asymptomatic until the late stages of the disease by which point significant visual field loss has already occurred. The necessity for early detection and management of the disease is therefore clear however this is hindered by our still relatively poor understanding of the pathogenesis of glaucoma, despite extensive research in the area. Multiple risk factors have been implicated in the development of glaucoma, the most acknowledged of which include advancing age ¹², elevated IOP ¹³, positive family history ¹⁴ and African descent ^{15, 16}. At the ocular level the presence of myopia ^{17, 18}, a large ONH ¹⁹ and/or a thinner central corneal thickness ^{15, 20, 21} have also been recognised to increase the risk of GON development as well as, hypertension ²², hypotension ^{23, 24}, reduced OBF ²⁵, vasospasm ²⁶, oxidative stress ^{27, 28} and cardiovascular disease history ²⁴ at the systemic level. With regard to NTG specifically, the occurrence of female gender ²⁹, Japanese ethnicity ³⁰ and optic disc haemorrhages ³¹ are factors additionally identified as increasing the risk of its development. Such differences in the risk factors associated with POAG and NTG suggests that these two forms of OAG may represent distinct clinical entities, each with their own pathogenesis; however the relationship between them is not this clear cut, as discussed in the following section.

1.1.2 POAG vs. NTG

Although the separation of OAG into two distinct clinical entities on the basis of IOP has been common practice for many years, a large number of overlaps between the

pathogenesis and features of both POAG and NTG have been identified more recently, throwing this concept of a distinct separation into dispute. Indeed it has even recently been suggested that the terms POAG and NTG should in fact be abolished, along with the 'arbitrary' 21mmHg IOP cut off value and that glaucoma should instead be considered as a disease continuum across which IOP and pressure-independent risk factors coexist with a varying degree of influence ³²⁻³⁵. In support of this, factors more traditionally associated with POAG, such as elevated IOP, have been firmly established as part of the pathogenetic process in NTG ³⁶⁻³⁸ and similarly vascular alterations and other IOP independent factors, more traditionally linked to NTG, have also been linked to POAG ³⁹⁻⁴³. On the contrary however, there are still a number of studies which describe subtle but important differences in both the structural and functional ONH changes, as well as the vascular risk between POAG and NTG patients ^{31, 44-47}. Indeed NTG patients have previously been demonstrated to show a greater degree of inferotemporal NRR thinning, notching and disc haemorrhaging in comparison to POAG patients ^{48, 49}, as well as deeper, steeper sided and more central visual field (VF) defects ^{50, 51}. Furthermore, with regard to vascular risk, stronger associations between the presence of vascular factors such as vasospasm, vascular dysregulation and hypotension have been made with NTG in comparison to POAG^{26, 42, 52-54} and additionally NTG patients have also been demonstrated to show a greater frequency of related conditions such as migraine and Raynaud's phenomenon ⁵⁵. The possibility that POAG and NTG may still represent two distinct clinical entities can therefore not be ruled out, however much of the literature is conflicting and whilst the similarities and differences discussed above have been identified by some studies, others have found no such relationships, making it difficult to draw any firm conclusions ^{32, 44-46}. Further investigation into the relationship between POAG and NTG and a thorough evaluation of their validity as distinct clinical entities would be beneficial in enhancing our understanding, diagnosis and future

management of these two conditions. A summary of the most commonly identified

similarities and differences between POAG and NTG is given in table 1.1.

	POAG	NTG
Onset and Symptoms	Gradual, asymptomatic	Gradual, asymptomatic
Clinical Profiles	Non-specific	 Lower body weight Detail-orientated and health conscious
Risk factors	 Age IOP* Gender (female) Family history African Descent Myopia Thinner corneal thickness Hypertension* Hypotension Altered OBF Vasospasm/Vascular dysregulation Oxidative stress Cardiovascular disease history* 	 Age IOP Gender (female)* Family history African Descent Japanese descent* Myopia Thinner corneal thickness Hypertension Hypotension* Altered OBF* Vasospasm/Vascular dysregulation* Oxidative stress Cardiovascular disease history
Structural ONH Change	 Progressive enlargement and deepening of the optic cup in conjunction with thinning of NRR Deep cupping 	 Progressive enlargement and deepening of the optic cup Inferotemporal NRR thinning predominantly Disc haemorrhages
Visual Field Defects	 Arcuate scotomas Paracentral locations involved at later stages of field loss 	 Deep, steep, paracentral defects
Nature of Progression	 More pronounced Correlates with level of IOP 	 Less pronounced or even non-progressive

Table 1.1: Comparison between POAG and NTG (*Indicates where a risk factor is more
predominant)

All in all, it is clear from the wide range of both ocular and systemic risk factors implicated in the development of both forms of OAG that it is a disease of multifactorial origin, the end stage of which is known to involve the apoptotic loss of retinal ganglion cells (RGC), tissue remodelling and excavation of the ONH. The mechanisms by which these multiple risk factors go on to contribute towards the classic glaucomatous ONH changes in both forms of the disease is however currently uncertain, with a number of different aetiological theories having been proposed. Gaining a better understanding of the exact pathophysiology of glaucoma could lead to huge advances in its diagnosis and treatment and significantly improve the prognosis for affected individuals. The current knowledge surrounding the aetiology of glaucoma and the involvement of both mechanical and vascular factors will now be discussed in detail, along with the anatomy and physiology of the ocular structures, which is necessary to aid the interpretation of this thesis.

1.2 Anatomy and Physiology of the Retina, Optic Nerve and Ocular Circulation

1.2.1 The Retina

The vertebrate retina is a complex multilayered structure consisting of five types of neuronal cells (ganglion, bipolar, amacrine, horizontal cells and photoreceptors) and one type of glial cell (Muller) inter-dispersed by synaptic connections ⁵⁶. Its anatomical structure is depicted in figure 1.2.

The inner most layer, closest to the vitreous body is termed the internal limiting membrane (ILM) and is formed from Muller cell end plates. These cells extend vertically through the retina from the ILM to the external limiting membrane (ELM) ⁵⁶ and act to maintain homeostasis and provide support and protection for the neurons. The ganglion cells lie immediately behind the ILM with their axons making up the retinal nerve fibre layer (NFL), and their nuclei forming the so called ganglion cell layer itself. Two other cellular layers are also present in the retina, namely the inner nuclear layer and the outer nuclear layer. The inner nuclear layer contains the nuclei

and cell bodies of the bipolar cells, the amacrine cells and, along its outer margin, the horizontal cells. The outer nuclear layer on the other hand contains the cell bodies of the photoreceptors ⁵⁶. Separating these two neural layers are two plexiform layers of synaptic connections; namely the inner plexiform layer and the outer plexiform layer ⁵⁶. The ELM separates the photoreceptor layer from the outer nuclear layer and in a similar way to the ILM is formed from Muller cell end plates ⁵⁶. Finally the retinal pigment epithelium (RPE) forms the outermost retinal layer, separating the neurosensory retina from the underlying choroid and providing support for the photoreceptors.

Ultimately, the RGCs are the means by which the retinal information is finally transferred to the relay stations in the brain for integration and processing ⁵⁷. This transfer of information is achieved by convergence of the RGC axons from across the inner retinal margin, onto the ONH, and along the optic nerve ⁵⁶ and only occurs following a series of complex interactions between the retinal photoreceptors (rods and cones) and the retinal neurons (bipolar, amacrine and horizontal cells). A full account of this is beyond the scope of this thesis but can be found in 'Webvision: The Organization of the Retina and Visual System [Internet] ⁵⁸. In short, the two types of retinal photoreceptors (rods and cones), which contain visual pigments (rhodopsin and opsin respectively), detect quanta of light and transfer information regarding this light signal to the outer plexiform layer ⁵⁹. At the level of the outer plexiform layer the photoreceptor processes synapse with the horizontal and bipolar cells ⁶⁰ and this information ultimately reaches the ganglion cells, via the inner plexiform layer through either the vertical pathways (photoreceptors - bipolar cells - ganglion cells) or the lateral pathways (photoreceptor – horizontal cells – amacrine cells – ganglion cells) 57.



Figure 1.2: Diagrammatic representation of the human retina

1.2.2 The Optic Nerve

The optic nerve or second cranial nerve is part of the CNS with a pathway extending from the retinal level to the primary visual centres in the brain. It is commonly subdivided into four segments termed the intraocular, intraorbital, canalicular and intracranial segments ⁶¹. The intraocular segment of the optic nerve is commonly referred to as the head and the anatomy and physiology of this portion of the optic nerve is particularly relevant with regard to the development of optic neuropathies such as glaucoma.

The ONH arises when the intraorbital portion of the optic nerve pierces the sclera, choroid and outer retinal layers and it is located approximately 4mm nasal to the posterior pole of the globe ⁶¹. The ONH itself can be further subdivided into three sections, termed the surface nerve fibre layer, the prelaminar region and the lamina cribrosa region (figure 1.3), with the remainder of the optic nerve being then termed the retrolaminar portion ⁶². The subdivisions of the ONH are outlined below:

- <u>The surface NFL</u>: a layer of compact nerve fibres formed from the convergence of the retinal nerves across the entire retinal surface towards the ONH before they bend to run backward along the optic nerve itself ⁶². It is the most anterior layer of the ONH and it is covered by the ILM of the retina.
- <u>The prelaminar layer:</u> lies immediately behind the surface NFL and consists of nerve fibres arranged in bundles and surrounded by glial tissue septa ⁶². Within the glial septa are capillaries and between the bundles any loose glial tissue form trabelculae ⁶². The network of glial cells provides support, protection and nutrition to the optic nerve fibres in this portion of the nerve ⁶³. The prelaminar layer is closely connected to the lamina cribrosa at its base and separated at its edge from the adjacent retina and choroid by a further layer of glial tissue ⁶².
- The lamina cribrosa: a band of dense compact connective tissue which bridges across the entire thickness of the optic nerve. It is lamellar in structure, being made up of alternating sheets of connective and glial tissue and containing many oval or rounded openings which are lined by glial cells and allow the transmission of the nerve fibre bundles through the lamina layer. It also has a central opening allowing the transmission of the central retinal vessels ⁶². These openings or pores are larger in the superior and inferior sections compared to the nasal and temporal sections of the ONH ⁶⁴. Each nerve fibre bundle is surrounded by a continuous glial membrane which separates it from the adjacent connective tissue and provides support, protection and nutrition. Capillaries, which lie within the fibrous septa, form a dense capillary plexus making the lamina region a highly vascular section of the ONH ⁶⁵.
- <u>The retrolaminar portion</u>: encompasses the intraorbital, canalicular and intracranial sections and is enclosed by dura, arachnoid and pia ⁶². In the intraorbital region the nerve fibres become myelinated and their bundles lie in polygonal spaces formed by the connective tissue septa. The septa contain blood vessels and are attached to the pia in the periphery, the connective tissue

⁶². From here the nerve fibres continue posteriorly through the canalicular and
^{intracranial} segments of the optic nerve to the primary visual centres in the brain.



Figure 1.3: Diagrammatic representation of the optic nerve head and its vasculature

1.2.2.1 The Healthy Optic Nerve Head on Clinical Examination

As well as the structural profile of the ONH it is important to also consider its clinical picture visible on examination. Indeed on clinical examination the ONH is vertically oval in form with a central depression, referred to as the cup, surrounded by a rim of neural tissue, termed the neuroretinal rim (NRR) (figure 1.4) ^{66, 67}.

In the healthy ONH the NRR is well perfused and orange/pink in colour. It has a characteristic physiological shape, defined by the 'ISNT rule', whereby its broadness decreases from being maximum in the inferior region, followed by the superior region, the nasal region and finally reaching its narrowest point in the temporal region of the ONH ⁶⁶. The size of the ONH itself can show great physiological variation between individuals, with its area having been shown to range from 0.80 mm² to 6.00 mm² in a normal white population ^{66, 68, 69}. The central cup is an area devoid of neural tissue and is usually horizontally oval and paler in colour compared to the NRR ⁶⁶. Its size

and depth can also show great inter-individual variation and it is commonly graded in terms of cup-to-disc ratio ⁶⁶. In the healthy ONH the area of the optic disc and the optic cup are correlated with each other, so larger optic discs are associated with larger optic cups ^{67, 69} and larger optic cups are associated with greater depths ⁶⁷. Also visible in the central portion of the healthy ONH is the entrance/exit points of the CRA, CRV and their branches. Variations away from this healthy clinical picture can be indicative of optic neuropathy and the relevance of this with regard to glaucomatous nerve damage is discussed later in section 1.3.



Healthy optic nerve head

Figure 1.4: Diagrammatic representation of the healthy optic nerve head

1.2.3 The Ocular Circulation

1.2.3.1 The Retrobulbar Vessels

The blood supply to the ocular tissues primarily arises from the internal carotid artery, which branches to form the ophthalmic artery (OA) and subsequently the central retinal artery (CRA) and posterior ciliary arteries (PCAs) (figure 1.5). The OA, CRA and PCAs are collectively referred to as the retrobulbar vessels and their anatomy is as follows:

a) Ophthalmic Artery (OA)

The OA is the first major branch of the internal carotid artery, arising, in most cases, just after the internal carotid leaves the cavernous sinus. From here it travels anteriorly, passing within the optic canal, usually inferolaterally to the optic nerve, until it enters the orbit at its apex ⁷⁰. Numerous branches arise from the OA along its pathway to the orbit and supply not only the orbit itself but also some structures in the face, nose and meninges ⁷¹. Anatomical studies have revealed multiple inter-individual variations in the site of origin of the OA itself, its branches and the course it follows, therefore it is common practice to discuss its anatomy and pathway in terms of that most commonly observed. The most relevant branches arising from the OA with regard to the ocular circulation are the CRA and PCAs.

a) Central Retinal Artery (CRA)

The CRA is one of the first branches of the OA. It penetrates the optic nerve approximately 8mm posterior to the globe and travels anteriorly along the centre of the optic nerve before ultimately dividing into four major branches which supply the inner retinal layers ⁷² (see section 1.2.3.2).

b) Posterior ciliary arteries (PCAs)

The PCA circulation is the major source of the blood supply to the ocular structures and ONH. One to five posterior ciliary arteries branch from the OA at a point distal to the origin of the CRA. They travel forward along the optic nerve and further divide into multiple branches which pierce the sclera, usually laterally or medially to the optic nerve ⁷³. These branches are of two types, termed short and long PCAs. The short PCAs (SPCAs), of which there can be 10-20, are also of two types, paraoptic and distal. Paraoptic SPCAs enter the sclera close to the optic nerve and contribute to the blood supply of the ONH, peripapillary choroid and the circle of Zinn and Haller, as well as sending recurrent branches to the retrolaminar ONH plial vascular plexus ⁷³. The distal SPCAs enter the sclera a short distance away from the ONH and run radially towards the equator. Each distal SPCA supplies a sector of the choroid

extending from the posterior pole to the equator and then further subdivides into smaller choroidal arterioles which ultimately supply a lobule of choriocapillaries ⁷³. The choriocapillaries exist as a single network of continuous capillaries located directly beneath the RPE, which demonstrate a segmental distribution and supply the outer portion of the retina extending from the RPE to the outer part of the inner nuclear layer ^{74, 75}. The long PCAs (LPCAs), of which there are two (medial and lateral), enter the sclera in the horizontal plane on the medial and lateral sides and run radially towards the iris. Each LPCA supplies a sector of the peripheral choroid as well as a small section of the ciliary body and iris ⁷³. The choroidal circulation accounts for 85% of the total ocular blood flow and is characterised by very high flow and low oxygen extraction ^{76, 77}. This high flow is facilitated by low resistance of the choroidal capillaries ⁷⁸ and unlike the retinal capillaries, the choriocapillaries are fenestrated, demonstrate less autoregulation and have a rich autonomic innervation ⁷⁹⁻⁸²



Figure 1.5: Diagrammatic representation of the retrobulbar circulation

1.2.3.2 Retinal Circulation

Normal retinal function requires a stable blood supply that does not interfere with the optics of the system but still provides adequate nutrient delivery and temperature control. The major source of blood supply to the inner retina (extending from NFL to inner section of inner nuclear layer) is the CRA, whereas the outer portion of the retina (extending from outer section of inner nuclear layer to RPE) receives its blood supply from the SPCAs, via the choriocapillaries ⁷⁵.

The CRA, on leaving the optic nerve, divides into four major branches which lie in the nerve fibre and ganglion cell layers and run from the posterior pole to the periphery, supplying the whole inner retina⁷⁵. Some arterioles branch further to form side arms and supply a complex retinal capillary network. The retinal capillary network is split into three main sections, termed the radial peripapillary capillaries (RPCs), the inner capillaries and the outer capillaries ⁷⁵. The RPCs lie within the inner part of the nerve fibre layer and run along the paths of the major CRA branches. The inner capillaries lie in the nerve fibre and ganglion cell layer, underlying the RPCs and form a complex capillary inner plexus. Finally the outer capillaries lie in the inner plexiform layer and run to the border of the outer plexiform layer ⁷⁵. This capillary network extends throughout the length of the retina, with the only exception being in the central macula region where a capillary free zone exists parafoveally ⁷⁵. The outer portion of the retina, extending from the RPE to the outer section of the inner nuclear layer, receives its blood supply from the choriocapillaries which form a continuous single network and lie in the inner most layer of the choroid ^{75, 83}.

Drainage of the retinal circulation is achieved via the central retinal vein (CRV), which travels centrally within the optic nerve, alongside the CRA, before exiting the nerve and ultimately draining into either the superior ophthalmic vein (SOV), other intraorbital venous branches or directly into the cavernous sinus ⁶¹. Drainage of the

choroidal circulation and hence the outer portion of the retina is via the vorticose veins.

On the whole, retinal circulation is characterised by a low level of flow and high level of oxygen extraction ^{78, 84}. It is autoregulated, allowing blood supply to be adjusted to metabolic demand, but it receives no autonomic innervation ⁸⁵⁻⁸⁷. Tight junctions exist between the capillary endothelial cells forming a blood-retinal barrier similar to the blood-brain barrier of the CNS. The RPE, which is considered the most posterior layer of the retina, forms an outer blood-retinal barrier which complements the inner blood retinal barrier formed by the retinal capillaries and prevents the passage of all but essential metabolites from the bloodstream to the retinal tissues ⁷⁸.

1.2.3.3 ONH Circulation

The ONH is a highly vascular structure which receives its primary blood supply from the SPCAs, either directly or via the peripapillary choroid. If present, contributions are also made from the intrascleral circle of Zinn and Haller⁶⁴.

The nature of the blood supply varies slightly between the different portions of the ONH as outlined below:

- <u>Surface NFL</u>: mainly supplied by the retinal arterioles and hence the CRA, in a manner continuous with the peripapillary retina ⁸⁸. The temporal portion may be supplied by the PCA circulation from the underlying prelaminar region and if a cilioretinal artery is present then it will also contribute in the corresponding area ⁶⁴.
- <u>Prelaminar region</u>: mainly supplied by centripetal branches from the peripapillary choroidal vessels, particularly on the temporal side, however some of the blood supply may also come from the vessels located in the

lamina region ⁸⁸. Blood supply is sectorial, in a similar way to the choroid, but no contribution is made by the CRA or the peripapillary choriocapillaries ⁶⁴.

- <u>Lamina cribrosa region</u>: supplied by centripetal branches arising either directly from the SPCAs or from the circle of Zinn and Haller, if present. The vessels form a dense capillary plexus making it a highly vascular region of the ONH ^{64, 88} however there is no contribution from the CRA.
- <u>Retrolaminar portion</u>: has a centripetal vascular system formed primarily by recurrent pial branches arising from the peripapillary choroid and the circle of Zinn and Haller. Additional pial branches may also arise from the CRA or other orbital arteries. This network of precapillaries and capillaries run centripetally within the connective tissue septa ⁶⁴.

Venous drainage of the ONH is primarily via the central retinal vein however in the prelaminar region drainage can also occur via the peripapillary choroidal veins (vorticose veins)⁶⁴.

The circulation of the optic nerve resembles the cerebral circulation and that of the rest of the CNS, in that, on the whole, tight junctions exist between vascular endothelial cells creating a blood-brain barrier (BBB)⁸⁹. This barrier prevents the passage of all but essential metabolites from the blood into the tissues, protecting them from the effects of foreign, potentially damaging, substances that could be present in the bloodstream as well as the effects of hormones or neurotransmitters that may be active in the rest of the body⁹⁰. The prelaminar portion of the ONH, however, has been to found to lack the BBB properties typical of the rest of the CNS and optic nerve, indicating that the vessels in the prelaminar region may demonstrate different permeability characteristics to that of the other regions of the ONH ⁸⁹. These permeability differences could be important with regard to regulation of blood flow
and disease pathology in this region due to the possibility of diffusion of vasoactive substances from the choroidal circulation into the ONH.

The ONH receives no direct autonomic innervation ^{91, 92}, however its blood flow is autoregulated ^{87, 93, 94}. This capacity for autoregulation is however considered to be less efficient than in the retina, but more efficient than in the choroid ⁹⁵.

1.2.3.4 Venous Drainage

Anatomical studies reveal the orbital venous system to be complex and highly variable ^{70, 96}. The SOV is generally accepted as the main route for venous drainage from the orbit. It is formed from the union of the supraorbital and angular veins just posterior to the trochlea and travels posteriorly, following the course of the OA, before ultimately leaving the orbit through the superior orbital fissure and draining into the cavernous sinus ⁹⁶. Multiple venous tributaries drain into the SOV including the superior vortex, lacrimal, muscular, inferior orbital and CRV ⁷⁰.

The CRV is the main source of venous drainage from the retinal circulation. It travels centrally within the optic nerve, exiting 2mm behind the CRA entry point before continuing to run posteriorly below it. The CRV then either drains into the SOV, into other intraorbital venous branches or directly into the cavernous sinus ⁶¹.

The vorticose veins, of which there are normally four, are the main source of venous drainage for the uvea. Each vorticose vein (superior and inferior lateral and medial) drains the corresponding quadrant of the choroid as well as the corresponding quadrant of the iris and ciliary body ⁸³. The superior medial and lateral vorticose veins ultimately drain into the SOV and the inferior lateral and medial vorticose veins drain into the inferior ophthalmic venous plexus ⁹⁶. The inferior ophthalmic venous plexus is the main component of the inferior orbital venous system. It runs posteriorly, close to

the orbital floor, and joins either the SOV or drains directly into the cavernous sinus. It receives contributing veins from the lower lid, lacrimal sac, inferior rectus and oblique muscles and the sclera (inferior vortex veins)^{70, 96}.

1.2.3.5 Assessment of Ocular Blood Flow

A variety of different techniques are available for the assessment of ocular blood flow (OBF) however no single technique is able to provide all the relevant information about the ocular vasculature in one reading as each tends to be designed to measure a specific aspect of ocular perfusion from one specific ocular vascular bed. The technique selected for the assessment of OBF in any research therefore needs to correspond with the vascular bed of interest. Indeed OBF can be assessed at the level of the retina, ONH, choroid or retrobulbar vessels and can either be measured directly or indirectly in terms of blood flow velocity, pulsatile OBF or vessel diameter changes. More recently enhanced techniques such as dynamic retinal vessel analysis (DVA), which enables retinal vessel reactivity to be assessed and retinal oximetry, which provides a direct assessment of retinal vessel metabolism, have also been introduced ^{97 98}. An overview of the techniques currently available for the assessment of OBF is given in table 1.2.

Method	Vascular bed and measurement	Advantages	Disadvantages
Color Doppler Imaging (CDI)	Retrobulbar vessels (OA, CRA, CRV, PCA)	Non-invasive, high reproducibility, quick	Poor inter-observer variability, retrobulbar assessment only, low resolution for smaller vessels
	Blood flow velocity		
Pulsatile OBF (POBF)	Choroid	Simple, non-invasive	Provides an approximation only and influenced by IOP and gender
(Langham OBF system; Laser interferometry)	Pulsatile OBF via IOP pulse wave or interferometry		
Laser Doppler flowmetry (LDF)	Choriod, ONH (depending on wavelength used)	Direct assessment of blood flow, non- invasive	Restricted to a small measurement area, exact volume of tissue measured unclear, inter- individual comparisons
	Capillary blood flow		poor
Laser Doppler velocimetry (LDV)	Retina	Simple, quick, quantitative,	Single vessel only
, (,	Blood flow velocity		
Heidelberg Retina Flowmeter (HRF)	Retina, ONH	Non-invasive, does not require dilation, quick, direct assessment of OBF	Requires clear media and good fixation. Very sensitive to illumination and measurement window changes, limited reproducibility
(= a scanning laser Doppler flowmeter)	Capillary blood flow		
Dynamic Retinal vessel analysis (DVA)	Retinal vessels (>60µm)	Non-invasive, high reproducibility, low variability	Requires clear media, good fixation and pupil dilation. More aimed at assessing retinal vascular function than OBF
	Retinal vascular diameter and reactivity		
Fluorescein angiography	Retina	Provides useful information on ocular perfusion	Debatable correlation of passage of time of dye to OBF. Invasive. Difficult to quantify
(in conjunction with scanning laser ophthalmoscope)	Blood flow velocity		
Indocyanine green angiography	Choroid	Provides useful information on ocular perfusion	Debatable correlation of passage of time of dye to OBF. Invasive. Difficult to quantify
scanning laser ophthalmoscope)	Blood flow velocity		
Laser speckle analysis	Retina, ONH	Provides overall map of retinal blood flow if used with scanning. Non-invasive	Not direct assessment of OBF
	Blood velocity		
Blue field entoptic technique	Foveal retina	Allows assessment of foveal perfusion, non- invasive	Dependent on patient cooperation, subjective, limited to foveal area, large inter-individual variation
Retinal Oximetry	Retinal vessels	Allows assessment of retinal metabolism	Risk of influence by external factors e.g light

Table 1.2: Overview of techniques available for assessment of ocular blood flow

1.3 Glaucomatous Optic Neuropathy

Whilst glaucomatous damage to the visual system has been demonstrated to involve pathological alterations in numerous areas, including the RGC bodies ^{99, 100}, photoreceptors ^{101, 102}, lateral geniculate nuclei (LGN) ^{103, 104} and visual cortex ¹⁰⁴, it is the lamina cribrosa of ONH which is considered the principal site of RGC axon insult ¹⁰⁵⁻¹⁰⁷. Indeed profound alterations within the prelaminar, laminar and peripapillary sclera tissues of the ONH have been identified by animal studies in the earliest detectable stages of experimental glaucoma ¹⁰⁸⁻¹¹⁰. Such alterations, combined with RGC loss, contribute towards the characteristic enlargement and deepening of the optic cup and thinning of the NRR observed clinically in GON (figure 1.6)^{111, 112}. Indeed prelaminar thinning, posterior deformation of the lamina cribrosa and excavation of the ONH have been identified as key features of GON ^{111, 113}. NRR thinning, indicative of nerve fibre loss, is unique to glaucomatous nerve damage and has a predilection for the inferotemporal and superotemporal regions of the ONH. extending to involve the temporal region as the nerve damage progresses before finally affecting the nasal region in advanced glaucoma^{114, 115}. Visual field and retinal NFL defects, characteristic of GON, develop in correspondence to the NRR thinning and subsequent excavation of the ONH 67, 116, 117.



Glaucomatous optic nerve head



Whilst the clinical picture and features of GON are fairly well evidenced, the causes of the primary insults which trigger the cascade of events leading up to its development are less clear and have subsequently become a key focus of current glaucoma research.

The evidence surrounding the potential role of elevated IOP in the developmental process will now be outlined in brief, followed by a detailed discussion of the vascular concept of glaucoma.

1.3.1 The Role of Intraocular Pressure

Traditionally the management of glaucoma has focused on the therapeutic reduction of IOP to below a certain target level with the aim of limiting the progression of ONH damage and visual field loss ¹¹⁸. There is no doubt that IOP is a relevant pathogenetic factor in the development of GON and since the introduction of the Schiotz tonometer in 1905, followed by the Goldmann applanation tonometer in 1954 ⁹, it has been the focus of the majority of diagnostic examinations and considered the only modifiable risk factor of glaucoma.

By definition IOP is a mechanical entity, referring to the normal force per unit area exerted by the intraocular fluids on the tissues that contain them ¹¹⁹. The normal range of IOP is defined as between 10-21 mmHg ¹¹⁸ and rather than being stable it exists in a state of constant flux, varying with cardiac and respiratory cycles and being influenced by factors such as posture and diet ¹²⁰. Numerous studies have explored the normal circadian variations in IOP that occur over a 24 hour period ¹²¹⁻¹²³ and whilst it was initially believed that IOP levels were at their highest in the morning, lower later in the afternoon and at their lowest at night ¹²⁰, other studies have suggested that IOP levels may in fact reach their highest levels nocturnally ¹²⁴⁻¹²⁶, regardless of whether measurements are being taken in the sitting or supine position

¹²⁷. Deviations away from this normal circadian rhythm have been demonstrated in those diagnosed with glaucoma and it is possible that abnormal fluctuations in IOP could increase the risk of GON development ¹²⁸⁻¹³⁰. This has led to the suggestion that 24 hour measurement of IOP may be beneficial in at risk patients ^{130, 131}.

By its very nature IOP exerts force on the lamina cribrosa, astroglia and axons of the ONH and when IOP is increased so is the strength of this mechanical force ¹³². As a result, the mechanical theory of glaucoma development proposes that in susceptible individuals abnormally elevated IOP damages the ONH by placing high levels of stress and strain on its tissues, ultimately leading to deformation of the cribriform plates of the lamina cribrosa and glial cell activation, followed by compression of the optic nerve fibre bundles and nerve fibre damage ^{109, 111, 133}. Indeed support for this theory comes from multiple studies which have demonstrated GON development in response to experimentally increased levels of IOP which progresses according to severity and duration of elevated IOP exposure^{8, 134, 135}. The overall susceptibility of the ocular structures to the effects of IOP however appears to vary between individuals as a function of the individual eye's anatomy and composition ^{111, 119} and such variation is perhaps demonstrated by the fairly high occurrence of the condition 'ocular hypertension' (OHT), whereby elevated IOP exists in the absence of GON and by the occurrence of NTG and of progressive glaucoma, in which nerve damage continues to progress despite therapeutically lowered IOP. Therefore, whilst there is undoubtedly a lot of evidence for the role of IOP in the pathogenesis of glaucoma⁸ and for the benefits of IOP reduction in the treatment of management of glaucoma ³⁶⁻ ³⁸ it is clear that alternative or additional causative factors need to also be considered with regard to the development of GON in the majority of cases. The most researched of these factors is the involvement of vascular abnormalities.

Disturbed vascular function is a concept which has long been recognised with regard to GON, with authors as early as 1925 proposing an alternative vascular theory of glaucoma development along with the aetiological involvement of microcirculatory disturbances and vascular dysregulation ^{136, 137}. The role of both ocular and systemic vascular factors in the development of GON has been subsequently explored by numerous researchers; however our understanding of the complex interactions between these and the many other factors implicated in the aetiology of the disease is still incomplete.

It is possible that rather than existing as two separate entities the so called mechanical and vascular theories of glaucoma could be intertwined and act synergistically to produce glaucoma, a concept first proposed by Flammer in 1985¹³⁸. Indeed elevated IOP can potentially influence ocular haemodynamics by subsequently raising venous pressure at the exit point of the eye and lowering the OPP (see section 1.4.1)¹³⁹. Furthermore large diurnal fluctuations in IOP could potentially alter the quality and stability of the blood supply to the ONH and also contribute to GON development¹⁴⁰. Alternatively it has been hypothesised that even if the primary insult of the ONH were not mechanical in origin, IOP related stress and strain could still contribute towards its final deformation¹¹¹. The vascular theory of glaucoma forms the focus of this thesis and will now be explored in detail.

1.3.2 The Vascular Theory

The concept that the quality of the blood supply to the ONH is altered in glaucoma is fairly well established and increasing amounts of evidence suggest that vascular insufficiency at the ONH plays an important role in GON development ^{95, 140-142}. It is hypothesised that an alteration in blood supply, which could result from the direct or indirect actions of a combination of risk factors, contributes to GON development through ischemic and hypoxic insult of the ONH tissues, including the RGC axons,

astrocytes, glial cells, pericytes and the central retinal vessels ¹⁴⁰. It is further hypothesised that the ischemic/hypoxic insult of the ONH tissues could produce its damaging effects through mechanisms of recurrent reperfusion injury (RRI) and oxidative stress ¹⁴¹. RRI is central to the current vascular theory of glaucoma and refers to the damage to a tissue caused when blood supply returns after a period of ischemia ¹⁴³. This can lead to increased production of harmful ROS, which, if not combated by antioxidant activity, can potentially damage the ONH cellular components, leading to apoptotic loss of RGC and their axons. This in combination with astrocyte activation and an increased production of vasoactive agents such as ET-1, is thought to contribute to the development of GON ¹⁴⁴. The concept of RRI and oxidative stress with regard to GON development is discussed further in section 1.10.3.2.

Confirming these vascular hypotheses and determining the combination of factors which contribute towards the reduction in the quality of the blood supply to the ONH in glaucoma is one of the main focuses of this thesis and current glaucoma research as a whole. Of particular interest is the exploration of the role that vascular factors play in NTG and progressive glaucoma development, where IOP involvement is less prominent. At the current stage aetiological roles for both ocular and systemic vascular abnormalities have been identified in the development of GON. These abnormalities extend from reductions in blood flow, through to disturbances in vascular regulation and increases in vessel rigidity at both levels. In addition systemic dysregulations such as peripheral vasospasm and endothelial and ANS dysfunctions, as well as high levels of oxidative stress, have been advocated to play an important part in the development of glaucoma. Alongside this alterations in factors such as BP and HR, which can influence overall vascular physiology, have also been implicated however many questions still remain. A summary of how these many ocular and systemic vascular abnormalities may link together in the development of GON is

given in section 1.11 at the end of this chapter (figure 1.16). The physiology of OBF along with the current evidence surrounding the involvement and interactions of each of these factors will now be explored in detail.

1.4 Ocular Blood Flow

The overall blood flowing through the body is determined by the cardiac output, which itself depends upon stroke volume and heart rate. The distribution of this cardiac output between vascular beds varies according to the hemodynamic situation of the body and is influenced by systemic factors such as the ANS and circulating hormones ⁷⁸. Within each vascular bed local factors strive to regulate the supplied blood flow to meet local needs. At the ocular level this regulation of blood flow has been identified as a multi-factorial and complex process ⁹⁵, dependant on both ocular perfusion pressure (OPP) and vascular resistance. Indeed the rate of blood flow through any vascular bed is known to be directly proportional to the pressure gradient and inversely proportional to resistance ¹⁴⁵.

1.4.1 Ocular perfusion pressure

OPP refers to the force of the blood flow through the intraocular vessels and is defined as the difference between the arterial and venous pressure. As a direct measure of arterial and venous pressure is not easily obtained from the retinal circulation, estimations have to be made in order to determine OPP. Experimentation has revealed that, due to the drop in BP between the heart and the OA, retinal arterial pressure can be estimated as 2/3 of the mean arterial pressure (MABP) and venous pressure can be taken to be approximately equal to the level of IOP ^{78, 146}, as a result OPP can be calculated as shown in equations 1.1 and 1.2

$$OPP = \frac{2}{3} MABP - IOP$$

OPP = Ocular perfusion pressure

MABP = Mean arterial blood pressure

IOP = Intraocular pressure

Equation 1.1: Calculation of ocular perfusion pressure

Where:

$$MABP = \frac{2}{3}DBP + \frac{1}{3}SBP$$

MABP = mean arterial blood pressure

DBP = Diastolic blood pressure

SBP = Systolic blood pressure

Equation 1.2: Calculation of mean arterial blood pressure

The relationship between blood flow and OPP in the eye is complex and as the equation above demonstrates both a lowering of MABP and an increase in IOP could potentially reduce OPP.

1.4.2 Vascular resistance

Vascular resistance, in addition to OPP, also plays a role in determining the amount of blood that flows through a vascular bed and itself is determined by blood viscosity, vessel length and vessel diameter ⁷⁸. In the retinal vascular bed, due to constant capillary perfusion and a lack of precapillary sphincters, alterations in vessel length are not thought to play an important part in the regulation of OBF ^{78, 147}. Furthermore, whilst alterations in blood viscosity have been found to substantially influence retinal blood flow ¹⁴⁸, such alterations to not tend to occur without the presence of underlying pathology, such as sickle cell anaemia and other blood disorders ¹⁴⁹, therefore the role that blood viscosity plays in the regulation of OBF in health is also not considered

significant. Vessel diameter on the other hand plays a major role in determining retinal vascular resistance and OBF. Vascular resistance is known to be inversely proportional to the fourth power of the radius of a blood vessel, hence only small changes in vessel diameter can significantly influence vascular resistance, with narrowing of the vessel diameter increasing resistance and reducing blood flow and widening of vessel diameter, reducing resistance and increasing blood flow. Reduced/unstable OBF could therefore occur, not only as a result of reduced OPP, but also as a result of an increase in vascular resistance due to abnormal alterations in vessel diameter, which themselves could occur as a result of either structural changes in the vessel wall or the presence of a functional dysregulation ⁹⁵.

Under normal physiological conditions the retinal vasculature exists in a state of partial constriction, referred to as the basal vascular tone. This state of partial constriction is one from which vessel diameter can be readily modified in response to hemodynamic alterations, such as a drop in OPP, to ensure maintenance of a constant blood supply to tissues and organs by a mechanism referred to as autoregulation ¹⁵⁰. Autoregulation is an important mechanism which acts in response to alterations in perfusion pressure through modification of vascular resistance. The autoregulation of blood flow will now be discussed in detail.

1.4.3 Autoregulation

Autoregulation refers to the ability of the cardiovascular system to modify vascular resistance in order to allow a constant blood supply to be maintained despite variations in perfusion pressure ¹⁵¹. It operates to ensure tissues and organs receive an adequate blood supply despite variations in hemodynamic conditions and has been demonstrated in both the retinal ^{86, 152} and ONH circulation ⁹⁴ as well as, to a lesser extent, in the choroidal circulation ⁸¹.

The most common method of evaluating autoregulatory function is through assessing the response of the ocular and systemic vasculature to provocation. This provocation can take the form of posture changes, artificial lowering of IOP, cold provocation, hand grip testing, flicker light stimulation or induced hypoxia and hypercapnia. Such provocations put the vascular system under stress and should evoke an autoregulatory response which allows maintenance of normal ocular perfusion, a failure to observe this autoregulatory response is indicative of disturbed autoregulation.

Although the exact mechanisms underlying autoregulation are still unclear, metabolic, myogenic, neurogenic and humoral factors are all known to trigger autoregulatory responses in the ocular circulation, as are endothelial derived vasoactive agents ^{151, 153}, as summarised in figure 1.7. These autoregulatory triggers and their responses are outlined in the following sections



Figure 1.7: Summary of the factors which trigger autoregulation of blood flow in the ocular circulation

1.4.3.1 Metabolic autoregulation

Metabolic autoregulation refers to the regulation of blood flow in accordance with tissue metabolite concentration. A tight coupling mechanism is thought to exist between tissue metabolism and ocular perfusion ¹⁵¹, with alterations in the local concentrations of metabolites including oxygen (O₂) and carbon dioxide (CO₂) ^{154, 155}, potassium (K⁺), hydrogen (H⁺) ¹⁵⁶ and adenosine ¹⁵⁷ having been shown to influence ocular vascular tone.

Partial oxygen pressure (pO₂) has been identified as one of the main driving forces of metabolic autoregulation ^{158, 159}. Under conditions of systemic hyperoxia and hypoxia autoregulatory mechanisms act to maintain retinal and ONH O₂ at constant levels. Hyperoxic conditions trigger retinal arteriolar vasoconstrictions, reducing retinal blood flow and pO₂ ^{154, 160} and hypoxic conditions trigger retinal arteriolar vasodilations, increasing retinal blood flow ¹⁶¹ and allowing normalisation of pO₂. The hemodynamic response of the retinal vasculature to hyperoxia is thought to be mediated by endothelin ¹⁶², whereas the hypoxia-induced vasodilation of the retinal vasculature is thought to involve endothelial derived prostaglandins and/or adenosine ¹⁶³⁻¹⁶⁶. The choroidal vasculature, in comparison, has been demonstrated to show little or no alterations in response to changes in blood oxygenation ^{167, 168}.

Under conditions of hypercapnia, in which the partial pressure of CO_2 (p CO_2) is increased, metabolic autoregulatory mechanisms have been demonstrated to function in the retinal, ONH and choroidal circulation bringing about a vasodilatory response of the vasculature which results in an increase in blood flow and p O_2 ^{167, 169,} ¹⁷⁰. The exact mechanism behind hypercapnia induced vasodilation is still the subject of debate, however it is thought that interactions between nitric oxide (NO) and endothelial derived prostaglandins play an important role ¹⁷¹.

1.4.3.2 Myogenic autoregulation

Myogenic autoregulation refers to the regulation of blood flow in response to alterations in systemic BP and allows a constant blood flow to be maintained despite variations in BP ¹⁵¹. The myogenic response was first described by Bayliss in 1902 and is characterised by a decrease in vessel diameter following an increase in transmural pressure ¹⁷². Its primary function in the body is to maintain the Starling equilibrium for capillary fluid exchange; however it is unclear whether this is also its primary function in the eye ¹⁷³. Myogenic autoregulation is on the whole considered to be mechanically independent of the endothelium and intrinsic to the vascular smooth muscle cells (vSMCs), whereby stretching of the vessel wall is thought to lead to depolarisation of the vSMC membrane and vascular constriction ¹⁷⁴. In the cerebral and renal circulation however myogenic induced vasoconstriction has also been suggested to be at least partly mediated by endothelial factors ¹⁷⁵. Whilst myogenic regulation has been demonstrated in the ONH and retina ¹⁷⁶ it is unclear whether myogenic mechanisms are also involved in the regulation of choroidal blood flow ¹⁷³.

1.4.3.3 Neurogenic control

The eye has a rich autonomic innervation however this only extends to the uvea, PCAs and the extraocular portion of the CRA ¹⁷⁷⁻¹⁷⁹ and does not include the retina and prelaminar portion of the ONH ^{179, 180}. Neuronal regulatory mechanisms are therefore suggested to play a key a role in the regulation of choroidal blood flow but have little effect on retinal or ONH blood flow ^{181, 182}, despite alpha and betaadrenergic receptors having been identified in the retinal vessels ^{183, 184}. Sympathetic stimulation of the choroid, via sympathetic nerves originating from the superior cervical ganglion, triggers constriction of the choroidal blood vessels and increases choroidal vascular resistance, reducing blood flow ^{80, 182}. It has been suggested that this vasoconstriction response of the uveal vasculature may function to protect the eye against overperfusion during periods of increased HR or BP ¹⁸⁵ and

indeed there is evidence demonstrating that under conditions of exercise, where sympathetic activity is increased, a regulatory vasoconstriction response of the choroidal vasculature allows maintenance of a constant blood flow in the face of an increase in OPP ^{81, 186}. Numerous agents have been implicated as mediators of this neuronal regulation including acetylcholine ¹⁸⁷, noradrenaline ¹⁸⁷, vasoactive intestinal polypeptide ¹⁸⁸, substance P ¹⁸⁹ and NO ¹⁹⁰, however the exact role played by each of these agents in ocular circulation physiology is still unclear.

Parasympathetic nerves reach the eye through the oculomotor nerve, facial nerve and through the ophthalmic and maxillary divisions of the trigeminal nerve ^{78, 178} and there is evidence to suggest that parasympathetic innervation can stimulate a vasodilation response in the choroidal vasculature and increase blood flow. This evidence is variable however, as whilst intracranial stimulation of the facial nerve has been demonstrated to cause significant vasodilation in the choroid ^{80, 191}, electrical stimulations of parasympathetic nerve fibres of the ciliary ganglion, although inducing intense miosis, have not been found to notably change the uveal vascular resistance ¹⁸². The role of parasympathetic innervation in ocular neurogenic regulation is therefore uncertain.

1.4.3.4 Humoral Control

Humural control refers to the potential regulatory influence of numerous vasoactive agents present in the circulating blood which, through either direct interaction with the vascular smooth muscle cells (vSMCs) and pericytes or through mediation of endothelial cells ¹⁹², could influence OBF. Angiotensin and catecholamines for example, which are both circulating hormones, have been suggested to influence retinal and choroidal circulation, however the evidence is variable. Indeed whilst angiotensin-II receptor binding sites have been identified in ocular tissue ¹⁹³, suggesting a role for the renin-angiotensin-system in OBF regulation ¹⁹⁴, the majority

of data seems to suggest that in human healthy subjects this system does not play a major regulatory role ^{173, 195}. Furthermore a similar uncertainty surrounds the role of catecholamines in OBF regulation after both a decrease ¹⁹⁶ and an increase ¹⁹⁷ in retinal perfusion following administration of adrenergic drugs has been demonstrated. Therefore whilst the potential influence of circulating and local hormones should not be discarded the evidence suggests that they are unlikely to have a major impact on the regulation of the choroidal and retinal blood flow. Furthermore the presence of the BBB prevents direct contact between the circulating blood and the retinal and ONH vSMCs indicating that the role of circulating hormones in the regulation of ONH blood flow in particular may be even less; however some diffusion of molecules from the choroidal vasculature may occur in the prelaminar region of the ONH due to the structural differences in this region (see section 1.2.3.3).

1.4.3.5 Endothelial dependent regulation of vascular tone

The vascular endothelium is an important mediator of vascular tone, releasing vasoactive agents both under basal conditions and in response to various chemical and mechanical stimuli. These vasoactive agents are commonly referred to as endothelial derived constricting factors (EDCF) and endothelial derived relaxing factors (EDRF) and they play an important role in the regulation of OBF ¹⁵⁶. The endothelium and its regulatory roles are discussed in more detail in the following section.

1.5 The Endothelium

The vascular endothelium plays a critical role in the regulation of blood flow at both the ocular and cerebral level and as such a dysfunction of the vascular endothelium is commonly implicated as a possible causative factor in vascular disease ^{198, 199}.

1.5.1 Background

The endothelium is a highly specialised monolayer of cells which lines the inner surface of all of the blood vessels in the circulatory system (figure 1.8) ²⁰⁰. Its strategic anatomical position between the blood components and the vSMCs and pericytes places it in an ideal position to monitor and regulate vascular homeostasis ¹⁹⁹ in response to mechanical (e.g. shear stress), chemical (e.g. oxygen tension) and biological stimuli (hormones and vasoactive agents). ²⁰¹



Figure 1.8 Diagrammatic representation of the structure of the vascular wall and the location of the vascular endothelium

The most critical role of the endothelium is as an active regulator of vascular tone ²⁰²⁻ ²⁰⁵ however its other functions include inhibition of vSMC proliferation, regulation of inflammation, thrombosis and platelet aggregation, angiogenesis and control of vascular permeability ^{199, 201, 206}. The endothelium exerts its functional roles through the release of a variety of regulatory agents following stimulation. Its critical role as an active regulator of vascular tone is achieved through the release of endothelial derived vasoactive factors (EDVF) which act on the underlying SMCs to bring about either vasodilation or vasoconstriction of the vessel wall (table 1.3). Of these EDVFs, nitric oxide (NO) has been identified as the most potent vasodilator and endothelin-1 (ET-1) as the most potent vasoconstrictor ²⁰⁷⁻²⁰⁹ (see sections 1.5.2 and 1.5.3)

Endothelial derived	Nitric oxide (NO), prostacyclin (PGI ₂), endothelial derived	
relaxing factors	hyperpolarising factor (EDHF), adenosine triphosphate (ATP),	
(EDRF)	substance P, acetylcholine, C-type natriuretic factor	
Endothelial derived	Endothelin-1 (ET-1), angiotensin II, thromboxane A ₂ ,	
constricting factors	prostaglandin H ₂ , superoxide anions, ATP	
(EDCF)		

Table 1.3: Summary of the vasoactive agents released by the vascular endothelium

1.5.2 Nitric oxide (NO)

NO was first recognised in 1980 by Furchgott and Zawadski as the primary vasodilating agent released by the vascular endothelium ²⁰². It is synthesised during the oxidation of the amino acid L-arginine to L-citrulline within the endothelial cell membrane, by the enzyme endothelial nitric oxide synthase (eNOS) ²¹⁰. Two other isoforms of nitric oxide synthase (NOS) also contribute to NO production, namely neuronal NOS (nNOS) which is found in some neurons of the central and peripheral nervous system ²¹¹ and inducible NOS (iNOS) which is only expressed after a challenge by immunological or inflammatory stimuli ²¹². All three isoforms of NOS have been identified in the ocular circulation ²¹³⁻²¹⁵.

There is a constant basal production of NO by the endothelium, which contributes to the maintenance of optimum vascular tone ^{207, 216}. However, on stimulation of the endothelial cell membrane by chemical, mechanical or biological factors, NO production is increased ²¹⁷⁻²¹⁹. This increase in production is achieved via activation of both a calcium/calmodulin complex and the NOS enzyme, following an influx of calcium across the cell membrane ²¹⁰. Diffusion of NO from the endothelium into the neighbouring SMCs ultimately brings about a vasodilatory response of the vessel wall through stimulation of the guanylate cyclise enzyme and increased production of cyclic guanine monophosphate (cGMP) ^{220, 221} (figure 1.9).



Figure 1.9: Mechanism of endothelial nitric oxide (NO) synthesis. Ca²⁺: calcium ions; eNOS: endothelial NO-synthase; O₂: oxygen; GTP: guanosine triphosphate; cGMP: cyclic guanosine monophosphate

NO plays an important role in the regulation of physiological functions in the cardiovascular system and the central and peripheral nervous systems. In addition to its vasodilatory role its range of functions extend to include anti-platelet, antithrombotic, anti-proliferative and anti-atherosclerotic actions ²²². At the ocular level, NO has been identified as a physiological mediator in not only the retinal circulation but also the OA, PCA and the choroid ^{222, 223}. Furthermore NO has also been suggested to play a functional role in the conjunctival vasculature, the cornea, the lens epithelium, the ciliary body and the neural retina ²²⁴. The NO mediated vasodilatory response is well evidenced in the ocular circulation and the important role played by NO in both the regulation of OBF and the maintenance of basal tone is well recognised ^{207, 225-227}. As such any disturbance in the production or release of NO by the endothelium can have a significant effect on the ocular haemodynamics. Indeed disturbances of the I-arginine/NO system have already been identified in numerous ocular diseases including diabetic retinopathy, glaucoma and retinopathy of prematurity ^{198, 222, 228}.

1.5.3 Endothelin-1 (ET-1)

ET-1 is the most important vasoconstrictive factor released by the endothelium and is active not only in the eye but throughout the rest of the body ^{229, 230}. It is synthesised from the pre-propeptide big endothelin via an endothelin-converting enzyme dependent pathway in response to physiological stimuli such as hypoxia, ischemia and shear stress or in response to circulating substances such as thrombin, angiotensin II and cytokines ^{230, 231} (figure 1.13). The majority of its secretion occurs abluminally however there is also an element of intraluminal secretion which leads to a certain concentration of ET-1 in circulating blood ^{232, 233}. Two other isoforms of endothelin also exist, namely endothelin-2 (ET-2) and endothelin-3 (ET-3), each of which are derived from different genes and different cell types, however their roles are more poorly understood and less studied in comparison to ET-1 ^{229, 230}.

The biological effects of ET-1 are governed by two different endothelin receptor subtypes, ET_A and ET_B. ET-1 has a higher affinity for ET_A receptors which are located primarily in vascular SMCs and are the main mediator of its potent vasoconstrictive effect ^{234, 235}. ET_B receptors, which are non-selective towards the endothelin subtypes, are located primarily on endothelial cells, neurons and glia however they can also be found in vascular SMCs where they contribute toward the endothelin induced vasoconstrictive response ²³⁶. Although at high concentrations ET-1 is known to bring about its potent and sustained vasoconstrictive effect, at lower concentrations it has in fact been demonstrated to produce a vasodilatory response ^{237, 238}. This vasodilatory response is mediated via the release of NO and/or prostacyclin following stimulation of the ET_B receptors located on the endothelial cells ²³⁹ (figure 1.10). Under physiological conditions the net effect of ET-1 therefore depends upon the balance between the expression of ET_A receptors in vSMCs and ET_B receptors on endothelial cells. In pathological states however an upregulation of ET_B receptors on vSMCs and a possible downregulation of ET_B receptors on

endothelial cells always results in a predominant ET-1 induced vasoconstrictive effect

231, 240



Figure 1.10: Mechanism of Endothelin-1 (ET-1) synthesis. ET_A: endothelin type-A receptors; ET_B: endothelin type-B receptors; cGMP: cyclic guanosine monophosphate

The functions of ET-1 extend from the regulation of vascular tone and local blood flow to neuronal support and signalling as well as the proliferation and migration of vSMCs $^{230, 241, 242}$. Clinically endothelin has been associated with many pathological conditions including hypertension, congestive heart failure and coronary artery disease $^{243\cdot245}$. At the ocular level the endothelin system is thought to play a significant role in both normal and pathological processing, with ET_A receptors having been identified in the retinal and choroidal vasculature as well as the iris and ET_B receptors having been identified in the retinal neurons, glia and the lamina cribrosa of the ONH $^{246\cdot248}$. Indeed multiple studies looking at the alterations in OBF which occur on administration of systemic ET-1 have revealed a dose-dependent reduction in choroidal, retinal and ONH blood flow when ET-1 levels are increased $^{249\cdot251}$. Furthermore the role of ET-1 in the regulation of retinal blood flow 252 and choroidal blood flow has also been confirmed, through for example, studies exploring the effect of ET_A receptor antagonist infusion on choroidal blood flow during isometric exercise ²⁵³. Due to its prominent role in the regulation of OBF, dysfunctions in ET-1 activity could potentially be implicated in the development of any ocular disease with an ischemic effect, such as glaucoma, via disturbances in autoregulation or vasospasm ²⁵⁴

1.5.4 Endothelial dysfunction

Endothelial dysfunction refers to an altered ability of endothelial cells to perform their normal physiological functions ^{255, 256}. Its occurrence has been linked to the presence of established cardiovascular risk factors, such as hypertension, smoking, hypercholesterolemia, diabetes mellitus and obesity, chronic infections such as herpes viruses and cytomegalovirus, aging, cardiovascular diseases such as chronic renal failure, congestive heart disease and coronary artery disease as well as environmental factors such as hypoxia, oxidants, drugs and nutrition ^{231, 257-259}. In the presence of endothelial dysfunction the vasculature exists in a pro-inflammatory and pro-thrombotic state accompanied by increased atherosclerotic plaque formation, increased vascular tone and reduced vasodilatory responsiveness ^{201, 256}. These features are the result of a pathological alteration in the balance of mediators produced by the endothelium, characterised by:

- A decreased biosynthesis and/or bioavailability of NO potentially resulting from either deficient NOS levels, increased levels of NOS inhibitors such as asymmetric dimethylarginine, apoptotic loss of NOS containing cells or rapid inactivation of NO after its release ^{198, 201}
- Increased production of vasoconstrictors such as ET-1 and angiotensin II
- An excess of oxidants

Endothelial dysfunction is known to be present at the earliest stages of a disease process and in addition to having been identified as an important early event in the pathogenesis of atherosclerosis ²⁵⁶ it has also been recognised as a preceding or predictive factor for the development of future cardiovascular disease ²⁶⁰⁻²⁶² and in the development of glaucoma ¹⁹⁸. Early assessment of endothelial function in the peripheral and/or coronary circulation of at risk individuals could therefore provide important prognostic information about the risk of future disease ^{256, 263}. This, in addition to the fact that endothelial dysfunction has been shown to be modifiable in its early stages, has led to it becoming a research topic of great interest.

The precise cause of endothelial dysfunction is currently unclear however oxidative stress and the production of ROS is one of the main aetiological factors implicated to play a central role in its development and is a common denominator in the majority of the above mentioned associated conditions ^{257, 264}. Indeed increased production of ROS in the vascular system has been found to affect both the synthesis and activity of NO ²⁶⁵ and to promote the contraction of vSMCs ²⁶⁶. Strategies aimed at reducing cardiovascular risk, such as smoking cessation and physical exercise as well as treatment regimes involving lipid-lowering therapy, angiotensin converting enzyme inhibitors and antioxidants have all shown to be effective at improving endothelial dysfunction to differing degrees ^{231, 267, 268}.

1.5.5 Assessment of endothelial function

The ability to reliably detect and assess endothelial dysfunction is important with regard to the diagnosis, understanding and treatment of neurodegenerative diseases such as glaucoma and AD. The methods available can generally be separated into three categories, the first being the determination of the presence of soluble circulating endothelial markers such as NO, ET-1 and von Willebrand factor (vWf), the productions of which are known to be disrupted in the presence of endothelial

dysfunction, the second being the functional measurement of endothelial dependent vascular tone at focal sites of circulation ²⁶⁹ and the third being measurement of morphological and mechanical characteristics of the vascular wall, which can be altered in the presence of endothelial dysfunction. A summary of these methods is given in table 1.4.

The initial invasive methods of assessing endothelial function, such as quantitative coronary angiography and strain gauge plethysmography, have now largely been superseded by the introduction of non-invasive techniques such as brachial artery flow mediated dilation (FMD). Brachial artery FMD is considered the gold standard method of systemic endothelial function assessment in clinical practice and has therefore been the technique of choice in this thesis. Its principles are discussed in more detail in Chapter 3: Subjects and Methods. As a supplement to this technique an analysis of the circulating endothelial marker, vWf, was also made in this thesis.

wWF is a large glycoprotein which is synthesised exclusively by vascular endothelial cells and circulates in the human plasma at a concentration of approx 10 μm/mL ²⁷⁰. It plays an important role in mediating platelet adhesion to damaged arterial walls ²⁷¹ and its production is known to be increased in the presence of damaged endothelial cells ^{272, 273}. Indeed elevated levels of vWF have been identified in a number of different conditions known to be associated with endothelial dysfunction, such as atherosclerosis ²⁷², diabetes ²⁷⁴, hypertension ²⁷⁵, cerebrovascular disease ²⁷⁶ and myocardial infarction ²⁷⁷. Furthermore correlations have been identified between impaired FMD, the gold standard technique for assessment of systemic endothelial dysfunction and increased levels of circulating vWF ^{278, 279}. It is therefore recognised as a useful biomarker for the presence of endothelial dysfunction ²⁷⁰, however it is important to note that some questions still remain around whether elevated levels of vWf can be considered a true indicator of impaired endothelial dysfunction in all

cases ²⁸⁰. This is largely due to its activity as an acute phase reactant, whereby its plasma levels may be increased in the presence of clinical conditions, such as infection or injury, which are not associated with endothelial dysfunction ^{280, 281}. In the absence of other acute phase markers, such as C-reactive protein however, elevated vWF levels can be more reliably attributed to endothelial dysfunction ²⁸².

Technique	Advantages	Disadvantages			
Circulating Markers					
Endothelin-1 (ET-1) Von Willebrand factor (vWf) Asymmetric dimethylarginine (ADMA) Tissue type plasminogen activator Plasminogen activator inhibitor-1 Intracellular adhesion molecules Vascular cell adhesion molecules E-selectin P-selectin	Decreased bioavailability can be one of earliest detectable signs Relatively simple technique	Can be difficult to distinguish between release via normal endothelial stimulation and endothelial damage Large variations can occur day to day based on diet etc			
Nitric oxide (NO) production assays					
Urine NO concentration (NO ₃ ⁻) Urine cGMP	Early indicator of endothelial dysfunction	Heavily affected by dietary habits			
Functional tests					
 Systemic Coronary angiography Forearm venous occlusion plethysmography 	Sensitive indicator of endothelial function	Invasive Poor repeatability and reproducibility			
 Coronary positron emission tomography 	Non-invasive				
 Brachial artery ultrasonography (Flow mediated dilation, FMD) 	Non-invasive Repeatable Sensitive indicator of endothelial function	Requires training and good control of testing environment to ensure reliability			
Ocular – Dynamic vessel analysis (DVA)	Non-invasive simple procedure. Good repeatability	Affected by high refractive error, hazy media and unstable fixation			
Mechanical and morphological vascular wall assessment					
 Pulse wave analysis Pulse contour analysis Pulse amplitude tonometry Intima-media thickness (IMT) 	Non-invasive, repeatable, simple techniques	More indirect method of endothelial dysfunction compared to other techniques – less sensitive			

1.6 Ocular Blood Flow and Glaucoma

Evidence of reduced OBF, compromised circulation and increased vascular rigidity has been discovered by multiple studies at the level of the ONH ²⁸⁵⁻²⁸⁹, retina ^{25, 290-295}, choroid ^{294, 296-298} and retrobulbar vessels ²⁹⁹⁻³⁰² in patients with glaucoma and these blood flow reductions have been found to be prevalent in NTG and progressive patients ^{294, 303} and to correlate with deteriorations in visual function ^{289, 304, 305}. Furthermore multiple population based studies have linked lower baseline OPP to increased prevalence of GON ^{22, 306-308}, with Tielsch et al ³⁰⁶ for example finding that those with perfusion pressures of less than 30 mmHg have a six times greater risk of developing POAG than those with perfusion pressures of more than 50 mmHg and a number of other longitudinal studies demonstrating baseline perfusion pressure as not only an important risk factor for the development of GON but also for the progression of the disease ^{15, 23, 24}. A full summary of these studies is given in appendix 1 and on the back of these findings reduced OBF is now considered an important IOP independent risk factor for the development of glaucoma. Concern was initially raised in the literature however about the possibility that the reduced blood flow detected in glaucoma patients, rather than being causative, could simply occur as a secondary effect to the loss of RGCs and reduction in OBF demand which occurs with the development of GON ⁹⁵. Multiple studies have however gone on to demonstrate reduced blood flow in glaucoma patients prior to the development of ONH damage ^{25, 286, 309} and further studies have linked reduced blood flow to disease progression ³⁰², indicating a primary causative role. Furthermore, moving away from the eye, an increased occurrence of ischemic lesions of the ear ³¹⁰, heart ³¹¹ and brain ^{312, 313} as well as reduced blood flow in the carotid arteries ³¹⁴ and peripheral capillaries ^{26, 315, 316} have also been identified in glaucoma patients. The co-existence of such systemic blood flow alterations is an important observation as it would not be

expected if the reduced blood flow demonstrated in glaucoma patients was simply occurring as a consequence of the nerve damage.

The cause of the blood flow alterations detected in glaucoma patients has been the subject of debate. Rather than a sustained reduction in blood flow evidence suggests that the development of GON is more closely linked to unstable or fluctuating OBF ³¹⁷. Indeed the finding that conditions known to cause chronic reductions in blood flow, such as arteriosclerosis, are only weakly related to the development of GON ^{22,} ³¹⁸ and that other conditions also known to cause a chronic reduction in blood flow, such as multiple sclerosis, often lead to atrophy but not excavation of the ONH, provide support for this theory ^{233, 319}. Unstable OBF has subsequently become one of the most researched hypotheses of glaucoma development. The primary cause of this unstable blood flow is thought to be a dysfunctional autoregulatory mechanism. Indeed in the presence of disturbed autoregulation the ONH is susceptible to changes in OPP, IOP and BP, as well as to increases in local metabolic demand (see section 1.4). In basic terms, if blood supply is insufficiently regulated then the ONH and ocular structures are at risk of ischemic damage, RRI and subsequent RGC loss in periods when demand for blood and oxygen is high. The concept of disturbed vascular autoregulation in glaucoma will now be explored in more detail.

1.6.1 Autoregulatory Dysfunction

When confronted with an alteration in OPP, a change in vascular resistance or an increase in metabolic demand, autoregulatory mechanisms should act to adjust blood supply to ensure the needs of the ocular tissues are still met. If these autoregulatory mechanisms are defective however then the ocular tissues are at risk of exposure to ischemic episodes and RRI whenever metabolic demand is increased or blood flow parameters are altered. Defective autoregulation, through its contribution to unstable blood flow, could therefore potentially be linked to RGC loss, excavation of the ONH

and the GON development. Indeed multiple studies have provided evidence of disturbed autoregulation in glaucoma patients, not only at the ocular level ^{39, 320-322} but also at the systemic level ^{323, 324} and this disturbance has been found to be particularly noticeable in those with NTG ^{325, 326} and progressive glaucoma ^{327, 328}.

The mechanisms by which OBF autoregulation may become defective are unclear, however it is important to consider that, in addition to the mechanism itself being defective, defects in the OBF autoregulation could also potentially occur if the capacity of normal autoregulation is exceeded. Indeed the capacity of autoregulation is not infinite and can only function within a certain range of OPPs^{87, 139, 156}. If this range is exceeded, for example when IOP is exceedingly high (> 30 mmHg) or BP is exceedingly low (see equation 1.1), even a normally functioning autoregulatory mechanism would be unable to act to normalise blood flow and blood supply would be insufficient ^{87, 329}. With regard to glaucoma, autoregulatory dysfunction has been identified in both NTG and progressive cases where IOP is considered within the normal range ^{294, 303}. It is therefore considered more likely that in glaucoma patients, rather than the capacity of a normally functioning autoregulatory system having been exceeded, the autoregulatory mechanism itself may be defective in the majority of cases ^{139, 330}. Indeed exploring the causes, mechanisms and implications of such autoregulatory dysfunctions is now one of the main focuses of glaucoma research. As mentioned previously autoregulation of OBF is a complex process involving numerous different systems including the ANS (neurogenic control), systemic BP (myogenic control) circulating hormones (humoral control) and the endothelium (see section 1.4.3) and underlying abnormalities such as endothelial dysfunction, vasospastic syndrome and ANS dysfunctions have been implicated to play a role in the development of autoregulatory abnormalities in glaucoma ³³¹. Research is however ongoing and many questions still remain. These possible causes of defective autoregulation in glaucoma are summarised in figure 1.11 and the

background of these mechanisms with regard to the development of GON will now be discussed.



Figure 1.11 Summary of the potential contributing factors to disturbed autoregulation and hence unstable ocular blood flow in glaucoma

1.6.2 Vascular dysregulation syndrome

Vascular dysregulation is defined as an inappropriate constriction or inadequate dilation of the microcirculation when stimulated, combined with simultaneous dilations of the arteries or veins of neighbouring tissues ²³³. It was a term first introduced in the context of glaucoma by Flammer in 1994, superseding the use of the term vasospastic syndrome, which it was decided did not encompass the full nature of the condition despite vasospasms still being considered responsible for inducing the majority of the associated symptoms ³³².

Vascular dysregulation syndrome refers to the presence of a 'global vascular dysregulation' which affects many organs simultaneously or sequentially, including the heart, brain, fingers and eye ^{233, 333}. These affected regions are subjected to

episodes of local vasospasm and/or disturbed autoregulation often simultaneously ³³⁴. This syndrome occurs in contrast to local dysregulations of blood flow which can develop in isolated regions in individuals as a result of pathological conditions such as endothelial or ANS dysfunction. Vascular dysregulation syndrome can be classified into two subcategories, namely primary vascular dysregulation (PVD) and secondary vascular regulation (SVD), with PVD being most relevant with regard to the development of GON.

1.6.2.1 Primary Vascular Dysregulation syndrome (PVD)

PVD refers to the occurrence of vascular dysregulation in the absence of any underlying disease ³³³. Individuals with PVD are usually otherwise healthy and do not require treatment ³³³. Under baseline conditions these individuals can hardly be distinguished from others as PVD syndrome has little influence on baseline blood flow, however under conditions of stress, such as cold, psychological or mechanical stress, there is an inborn tendency for the vascular systems of these individuals to respond differently, showing a more frequent and intense vasospastic response which can result in reduced blood flow ^{233, 333}. This difference in response is thought to be due to a disturbance of the normal autoregulatory mechanisms in these individuals ³³².

PVD occurs more frequently in women ³³⁵ and in those with low body mass index, type A personalities ³³³ and a Japanese ethnicity ^{336, 337}. Individuals with PVD syndrome have a tendency towards cold hands and feet ³³³, slower sleep onset times ³³⁸, low BP, particularly at night and when young ³³⁹ and a reduced feeling of thirst (drinking because they know they have to rather than because of thirst) ³⁴⁰. They have also been shown to have a good sense of smell ¹⁴¹ and suffer more frequently from migraine than non-PVD individuals, however this is not a well established relationship ³³³. Some individuals also show an altered sensitivity to drugs, such as

an increased response to calcium channel blockers and systemic beta blockers, possibly related to the lack of an ABC transporter protein ³⁴¹. Generally, all of these symptoms will first manifest in puberty and then decrease with age, showing a marked reduction after the menopause in women, but a possible increase again if they are treated with oestrogen-replacement therapy ¹⁴¹.

There is a large body of evidence associating PVD syndrome with the ocular circulation, with the first links being made in the early 1980's when diffuse or glaucomatous-like visual field defects, which fluctuated markedly and could spontaneously disappear, were noted in PVD sufferers ^{342, 343}. This occurrence was at the time referred to as ocular vasospastic syndrome and was the first indicator that the eye may be affected by what is now referred to as vascular dysregulation syndrome ³⁴⁴. Further to this Gherghel et al ³³⁴, using Color Doppler imaging, found that individuals with vasospasm had decreased mean OPP and an increased resistivity index in the CRA compared to non-vasospastic individuals, providing evidence of both involvement of the retinal circulation and disturbed autoregulation in PVD sufferers. Indeed a number of ocular diseases have now been linked to the occurrence of PVD including, central serous chorioretinopathy ³⁴⁵, venous and arterial occlusions at a younger age ³⁴⁶, anterior ischemic optic neuropathy ³⁴⁷ and Susac syndrome ³⁴⁸. Of relevance here however is the large body of evidence linking PVD syndrome to the development of GON, as discussed in the following section.

1.6.2.2 PVD and glaucoma

The first associations, made between the presence of PVD and the development of glaucoma, arose from the finding that a minority of patients with the so called ocular vasospastic syndrome went on to develop GON ³³³. The possibility that PVD and glaucoma could share a common underlying mechanism was suggested on the back of multiple studies demonstrating disturbed autoregulation and altered blood flow in

PVD sufferers, similar to that found in glaucoma patients. Indeed there is now a large body of evidence supporting the role of PVD in glaucoma development, particularly with regard to NTG and an equally large body of evidence demonstrating increased tendency towards vasospasm in glaucoma patients. The role of PVD in the development of NTG is supported by the finding that NTG and PVD share a number of similarities, for example both occur more frequently in women ^{29, 335} and those of Japanese ethnicity ^{30, 337}. Furthermore NTG patients suffer more frequently from migraine ⁵⁵ as do individuals with PVD ³⁴⁹ and they suffer more commonly from silent myocardial ischemia ^{311, 350}, of which PVD syndrome is a cause ³⁵¹. Additionally, the occurrence of flame haemorrhages at the optic disc, which develop more commonly in NTG $^{\rm 31}$ has been linked to the presence of PVD $^{\rm 352}$. Other studies have identified direct signs of vasospasm or vascular dysregulation in NTG patients, for example, the presence of peripheral vasospasm in NTG patients, using nail fold capillaroscopy and cold provocation has been demonstrated on numerous occasions ^{26, 353, 354}. Furthermore, findings such as raised levels of ET-1, a potent endothelial derived vasoconstrictor ³⁵⁵⁻³⁵⁷, and of associations between NTG and the presence of migraine and Raynaud's phenomenon ⁵⁵, also highlight the potential role of vasospasm in the development of NTG. Similarly, indirect evidence of vasospasm in NTG patients has been found by studies assessing vascular and functional responses to breathing the vasodilator CO₂, ^{358, 359} whereby it is hypothesised that significant increases in OBF parameters, recorded in response to CO₂, are suggestive of the vasculature having been in an initial vasospastic state.

The presence of PVD and vasospasm have also additionally been linked to the development of POAG, for example Goldberg et al ⁴⁵⁹ using the Hettinger Hand Vibration Test and Butt et al ³⁶¹, both found signs of vasospastic tendencies in NTG and POAG patients, although the pattern of disturbance was slightly different between the two. Furthermore Rankin et al ³⁰¹ found evidence of increased mean

resistivity index and decreased mean flow velocity in the OA and CRA of patients with POAG which were similar to those found in NTG patients. Other studies looking specifically at POAG patients have found that they show larger diurnal fluctuations in OBF parameters compared to controls, indicating unstable ocular perfusion and supporting the hypothesis that there may be an underlying vascular dysregulation in these patients as well ⁴⁰. Additionally, Hosking et al ³⁶² assessed BP and blood flow parameter changes to induced hyperoxia and hypercapnia and found that POAG patients showed a larger increase in blood flow velocity in response to hypercapnia, a vasodilator, compared to controls but no significant change in blood flow in response to hyperoxia, a vasoconstrictor, to which controls showed a decrease in blood flow, again suggestive that there may be an initial state of vasospasm in POAG patients similar to that mentioned previously with regard to NTG.

It is possible that, in addition to disturbing blood flow regulation and contributing to ischemic damage, PVD may also render the eye more susceptible to changes in IOP. Indeed Schulzer et al ³⁶³ identified a distinct group of patients with vasospastic tendencies, which included both NTG and POAG patients, in whom a high positive correlation was found between mean deviation index of field severity and highest IOP. No such association was found in the other group who did not show vasospastic tendencies. Furthermore Hafez et al ³⁶⁴ found that vasospastic POAG and OHT patients showed the greatest improvement in NRR blood flow after sustained IOP reduction, compared to non-vasospastic patients and healthy controls, suggesting a closer relationship between OBF parameters and IOP level in patients with PVD syndrome compared to those without. Finally, Gugleta et al ³²⁸ found that progressive glaucoma patients who exhibited a vasospastic response which decreased choroidal blood flow by at least 10%, had lower IOPs compared to those with a milder vasospastic response, possibly suggesting that those with a greater degree of

vasospasm are more susceptible to IOP and therefore more likely to progress despite IOP being normalised.

On the basis of this plentiful evidence it has been suggested that PVD syndrome should be considered an independent risk factor for the development of glaucoma and its presence could potentially interfere with OBF either through the manifestation of disturbed autoregulation, or through the combined effects of hypotension, which is more prevalent in PVD, and reduced OPP ⁹⁵. A clinical diagnosis of PVD is largely based on history and symptoms and as such enquiries aimed at determining its presence, such as the occurrence of cold hands and feet or delayed sleep onset, are now recommended as routine practice in glaucoma ²³³. A more reliable diagnosis of PVD can be made using nail fold capillaroscopy and cold provocation, whereby capillary blood flow is measured in the nail fold area before, during and after cooling to -15°C and a reduced baseline blood flow velocity or a prolonged flow stop after cold provocation is considered indicative of PVD ^{349, 354, 365}.

1.6.2.3 Secondary Vascular Dysregulation

Secondary vascular dysregulation refers to the presence of a local or systemic vascular dysregulation which has developed secondary to an underlying disease, such as multiple sclerosis, rheumatoid arthritis or giant cell arteritis ^{141, 233}. Under conditions of stress or on stimulation, the vasculature of those with SVD exhibits a vasospastic response. The exact mechanism by which the underlying disease evokes this secondary vasospastic response is unclear, however a marked increase in levels of circulating ET-1, a potent vasoconstrictor released by the vascular endothelium, in response to inflammatory stress, has been identified in the majority of cases ^{233, 333}.

SVD tends to cause a more or less constant reduction in baseline blood flow and has little effect on autoregulation, making it less relevant to glaucoma development ¹⁴¹. At

an ocular level the effects can range from minimal to mild visual field reduction and a pale ONH ²³³. As SVD syndrome does not interfere with autoregulation it is considered only a minor risk factor for the development of glaucoma ³⁶⁶ in comparison to PVD.

1.6.3 Endothelial dysfunction

Endothelial dysfunction could potentially occur in isolation or in combination with PVD and/or ANS dysfunction in glaucoma patients ³⁶⁷. The concept and features of endothelial dysfunction were discussed in section 1.5.4 and there is now a fairly large body of evidence indicating the involvement of both ocular and systemic endothelial dysfunction in the dysregulation of blood flow and pathogenesis of both NTG and POAG. This endothelial dysfunction commonly takes the form of an imbalance in NO and ET-1 production, leading to reduced vasodilation, increased vasoconstriction and disturbed autoregulatory mechanisms in response to increased demand. The observation of both altered systemic and ocular endothelial dysfunction in glaucoma emphasises that, rather than just being a local ocular phenomenon, glaucoma may actually form part of a more global vascular dysfunction. The evidence linking endothelial dysfunction at both the systemic and ocular level to the development of glaucoma will now be discussed.

1.6.3.1 Ocular Endothelial Dysfunction and glaucoma

Direct assessment of endothelial dysfunction at the ocular level is more challenging than that at the systemic level as the ocular vasculature is less accessible. Nevertheless, multiple studies have found evidence of reduced NO ^{368 369} and increased ET-1 levels ^{357, 370, 371}, characteristic of endothelial dysfunction, in the aqueous humour of both NTG and POAG patients, however as well as relating to the vascular endothelium, these findings could also be indicative of endothelial
dysfunction in the cells lining the trabecular meshwork and Schlemm's canal. Stronger support for the presence of ocular vascular endothelial dysfunction in glaucoma therefore comes from functional studies. With regard to NO, abnormal OBF responses to systemic NOS inhibition, suggestive of reduced constitutive NOS at the ocular vascular level have been identified in POAG patients ³⁷². Furthermore abnormal or reduced neurovascular coupling responses of the retinal and ONH vasculature have been identified in early glaucoma patients following exposure to flicker light stimulation via dynamic retinal vessel analysis (DVA) ^{41, 373}. The concept of DVA, flicker light stimulation is discussed in more detail in Chapter 3: Subjects and Methods.

With regard to ET-1, in addition to raised levels in the aqueous humour, chronic ET-1 mediated ischemia of the ONH has been shown to induce glaucoma-like optic neuropathy in animals ³⁷⁴⁻³⁷⁶ suggesting a causative role. Furthermore the high sensitivity of the ocular circulation to changes in local ET-1 concentration has been demonstrated by the finding that intravenous application of ET-1 reduces pulsatile blood flow in the choroid and ONH at doses which show no effect on systemic haemodynamics ²⁵¹.

The ability to assess the ocular endothelium is improving through the development of techniques such as DVA, however due to the wide variety of procedures available and the easy accessibility of the systemic circulation more plentiful evidence exists for the presence of endothelial dysfunction at the systemic level in glaucoma patients.

1.6.3.2 Systemic Endothelial Dysfunction and glaucoma

A variety of different methodologies have demonstrated impaired NO activity and increased ET-1 activity along with reduced vasodilation and increased vasoconstriction responses, characteristic of endothelial dysfunction, in the systemic

vasculature of glaucoma patients. Considering NO first, reduced plasma levels of NO, cGMP and nitrite (NO_2) , have been demonstrated in both POAG ³⁷⁷ and NTG patients ³⁷⁸. Furthermore a higher plasma C-reactive protein level, a marker of reduced eNOS activity ³⁷⁹, has also been demonstrated in NTG patients ³⁸⁰, however this finding should be considered with caution as increased or normalised systemic levels of such markers have been demonstrated in glaucoma patients by other studies ³⁸¹. In addition to this biological evidence, functional assessment of the system.ic endothelium in glaucoma patients has also revealed evidence of NO mediated dysfunction. Indeed using venous occlusion plethysmography, a reduced vasodilation response in the forearm to both acetylcholine ³⁸² and ET_A receptor antagonism has been identified in NTG patients ^{383, 384}. Furthermore reduced brachial artery flow mediated dilation responses, indicative of impaired NO mediated vasodilation, have also been demonstrated in NTG patients ⁵² as well as POAG patients ³⁸⁵, although possibly to a lesser extent in this second group ⁴². (These measurement techniques were introduced in section 1.5.5 and are discussed further in Chapter 3: Subjects and Methods, section 3.3.4)

With regard to ET-1, again from a biological point of view multiple studies have found increased levels in the plasma of NTG patients ^{355, 356} and progressive glaucoma patients ³⁸⁶. However, the majority of studies report no difference in plasma ET-1 levels between POAG and healthy controls ^{357, 370, 387-389}. This suggests systemic vasoconstriction or vasospastic tendencies play less of a role in the development of POAG in comparison to NTG and progressive cases. Indeed, a further illustration of ET-1 abnormalities in NTG patients comes from functional testing, whereby abnormal changes in plasma ET-1 levels, indicative of disturbed autoregulatory mechanisms and systemic endothelial dysfunction have been identified in response to posture changes ³⁹⁰ and following peripheral cooling in NTG patients ³⁹¹, suggesting abnormal ET-1 mediated vasoreactivity . This also ties in with the findings of higher occurrence

of vasospastic related conditions such as migraine ⁵⁵, Raynaud's phenomenon and myocardial ischemia ^{311, 350} in NTG patients.

Plentiful evidence currently exists therefore for the presence of endothelial dysfunction at the systemic level in glaucoma patients. In addition to the above findings which relate to NO and ET-1, plasma levels of circulating von Willibrand factor (vWf), an established marker of endothelial damage, have been found to be abnormal in both POAG and NTG patients providing further evidence for the presence of systemic endothelial dysfunction in glaucoma ³⁹². Additionally, genetic studies have found evidence of polymorphisms in the eNOS gene in some familial POAG patients ³⁹³⁻³⁹⁵ and in the ET_A receptor gene of some NTG patients ^{396, 397} suggesting there a possible genetic involvement in the development of endothelial dysfunction in certain cases.

1.6.3.3 Summary of the role of endothelial dysfunction in glaucoma

A combined state of impaired vasodilation and increased vasoconstriction in the ocular and/or systemic vasculature of patients with glaucoma, as a result of endothelial dysfunction, is therefore hypothesised to play a key role in GON development through its significant impact on OBF regulation. Whether endothelial dysfunction occurs in isolation or in combination with other factors such as PVD or ANS dysfunction is currently unclear and little research has been conducted into the occurrence of simultaneous dysfunction of the ocular and systemic circulation in individual patients. Exploring the associations of endothelial dysfunction and determining whether a global dysfunction, affecting both the systemic macrovasculature and the ocular microvasculature, exists in glaucoma patients could lead to significant advances in our understanding of the development of the disease.

1.6.4 The Autonomic Nervous System (ANS)

The hemodynamic situation of the body is supervised and influenced by the ANS via constant regulation and control of heart rate (HR) and blood pressure (BP), amongst other factors ^{398, 399}. As such a dysfunction of the ANS system can impact on the regulation of blood flow in tissues in the eye and throughout the rest of the body and potentially play a part in the development of GON.

1.6.4.1 Background

The ANS itself forms part of the efferent division of the peripheral nervous system and is split into two systems, namely the parasympathetic nervous system (PNS) and the sympathetic nervous system (SNS), both of which innervate smooth muscle, cardiac muscle and glands ^{145, 400}.

The SNS promotes responses that prepare the body for strenuous physical activity in emergency or stressful situations ('fight or flight'). The release of adrenergic agents, such as noradrenaline, from the postganglionic fibres of the SNS, results in an increased HR and force of contraction, vasoconstriction, pupil dilation, stimulation of sweat glands and a breakdown of glycogen and fat stores ⁴⁰⁰. The PNS on the other hand dominates under resting, relaxed, non-threatening circumstances, with the release of acetylcholine from the postganglionic fibres of the PNS leading to a decrease in HR and force of contraction, pupil constriction and constriction of the bronchioles ⁴⁰⁰.

Under normal conditions both systems are partially active, giving rise to a basal sympathetic/parasympathetic tone. This tone can be involuntarily modified according to hemodynamic needs, allowing one system to dominate over the other so that the required response can be brought about ⁴⁰⁰. Most organs or tissues throughout the

body have both SNS and PNS innervations which are under reciprocal control; the exception to this is innervated blood vessels which only have sympathetic innervation.

1.6.4.2 Role of the ANS in Regulating Cardiac Rhythm

Among the hemodynamic factors regulated by the ANS is the cardiac rhythm. Rhythmical atrial and ventricular contractions pump blood out of the heart and into the systemic and pulmonary circulations as part of the normal cardiac cycle ¹⁴⁵. Both electrical and mechanical events play a role in producing these rhythmical contractions.

Electrical events

Contraction of the cardiac muscle (myocardium) is initiated by spontaneous, electrical impulses which originate from the so called cardiac pacemaker cells. The sinoatrial node (SA node) is the primary pacemaker cell, located in the right atrial wall. It undergoes spontaneous depolarisation approximately 70 times per minute and brings about atrial contraction. The electrical impulse generated by the SA node is conducted via the internodal conduction pathway to the atrioventricular node (AV node), the secondary pacemaker cell. Subsequent conduction of the electrical impulse from the AV node to the ventricle walls, via the bundle of His and purkinje fibres, ultimately leads to ventricular contraction ¹⁴⁵. The SA node receives extensive autonomic innervation from both the SNS and PNS meaning HR can be regulated by the ANS. Parasympathetic innervation of SA node initiates a rapid response which leads to a decrease in HR, whereas sympathetic innervation acts over a longer time course, bringing about increased HR and contractility of the cardiac tissue ³⁹⁸. At any given moment the overall HR is therefore determined by the balance between the PNS and SNS and under resting conditions it is the PNS which dominates, generating an average resting HR of 70 beats per minute ¹⁴⁵.

Mechanical events

The cardiac cycle begins with ventricular contraction, a phase referred to as systole. As the ventricles contract the pressure inside them rises rapidly and once it exceeds that inside the aorta and pulmonary artery, blood is ejected from the ventricle into the systemic and pulmonary circulations. The pressure subsequently decreases and the ventricles refill with blood, a phase referred to as diastole ¹⁴⁵. The cycle then repeats itself.

1.6.4.3 Heart Rate Variability (HRV)

Under normal resting conditions the sinusal rhythm is highly irregular, varying as a result of changes in physiological parameters such as respiration, BP, body temperature, metabolic rate, hormone levels and sleep cycles ⁴⁰¹. This high irregularity is evident when HR is examined beat by beat and is termed HRV. As the SA node, whose activity determines HR, is densely innervated by both the SNS and PNS, HRV reflects the modulating effect of the ANS on the intrinsic firing rate of the cardiac pacemaker cells ³⁹⁸. As such assessment of HRV can provide an indirect assessment of the autonomic control of the heart and a change in HRV pattern is recognised as an early sensitive indicator of compromised health ³⁹⁸. Indeed a low HRV, suggestive of a poor adaptability of the ANS, has been correlated with increased mortality ⁴⁰². HRV is therefore commonly used for the investigation of normal physiology and pathological conditions and can be assessed using either a frequency domain or time domain analysis, as outlined below.

Frequency domain analysis

This technique involves spectral analysis of the arterial pulse wave in order to evaluate the predominance of the sympathetic and parasympathetic divisions of the ANS and their effects on HR ⁴⁰³. In normal individuals cyclic variations in HR occur in association with respiration and BP fluctuations. Respiratory related cyclic variations

are mediated by the PNS and occur at a high frequency (HF) (0.2-0.4 Hz) ^{398, 404}. Conversely, cyclic variations related to fluctuations in BP and the subsequent changes in baroreceptor activity, typically occur at low frequency (LF) (0.0-0.04 Hz) and are mediated by the SNS ⁴⁰⁵. As well as considering these parameters individually the dynamics of the HRV signal and the sympathovagal balance of ANS function, can also be assessed through the evaluation of the LF/HF ratio ^{398, 406}.

Time domain analysis

This technique involves the analysis of an individual's electrocardiogram (ECG) profile and the subsequent determination of the average normal-to-normal heart beat interval (mean NN interval). This is then converted into a measure of HRV via simple mathematical evaluation ^{398, 407}.

1.6.4.4 Role of ANS in Regulating Blood Pressure

Alongside modifying HR and influencing cardiac output, the ANS also plays a role in regulating MABP (see equation 1.2). MABP is influenced by both cardiac output and total peripheral resistance, with an increased cardiac output and/or increased peripheral resistance leading to the elevation of MABP. Baroreceptors located within the circulatory system constantly monitor MABP and if any deviation away from normal is detected a series of reflex responses, including stimulation of the ANS, are initiated ¹⁴⁵. Subsequent autonomic innervation of the heart, arterioles and veins then leads to either adjustment of cardiac output or modification of peripheral resistance in order to normalise MABP accordingly. A detected increase in parasympathetic activity and an increase in parasympathetic activity and an increase in parasympathetic output and an increase in peripheral vasodilation. A fall in MABP leads to the opposite effect.

Both BP and HR follow a normal circadian rhythm which is dependent on the ANS ^{145,} ⁴⁰⁸. ANS activity itself is also thought to follow a circadian rhythm, with sympathetic activity having been shown to be lower at night, along with HR and BP, however this can also vary according to sleep cycles ⁴⁰¹. Through its influence on HR and BP a dysfunction or chronic imbalance of ANS activity could therefore significantly impact the regulation of blood flow not only in the heart, brain and/or peripheral vasculature, but also in the ocular circulation, due to the close proximity of the eye to the heart and brain via the carotid artery ⁴⁰⁹. The relevance of this with regard to the development of GON is discussed in the following section

1.6.4.5 Autonomic Nervous System Dysfunction and Glaucoma

Due to the important role that the ANS plays in the maintenance of blood flow physiology and the regulation of variables such as HR and BP, it is clear that a dysfunction of the ANS could have significant adverse hemodynamic effects. With regard to GON development both systemic parasympathetic and sympathetic neuropathies have been reported in those with POAG^{43, 410-412} and those with NTG ^{410, 413-415} using a variety of different assessment techniques.

Studies monitoring 24 hour BP and HRV have demonstrated greater low frequency (LF) values in NTG patients ⁴¹³ and greater LF values and LF/HF (low frequency/high frequency) ratio in POAG patients ⁴³, both diurnally and nocturnally, suggestive of a high sympathetic tone. Furthermore abnormalities suggestive of altered systemic ANS function have been linked to abnormal OBF regulation in POAG patients using cold pressor testing ⁴¹⁶ and on this basis it could be hypothesised that disturbed ANS function may contribute to the disturbed regulation of OBF and subsequently to the development of GON.

When considered as a whole the most recent evidence indicates the presence of a heightened sympathetic activity and suppressed parasympathetic activity in glaucoma patients; however some studies have found parasympathetic activity to also be increased. The exact nature of ANS dysfunction in glaucoma is therefore still unclear and the decision on whether it should be considered an independent risk factor for the disease or whether it simply occurs in coexistence with other factors has not been made. As mentioned in previous sections it is only the choroid and retrobulbar vessels which receive autonomic innervation at the ocular level, however multiple studies have demonstrated impaired regulation and reduced blood flow in both the retinal and ONH vessels as well. It is therefore likely that, if ANS dysfunction does play a role in the development of GON, it does not act in isolation. Links have been made in cardiovascular research between the coexistence of impaired ANS function and impaired endothelial function and it is possible that, due to the links between GON and endothelial dysfunction, similar dual impairments could be occurring in glaucoma patients ³⁶⁷.

1.6.4.6 ANS and endothelial dysfunction

Under normal conditions vascular tone is maintained by both the ANS and the endothelium, which work together in opposition. The release of vasodilating factors from the endothelium is balanced with the release of vasoconstricting factors from the sympathetic nerve terminals ³⁶⁷ and these vasoactive factors act together on the vascular smooth muscle cells to maintain vascular tone.

The ANS and the endothelium are not completely separate systems. Endothelial cells possess receptors for both SNS and PNS neurotransmitters and therefore ANS activity can directly influence the endothelium ^{218, 417}. As well as this direct influence, the ANS can also influence the endothelium indirectly through the release of neurotransmitters from its nerve terminals which influence the release of vasoactive

factors from the endothelium ⁴¹⁸. Conversely the endothelium can also directly influence ANS function, with the increased release of NO from the endothelium being found to inhibit the vasoconstricting effects of the SNS ^{418, 419} and the increased release of ET-1 from the endothelium being found to increase the vasoconstricting effects of the SNS ⁴²⁰. Both the ANS and the endothelium therefore interact with each other under normal conditions to ensure optimum vascular tone is achieved.

It is possible that, due to this close working relationship between the ANS and the vascular endothelium, if a dysfunction occurs in one system then the other system may also be affected. A number of diseases, including diabetes, cardiovascular disease, hypertension and congestive heart failure, have been associated with both abnormalities of ANS regulation ⁴²¹⁻⁴²³ and abnormalities of endothelial function ⁴²⁴⁻⁴²⁶. Whether a dysfunction in one system may have driven a dysfunction in the other, or whether both systems have developed a dysfunction independently as part of the disease process, for example due to the effects of factors such as oxidative stress, aging or insulin resistance, is difficult to determine however.

In patients with congestive heart failure, studies have found that endothelin levels correlate negatively with some measures of HRV ³¹² and in patients with diabetes associations have been found between lower HRV and higher levels of von Willibrand factor, a marker of endothelial dysfunction ⁴²⁷. No such data has been collected for patients with glaucoma to date and it is therefore unknown whether similar coexisting dysfunctions are occurring in these patients. Determining this would provide important information that could assist in determining the aetiology of GON.

1.7 The Role of Systemic Blood Pressure

Variations in systemic BP are among the vascular risk factors implicated in the development of GON ⁴²⁸. Regulation of BP is globally controlled by the ANS and follows a normal circadian rhythm ⁴⁰⁸. Due to close relationship between BP, IOP and OPP (equation 1.1), abnormal fluctuations in BP can significantly impact the ocular circulation and in the presence of disturbed autoregulation, due to, for example, PVD syndrome, endothelial dysfunction and/or ANS dysfunction (see section 1.6), the impact of abnormal BP would be increased. Both hypertension and hypotension have been linked to the development of GON and the evidence surrounding their involvement along with an outline of the normal systemic BP physiology is given in the following sections.

1.7.1 Systemic Blood Pressure - Background

BP, in basic terms, defines the force exerted by the blood against the vessel wall and is determined by both the volume of blood in the vessel and the distensibility of the vessel wall. It is commonly described in terms of systolic (SBP) and diastolic (DBP) blood pressure, with systolic being the highest measured pressure, corresponding to ventricular contraction and diastolic being the lowest measured pressure, corresponding to ventricular relaxation and refilling. It is measured in milligrams of mercury (mmHg) from the brachial artery in the upper arm using a sphygmomanometer. The most recent World Health Organisation (WHO) and International Society of Hypertension guidelines, which are based on the 1999 publication for the classification of hypertension, are outlined in table 1.5.

Category	Systolic (mmHg)	Diastolic (mmHg)
Optimal	<120	<80
Normal	120-129	80-84
High Normal	130-139	85-89
Hypertension:		
Grade 1 (mild)	140-159	90-99
Grade 2 (moderate)	150-179	100-109
Grade 3 (severe)	≥ 180	≥ 110
Isolated systolic hypertension	≥ 140	< 90

Table 1.5 Classification of hypertension (World Health Organisation & International society of Hypertension)

Another term used to describe BP is 'mean arterial blood pressure' (MABP), which was defined in equation 1.2. MABP defines the average BP over a single cardiac cycle and is constantly monitored and regulated by the body through control of cardiac output and total peripheral resistance. The regulation of MABP by the ANS was discussed in section 1.6.4.4.

1.7.1.1 Normal Circadian Rhythm

BP follows a distinctive circadian rhythm characterised by a decline in both SBP and DBP during sleep, reaching their lowest point between 2.00 am and 4.00 am, followed by a transient spike in arterial pressure in the early morning, soon after waking, which corresponds with the peak occurrence of cardiovascular incidents ⁴²⁹⁻⁴³¹.

As indicated by this circadian rhythm, it is normal to observe a physiological dip in BP nocturnally. This dip is related to the reduced need for oxygen and nutrients in peripheral tissues during sleep and is brought about by a number of factors including, the normal decline in sympathetic nervous system activity over night, translocation of blood to the peripheral circulation, resulting in a lower venous return, and the effects of the recumbent posture taken on during sleep. The nocturnal dip in BP is traditionally categorised as being either a physiological dip, a non-dip or an excessive

dip ^{429, 432, 433}. A drop in BP of around 10-20% of the average daytime level is considered physiological and occurs in approximately 2/3 of the healthy population ⁴³⁴. Of the remaining population, those with a nocturnal dip in BP of less than 10% are classified as non-dippers and those with a nocturnal dip in BP of greater than 20% are classified as extreme dippers ⁴³³.

Non-dippers tend to have an increased risk of cardiovascular morbidity, myocardial ischemia and cerebrovascular damage such as stroke, haemorrhages and thrombosis ^{435, 436}. This is likely to be due to their sustained exposure to high BP levels over the 24 hour period compared to physiological dippers and extreme dippers ⁴³⁴. Extreme dippers on the other hand tend to have an increased risk of ischemic conditions such as nocturnal myocardial infarction and silent cerebrovascular damage, as significantly reduced BP levels can have a significant effect on perfusion levels.

In order to fully assess an individual's BP profile (or dipping status), 24 hour ambulatory BP monitoring (ABPM) is required. The superiority of 24 hour ABPM in comparison to conventional methods of BP assessment has been well demonstrated with regard to enhanced prediction of cardiovascular mortality and ability to overcome the so called 'white coat syndrome', whereby BP readings are recorded artificially higher in a clinical setting ⁴³⁷⁻⁴⁴⁰. The use of ABPM to evaluate BP in this thesis is discussed further in Chapter 3: Subjects and Methods, section 3.3.4.5.

1.7.2 Hypertension and glaucoma

Whilst the presence of hypertension has been associated with the development of GON, the evidence is somewhat variable. Some studies have identified hypertension as a potential risk factor for both development and progression of POAG ^{441, 442}, especially if poorly controlled ⁴⁴³. Furthermore other studies have found that patients

with POAG in particular, tend to have slightly higher BP than controls ⁴⁴⁴⁻⁴⁴⁸ suggesting a possible causative link. In contrast to these findings however a number of other studies have found either no association between hypertension and glaucoma ^{307, 449}, or a positive association, whereby hypertension has been suggested to have a protective against the development of glaucoma. Leske et al ²³, for example, found that those with systemic hypertension at baseline had half the relative risk of developing glaucoma after four years compared to those without hypertension and similarly Leske et al ²⁴ found that those with higher SBP at baseline had a significantly lower risk of disease progression and these findings were then also confirmed by Leske et al ¹⁵.

It is clear that the evidence linking hypertension and glaucoma is therefore very variable, with both positive and negative associations having been identified. On the whole, the literature suggests that hypertension should therefore not be considered to be a significant risk factor in the development and progression of glaucoma in the majority of patients. However, on saying that, with regard to circadian rhythm, there is evidence to suggest an association between progression of glaucoma and having an absent or reduced nocturnal dip in BP (less than 10%) i.e. having sustained exposure to high BP levels nocturnally may increase the risk of developing glaucoma ^{433, 450-452}. Tokunaga et al ⁴³³ hypothesised that this may be associated with impaired microcirculation around the ONH, in the form of increased peripheral resistance and impaired blood flow, as a result of the increased BP and an increased production of free radicals or other toxic substances, however more research would be needed to confirm this.

1.7.3 Hypertension and IOP

There is some evidence to suggest that a positive correlation exists between systolic and diastolic BP and level of IOP introducing the possibility that hypertension may

contribute to the development of POAG through the elevation of IOP ^{22, 306, 453, 454}. The magnitude of change in IOP with increasing BP however is very small, with a 10mmHg increase in SBP leading to only around a 0.2 to 0.44 mmHg increase in IOP ⁴⁵⁵. Nevertheless theories to explain the associations between BP, IOP and the development of POAG have been put forward, relating to the activities of the ANS and the renin-angiotensin system, both of which have dual control of BP and IOP regulation and to alterations in choroidal volume which can occur in response to elevated BP and impact on IOP ^{401, 456-458}. Regardless of this, due to the weak relationship, the role that elevated BP plays in increasing IOP is considered unlikely to make a significant contribution to the disease process.

1.7.4 Hypotension and glaucoma

The evidence linking hypotension and glaucoma is more consistent and the majority of studies suggest that hypotension, particularly nocturnal hypotension should be considered an important risk factor for the development and progression of glaucoma, particularly with regard to NTG. The Early Manifest Glaucoma trial for example identified low SBP as a risk factor for the progression of glaucoma ²⁴ and a number of further studies have found nocturnal BP levels to be significantly lower in both NTG and progressive glaucoma patients ^{54, 130, 459-462}, suggesting a possible causative link. Furthermore, a large number of studies have found associations between large nocturnal dips in BP (>20%) and the progression of glaucoma ^{313, 433, 450, 463-466}, and increased variability in nocturnal BP parameters and the presence of NTG ^{53, 467, 468}.

The mechanisms by which nocturnal hypotension and large nocturnal BP dips may lead to the development and progression of glaucoma is not entirely clear; however it is thought to largely relate to the impact that reduced BP has on OPP. Indeed reduced OPP has been identified as an independent risk factor for the development

of GON (section 1.6) and, as equation 1.1 shows, in the presence of reduced BP, OPP would also be reduced. It could be hypothesised therefore that such a reduction in OPP, if combined with disturbed autoregulation, occurring either as a component of a PVD syndrome, endothelial dysfunction or as a result of disturbed ANS function affecting the normal circadian rhythm, could result in ischemic damage of the ONH and subsequent development of GON in at risk individuals, via RRI. This hypothesis is supported by a study conducted by Gherghel et al ⁴⁶⁹ which found that glaucoma patients with marked nocturnal dips in BP also had altered retrobulbar blood flow parameters. Furthermore, as IOP also follows a circadian rhythm, with levels being found to be highest nocturnal IOP could further reduce OPP beyond the capacity of the available autoregulatory system and contribute to the development of ischemia and optic nerve damage in susceptible individuals.

An important factor to consider when exploring the presence of large nocturnal dips in BP is the role of antihypertensive medications. Indeed it is proposed that the increased risk of GON development may be associated with, or exacerbated by, the use of antihypertensive medications, which whilst effectively reducing diurnal BP, may lead to nocturnal hypotension ^{433, 470, 471}, however the evidence supporting this association is variable. There are a number of studies, for example, which have demonstrated an association between the use of antihypertensive medications and glaucoma ^{307, 470, 471}, however a number of other studies have found either no association or that antihypertensive medications may actually have a protective effect against the development of glaucoma ^{23, 448, 472}. The evidence is therefore clearly variable however it is generally considered worthwhile to consider the possible role that antihypertensive medications may be playing on glaucoma, particularly in those patients who are progressing.

1.8 The Role of Cardiovascular Risk Factors

The presence of cardiovascular disease and structural changes to the vascular wall have been variably linked to the development and presence of glaucoma and could potentially contribute to the alterations in vascular function and OBF regulation observed in glaucoma patients. The following sections outline the anatomy and physiology of the cardiovascular system and go on to discuss the role that systemic arterial stiffness and cardiovascular disease may play in the development or progression of glaucoma.

1.8.1 Physiology of the Cardiovascular System

The principle function of the cardiovascular system is to maintain adequate blood flow to all tissues, ensuring their oxygen and nutrient demands are met and waste products removed. Blood flows through the cardiovascular system primarily as a result of the pressure produced by the contraction of the heart ventricles, referred to as the systemic BP ⁴⁷³. Both systemic BP and HR are constantly regulated and controlled by the ANS in accordance with the hemodynamic needs of the body. The background and relevance of systemic BP and the ANS with regard to cardiovascular physiology is outlined in the following sections.

1.8.2 Anatomy of the Cardiovascular System

The cardiovascular system is composed of the heart and circulatory blood vessels and plays a primary role in the transport of materials to and from all regions of the body, including the eye and brain. On leaving the heart, the blood is distributed between two principle cardiovascular pathways, namely the pulmonary circuit and the systemic circuit ⁴⁷³. An overview of these circuits with relevance to this thesis is given in the following sections.

1.8.2.1 The Heart

The human heart resembles a double pump system consisting of four chambers, namely the right and left atria and the right and left ventricles (figure 1.12). The atria function primarily as reservoirs, collecting blood returning in the veins and feeding it into the ventricles. The ventricles are the major muscular pumps of the heart, ejecting blood from the heart and into the circulatory system. Each ventricle serves as a pump for a specific cardiovascular pathway. The right ventricle pumps deoxygenated blood received back from the systemic circulation into the pulmonary circuit. Following the removal of carbon dioxide and the addition of oxygen, the left ventricle then receives the oxygenated blood from the systemic circuit via the aorta ⁴⁷³.

1.8.2.2 Coronary arteries

The blood supply to the metabolically active cardiac muscle, which lines the wall of the heart, is provided by the coronary arteries. The left coronary artery originates on the left side of the aorta and separates into three branches which supply much of the anterior wall of the heart and most of the left ventricle. The right coronary artery originates on the right side of the aorta and separates into two main branches which supply most of the wall of the right ventricle. The coronary arterial blood supply to the heart muscle can be described as an 'end circulation' as it represents the only source of blood supply to the myocardium, as such any blockage or dysfunction of the coronary arteries, for example through atherosclerosis, poses a critical threat to heart function, and could potentially lead to heart attack and/or myocardial infarction ⁴⁷⁴.



Figure 1.12: Diagrammatic representation of the heart

1.8.2.3 Systemic Circulation: Carotid, Brachial and Radial Arteries

The systemic circulation refers to the flow of blood from the left ventricle of the heart to the tissues of the body and back to the right atrium. All of the arteries which make up the systemic circulation branch from the aorta, which itself can be considered in three parts, namely the ascending aorta, the aortic arch and the descending aorta. The blood supply to the head and neck, which ultimately supplies the vascular networks of the eye and brain, originate from the aortic arch. Of primary interest to this thesis are the right and left common carotid arteries. The right common carotid artery originates from the first branch of the aortic arch, referred to as the brachiocephalic artery, whereas the left common carotid arteries extend superiorly within the corresponding parts of the neck before branching to form the right and left, internal and external carotid arteries ⁴⁷³ (figure 1.13).

Other systemic arterial branches referred to in this thesis include the brachial artery and the radial artery. The brachial arteries are located in the upper arm on both sides and are the common site for BP measurement. They arise from the subclavian artery, which itself arises either directly or indirectly from the aortic arch. The brachial artery branches at the elbow to form the radial artery which supplies the forearm and the hand and is common site for taking a pulse.



Figure 1.13 Diagrammatic representation of the arteries of the neck

1.8.2.4 Structure of the Arterial Walls

On leaving the heart the blood flows from the aorta, into the systemic arteries, which then repeatedly branch throughout the body forming progressively smaller arterioles and capillaries. As the vessels branch the structure of the arterial wall undergoes a gradual transition from high elasticity, low smooth muscle cell (SMC) content to high SMC content and low elasticity. The walls of the arteries and arterioles are composed of three layers referred to as the tunica intima, the tunica media and the tunica adventitia, as depicted in figure 1.14. The tunica intima is collective name for the innermost layers of the vessel wall, encompassing the endothelium and connective tissue, the tunica media is the middle and thickest layer and the tunica adventitia is the outermost layer ⁴⁷³.



Figure 1.14 Diagrammatic representation of the artery wall

The structure of the smallest vessels, referred to as the capillaries, is somewhat different from that of the arteries and arterioles in that their wall consists only of a thin layer of endothelial cells, accompanied by an underlying basement membrane. This structure facilitates the exchange of materials between the capillaries and surrounding tissues.

1.8.2.5 Atherosclerosis

The arterial walls of the circulatory system undergo changes as they age, the most significant of which is a loss of elasticity. This loss of elasticity or 'hardening of the arteries' is termed arteriosclerosis and primarily effects the large elastic arteries such as the aorta and carotid arteries. A specific form of arteriosclerosis, termed atherosclerosis, refers to the hardening of the artery wall as a result of the deposition of fatty materials and plaque formation on the inner surface of the wall. This deposition leads to narrowing of the arterial wall, increasing resistance to blood flow and impairing blood circulation ⁴⁷³. It also increases the risk of vessel occlusion through thrombus formation. The severity and rate at which atherosclerotic changes develop has been linked to factors such as lack of exercise, obesity, smoking and high cholesterol diets but may also have a genetic influence ⁴⁷⁵. The detection of atherosclerotic vessel changes and the adjustment of modifiable risk factors is recommended in the treatment and prevention of cardiovascular disease. A summary

of the techniques available for detecting and/or monitoring atherosclerotic changes is

given in table 1.6

Method	Advantages	Disadvantages
Angiography	Original technique	Detects only severe narrowing not the underlying atherosclerotic
	stenosis	Invasive
Stress testing	Allows a general assessment of physical condition of patient	Detects only severe narrowing not the underlying atherosclerotic disease Invasive
Anatomic methods:		
 Coronary Calcium scoring by CT Carotid intima-media thickness measurement by ultrasound Intravascular ultrasound 	Directly measure aspects of the actual atherosclerotic disease process itself Allow detection of atherosclerotic changes before patient becomes symptomatic Allow tracking of disease progression	More expensive Can be invasive
Physiological methods:		
Lipoprotein subclass analysis	Cheap Safe	Don't allow the state of the disease to be quantified
HbA1c analysisC-Reactive protein	Modification of abnormal parameters can slow progression	Don't allow progression of disease to be tracked
analysis (hsCRP)Homocysteine analysis		

 Table 1.6: Overview of the techniques available for the detection of atherosclerosis

1.8.3 Cardiovascular risk and Glaucoma

There is evidence to suggest a higher incidence of cardiovascular disorders such as coronary artery disease, cardiac arrhythmias, atrial fibrillation, congestive heart failure and hemodynamic crisis in NTG patients compared to the average population ^{142, 350, 444, 476}; however this is not confirmed by all studies ^{477, 478}. Furthermore POAG patients have been identified as having a higher cardiovascular risk than control patients ⁴⁴⁶ and cardiovascular disease history has been identified as a risk factor for POAG

progression ²⁴. Additionally an increased cardiovascular mortality has been described in those previously diagnosed with open angle glaucoma ⁴⁷⁹.

Whilst these findings obviously support the role that systemic vascular disease may play in the development of glaucoma, other studies directly assessing arterial stiffness or the presence of arteriosclerosis and its risk factors in those diagnosed with glaucoma have given more mixed results, with some finding strong associations ^{307, 480-483} and others finding no association at all ^{318, 484, 485}. Furthermore risk factors classically associated with cardiovascular disease and arteriosclerosis, such as obesity, smoking and hypercholesterolemia have not been found to show any strong links to the development of GON ⁴⁸⁵⁻⁴⁸⁷. The role that systemic arterial stiffness and cardiovascular disease may play in the development or progression of glaucoma is therefore still uncertain.

Increased arterial stiffness, an independent predictor of cardiovascular disease and stroke ⁴⁸⁸, can arise as a result of an age related breakdown of the elastin structures in the arterial walls, damage to the endothelium/smooth muscle system or an increase in MABP. Interestingly these are all factors that have been individually associated with GON development. Furthermore endothelial dysfunction has also been identified as the first detectable change in an atherosclerotic vessel so it is not unreasonable to hypothesise that, due to the fairly strong links between glaucoma and endothelial dysfunction; vessel wall changes such as atherosclerosis could contribute towards vascular dysfunction and GON development in certain individuals. Interestingly a recent study by Oettli et al ²⁹⁵ was able to identify increased arterial stiffness at the retinal level in NTG patients and found this correlated with level of glaucomatous damage, suggesting a possible role for both ocular and systemic arterial wall changes in the development of GON.

The possibility that cardiovascular disease and arterial stiffness may contribute towards the development of GON can therefore not be ruled out however further research is needed to elicit the exact nature of their association. Systemic arterial stiffness and cardiovascular risk can be assessed by means of central pulse pressure, ultrasound measurement of distensibility and compliance, pulse wave velocity assessment, pulse wave analysis (PWA) techniques and carotid artery intima-media thickness (IMT) measurement ⁴⁸⁹. The most commonly used indices of arterial stiffness are detailed in table 1.7. Both PWA and IMT were conducted in this thesis and are discussed further in Chapter 3: Subjects and Methods

Indices of arterial stiffness	Definition
Arterial distensibility	Relative diameter or area change for a pressure increment
Arterial compliance	Absolute diameter or area change for a given pressure step at fixed vessel length
Elastic Modulus	The pressure step required for 100% stretch from resting diameter at fixed vessel length
Pulse wave velocity (PWV)	Speed of travel of pulse along an arterial segment
Pressure augmentation	Increase in aortic or carotid pressure after the peak of blood flow in the vessel
Young's Modulus	Elastic modulus per unit area; pressure step per square cm required for 100% stretch from resting length

Table 1.7: Overview of commonly used indices of arterial stiffness 490

Oxidative stress and the concept of reperfusion injury 1.9

Regardless of the cause, the final common pathway in the development of GON remains the apoptotic loss of RGCs, tissue remodelling and excavation of the ONH. Increasing evidence suggests that oxidative stress, in combination with the more extensively researched causative factors discussed in previous sections, could also be a contributing factor in the development of GON.

1.9.1 Background

In simple terms, oxidative stress refers to an imbalance between the production of oxidants and antioxidants, in favour of the former, with the potential for damage ⁴⁹¹. In biochemical terms 'oxidation' is a harmful process which refers to the loss of electrons and/or gain of oxygen by a molecule and is brought about by the action of so called 'oxidising agents' or 'oxidants' ⁴⁹². Oxidants take the form of either free radicals, which are usually highly reactive and have one or more unpaired electrons, or non-radical species and can be derived from either oxygen (reactive oxygen species, ROS) or nitrogen (reactive nitrogen species, RNS), as outlined in table 1.8 493

Features
Oxygen molecule in its excited state. Generated through photodynamic reactions ⁴⁹⁴
Generated continuously by normal cellular processes from the reduction of O ₂ . Converted to hydrogen peroxide by superoxide dismutase enzyme ⁴⁹³
H ₂ O ₂ only weak oxidising agent, but forms the more potent OH ⁻ on crossing cell membranes and reacting with copper/iron ions ⁴⁹³
Features
NO is a free radical and reacts strongly with superoxide (O_2) to form ONOO. NO has both oxidant and antioxidant properties. ⁴⁹⁵

ROS are potentially harmful and can be generated as a by-product of normal O₂ metabolism by a cell. Therefore, although O₂ is essential for maintenance of normal life in aerobic organisms, it also has the potential to cause harm ⁴⁹¹. The eye, due to its unique constant exposure to light, atmospheric oxygen, environmental chemicals and physical abrasion is particularly susceptible to ROS formation ³⁶⁶. Under optimal conditions the rate and magnitude of ROS formation is balanced by the rate of ROS elimination through the action of antioxidants and minimal tissue damage occurs ⁴⁹². Such antioxidants act by significantly delaying or preventing the oxidation of a substrate and form part of the body's natural defence mechanism against the action of ROS ⁴⁹¹. They take a variety of different forms and can be categorised as either enzymatic or non-enzymatic. These are summarised in table 1.9.

Enzymatic antioxidants	Features	
Superoxide Dismutase (SOD)	Scavenges the superoxide anion (O_2) : catalyses the dismutation of O_2 to H_2O_2 and water ⁴⁹⁶	
Catalase (CAT)	Important role in removing H_2O_2 : catalyses H_2O_2 to water and oxygen ⁴⁹⁷	
Glutathione Peroxidase (GPX)	Important role in removing H_2O_2 : reduces H_2O_2 to water by oxidising glutathione (GSH). Oxidised from of glutathione (GSSG) is then catalysed by glutathione reductase ⁴⁹⁸	
Non-enzymatic antioxidants	Features	
Glutathione (GSH)	Strong antioxidant properties - donates electrons readily converting it to its oxidised from GSSG. GSH/GSSG ratio of a cell is good indicator of cellular redox balance ^{499, 500}	
Heat shock proteins	Produced by cells in response to stress. Regulate the function of other proteins ⁵⁰¹	
Transferrins	Bind free iron or metal irons in forms that will not stimulate free radical reactions ⁵⁰²	
Haptoglobins	Haemoglobin-binding proteins – decrease effectiveness of lipid peroxidation ⁵⁰³	

Table 1.9 Summary of enzymatic and non-enzymatic antioxidants

If the oxidant/antioxidant balance is disturbed so that the level of oxidants exceeds the antioxidant capacity a state of oxidative stress is reached. The body is able to respond to a small level of oxidative stress by increasing its defence and repair mechanisms ⁵⁰⁴. However under conditions of sustained oxidative stress or excessive

ROS production the damage induced may exceed the capacity of not only the antioxidants but also the bodies repair mechanisms, leading to more permanent macromolecular damage. Proteins, lipids, sugar residues and DNA are at risk of damage, potentially resulting in growth arrest, growth modulation, cell death and disease formation ⁴⁹³. Indeed oxidative stress has been implicated as a causative factor in a number of pathological conditions including diabetes ⁵⁰⁵, atherosclerosis ⁵⁰⁶, rheumatoid arthritis ⁵⁰⁷, malignant disease ⁵⁰⁸, human immunodeficiency virus (HIV) ⁵⁰⁸ and age-related macular degeneration (AMD) ⁵⁰⁹ and the bodies capacity for dealing with oxidative stress shows a normal age-related decline ⁵¹⁰.

A determination of the presence of oxidative stress can either be made through evaluation of oxidative stress related tissue damage, such as increased DNA breaks and increased proteosome activity, or through the more common method of analysing the circulating plasma oxidant/antioxidant balance which can be disrupted in the presence of oxidative stress. One such antioxidant is glutathione (GSH). GSH is the major low-molecular mass thiol compound in plants and animals and is among the most efficient substances that cells and tissues can use in their defence against oxidative stress ⁵¹¹. It is a tripeptide consisting of glycine, cysteine and glutamic acid, as depicted in figure 1.15 ⁵¹² and is active in the eye. As an antioxidant GSH can prevent the devastating effects of ROS either directly through its oxidisation, or indirectly by maintaining other cellular oxidants in a functional state ⁵¹³.



Figure 1.15: Chemical structure of the tripeptide glutathione

Glutathione is present in two forms in healthy cells and tissues, namely a reduced form, glutathione (GSH), and an oxidised form, glutathione disulphide (GSSG) and GSH is constantly being broken down to GSSG and re-synthesised ^{512, 514} (equation 1.3). As such, GSSG, which is formed on oxidation of GSH following its interaction with harmful ROS, accounts for only around 10% of the total glutathione pool of a healthy cell, with GSH accounting for around 90% ⁵¹⁴. Under conditions of oxidative stress however, ROS levels are raised and hence the production of GSSG is increased. As such determining the GSH:GSSG ratio, or glutathione status of a cell is considered a good indicator of cellular redox imbalance and oxidative stress ⁵¹⁵, especially as maintaining an optimal GSH:GSSG ratio is critical for cell survival. A reduced GSH:GSSG ratio would be considered indicative of oxidative stress ⁵¹⁶

$$2 GSH + H_2O_2 \xrightarrow{GPase} GSSG + H_2O$$

$$GSSG + NADPH + H^+ \xrightarrow{GRase} 2 GSH + NADP^+$$

Equation 1.3: GSH: reduced glutathione; H_2O_2 : hydrogen peroxide (ROS); H_2O : water, GSSG: oxidised glutathione; NADPH, NADP⁺: nicotinamide adenine dinucleotide phosphate. GPase: glutathione peroxidise, GRase: glutathione reductase

1.9.2 Oxidative Stress and Endothelial Dysfunction

Over recent years oxidative stress has been increasingly linked to the development of endothelial dysfunction and has been implicated as a causative factor in the pathogenesis of multiple associated diseases, including neurodegeneration and atherosclerosis ⁵¹⁷⁻⁵¹⁹. The predominant pathway by which oxidative stress is thought to promote endothelial dysfunction is through reduction in the bioavailability of the vasodilator NO, resulting primarily from its increased reaction with ROS such as O_2^{-1} under conditions of oxidative stress ⁵¹⁹. Disturbances in the functioning of NOS, or disruptions to the interactions between NO and the vSMCs, both of which are induced by an increased presence of ROS, can however additionally contribute to the reduced bioavailability of NO ^{519, 520} in oxidative stress. Interestingly the modification or correction of oxidative stress has been shown to alleviate induced endothelial dysfunction ⁵²¹⁻⁵²³. As such, the detection and consideration of oxidative stress and the coexistence of endothelial dysfunction is important in the management of associated conditions, including neurodegenerative disease. The mechanisms by which oxidative stress and endothelial dysfunction may contribute to ocular neurodegeneration, in the form of glaucoma, are therefore discussed fully in the following sections.

1.9.3 Oxidative stress and glaucoma

The RGCs and axons of the ONH are considered particularly susceptible to the effects of oxidative stress due to their direct exposure to light, their high proportion of polyunsaturated fatty acids and their very high levels of O₂ consumption, resulting from their lack of myelin sheaths and high concentration of mitochondria. Indeed multiple studies have found characteristic indicators of oxidative stress in glaucoma patients, including an increased number of DNA breaks ⁵²⁴, an upregulation of ET-1 ³⁸⁶ and metalloproteinase-9 (MMP-9) ⁵²⁵ and increased proteosome activity ⁵²⁶.

Furthermore, low plasma levels of the antioxidant glutathione (GSH) and total glutathione (t-GSH) have been demonstrated in glaucoma patients compared to controls ²⁷ as well as a decreased activity of various other antioxidants and of the total serum antioxidant status ^{527, 528}, indicative of an increased oxidative burden. It is therefore no surprise that oxidative stress has been linked to the development of GON and as well as being implicated in the so called vascular theory of the disease it has also been linked to the development of GON through mechanical means as discussed in the following section.

1.9.3.1 Oxidative stress and IOP

With regard to the mechanical theory, oxidative stress has been implicated to play a causative role in the elevation of IOP and subsequent development of GON. It is proposed that increased levels of oxidative stress in the aqueous humour, occurring as a result of light catalysed reactions, metabolic pathways or inflammation, leads to cellular loss and alterations of glycoprotein structure in the extracellular matrix of the trabecular meshwork (TM) cells. This is followed by an alteration in TM function, impaired aqueous outflow and increased IOP ^{28, 529, 530}. Increased IOP, as discussed in section 1.9.4, can then contribute to the development of GON through either direct mechanical insult of the optic nerve fibres, inhibition of retrograde neurotrophin support to the RGCs, or alternatively through acting in combination with vascular alterations at the ONH.

Multiple studies have demonstrated depletion of total antioxidant potential and enhanced antioxidant activity in the aqueous humour of glaucoma patients, suggestive of increased oxidative burden ^{527, 531-534}, as well as the presence of oxidative damage in the TM cells ^{535, 536}, which could potentially impede aqueous outflow. Oxidative stress can therefore be considered a potential risk factor for the development of POAG in particular ³⁶⁶.

1.9.3.2 Oxidative stress and reperfusion injury

The vascular concept of glaucoma centres around the presence of unstable blood flow and/or altered autoregulation at the ocular and systemic level in susceptible patients. Unstable blood flow is characterised by a repeating cycle of normal then reduced perfusion and could potentially lead to cellular damage through RRI and oxidative stress. RRI refers to the damage to a tissue caused when blood supply returns after a period of ischemia ¹⁴³. During ischemia, the absence of oxygen and nutrients from a cell impairs electron transport in the mitochondria, resulting in inefficient energy production and the presence of a number of spare electrons. When perfusion then returns to normal these spare electrons react with the now plentiful supply of O₂ molecules, leading to the formation of damaging ROS ⁵³⁷. Mitochondria are abundant in the RGC axons of the ONH putting it at high risk of damage by ROS and it is hypothesised that, if mild RRI continues over a sustained length of time chronic oxidative stress, endothelial dysfunction, ONH damage and the development of GON would occur, regardless of antioxidant activity ^{141, 366}.

Apoptosis, referring to programmed cell death without necrosis ⁵³⁸, has been strongly implicated as the ulitmate pathway for RGC loss in glaucoma ^{100, 539, 540}. The exact mechanism by which RRI and oxidative stress may contribute to this process however is still unclear. All cells have a genetically predetermined programme of death that is part of the normal cell life cycle and enables the maintenance of homeostasis and removal of pathologically altered cells. If this programmed cell death is activated inappropriately however, then pathological loss of cells and the development of associated disorders can occur. Both RRI and the subsequent activation of astrocytes are thought to play an important role in initiating inappropriate apoptosis of RGCs in glaucoma ⁵⁴¹.

Astrocytes are found in the ONH and provide biochemical support to the RGC axons ⁵⁴¹. They have a high susceptibility to ROS and on becoming activated change appearance and start producing a variety of abnormal molecules, including ET-1 and NO, which can alter the microenvironment of the cell ¹⁴⁴. Diffusion of NO from the astrocytes into the neighbouring RGC axons, where levels of ROS such as O₂ are high as a consequence of RRI, can lead to the formation of very damaging peroxynitrate (ONOO⁻) ⁵⁴². It is proposed that subsequent diffusion of both O_2^- and ONOO⁻ within the RGC axons, towards the retina and LGN could then trigger the apoptotic loss of RGCs ^{141, 543, 544}. In parallel to the loss of RGCs, ONH tissue remodelling is thought to occur as a consequence of not only mechanical force, but also an active biological process including the effects of MMP-9¹⁴¹. Indeed diffusion of ET-1 and MMP-9, which are present in increased concentrations in the bloodstream of glaucoma patients, from the choroid into the surrounding ONH, can result in vasoconstriction and weakening of the BBB ⁵⁴⁵, with an increased risk of subsequent damage. In addition to via RRI, astrocyte activation may also arise as a direct result of the presence elevated IOP, through either its direct mechanical effect, or through its contribution to the apoptosis of RGC axons via disruption of axoplasmic transport and deprivation of essential growth factors ^{546, 547}. Additionally raised ET-1 levels, which may occur as a result of factors such as raised IOP, hypoxia or endothelial dysfunction, could also trigger the activation of astrocytes and initiate the apoptosis of RGC and their axons in the presence of oxidative stress ¹⁴⁴

All in all the concept of RRI and oxidative stress, combined with the subsequent activation of astrocytes and apoptotic loss of RGCs is the most widely accepted current pathogenetic concept for the development of GON. Indeed further evidence for the potential role of RRI in the pathogenesis of GON comes from its associations with sleep apnea and reversible shock-like states, both of which are conditions whose pathologies are linked to the presence of RRI ^{548, 549}. It is the ocular and systemic

vascular mechanisms which contribute to this pathogenesis which is still the subject of debate.

1.10 Overall summary of the current pathogenetic concept of glaucoma

From the evidence discussed in the previous sections it is clear that open angle glaucoma is a disease of multifactorial origin and that vascular alterations appear to play a key part in its development. Figure 1.16 summarises the current pathogenetic concept of GON development, the mechanisms thought to be involved and their interactions. Current thinking suggests a synergistic involvement of both mechanical and vascular factors in the aetiology of GON however the extent of involvement and coexistence of the many implicated factors is still uncertain and may vary between individuals



Figure 1.16 Summary of the current pathogenetic concept of glaucomatous optic neuropathy development

1.11 Cerebral Neurodegeneration: Alzheimer's disease

1.11.1 Dementia overview

AD is a form dementia. Dementia describes a progressive decline in cognitive function in a previously unimpaired person, beyond that which is expected through normal aging ⁵⁵⁰ and is a broad term encompassing a large range of different cognitive impairments ⁵⁵¹⁻ ⁵⁵³. The most common form of dementia is AD, which accounts for approximately 80% of all cases and is the main focus of this thesis. An overview of the subcategories of dementia is given in table 1.10

Dementia Classification	Examples	
Primary degenerative disorders (progressive with no associated disease or cause)	Alzheimer's disease Lewy body dementia Pick's disease (fronto-temporal dementia)	
Vascular dementias (cerebrovascular disorders with dementia)	Multiple-infarct syndrome (cortical) Small vessel dementia (subcortical ischemic vascular disease) Strategic infarct dementia	
Secondary dementias (secondary to other physical disease, infection or injury)	Huntington's chorea related Parkinson's disease related HIV related Creutzfeldt-Jakob disease related	
Reversible/Treatable dementia	Normal pressure hydrocephalus related Brain tumour related Hypothyroidism related Vitamin B-12 deficiency (Korsakoff's syndrome) Neurosyphilis related	

Table 1.10: Overview of the subcategories of dementia

1.11.2 Clinical presentation, diagnosis and treatment of AD

AD is a progressive neurodegenerative disorder, first described by Alois Alzheimer in 1906 ⁵⁵⁴, which affects over 35 million people worldwide ² and increases in prevalence after the age of 65 ⁵⁵⁵. It is characterised clinically by a gradual decline in cognitive function to include memory impairment, confusion, disorientation, mood swings and in the later stages, speech abnormalities and a gradual loss of bodily function ^{556, 557}. From

a neuropathological point of view two major hallmark lesions have been identified in the brain of AD sufferers, namely the formation of extracellular beta-amyloid senile plaques and the formation of intracellular neurofibrillary tangles, as a result of abnormal tau protein phosphorylation ⁵⁵⁸⁻⁵⁶¹. These classic pathological lesions are thought to contribute towards the characteristic neuronal degeneration and cerebral atrophy which ultimately occurs in the AD brain; however their aetiology is still the subject of debate. A number of different theories for their development have been proposed including, damage through exposure of the CNS to increased mechanical stress ^{463, 562}, inflammation ⁵⁶³, mitochondrial dysfunction ⁵⁶⁴ and oxidative stress ⁵⁶⁵. Of particular interest in this thesis however is the involvement of vascular factors, which have also been strongly hypothesised to play an aetiological role in AD ⁵⁶⁶.

1.11.2.1 Diagnosis

AD predominantly develops sporadically, with only 2.5% of all cases having being deemed to be of genetic origin ^{567, 568}. The need for early and accurate diagnosis of the condition is therefore well recognised, particularly with regard to ensuring optimum treatment, care and prognosis for sufferers, however it still remains a challenge. A definitive diagnosis of AD can only currently be made post-mortem by the histopathological confirmation of the presence of senile plaques and neurofibrillary tangles in the brain. This is in accordance with the National Institute of Neurological Disorders and Stroke-Alzheimer Disease and Related Disorders (NINCDS-ADRDA) working group, which, along with the Diagnostic and Statistical Manual of Mental Disorders, VIth edition (DSM-IV-TR), forms the basis of AD diagnosis in research ⁵⁶⁹.

The NINCDS-ADRDA, first proposed in 1984, is among the most used diagnostic criteria for AD ⁵⁷⁰. It allows the diagnosis of definitive, probable, possible or unlikely AD, as outlined in table 1.11. A diagnosis of probable AD is the maximum that can be made
clinically and requires the presence of progressive memory impairment with involvement of at least one other cognitive domain and the lack of other systemic or brain diseases that may account for the cognitive deficits ⁵⁷⁰. The DSM-IV-TR criteria, published in 2000, also requires the presence of both memory impairment and a disorder in at least one other cognitive domain for the probable diagnosis of AD; however it additionally requires that these features also interfere with social function or activities of daily living ^{569, 571}.

Whilst both the NINCDS-ADRDA and DSM-IV-TR have been validated against gold standards and have demonstrated high levels of sensitivity and specificity in distinguishing those with AD from those without dementia ^{572, 573}, significant advances in the scientific knowledge and understanding of pathogenic events in the disease over recent years has lead to the proposal of revised diagnostic criteria which consider the presence of biomarkers of AD and centre around the detection of early memory impairment ⁵⁶⁹. Current biomarkers include an abnormal cerebral spinal fluid (CSF) beta-amyloid and tau protein profile, structural brain changes on magnetic resonance imaging (MRI), hypometabolism or hypoperfusion changes on positron emission tomography (PET) scanning and the presence of pathogenic gene mutations ^{557, 574}. Determining the presence of biomarkers allows in vivo biological evidence of AD pathology to be gained, however the methods by which they are obtained are often invasive so development of alternative non-invasive techniques would be beneficial.

A large degree of variation exists in the presentation and course of AD between individuals and furthermore the disease is thought to exist in an undiagnosed state for an indeterminate period of time before its symptoms become apparent ⁵⁷⁵. Enhancing the ability to detect AD in its earliest stages has therefore become a research area of intense interest over the last few years due to its obvious benefits with regard to patient prognosis and management. As such several different stages of AD or memory

impairment have now been defined in order to document the progression of the disease. These are outlined in table 1.12 and are variably referred to as mild cognitive impairment (MCI), preclinical AD, prodromal AD and finally AD dementia, which itself can be further separated into mild, moderate and advanced AD ⁵⁶⁹. It is recommended that such further separation of AD dementia into differing degrees of cognitive impairment is made using formal standardised cognitive tests such as the Mini Mental State Examination (MMSE) ^{576, 577}. This is a simple, widely used and validated test which takes only 10 minutes to complete and provides a global assessment of cognitive function. Out of a maximum score of 30, those achieving 25 or higher are generally considered as normal and those scoring less than 10 are considered to have a severe impairment ⁵⁷⁷. A score of around 19-24 is generally considered to indicate mild AD, however as the MMSE result has been shown to be influenced by factors such as age, ethnicity and education, it is recommended that adjustments are made where necessary to account for these factors ^{578, 579}.

Diagnosis	Features
Probable AD	Clinical diagnosis: • Dementia established by clinical examination and documented by the Mini- Mental Test (MMSE) or similar and confirmed by neurophysical tests • Deficits in two or more areas of cognition • Progressive worsening of memory and cognitive decline • Normal level of consciousness • Onset between ages 40-90 • No other possible medical or neurological explanation Supported by presence of: • Progressive aphasia, apraxia and agnosia; impaired activities of daily living; family history of similar disorder; brain atrophy on CT/MRI; normal CSF Other consistent clinical features: • Plateaus in the course of progression • Associated symptoms, such as depression, insomnia, incontinence, illusions, weight loss, emotional or physical outbursts • Seizures in advanced stages • CT normal for age Exclusion factors: • Acute onset • Focal neurological findings • Seizures or gait disorders at early stages of disease
Possible AD	 The presence of dementia with an atypical onset or course that occurs in the absence of any other medical/neuropsychiatric explanation Dementia in the presence of another brain or systemic disease not considered to be the cause of the dementia Gradually progressive severe cognitive deficit in the absence of other identifiable cause (research purposes only)
Definite AD	All the criteria of 'probable AD' plus histopathological evidence from biopsy or autopsy

Table 1.11: NINDS-ADRDA Diagnostic criteria for AD

Stage	Description	
Mild Cognitive Impairment (MCI)	 Subjective memory and/or cognitive symptoms Objective memory and/or cognitive symptoms Normal, unaffected activities of daily living Do not meet criteria for dementia or AD 	
Preclinical AD	 Long asymptomatic period between the first brain lesion and the first appearance of symptoms Later go on to fulfil diagnostic criteria for AD 	
Prodromal AD	 Symptomatic pre-dementia phase Often included within MCI category Symptoms not yet severe enough to meet AD criteria 	
AD dementia	 Symptoms and features sufficient to meet accepted diagnostic criteria for dementia and AD Progresses from mild to moderate and then advanced as symptoms worsen 	

Table 1.12: Clinical stages of Alzheimer's disease 569

As mentioned previously a number of different theories have been proposed for the aetiology of the classic pathological brain lesions associated with AD. Of primary interest in this thesis is the role that vascular factors may play in the development of AD and the links between the ocular and cerebral vasculature. The following sections will discuss the cerebrovascular anatomy along with the current evidence surrounding vascular involvement in AD at both the ocular and systemic level.

1.12 Anatomy and Physiology of the Cerebral Vasculature

The cerebral circulation can be broadly separated into the macro-vascular system and the microvascular system. These will now be discussed in turn.

1.12.1 Macrovasculature

The arterial blood supply to the brain is received is via two major vessels, the internal carotid artery (ICA) and the vertebral artery. These arteries enter the skull at the base of the brain and branch dorsally, spreading over the surface of the cerebrum in the subarachnoid space above the pia mater. The ICA is primarily responsible for the anterior circulation of the brain and gives rise to the anterior and middle cerebral arteries, which supply 80% of the blood that reaches the cerebral hemispheres ⁵⁸⁰. The vertebral arteries, on the other hand, fuse within the cranium to form the basilar artery which is primarily responsible for supplying the posterior circulation of the cerebral hemispheres ⁵⁸⁰.

The anterior and posterior circulatory regions do not act independently of each other and are interconnected by a series of communicating arteries which create the so called circle of Willis at the base of the brain ⁵⁸¹. This network of vessels consists of the right and left ICAs, anterior cerebral arteries, posterior cerebral arteries, posterior communicating arteries and the anterior communicating artery (figure 1.17). This intercommunication between the anterior and posterior regions is advantageous as if blood flow becomes insufficient in either area it is able to be redistributed, via the communicating arteries of the circle of Willis, according to requirement ⁵⁸⁰.



Figure 1.17: Diagrammatic representation of the cerebral macrovasculature

The cerebral macro-circulation is crucial in the conductance of blood flow to the various regions of the brain. The cerebral microcirculation then acts to regulate local blood flow and vascular tone within these regions.

1.12.2 Microvasculature

The cerebral microvasculature consists of a dense three-dimensional network of fine capillaries which act to maintain the BBB and sustain continuous oxygen, nutrient, electrolyte and waste product transfer between the blood and the cerebral tissues ⁵⁸⁰. The density of the capillary network is not uniform throughout the cerebrum and has been shown to be greater in areas of increased metabolic activity, in the grey matter compared to the white matter and in sensory and association centres compared to motor centres ^{580, 582, 583}.

The cerebral capillaries have a specific structure which enables them to fulfil the crucial role of maintaining normal BBB function (figure 1.18). In a similar way to in the ONH (see section 1.2.3.3), the cerebral BBB prevents the passage of all but essential metabolites from the blood into the cerebral tissues, protecting them from the effects of foreign, potentially damaging substances that could be present in the bloodstream as well as the effects of hormones or neurotransmitters that may be active in the rest of the body 90 .





A key feature of the cerebral capillary structure is the tight junctions that exist between the vascular endothelial cells. These form the basis of the BBB by preventing the free passage of metabolites from the blood stream into the cerebral tissues. Additionally the lack of endothelial cell fenestrations and the high level of mitochondria within the endothelial cells also assist in maintaining the structure of the BBB and the functioning of the specific BBB transport proteins, as does the basement membrane surrounding the endothelial cells. The basement membrane is lamellar in structure and has also been identified to provide physical support to the microvessels, control cellular migration, influence endothelial function, promote cell adhesion and protect the brain against circulating proteins ⁵⁸⁴. Another key feature of the cerebral capillary structure is the astrocytic processes which are apposed to the abluminal surface. These act as a link between synaptic activity and the cerebrovascular cells and play a key role in the maintenance of BBB function ⁵⁸⁵ and in the regulation of vascular tone ^{586, 587}.

1.12.3 Regulation of cerebral blood flow

As discussed in section 1.4 the blood flow through any microvascular bed depends on both perfusion pressure and vascular resistance and the maintenance of normal tissue structure and function depends upon a continuous and well regulated blood supply. The cerebral blood flow (CBF), in a similar way to OBF, is regulated according to changes in perfusion pressure (autoregulation) and also to changes in neural activity (neurovascular coupling) so that blood supply can always meet demand ⁵⁸⁸.

1.12.3.1 Autoregulation

Autoregulation, which was discussed in section 1.4.3, refers to the ability of the cardiovascular system to modify vascular resistance in order to allow a constant blood supply to maintained despite variations in perfusion pressure ¹⁵¹. It was first demonstrated in the cerebral circulation by Fog in the 1930s ^{589, 590} and later established by Lassen in 1959 ⁵⁹¹. Whilst the exact mechanisms underlying cerebral autoregulation are still unclear it is known to be triggered by metabolic, myogenic, neurogenic and to a lesser extent, humoral factors, as well as by endothelial derived vasoactive agents ^{588, 592, 593}. These triggers were discussed in detail in section 1.4.3 and act on the cerebral vasculature in a similar manner to that discussed with regard to the ocular circulation. Worth noting is the fact that the cerebral vessels, in a similar way to the choroidal vessels and unlike the retinal vessels, have a rich autonomic innervation which allows maintenance of blood flow in the presence of altered BP or HR. Sympathetic stimulation, although having little effect on regulation of blood flow under normal conditions, attenuates the increase in blood flow observed when BP is raised ^{594, 595} and parasympathetic stimulation acts to increase cerebral blood flow when required ⁵⁹⁶.

1.12.3.2 Neurovascular Coupling

The other means by which CBF is regulated is via neurovascular coupling mechanisms. Neurovascular coupling refers to the ability to regulate blood flow according to changes in neural function and was first identified in the cerebral circulation in the late 1800s ^{597,} ⁵⁹⁸. The normal neurovascular coupling response at the cerebral level is characterised by the rapid dilation of arterioles and capillaries in a restricted brain region in response to a local episode of increased neural activity ⁵⁹⁹. This dilation response leads to a rapid increase in blood flow and oxygen supply to the active brain regions, ensuring the increased metabolic demand is met. The exact mechanisms which drive the neurovascular coupling response are still unclear, however it is thought to involve a complex interplay between neurons, astrocytes and vascular cells (endothelial cells, pericytes and vascular smooth muscle cells) ^{600, 601}, with the vasodilator nitric oxide (NO) having been identified as one of the important mediators of the response ^{602, 603}. The key components of the neurovascular coupling response, namely neurons, astrocytes and vascular cells are commonly referred to collectively as the 'neurovascular unit' (NVU) and they are not only involved in the regulation of blood flow but also work to maintain the homeostasis of the cerebral microenvironment through controlling BBB exchange and contributing to immune surveillance ⁶⁰⁴. As well as having been recognised at the cerebral level, neurovascular coupling has also been demonstrated at the ocular level in both the retinal and ONH circulations⁶⁰⁵⁻⁶⁰⁷, whereby an increase in retinal neuronal activity stimulates the dilation of the retinal arteries and capillaries, increasing blood to flow to the excited regions. Further similarities in the functioning of the ocular and cerebral microcirculations are discussed in the following section.

1.12.4 The Vascular Theory of Alzheimer's disease

As discussed previously the formation of extracellular beta-amyloid senile plaques and intracellular neurofibrillary tangles, as a result of abnormal tau protein phosphorylation, are considered the hallmark lesions of AD ⁵⁵⁸⁻⁵⁶¹. However whilst there is clear evidence that these features contribute towards the degeneration of neurons and synapses in the AD brain, the mechanisms which trigger their development is currently unclear. The potential role played by vascular factors and microvascular dysfunction in the development of such features has long been recognised and increasing amounts of evidence for a vascular aetiology of AD has accumulated over recent years. Indeed the presence of traditional vascular risk factors such as obesity 608, smoking 609, hypertension ⁶¹⁰, hypercholesterolemia ⁶¹¹, diabetes ^{612, 613} and alcohol consumption ⁶¹⁴, in both mid-life and in the elderly, as well as the occurrence of vascular diseases, such as heart disease, atherosclerosis, stroke and transient ischemic attacks ^{615, 616}, have all been identified as established associates of AD. The so called vascular theory of AD development, first introduced by de la Torre et al ⁵⁶⁶, proposes that the presence of such cardiovascular risk factors could contribute to the development of AD through triggering either dysregulation of CBF and exposure of the cerebral tissues to repeated ischemic episodes, the development of chronic brain hypoperfusion ^{566, 617, 618}, a disturbance of BBB function or a direct alteration in beta-amyloid regulation and function ⁶¹⁹⁻⁶²¹. Indeed, in support of this, numerous studies have found evidence of reduced CBF and glucose metabolism in AD patients 622-625, as well as disturbed cerebral autoregulation 567, 604, 617, ⁶²⁰ and altered neurovascular coupling mechanisms ^{567, 626-630}. Additionally impaired vascular function has been identified at the systemic level in AD patients ⁶³¹ along with the presence of endothelial dysfunction and disturbances of the BBB ^{619, 632}. Links have also been made between cerebrovascular dysfunction and the abnormal regulation of beta-amyloid, a key feature of AD ⁶⁰⁴ and furthermore the distribution of beta-amyloid deposits across the cerebral tissues, has been shown to map the pattern of

microvascular damage in AD ⁶³³. The relationship between beta-amyloid and microvascular dysfunction however is not straight forward as it is possible for either of them to drive the development of the other. The accumulation of beta-amyloid, for example, is known to have a detrimental effect on the cerebrovasculature and to act as a potent vasoconstrictor; if occurring first it could therefore potentially contribute towards the development of cerebrovascular dysfunction ⁶²⁰. The initial presence of cerebrovascular dysfunction or ischemia on the other hand could promote the accumulation of beta-amyloid and, through disruption of the BBB, impair its regulation and removal from the cerebral tissues, contributing towards beta-amyloid plaque formation ⁶²⁰. Both mechanisms would lead to the development of AD pathology and recent evidence suggests their relationship is likely to be synergistic ⁶⁰⁴.

Further evidence in support of cerebrovascular dysfunction in AD comes from studies looking at structural alterations in the cerebral microvasculature in AD patients, which could disrupt the cerebral microenvironment homeostasis and promote neuronal dysfunction ⁶⁰⁴. Indeed decreased microvascular density, increased tortuousity, basement membrane thickening, degenerative wall changes and the accumulation of beta-amyloid has been evidenced in the cerebral arterioles and capillaries of those with AD ^{580, 621, 634}, along with the presence of atherosclerotic plaques in the cerebral macrovasculature ^{635, 636}.

The concept that a dysregulation of CBF, in the form of either disturbed autoregulation or neurovascular coupling, may be associated with the development of AD is therefore well evidenced. It has been proposed that the resultant exposure of the cerebral tissues to repeated ischemic/hypoxic episodes, if sustained over a long period of time, could potentially lead to neurodegeneration and the development of AD-type brain lesions, through the mechanism of RRI, astrocyte activation and exposure to oxidative stress ⁶¹⁷, in a similar manner to that discussed with regard to glaucoma (section 1.9.13). Indeed

oxidative stress induced tissue damage in the form of lipid, nucleic acid and DNA modifications has previously been demonstrated in the cerebrovasculature of those with AD and also linked to the cerebral accumulation and dysfunction of beta-amyloid, a key feature of the disease ^{565, 619, 620, 637, 638}. Interestingly the development of cerebrovascular dysregulation, if present, is thought to occur as an early event in the pathogenesis of AD, preceding symptoms and the characteristic progressive decline in cognitive function ^{632, 639}. The detection of such dysregulation at the earliest stages in at risk individuals could therefore be critical in the early diagnosis and management of the disease ⁶⁴⁰.

Overall the evidence linking vascular dysfunction and cardiovascular risk factors to the development of AD is fairly strong. Indeed, on the back of recent evidence, questions have been raised as to whether AD should even still be considered as a primarily neurodegenerative disorder or in fact should be considered a primary vascular disorder ⁶³⁰. Despite its apparently strong vascular component however it is important that AD is still distinguished from vascular dementia (VaD), the second most common type of dementia accounting for around 15% of cases, in comparison to AD's 75-80% ^{641, 642}. For clarity an overview of VaD and its characteristics in comparison to AD is given in the following section.

1.12.4.1 Alzheimer's disease vs. Vascular dementia

VaD can be defined as a cognitive impairment resulting from cerebrovascular disease and ischemic and/or haemorrhagic brain injury ⁶⁴³. Where the hallmark characteristic of AD is the formation of extracellular beta-amyloid senile plaques and intracellular neurofibrillary tangles ⁵⁵⁸⁻⁵⁶¹, the hallmark characteristics of VaD is ischemic damage and the presence of multiple infarctions due to cerebral vessel occlusion ⁶⁴⁴. The relationship between AD and VaD has become complex due to the increasing amounts of evidence for vascular involvement in AD and the considerable overlap in the symptoms and pathophysiology of the two conditions. Indeed, it has recently been suggested that rather than considering these conditions as two separate entities they may in fact represent a dementia continuum, extending from pure AD through to pure VaD, with a 'mixed' component separating the two extremes ^{645, 646}. Nevertheless and despite their similarities there are a number of distinctions that can be made between AD and VaD and this is particularly important for research purposes. These similarities and differences are summarised in table 1.13.

	Alzheimer's disease (AD)	Vascular Dementia (VaD)
Onset	Gradual, often present for many years before diagnosis	Abrupt
Risk factors	 Stroke Hypertension Heart disease Diabetes Hyperlipidemia Atherosclerosis Smoking Obesity Alcohol Age Education 	 Stroke Hypertension Heart disease Diabetes Hyperlipidemia Atherosclerosis Smoking Obesity Alcohol Age Education
Characteristic features / hallmark lesions	 Beta-amyloid plaques Neurofibrillary tangles 	 Primary damage is ischemic Physical and imaging evidence of cerebrovascular disease
Clinical signs/symptoms	 Cognitive decline and functional deterioration Memory and language deficits more common 	 Cognitive decline and functional deterioration Attention, planning and speed of mental processing deficits more common Earlier and more severe mood/personality changes
Diagnostic criteria	NINDS-ADRDA (National Institute of Neurological Disorders and Stroke – Alzheimer's disease and related disorders association)	NINDS/AIRENS (National Institute of Neurological Disorders and Stroke / Association Internationale pour la Recherche et l'Enseignement en Neurosciences)
Progression / Course	 Predictable pattern of cognitive impairment Predictable spread of cortical neuronal death Predictable rate of cognitive decline 	 Fluctuating and variable Unpredictable Shorter life expectancy than AD but similar rate of decline

Table 1.13 Features of Alzheimer's disease vs. Vascular dementia

1.12.5 Cerebral and Ocular Microcirculations: Associations

The cerebral and ocular microcirculations share a large number of similarities in their anatomy, physiology and embryology, with both forming part of the CNS⁴. The tissues they supply, namely the brain and retina, are both highly metabolically active and rely upon the integrity of their respective microcirculations for maintenance of normal function. Blood flow is actively regulated at both sites via the mechanisms of autoregulation and neurovascular coupling 647 and maintenance of both normal brain and retinal tissue function is reliant on the integrity of the BBB, or blood-retinal-barrier as it is commonly referred to at the ocular level ⁶⁴⁸. As discussed in section 1.12.2, the cerebral capillaries have a characteristic structure which allows formation and maintenance of the BBB and this structure is replicated in the inner retinal capillaries. Additionally, both the cerebral and retinal microcirculations have been shown to undergo similar changes with aging, whereby a decrease in both CBF and OBF is observed along with disturbances in the structural integrity of the microvessels ^{4, 580, 649}. These plentiful similarities between the anatomy and physiology of the cerebral and ocular circulations have inevitably led to the cerebral vasculature being considered in ocular disease and the ocular circulation being considered in cerebral disease.

1.12.6 The Concept of using the 'Eye as a Window to the Brain'

With the increasing amount of evidence linking vascular dysfunction to AD, the development of new or enhanced methods of detecting such vascular dysfunctions in the earliest stages of the disease process would be beneficial with regard to its diagnosis and management. It is with this in mind that AD research has expanded to look, not only at the cerebral vasculature, but also at the significantly more accessible ocular vasculature. Indeed the concept of using the 'eye as a window to the brain' has been increasingly recognised and explored ³ and it is proposed that through imaging and

analysis of the easily visible retinal vessels, information regarding the state and functioning of the cerebral vessels could be obtained non-invasively and diagnosis and understanding of cerebral diseases such as AD could be enhanced.

The relevance of the retinal circulation in cerebral disease is highlighted by the presence of retinal microvascular abnormalities, including arterial narrowing, arteriosclerosis and the presence of retinal exudates or micro-aneurysms, in various forms of cognitive impairment ⁶⁵⁰⁻⁶⁵². With regard to AD in particular, alterations in numerous aspects of ocular function, including visual changes, structural retinal vessel changes and OBF alterations, have been identified, highlighting the probable involvement of the ocular circulation in the AD disease process ³. Indeed difficulty with reading and finding objects, colour recognition problems, abnormalities in visual attention, memory, depth and motor perception, as well as reduced contrast sensitivity have been previously identified in AD patients ⁶⁵³⁻⁶⁵⁵ in the earliest stages of the disease, often before diagnosis has been firmly established, but these are primarily been linked to higher visual cortex changes rather than changes at the ocular level. The exception to this is reduced contrast sensitivity which has also been linked with alterations in the retinal ganglion cells and RNFL ^{653, 656, 657}. At the retinal level, multiple studies have demonstrated structural and functional changes in AD patients including RGC degeneration ^{427, 658}, a reduction in RNFL thickness and macular thickness ^{659, 660}, RGC beta-amyloid deposition ^{661, 662}, narrowed retinal blood vessels and reduced retinal blood flow ⁶⁶³. Furthermore increased cupping and reduced NRR volume and area has also identified at the ONH in AD patients ⁶⁶⁴⁻⁶⁶⁶ and away from the retina, beta-amyloid deposition has been identified in both the intraocular lens and aqueous humour ^{3, 667}.

Ocular involvement in AD is therefore fairly well evidenced and assessment of the retrograde loss of retinal and ONH nerve fibre layer tissue has even been proposed as a potential early biomarker of AD ⁶⁶⁸. The importance of considering the ocular circulation

in AD is supported further by the associations found between AD and the presence of ocular diseases such as glaucoma and AMD ⁶⁶⁹⁻⁶⁷¹. Indeed the presence of beta-amyloid has even been identified in the drusen of patients with AMD ⁶⁷². Of relevance to this thesis is the possible aetiological associations between AD and glaucoma, this is discussed in more detail in the following section.

1.12.7 Alzheimer's disease and glaucoma

The possibility that AD and glaucoma may share a common underlying mechanism has been increasingly realised over recent years to the extent that glaucoma has been suggested as an ocular form of AD and AD as a cerebral form of glaucoma ^{671, 673}. The two conditions have obvious associations in that they are both chronic neurodegenerative diseases, strongly related to aging, which involve the loss of nerve cells by apoptosis. The end stages of both conditions are fairly well evidenced, with RGC loss and ONH excavation being the key feature of glaucoma and the formation of extracellular beta-amyloid senile plaques and intracellular neurofibrillary tangles being the hallmark cerebral lesions of AD. The aetiological mechanisms which lead to the development of these characteristic lesions is currently uncertain for both diseases, however vascular dysregulation and oxidative stress has been strongly linked to both ^{141, 565} (see sections 1.9 and 1.12.4).

Evidence for an association between AD and both POAG and NTG comes from epidemiological, pathological and vascular studies. Firstly from an epidemiological point of view, multiple studies have demonstrated an increased rate of occurrence and progression of OAG (POAG or NTG) in patients diagnosed with AD ^{400, 436, 449, 674} and an increased prevalence of glaucoma in those with cognitive impairment or dementia ^{435, 675}. Other studies, however have failed to demonstrate such an increased rate of POAG and NTG in those with AD ^{426, 676}, making it difficult to draw any firm conclusions. The differences between these studies however could be accounted for by their differing

nature of evaluation, with the studies finding no increased rate being of a retrospective and register-based nature, which is a less favoured method of evaluation and has received some criticism ⁶⁷⁷. Secondly, from a pathological point of view, classic lesions commonly associated with AD have been reported at the ocular level in glaucoma patients and vice versa. Decreased levels of beta-amyloid and increased levels of tauprotein, a feature commonly demonstrated in the CSF of AD patients, has for example been demonstrated in the vitreous humour of glaucoma patients ⁴²³. Furthermore, in glaucoma patients with unregulated IOP, an increased presence of abnormal tau protein molecules has been shown in the posterior retina ⁶⁷⁸ and in experimental glaucoma an increased aggregation of beta-amyloid and the presence of amyloid precursor proteins has been demonstrated in the RGC and ONH of rat eyes ⁴¹⁹. On the other hand, reduced retinal perfusion and increased ONH cupping, characteristic of glaucoma, has been identified at the ocular level in AD patients ^{663, 665}. This discovery of coexisting lesions in AD and glaucoma patients provides more direct support for the concept that both diseases may share a common underlying pathology. The nature of this common underlying pathology is unclear; however due to the strong vascular links outlined in both conditions individually and the similarities in the ocular and cerebral microcirculation, it is highly possible that this shared pathology may be vascular in origin. Indeed, evidence of increased cerebral small vessel ischemia ³¹³ and AD-like perfusion patterns ⁴⁶⁸, have been demonstrated in NTG patients and evidence of disturbed cerebral autoregulation, in association with altered sympathetic innervation or endothelial function, has been found in both POAG and NTG patients ³²³. As both ischemia and cerebrovascular dysfunction have been previously linked to the development of AD, these finding suggest that microvascular abnormalities present at the ocular level in glaucoma patients could potentially extend to the cerebral level and be associated with the development of AD.

Aside from vascular factors, other aetiological theories linking AD and OAG have also been proposed relating, for example, to decreased cerebrospinal fluid pressure in both conditions ⁶⁷⁹, raised intracranial pressure in association with raised IOP ⁵⁶² or the presence of common genetic risk factors ^{680, 681}.

All in all the associations between AD and glaucoma, although evident, are still surrounded by uncertainty. Establishing the nature of their relationship could be beneficial to the early diagnosis, treatment and pathological understanding of both conditions. Furthermore, exploring the relationship between the cerebral and retinal microcirculation and the concept of using the 'eye as a window to the brain' could offer new methods of assessment and diagnosis in AD. Both of these issues are addressed in this thesis

2. Research Rationale

Central to both current glaucoma and AD research is the consideration that, aside from the local ocular or cerebral circulations, multiple additional systemic sites may also be involved in the aetiology of these diseases and that a generalised vascular dysfunction, affecting the circulatory system as a whole, may exist in diagnosed individuals. Exactly how such vascular alterations at multiple sites may inter-relate and their impact on the neurodegenerative disease process is however still poorly understood. Elucidating these inter-relations in both AD and glaucoma individually and establishing their relative impact in different subcategories of these diseases, such as POAG and NTG, could not only enhance our understanding of disease aetiology but also open up the possibility for new and alternative diagnostic and therapeutic avenues.

As well as the interrelationships between vascular alterations at multiple sites in AD and glaucoma individually, the interrelationships between these two neurodegenerative diseases themselves, along with the potential benefits of utilising more accessible regions of the CNS, such as the ocular circulation, to gain an insight into the pathological mechanisms which may be occurring at the cerebral level, have been increasingly realised over recent years. Such utilisation of the ocular circulation could significantly aid the assessment and diagnosis of vascular alterations in AD and enhance our understanding and management of the disease. Furthermore if the exact nature of the relationship between AD and glaucoma could be established, the relevance of considering each disease process in the presence of the other could be highlighted and the diagnosis, management and aetiological understanding of both conditions could potentially be enhanced. At the present date, direct evidence of vascular dysfunction at the ocular level, comparable to that at the cerebral level is yet to be demonstrated and although a number of similarities between AD and glaucoma have been acknowledged in recent years, the exact association between these two

neurodegenerative diseases is still surrounded by uncertainty. Furthermore the concept of whether AD and glaucoma share a common underlying aetiology, comparable in both POAG and NTG, has not been fully explored. Through the development of new technologies aimed specifically at assessing ocular vascular function, such as dynamic retinal vessel analysis (DVA), there is however now an increased potential for this.

Taking all of this into consideration, the principle purpose of this research was to fully establish and explore the presence of vascular alterations at both the ocular and systemic level in AD, POAG and NTG patients and to evaluate the potential impact that these vascular alterations may have, both on these disease processes individually and on the inter-relationships between them. In line with this the overall aims of this research are as outlined in the following section.

2.1 Aims

- To investigate the presence and impact of ocular and systemic vascular alterations in AD and to explore the concept of using the 'eye as a window to the brain'
- To investigate the presence and impact of ocular and systemic vascular alterations in NTG
- To investigate the possibility of a shared vascular aetiology, involving both the ocular and systemic circulations, in AD and both POAG and NTG.
- To compare and contrast vascular alterations at both the ocular and systemic level in POAG and NTG and to explore their validity as distinct clinical entities

3. Subjects and Methods

This chapter outlines the recruitment procedure, inclusion and exclusion criteria for the AD, POAG, NTG and control participants involved in this research. It then goes on to outline the study protocol and investigative techniques used throughout this thesis for the assessment of ocular and systemic vascular function. All disease participants recruited for this research were newly diagnosed and previously untreated. This ensured any vascular alterations identified were less likely to have occurred as a secondary effect of the disease process and could be more reliably attributed to the development or pathogenesis of the disease itself.

3.1 Patient recruitment

3.1.1 Recruitment of mild newly diagnosed AD patients

Successive newly diagnosed mild AD patients were recruited from the Birmingham and Solihull Mental Health NHS Trust (BSMHT, UK) by a team of dementia specialists. Posters advertising the study were displayed in the clinic waiting areas along with information leaflets and flyers. The dementia team were fully briefed on the study protocol, inclusion and exclusion criteria.

3.1.1.1 AD inclusion criteria

Only those patients diagnosed with AD in accordance with the NINCDS-ADRDA and DSM-IV-TR criteria for diagnosis (see section 1.10.2) were considered for this study ^{569,} ⁶⁸². Of these diagnosed patients only those classified as having mild AD, based on a MMSE score of between 18 and 24, were included in the final study. Following the identification of a suitable participant by the dementia team, the study was discussed with them and a detailed information booklet provided. 24 hours later the potential participants were contacted by the author to further discuss the study, answer any questions and confirm enrolment in the study for those who were willing.

3.1.2 Recruitment of newly diagnosed POAG patients

Successive, early stage, newly diagnosed and previously untreated POAG patients were recruited from the glaucoma clinics at both the Heart of England and Sandwell and West Birmingham NHS Trusts. Posters advertising the study were displayed in the clinic waiting areas along with information leaflets and flyers. The diagnosis of early POAG was made by a team of ophthalmologists working under the supervision of two glaucoma consultants. Following diagnosis suitable participants were provided with detailed information about the study by the author and allowed at least 24 hours to consider their enrolment.

3.1.2.1 POAG inclusion criteria

Only those patients identified as having glaucomatous cupping of the optic disc, normal open anterior chamber angles and visual field (VF) defects consistent with the diagnosis of early glaucoma using program 24-2 of the Humphrey visual field analyser (HFA:Zeiss-Humphrey, San Leandro, CA) were included. An early glaucomatous VF defect was defined as a mean deviation (MD) score of \geq -6.00dB along with either a glaucoma hemifield test outside normal limits and/or a corrected pattern standard deviation (CPSD) with p-value<0.05⁶⁸³. Only reliable VF plots, with <20% fixation losses and <33% false positive and false negative responses, were considered. Classification as POAG was based on an IOP measurements consistently above 21 mmHg on diurnal testing (measurements every 2 hours over an 8 hour period) with applanation tonometry. All patients were newly diagnosed and had no current or previous treatment with IOP lowering drops.

3.1.3 Recruitment of newly diagnosed NTG patients

Successive, early stage, newly diagnosed and previously untreated NTG patients were recruited from the glaucoma clinics at both the Heart of England and Sandwell and West Birmingham NHS Trusts. Posters advertising the study were displayed in the clinic

waiting areas along with information leaflets and flyers. The diagnosis of early NTG was made by a team of ophthalmologists working under the supervision of two glaucoma consultants. Following diagnosis suitable participants were provided with detailed information about the study by the author and allowed at least 24 hours to consider their enrolment.

3.1.3.1 NTG inclusion criteria

Only those patients identified as having glaucomatous cupping of the optic disc, normal open anterior chamber angles and visual field (VF) defects consistent with the diagnosis of early glaucoma using program 24-2 of the Humphrey visual field analyser (HFA:Zeiss-Humphrey, San Leandro, CA) were included. An early glaucomatous VF defect was defined as a mean deviation (MD) score of \geq -6.00dB along with either a glaucoma hemifield test outside normal limits and/or a corrected pattern standard deviation (CPSD) with p-value<0.05⁶⁸³. Only reliable VF plots, with <20% fixation losses and <33% false positive and false negative responses, were considered. Classification as NTG was based on an IOP measurement consistently less than or equal to 21 mmHg on diurnal testing (measurements every 2 hours over an 8 hour period) with applanation tonometry. All patients were newly diagnosed and had no current or previous treatment with IOP lowering drops.

3.1.4 Recruitment of healthy controls

Age-matched healthy controls were recruited by inviting the participation of patients' spouses, same-generation relatives and friends, as well as through promotion of the study at the Aston University Health Clinics via posters, information leaflets and flyers displayed in the waiting areas.

3.1.4.1 Healthy control inclusion criteria

All healthy controls were screened for ocular disease and dementia, using the Addenbrooke's Cognitive Examination-Revised (ACE-R; appendix 2) ⁶⁸⁴, before inclusion in the study. Only healthy individuals with a normal ocular examination (fundus assessment, IOP measurement and visual fields) and an ACE-R score of at least 88 were then included in the study ⁶⁸⁴.

3.1.5 Exclusion criteria for all groups

Patients with closed iridocorneal angles, evidence of secondary glaucoma, pseudoexfoliation, pigmentary dispersion, history of intraocular surgery or any form of retinal or neuro-ophthalmological disease that could result in visual field defects were excluded from the study. AD, POAG, NTG and healthy participants were excluded if they were smokers or had a positive diagnosis of cardio- or cerebro-vascular disease. (coronary artery disease - CAD, heart failure, arrhythmia, stroke, transient ischaemic attacks), peripheral vascular disease, severe dyslipidaemia (defined as plasma triglycerides>6.00mmol/L or cholesterol levels>7.00mmol/L), diabetes, as well as other metabolic disorders or chronic diseases that required treatment. Participants were screened for ocular disease and were excluded from the study if they had a refractive error of more than $\pm 3DS$ and more than $\pm 1DC$, IOP>24 mmHg, cataract or any other media opacities, as well as if they had a history of intraocular surgery or any form of retinal or neuro-ophthalmic disease affecting the ocular vascular system. Well controlled systemic hypertension was neither an inclusion nor exclusion criteria; however any individuals taking other medications which could potentially influence vascular function were excluded.

3.2 Ethical approval

Prior to the study ethical approval was received from South Birmingham local research ethics committee, Heart of England and Sandwell and West Birmingham NHS Research Ethics Committees as well as the Aston University Life and Health Sciences Ethics Committee. Written informed consent was received from all subjects before entry into the study and all procedures were designed and conducted in accordance with the tenets of the Declaration of Helsinki.

3.3 Methods

The investigative techniques used throughout this thesis were carefully selected to ensure the most accurate and reliable assessment of vascular function at both the ocular and systemic level could be obtained. A summary of these techniques is given in table 3.1. All techniques were performed by the author with the exception of the blood analyses for von Willebrand factor and glutathione status, which were performed by Dr Lu Qin, an experienced lab technician. Prior to commencing recruitment the author was trained in the use of each technique by experienced colleagues and conducted a series of preliminary examinations on at least 15 volunteers to ensure an adequate level of competency was achieved. Across this series of preliminary examinations the importance of careful patient instruction, optimal patient adjustment and optimal examiner positioning was realised and the measurement procedures adjusted accordingly.

Technique	Purpose
Ocular level	
Dynamic retinal vessel analysis (DVA)	Assessment of ocular vascular function
Systemic level	
Flow mediated dilation (FMD)	Systemic endothelial dysfunction
Ambulatory blood pressure monitoring (ABPM)	24 hour blood pressure profile
24 hour Heart rate variability assessment (HRV)	Autonomic nervous system assessment
Pulse wave analysis (PWA)	Systemic arterial stiffness
Intima-media thickness measurement (IMT)	Cardiovascular risk / atherosclerosis
Blood analysis	
Von Willebrand factor (vWf)	Systemic endothelial function
Glucose, Cholesterol, Triglycerides	Cardiovascular risk
Glutathione status (GSH vs. GSSG)	Oxidative stress

Table 3.1 An overview of the investigative techniques conducted in this thesis

3.3.1 Experimental Protocol

All measurements were performed between 8 am and 11 am following a 12 hour overnight fast, which included no alcohol or caffeine. All procedures were conducted in a consistent order outlined below and detailed in the following sections:

- 1. Suitable participant identified, approached and provided with the study information pack
- 2. Procedures and risks explained, concerns addressed
- 3. Consent form read, understood, completed and signed by participant
- 4. Preliminary assessments:
 - Demographic questionnaire
 - Addenbrooke's Cognitive Examination-Revised (ACE-R) questionnaire
 - Ocular assessment (Indirect ophthalmoscopy and visual fields)
 - IOP measurement
 - Height and weight measurement
 - Baseline BP readings
- 5. 1% Tropicamide inserted into randomly selected eye
- 6. Fasting venous blood sample obtained by venepuncture
- 7. 24 hour BP and HRV monitor fitted (POAG, NTG, Controls only)
- 8. Assessment of retinal vessel reactivity (DVA)
- 9. Pulse wave analysis (POAG, NTG, Controls only)
- 10. Intima-media thickness measurement (POAG, NTG, Controls only)
- 11. Assessment of systemic endothelial function (FMD)
- 12. Final BP measurement

3.3.2 Preliminary Assessments

Demographic data regarding age, gender, ethnicity and current medication were collected by means of a short questionnaire. In order to ensure our healthy control and glaucoma patients were not suffering from any undiagnosed cognitive impairments which could have affected the reliability of our results, participants from these two groups were also screened for dementia prior to inclusion in the study using the ACE-R test (appendix 2) ⁶⁸⁴. In accordance with guidelines only healthy individuals with ACE-R score of at least 88 were included in the study ⁶⁸⁴.

IOP was measured using Goldman applanation tonometry following instillation of 0.4% oxybuprocaine hydrochloride and fluorescein. Visual field analysis was conducted using the SITA 24-2 program of the Humphrey visual field analyser and the fundus was assessed using indirect ophthalmoscopy. Weight and height were recorded and BMI calculated (equation 3.7, section 3.3.5.4). EDTA blood samples were obtained from the antecubital fossa vein of all participants and were tested immediately for fasting triglycerides (TGs), glucose and total and HDL cholesterol, using a Reflotron Desktop Analyser (Roche Diagnostics, UK) and later analysed for vWf and oxidative stress. SBP and DBP were measured using an automatic BP monitor (UA-767, A&D Co. Ltd, UK).

3.3.3 Assessment of Ocular Vascular Function: Dynamic retinal vessel analysis (DVA)

As discussed in section 1.9.5, the presence of reduced/unstable blood flow in the retina, choroid, ONH and retrobulbar vessels of patients with glaucoma is already fairly well established; however the mechanisms by which the blood flow becomes reduced / unstable are less established. With regard to this thesis therefore, techniques aimed at assessing ocular vascular function, as opposed to simply measuring OBF, were of much greater interest. The dynamic retinal vessel analyser (DVA) is currently the most widely

used method for assessing retinal microvascular reactivity in health and disease and is considered to be a sensitive indicator of retinal microvascular function ^{685, 686}; it has therefore been the technique of choice for assessing ocular vascular parameters in this thesis. DVA allows the dynamic behaviour of the retinal vessels to be assessed through the continuous online measurement of vessel diameter changes in response to provocation ⁶⁸⁷. As discussed in section 1.4, adjustments in vascular diameter and hence adjustments in vascular resistance play an important role in the local autoregulation of OBF. As disturbed autoregulation has been identified as a key risk factor in the development of glaucoma and AD, (see sections 1.9.7 and 1.10.3), assessment of the alterations in retinal vessel diameter in response to provocation, by means of DVA, can provide useful diagnostic information about autoregulatory function in such individuals.

3.3.3.1 Basic principles

The setup of the dynamic retinal vessel analyser device (DVA, IMEDOS, Germany) consists of a fundus camera (FF450, Zeiss Jena, Germany), a charged coupling device (CCD) camera, a high resolution video recorder, a real time monitor and a personal computer with analysis software, as illustrated in figure 3.1.





The DVA is capable of measuring retinal vessel diameter continuously along a selected vessel segment over a specified time. This is achieved via analysis of the brightness profile of the retinal vessels. Brightness profile is based on the absorbing properties of the red blood cells (RBC) within a vessel. With regard to retinal vessels, maximum absorption of light occurs at a wavelength of 400-620nm and by comparing the brightness profile of the RBC column within the vessel with that of the surrounding tissue a continuous assessment of vessel diameter can be made. DVA therefore measures vessel diameter as the width of the red blood cell column within the selected vessel. It achieves this as follows:

- The illumination light of the fundus camera enters the eye through a dilated pupil and is reflected by the different layers of the retina and retinal vessels.
- It is then delivered via the observation pathway to the CCD camera
- The brightness profile data is then analysed by the computer system and simultaneously recorded by a high quality video recorder so that off-line analysis can be conducted at a later date if necessary.

To ensure optimal alignment and set up the operator can observe an image of the fundus on the computer display (figure 3.2) and to ensure optimal contrast for vessel visualisation a green filter is inserted into the illumination pathway of the fundus camera. Furthermore, the device is equipped with a series of adaptive algorithms to compensate for any disruption of the vessel brightness profile, by the presence of either shadowing structures from the background or reflections from the vessel surface. It also has the capability to correct for slight eye movements during assessment and can continuously monitor image quality, according to image contrast and then automatically remove any inadequate measurements from the analysis ^{687, 688}.



Figure 3.2 Dynamic retinal vessel analyser (DVA)

The technical specifications of the DVA are summarised in table 3.2 ⁶⁸⁷. The resolution of the device is such that accurate measurements cannot be achieved from vessels with diameters < 90 μ m and the temporal resolution of the device is 40 ms, such that 25 video frames are captured per second (i.e. sampling rate = 25Hz). The image field or camera angle should be set at 30⁰ and a clear fundus image, with good contrast and even illumination should be obtained through a fully dilated pupil. All size related measurements are expressed in 'units of measurement' (UM), whereby 1 UM is equivalent to 1 μ m in a normal emmetropic eye ⁶⁸⁷.

Parameter	Value
Measurement range	90 µm
Measurement resolution	< 1 µm
Temporal resolution	≥ 40 ms
Image field angle	30°
Measuring time	350 seconds (but can be up to 10 mins)
Maximum length of vessel segment	3 mm
Spatial resolution (along vessel segment)	180 µm
Measuring sensitivity	1 MU/1 µm

Table 3.2: Technical specification of the DVA 687

Over the last few years DVA has become a widely used method for assessment of retinal vascular function however it does have some limitations which are summarised below, along with its relative advantages:

3.3.3.2 Advantages

- Non-invasive
- Continuous recording of vessel diameter allows quantification of the effects of provocation with high time resolution
- Vessel segments and different retinal vessels can be investigated simultaneously
- High reproducibility 689
- Low variability ⁶⁹⁰

3.3.3.3 Limitations

- Reliant on patient having clear media to gain an image of sufficient quality
- Reliant on good patient fixation over 350 second testing period
- Requires full pupil dilation with Tropicamide 1% only (to avoid adverse vascular effects)
- Assumes the eye has no refractive error
- Doesn't measure absolute retinal vessel size, instead uses standardised or relative units

As mentioned previously in order to assess the dynamic behaviour and autoregulatory capacity of the retinal vessels the vessels need to be provoked. The DVA can be operated in conjunction with a variety of provocation devices including suction cup IOP enhancement, pure oxygen breathing, inhalation of CO_2 and flicker light stimulation 687 . For the purposes of this thesis flicker light stimulation has been used to assess dynamic retinal vessel reactivity and this will be discussed in more detail in the following section.

3.3.3.4 Flicker light stimulation

Flicker light can be considered the most natural provocation method for assessing dynamic retinal vessel behaviour and has the advantage of stimulating the retina exclusively, without the involvement of any other vascular bed. The normal vascular response to flickering light has been widely studied ^{686, 691} and there is plentiful evidence to indicate that, under normal circumstances flicker stimulation should lead to an increase in vessel diameter, retinal blood flow and ONH blood flow in humans ⁶⁹².

Flicker light can be defined as illumination which alternates in brightness or colour at a frequency of approx 1-50 Hz⁶⁹¹. Electrophysiological studies have shown that the maximum sensitivity of the human visual system to flicker stimulation is obtained with a flicker frequency of between 10-20 Hz⁶⁸⁵. The DVA device used throughout this thesis was equipped to generate flickering light at a sampling rate of 12.5 Hz via an optoelectronic shutter placed in the optical pathway of the camera. A sampling rate of 12.5 Hz lies within the optimum flicker frequency range and has been shown to provide appropriate retinal stimulation by numerous studies ^{685, 691, 693}.

The measurement protocol used to assess retinal microvascular reactivity to flicker light in this thesis is in accordance with that introduced by Nagel et al ⁶⁸⁶. This protocol is widely used and recommended ⁶⁸⁸ and is outlined in the following section.

3.3.3.5 Measurement protocol

- The fundus camera is positioned so as to obtain a uniformly illuminated fundus image through the fully dilated pupil without unwanted reflections and the brightness is adjusted to ensure optimal contrast
- The patient's fixation is directed, with the use of a fixation needle, so that the measurement area of interest lies in the centre of fundus picture

- A region of interest is defined on the retina by the user selecting a rectangular area on the real time monitor which outlines the region to be studied (usually inferior to ONH).
- From within this area a section of the inferior temporal retinal artery and a section of the inferior temporal retinal vein, located approximately 1.5 disc diameters from the ONH and approx 0.5-1 disc diameters in length and a reasonable distance apart, are selected for analysis and monitoring (figure 3.3).



Figure 3.3 Example of retinal vessel selection prior to dynamic retinal vessel analysis

Measurement then starts automatically and vessel diameters are continuously calculated along the length of the selected vessels over a 350 second testing period. This consists of 50 seconds of baseline measurements under still illumination (25Hz), followed by 3 cycles of 20 second flicker stimulation (optoelectronically generated at 12.5 Hz) each interrupted by 80 seconds of still illumination (recovery), as outlined in figure 3.4.



Figure 3.4 Breakdown of the 350 second DVA testing period

All measurements are performed in a quiet, temperature-controlled room (22°C) following full dilation of one unselected eye (1% tropicamide, Chauvin Pharmaceuticals Ltd).

The use of a 350 second testing period with three sequential flicker cycles was initially introduced so that an averaged vessel response could be calculated and analysed to ensure stable results could be obtained over a testing period of tolerable length ^{686, 694}. More recently however new techniques of analysis have been introduced which also consider the vessel responses to each flicker cycle individually as well as the overall average ⁶⁹⁵. This is discussed in more detail in section 3.3.3.8.

3.3.3.6 Normal vascular response

The normal retinal vessel response profile to flicker light stimulation by the DVA is illustrated in figure 3.5. Previous studies have shown that the maximum vessel response to flickering light typically occurs within 20 seconds ⁶⁹³. After this time period only small increases in diameter occur. Once the flicker stimulation ends, dilation ceases immediately but rather than simply returning to baseline, the baseline is usually overshot and a vasoconstriction occurs. This overshoot has been found to start within approximately 6-10 seconds following the end of the flicker period, reaching its minimum diameter between 10-40 seconds following the end of flicker ^{691, 693}. The vessel diameter then returns to its baseline level. Both retinal arteries and veins respond to flicker light stimulation, however arteries tend to show a more pronounced diameter change in
comparison to veins and whilst arteries start to react immediately veins have been shown to have approximately a 5 second delay before a response is seen ^{685, 693}.



Figure 3.5 Diagrammatic representation of a typical retinal vessel response to flicker light on dynamic retinal vessel analysis

Deviation away from this normal vascular response profile can be indicative of vascular disease and indeed numerous studies have already provided evidence of altered vascular response to flickering light in both ocular and systemic disease, including POAG ⁴¹, ARMD ⁶⁹⁶, diabetes ⁶⁹⁷⁻⁶⁹⁹ and hypertension ⁶⁸⁶. In order to understand the implications of an impaired or altered vascular response to flicker light with regard to the development of disease, it is necessary to try and understand the mechanisms by which flicker light provokes the retinal vessels.

3.3.3.7 Reaction mechanism

Flicker light stimulation increases the neural demand of the retina, which, under normal circumstances should trigger a neurovascular coupling response of the retinal microvasculature, resulting in vasodilation ⁶⁹² (see section 1.5.3.2) . In general terms an increase in the metabolic rate of the photoreceptors, following stimulation by flickering light, is thought to trigger the release of NO from the retinal vascular endothelium, bringing about an increase in vessel diameter followed by an increase in blood supply to meet the increased demand ²²⁶. An altered vascular response to flicker light therefore could be indicative of impaired autoregulatory mechanisms and/or endothelial dysfunction in the form of reduced bioavailability of NO (see section 1.6). However, due to the complexity of the neurovascular coupling response, it is highly likely that other causative factors such as altered ET-1 levels, altered astrocyte activity or changes in the basal tonus of the vessels could also play a role in altering the retinal vascular response to flicker. The role of the these alternative factors may be particularly relevant when considering the vascular constriction response following flicker ⁶⁹⁶.

3.3.3.8 Data analysis

Of primary interest when analysing the retinal vascular response profile to flicker light is the percentage dilation of the vessel in response to the stimulus and the time scale across which this happens, along with the percentage constriction or overshoot of the vessel following cessation of flicker and again the time scale across which this happens. In this thesis such analysis of the retinal vascular response was conducted using elements of a newly defined method of DVA analysis, termed 'Sequential and Diameter Response Analysis' (SDRA) in conjunction with our own novel vascular profile imaging methods. The inbuilt software of the DVA device itself does have the capability to provide an analysis of the retinal vascular response to flicker light, however this has been identified to have a number of shortfalls and the need to move away from this traditional inbuilt software analysis and to evaluate the retinal vascular response profile

in more detail has been increasingly realised by numerous authors in recent years ^{688,} ^{695, 696}. Indeed SDRA, first introduced by Heitmar et al ⁶⁹⁵, was primarily designed to overcome the shortfalls of the inbuilt software analysis programme. An overview of this inbuilt DVA analysis and its limitations, along with a summary of SDRA and the novel imaging analysis methods used in this thesis is given in the following sections.

3.3.3.9 Previous methods of analysis

The inbuilt DVA software calculates the vascular dilation response to flicker light by averaging all three of the stimulation cycles and then taking the average diameter from the last +/- 3 seconds of flicker stimulation as the maximum diameter response to flicker (i.e average diameter reached between 17-23 seconds from start of flicker taken as maximum diameter response). The shortfalls of this method largely arise due to its incorporation of both time and diameter responses and its inaccurate assumptions about the nature of the vascular response ⁶⁹⁵. Indeed subjects who reach their maximum dilation outside the 17-23 second window would have their maximum dilatory response underestimated by this technique and furthermore by averaging the results from all three flicker cycles differences in the reaction pattern or time course of each individual cycle cannot be determined ⁶⁹⁵. Additionally this method of analysis does not take into consideration baseline fluctuation in vessel diameter (BDF), a parameter first highlighted as important in DVA analysis by Nagel et al ⁶⁸⁶. BDF refers to the spontaneous variations in vessel diameter, which occur under normal resting conditions, as a result of vascular tone and arterial pulsation and are superimposed on the vascular response profile ⁶⁸⁶. In order to account for the influence of BDF, Nagel et al ⁶⁸⁶ introduced the concept of baseline corrected flicker response (BFR), where BDF is accounted for by subtracting it from the dilation amplitude (DA) of the vascular response, as shown in equations 3.1 and 3.2.

BFR = DA - BDF

BFR = Baseline corrected flicker response

DA = Dilation amplitude

BDF = Baseline diameter fluctuation

Equation 3.1 Baseline corrected flicker response

Where:

$$DA = MD - MC$$

DA = dilation amplitude

MD = maximum dilation

MC = maximum constriction

Equation 3.2 Dilation amplitude

3.3.3.10 Sequential and Diameter Response Analysis (SDRA)

SDRA has the advantage of utilising the raw data set generated by the DVA device and allowing each individual flicker cycle to be considered separately. This enables a more accurate assessment of dynamic vessel response to be obtained and enables the inclusion of the parameters BDF, BFR and DA on top of the standard percentage dilation and constriction response parameters. Additionally it allows the time taken to reach maximum dilation and the time taken to reach maximum constriction to be determined for both the artery and vein for each individual flicker cycle. A summary of the SDRA parameters and how they relate to the vascular response profile is illustrated in figure 3.6 and summarised in table 3.4.



Figure 3.6 Diagrammatic representation of the parameters calculated from the dynamic retinal vessel analysis (DVA) response profile

Parameter	Acronym	Explanation
Baseline diameter fluctuation	BDF	Calculated as the maximum range in vessel diameter during first 30 seconds of baseline readings (i.e. Difference between max diameter and min diameter at baseline)
Percentage dilation	MD%	Calculated as the percentage change in vessel diameter from baseline to maximum following onset of flicker
Baseline corrected flicker response	BFR	Percentage change in vessel diameter after taking into consideration the baseline diameter fluctuation (equation 3.1)
Reaction time	RT	Time taken to reach maximum diameter from the onset of flicker
Percentage constriction	MC%	Percentage constriction of vessel diameter below baseline following cessation of flicker. Calculated as smallest vessel diameter reached following cessation of flicker subtracted from average baseline diameter
Constriction time	tMC	Time taken to reach the point of maximum constriction following cessation of flicker
Dilation amplitude	DA	Amplitude of the dilation response, calculated as the difference in diameter between the maximum and minimum points (equation 3.2)

Table 3.3 Summary of the parameters which can be calculated and in used for analysis of the dynamic retinal vessel response profile

It is important to note that analysis of individual flicker cycles is reliant on a full data set having been obtained from the participant on each subsequent cycle. If this is not achieved, for example due to poor patient fixation, loss of concentration or excessive blinking, calculating and analysing the average data set using SDRA is considered more reliable.

Overall the SDRA method has been validated and shown to be a sensitive measure of the vascular response to flicker light with good coefficients of variation ⁶⁹⁵. Furthermore it is able to overcome a number of the shortfalls of the inbuilt RVA analysis software. It has therefore been the analysis method of choice for this thesis; however we have taken it one step further and using the principles of SDRA have developed an additional way of imaging the retinal vascular profile to allow further aspects of the vascular response to be explored.

3.3.3.11 Novel Analysis

Whilst SDRA overcomes many of the limitations of the inbuilt RVA software analysis it still does not allow visualisation of the entire dynamic retinal vessel response profile. Furthermore, it has been suggested that, in addition to the parameters illustrated in figure 3.6 and summarised in table 3.3, evaluation of the slope of both the dilation and constriction responses to flicker light could give additional important information about the state of the retinal microvasculature in health and disease. In order to address this we have developed a new method of analysing and interpreting the retinal vascular response to flickering light using Matlab (MATLAB R2010a; MathWorks Inc., Natick, MA). Our method expands on the SDRA methodology by extracting the raw response data and applying a statistical polynomial regression algorithm, implemented using the polyfit and polyval functions of the Matlab high-level programming language (MATLAB R2010a; MathWorks Inc., Natick, MA). Dr Aniko Ekart, an experienced

mathematician and Matlab user assisted the author with the construction of the statistical algorithm used here.

Given the measurements y_i at times t_i , i = 1,...,T, we approximated y = f(t) by a polynomial of degree n as

$$p(t) = p_1 t^n + p_2 t^{n-1} + \dots + p_n t + p_{n+1}$$

in a least squares sense.

The polyfit function locates the coefficients $p_1, p_2, ..., p_n, p_{n+1}$ such that the error

 $\sum_{i=1}^{T} (y_i - p(t_i))^2 \text{ is minimized.}$

This involves solving the system of equations

$$\begin{cases} p_{1}t_{1}^{n} + \dots + p_{n}t_{1} + p_{n+1} = y_{1} \\ \vdots \\ \vdots \\ p_{1}t_{T}^{n} + \dots + p_{n}t_{T} + p_{n+1} = y_{T} \end{cases}$$

If we denote $t_i^{n-j+1} = v_{ij}$ then $V = (v_{ij})$ is the Vandermonde matrix and the least squares

problem to be solved can be written as Vp = y where the vectors $p = \begin{pmatrix} p_1 \\ \cdot \\ \cdot \\ p_{n+1} \end{pmatrix}$ and

$$y = \begin{pmatrix} y_1 \\ \cdot \\ \cdot \\ \cdot \\ \cdot \\ \cdot \\ y_T \end{pmatrix}$$

The polyval function was used to calculate the fitted polynomials which ultimately provided us with curves representative of the dynamic vascular response profile which

could then be used for analysis. The polynomial regression algorithms were performed on the averaged data for each individual patient. The degree of the polynomial, *n*, is an adjustable parameter. We applied n = 15, as this value provided the closest fit polynomials on our data points.

As well as the original SDRA parameters of baseline diameter fluctuation (BDF), maximum dilation (MD), maximum constriction (MC), reaction time (RT), dilation amplitude (DA) and baseline corrected flicker response (BFR), the polynomial fitted curves allowed the nature of the dynamic response profile and the slope of the vascular dilation and constriction responses to be calculated and compared between study groups. The slope may be an important parameter as it describes the interaction between the change in vessel diameter and the rate at which this change occurs, both of which are parameters that have shown to be altered in disease states individually. Dilation and constriction slope for both the arteries and veins were calculated as shown in equation 3.3.

$$Dilation \ slope = \frac{MD - Av \ Baseline}{RT} \qquad Constriction \ slope = \frac{MC - MD}{tMDMC}$$
Where: MD = maximum dilation
RT = reaction time
MC = maximum constriction
tMDMC = time between MD and MC

Equation 3.3 Dilation and constriction slope

3.3.4 Assessment of systemic parameters

3.3.4.1 Endothelial function: Flow mediated dilation (FMD)

Determination of the presence of endothelial dysfunction at the systemic level in neurodegenerative disease could provide important information about the involvement of macro-vascular function in the pathogenesis of the disease.

FMD is considered the gold standard technique for assessing systemic endothelial function ¹⁹⁹. It is a well-established technique that has been widely used in clinical research to assess peripheral vascular function and cardiovascular disease risk factors ^{700, 701}. Its main advantage over other techniques is that it is non-invasive and by nature of this, allows repeated measurements to be taken either sequentially or over time with a high level of patient acceptance ⁵². Its only disadvantage is that it requires practice to master the technique and carefully controlled experimental conditions are required to ensure reproducibility and reliability of results ^{52, 702}.

FMD refers to the dilation of a vessel in response to increased blood flow. A sudden increase in blood flow is known to exert a shear stress stimulus on the vessel wall which, under normal circumstances, triggers the release of the vasodilator NO from the endothelium leading to an increase in vessel diameter to accommodate the increased blood flow. A failure to observe this vascular dilation response to increased flow is considered indicative of endothelial dysfunction and hence important information about the functioning of the endothelium can be gained by assessing the dilation response of a systemic artery to an experimentally produced increase in blood flow in FMD.

3.3.4.2 Protocol

Measurement of endothelium dependent FMD in this thesis has been conducted in accordance with the guidelines for assessment published by Corretti et al ⁷⁰².

The procedure involves ultrasound imaging of the brachial artery of the upper arm, with an ultrasound probe positioned in the longitudinal plane, 5-10cm above the antecubital fossa, using a 2D colour Doppler ultrasound system (CDI).

Prior to all FMD assessments conducted in this thesis, patients were required to undergo a 12 hour fast due to the potential influence that food intake and caffeine can have on flow mediated vascular reactivity ⁷⁰². Furthermore, as temperature can also influence vascular reactivity all procedures were conducted in a quiet, dark, temperature controlled room at 22^oC. The measurement procedure is summarised as follows:

- The patient is positioned supine and allowed to rest in this position for 10 minutes prior to first scan.
- Their arm is then positioned comfortably in an extended position and an ultrasound image of the brachial artery is obtained from the upper right arm
- A segment of the artery with clear anterior and posterior intimal-lumen interfaces is selected on the imaging screen to ensure optimum recording of vessel diameter changes (figure 3.7)



Figure 3.7 Imaging screen in FMD with brachial artery segment highlighted

- 2 minutes of baseline vessel diameter readings are taken from the brachial artery
- A 5.6 inch wide BP cuff (sphygmomanometer) positioned at the forearm is then inflated to 50mmHg above the patients recorded SBP and is kept inflated for 5

minutes. The inflated cuff effectively occludes the blood flow through the brachial artery, inducing hypoxia and causing dilation of downstream resistance vessels.

 After 5 minutes the cuff is rapidly deflated and the brachial artery is imaged for a further 2 minutes to assess the hyperaemic response. (figure 3.8)



Figure 3.8 Breakdown of the 7 minute FMD examination protocol

On deflation of the cuff, the brachial artery is no longer occluded and there is a sudden increase of blood flow through it (hyperaemia). As mentioned previously this increased blood flow produces a shear stress stimulus on the vessel wall which should trigger the endothelium to release NO, possibly in combination with other vasodilators, and an increase in vessel diameter should be seen. A failure to observe a good vasodilation response would indicate systemic endothelial dysfunction.

Following FMD assessment and after a 10 minute rest period, the procedure is repeated using a sublingual tablet of nitroglycerin (GTN 0.3mg) in place of the sphygmomanometer. GTN is converted by the body into NO and this then acts directly on the vascular smooth muscle cells to bring about vasodilation, independent of the endothelium. GTN is therefore used to confirm whether any impaired dilation response detected by FMD can definitely be attributed to a dysfunctional endothelium and is not simply the result of a vascular smooth muscle dysfunction. It would be expected that, even in the presence of endothelial dysfunction, nitroglycerin mediated vasodilation (NMD) should be unchanged. If NMD were significantly impaired to the same degree as FMD it would suggest the vascular smooth muscle is not functioning properly and any impaired FMD responses could not be convincingly attributed to endothelial dysfunction.

3.3.4.3 Analysis

Flow mediated dilation is calculated as the percentage change in brachial artery diameter in response to hyperaemia (equation 3.4). Baseline vessel diameter is taken as the average diameter from the first 2 minutes of brachial artery ultrasound imaging and the vessel diameter following hyperaemia is taken as the maximum diameter reached in the 2 minutes following cuff deflation.

$$FMD = \frac{VD \ hyperaemia - Baseline \ VD}{Baseline \ VD} \qquad x \ 100$$

FMD = Flow mediated dilation VD = vessel diameter

Equation 3.4 FMD dilation response

Previous studies have suggested that normal values for FMD should be considered to be between a 5-15% increase in brachial artery diameter and a 5-6 fold increase in blood flow on release of the BP cuff. A dilation response of between 0-5% indicates an impaired flow mediated dilation response ^{703, 704}.

Nitroglycerin mediated dilation is calculated in the same way but as the percentage change in brachial artery diameter in response to nitroglycerin (equation 3.5).

$$NMD = \frac{VD \ nitroglycerine - Baseline \ VD}{Baseline \ VD} \qquad x \ 100$$

NMD = Nitroglycerine mediated dilation
VD = vessel diameter

Equation 3.5 NMD dilation response

3.3.4.4 Circulating markers – von Willebrand factor

vWF levels were determined in this thesis via the analysis of fasting venous blood samples obtained by the author. The analysis was conducted by an experienced lab technician (Dr Lu Qin) according to the methodology outlined below:

- Following the collection of fasted venous blood samples into citrate tubes, the samples were centrifuged at 3000rpm for 15 minutes and the supernatant was aliqouted and stored in a -80°C freezer.
- The citrated plasma was then thawed and analysed for vWf levels using a standardised ELISA kit which was optomised according to previously established methods ^{705, 706} and conducted as follows:
 - A microtitre plate was coated with 100μl of dilated primary antiserum solution (30μl in 20.5ml coating buffer at pH 9.6) at room temperature and then refrigerated for a minimum of 60 minutes to over night
 - 2. The microplate was washed 4 times with 250μ l of wash buffer per well before 100 μ l of substrate was added with working strength detection antibody dilutant and incubated for 60 minutes at room temperature
 - The microplate was then washed 3 times with wash buffer before 100µl of secondary antiserum was added and incubated at room temperature for 45 minutes
 - The microplate was then washed again for a final 3 times and 100μl of substrate was added and then incubated at room temperature for 20 minutes
 - The enzymatic reaction was then stopped by adding 50µl of hydrosulphuric acid
 - The absorbance of the solution was then immediately read on a microwell plate reader set at 492nm

3.3.4.5 Ambulatory blood pressure monitoring (ABPM)

ABPM was the technique of choice for the assessment of systemic BP in this thesis and was conducted using the Cardiotens-01 device which is a computer operated ambulatory BP and ECG monitor (Cardiotens-01, PMS Instruments, Maidenhead, UK). This device has been validated in accordance with recommended protocols and has been widely used in previous clinical studies ⁷⁰⁷⁻⁷¹⁰. Its set up is such that a blood pressure cuff is positioned around the upper left arm and connected to a personal monitoring device worn around the waist, via a fibre optic cable. The device was programmed prior to use using the BP monitoring software Cardiovisions 1.7.2 (PMS Instruments, Maidenhead, UK) and customised to each individual, with regard to sleep and wake times and measurement intervals, amongst other factors. Measurements of BP are obtained automatically using an oscillometric method and in the event of a faulty reading the device is programmed to re-inflate a second time in order to avoid missed data points. For the purposes of this thesis the device was programmed to take readings at intervals of 15 minutes during daytime and intervals of 30 minute nocturnally. Patients were advised to carry out their normal daily activities and to complete a diary outlining any periods of unusual exertion or changes in activity for consideration when evaluating their BP variations. After the 24 hour period the BP data was downloaded and analysed using the 'Medibase' software program (Meditech, version 1.42). Maximum and minimum, diurnal and nocturnal, SBP, DBP and MABP were recorded and the mean nocturnal dip in BP calculated (equation 3.6). Furthermore the short-term variability in SBP was determined for both the diurnal and nocturnal periods through calculation of the average coefficient of variation for each group (equation 3.7).

 $Nocturnal BP dip = (Diurnal MABP - Nocturnal MABP) \times 100$

Where: MABP = mean arterial blood pressure, calculated according to equation 1.2

Equation 3.6 Nocturnal blood pressure dip

$Coefficient of variation = \frac{Standard \ deviation}{Mean} \ x \ 100$

Equation 3.7 Coefficient of variation

3.3.4.6 Autonomic Nervous system assessment – Heart rate variability

Assessment of ANS function commonly falls into two categories, the first being the evaluation of BP and HR responses to provocative stimuli designed to test baroreflex sensitivity and the second being the analysis of systemic BP, resting HR and HRV over a 24 hour period ³⁹⁹. For the purposes of this thesis ANS function was assessed using 24 hour ECG monitoring and a frequency domain analysis of HRV using the Cardiotens-01 device and 'Medibase' software introduced in the previous section (3.3.4.5). This device records continuous real time beat-to-beat ECG analysis via two independent channels following the precise placement of electrodes on the patient's chest in the positions illustrated in figure 3.9. The electrodes are connected via a fibre optic cable to the personal monitoring device worn around the patient's waist.



Figure 3.9 Diagrammatic representation of the electrode positioning for 24 hour ECG recording with the Cardiotens-01 device

Following 24 hours of ECG recordings the data can be downloaded from the device and a full frequency domain HRV analysis performed by the 'Medibase' software using a series of validated algorithms. An assessment of HRV and ANS function can then be made through analysis of the LF and HF values and the LF/HF ratio (see section 1.7.2), recorded during the diurnal (active) phase, nocturnal (passive) phase and over the entire 24 hour period.

3.3.5 Assessment of Cardiovascular risk

As discussed in section 1.9.12 there is a possible role for increased ocular and systemic vascular stiffness and the presence of cardiovascular disorders in the development of GON, however the current evidence is variable. In order to explore this relationship further an assessment of systemic arterial stiffness, atherosclerotic vessel changes and cardiovascular risk has been made in this thesis using pulse wave analysis, intima-media thickness measurement, calculation of Framingham risk score, BMI and routine blood analysis as outlined in the following sections.

3.3.5.1 Assessment of arterial stiffness: Pulse wave analysis (PWA)

The method of choice for the assessment of arterial stiffness in this thesis was pulse wave analysis (PWA) which was performed using the SphygmoCor device in accordance with a well-established protocol ^{711, 712}. This is a simple, non-invasive, validated device which allows an assessment of arterial stiffness to be made with high repeatability ⁷¹¹⁻⁷¹³. The procedure involves the placement of a high fidelity pressure sensor on the outside of the skin overlying the radial artery at the wrist which generates a signal representative of the intravascular pulse (figure 3.10). From this signal the SphygmoCor device generates a peripheral pressure waveform and then converts it, using measured values of brachial SBP and DBP, into a central aortic pressure wave from which arterial stiffness parameters, such as augmentation index (Alx) can be

calculated by the software. The SphygmoCor software also generates quality control indices and therefore, to ensure reliability of measurements, in this thesis only those readings obtained with an operator index of greater than 80 were accepted.



Figure 3.10: SphygmoCor device set up

The principle of PWA is therefore the generation and analysis of the central aortic pressure waveform, the profile of which can provide important information about the stiffness of the systemic vasculature. This aortic pressure waveform is derived from two separate components, namely a forward pressure wave created by ventricular contraction and a backward or reflected pressure wave created by the reflection of the incident wave back from bifurcations and peripheral vascular beds.



Figure 3.11 a,b,c: forward aortic waveform, reflected waveform, complete aortic waveform

In elastic vessels the reflected wave returns to the aortic root in diastole, supplementing coronary perfusion in this phase and creating the waveform profile illustrated in figure 3.11c. In stiff vessels the reflected wave can return earlier and therefore in systole, causing augmentation of systolic pressure and a decrease in diastolic pressure, along with decreased coronary perfusion, as illustrated in figure 3.12b.



Figure 3.12b: Aortic waveform in presence of arterial stiffness: increased systolic peak, shoulder in ascending limb indicating early return of reflected wave, steeper descending limb with loss of notch

Figure 3.12 a: Normal aortic pulse waveform; b: Aortic waveform in presence of arterial stiffness

waveform: smooth ascending limb,

notch in descending limb just after

systole due to wave reflection.

The Alx parameter generated by the SphygmoCor was used as the marker for systemic arterial stiffness in this study. Alx is a widely used research parameter and is considered a sensitive indicator of arterial stiffness ⁷¹². It indicates the amount by which the aortic pressure is increased by the peripheral reflection of blood flow and it is given as a percentage value corrected for a HR of 75 beats per minute ⁷¹⁴ (figure 3.13). Alx increases with increasing arterial stiffness due to an increased speed of reflection and has been shown to vary with both age and gender, being greater in females and older persons ⁷¹⁵.



Figure 3.13: Diagrammatic representation of the aortic pressure waveform and augmentation index

Normal or reference values for Alx in healthy individuals have been reported for numerous different ethnic groups. The reference values given by the SphygmoCor group for healthy individuals according to age are shown in table 3.4.

Age	Mean (%)	Lower 5% CI (%)	Upper 5% CI (%)
20	-4.67	-23.27	16.87
30	3.03	-15.57	24.57
40	10.73	-7.87	32.27
50	18.43	-0.17	39.97
60	26.13	7.53	47.67
70	33.83	15.23	55.37
80	41.53	22.93	63.07

Table 3.4: Alx Reference values for healthy individuals provided by SphygmoCor

One potential limitation of PWA stems from the suggestion that the estimation of the central pressure waveform alone may be insufficient to accurately quantify the magnitude of pressure waveform reflection and therefore the true accuracy and reliability of any results generated through PWA may be limited ⁷¹⁶. However, the majority of reports indicate that PWA and AIx can be considered good indicators of arterial stiffness and they are amongst the most widely used assessment parameters ⁷¹¹.

3.3.5.2 Intima-media thickness (IMT) measurement

Measurement of carotid artery IMT allowed a simple and non-invasive assessment of early arterial wall changes to be made in this thesis. IMT is a parameter widely used in clinical research and has been shown to provide a direct measure of carotid arteriosclerosis and an indirect measure of generalised atherosclerosis ⁷¹⁷. Furthermore it has been demonstrated to offer predictive value with regard to future cardiovascular complications ⁷¹⁸ and has shown a close relationship with the presence of cardiovascular risk factors ⁷¹⁹⁻⁷²¹ and calculated risk scores such as the Framingham risk score ⁷²² (see section 3.3.5.3).

Measurement of IMT (figure 3.14; 3.15) was conducted in this thesis using a wellestablished protocol ^{717, 723}. Patients were positioned supine with their head turned away from the measurement site and their neck slightly extended. High resolution B-mode ultrasonography (Siemens; Acuson Sequoia, UK) was used to obtain an image of the right common carotid artery at the level of the carotid bifurcation. IMT measurements were then taken from the central region of the inferior wall of the artery, using the inbuilt software calliper system, at a site proximal to the artery bifurcation. The presence of any arteriosclerotic plaques along the vessel wall was also noted and measured. A normal IMT measurement is considered to be anything below 0.1 cm ⁷¹⁷.



Figure 3.14 Diagrammatic representation of the carotid artery wall



Figure 3.15 Ultrasound image indicating carotid artery intima-media thickness measurement site. C-IMT: carotid-intima media thickness

Ultrasound IMT measurement is generally considered a non-invasive and safe technique that is low cost and offers highly reproducible results. Questions have been raised about the possible subjective nature of the measurements and the risk of high between-observer variability, with the potential benefits of alternative automated analysing systems having been discussed ^{723, 724}. However the within-observer variability in measurements has been shown to be small and hence manual IMT measurement is

still considered a very valid technique ⁷²³. All IMT measurements in this thesis were performed by a single operator.

Research has suggested that IMT can be influenced by age, systolic BP and weight. These are parameters that therefore all need to be controlled when considering the results. Furthermore relationships have been identified between IMT and FMD, with multiple studies suggesting an inverse correlation between the two, in that a greater IMT is associated with a lower FMD result ⁷²⁵. This may be unsurprising as endothelial dysfunction has been shown to be an early occurrence in the development of atherosclerosis, of which IMT is a measure. Other studies have shown no correlation between the two measures suggesting that maybe they represent different stages of the atherosclerotic process, with endothelial dysfunction occurring in the earlier stages and increased IMT developing after more prolonged exposure to risk factors ⁷²⁶. Nevertheless determining the presence of any correlations between IMT and FMD in this thesis could provide a further insight into the role of arterial wall changes in GON.

3.3.5.3 Framingham risk score

Determination of cardiovascular risk is often made on the basis of lifestyle and history questionnaires which aim to determine the presence of factors such as obesity, hypercholesterolemia, diabetes and smoking. A more quantitative assessment however can be made through the calculation of risk scores such as the Framingham risk score. The Framingham risk score provides an estimate of the 10 year absolute risk of an individual developing coronary heart disease ⁷²⁷. It is sex specific, taking into account age, total cholesterol, HDL cholesterol, BP, diabetes and smoking and has been shown to be validated for use across different ethnic groups ⁷²⁸. The reference table on which the calculation is based is given in appendix 3. A Framingham risk score was calculated for all study participants in this thesis to provide an additional quantified measure of cardiovascular risk.

3.3.5.4 Body mass index (BMI) and blood analysis

In addition to the above procedures a further assessment of cardiovascular risk was made through the recording of both weight and height measurements for all participants, followed by the calculation of body mass index (BMI) (equation 3.8)

 $BMI = \frac{Weight(kg)}{Height^2(m)}$

Equation 3.8: BMI

Furthermore fasting EDTA blood samples were obtained from the antecubital fossa vein of all participants and tested immediately for fasting triglycerides (TGs), glucose levels and total and HDL cholesterol, using a Reflotron Desktop Analyser (Roche Diagnostics, UK).

3.3.6 Evaluation of Oxidative Stress

The potential role of oxidative stress in the development of neurodegenerative diseases such as glaucoma and AD was discussed in previous sections (see sections 1.10 and 1.12.4). An evaluation of oxidative stress was obtained in this thesis through the determination of circulating levels of the antioxidant glutathione (GSH, L- γ -glutamyl-L-cysteinyl-glycine) from fasting venous blood samples obtained by the author. The analysis was conducted by an experienced lab technician (Dr Lu Qin) according to protocols optimised in house according to previously reported and validated methods ⁷²⁹. This methodology is outlined below:

- Within 10 minutes of collection, 30µl of the fasting venous EDTA blood sample.
 combined with 33.3µl of sulfosalicylic acid and 936.7µl of sodium phosphate
 stock buffer solution, was centrifuged at 15000rpm for 5 minutes
- 150µl of the supernatant was then aliqouted and cooled to -80°C for later analysis

- The analysis of glutathione levels is then determined using an enzymatic reaction created by:
 - 1. Adding 150µl of daily buffer to 50µl of DTNB solution in each microwell
 - 2. Then adding 25μ I of the prepared plasma samle to each well and incubating this at 37° C for 3 minutes
 - 25μl of GSR solution is then added to the previous mixture and read using a microplate reader set at 410nm.

Following determination of the plasma GSH and GSSG levels the redox status or GSH:GSSG ratio was additionally calculated, along with the total plasma glutathione level (total-GSH, t-GSH) (see equation 3.9) ^{512, 729}.

 $tGSH = GSH + (2 \ x \ GSSG)$



4. Is the eye a window to the brain? Retinal vascular dysfunction in Alzheimer's disease

4.1 Abstract

Purpose: To assess ocular and systemic vascular function and its relationship to the degree of cognitive deficit in newly diagnosed AD patients without overt systemic vascular disease.

Methods: Retinal vascular reactivity to flickering light was assessed in 10 AD patients and 28 age-matched healthy individuals by means of Dynamic Vessel Analysis (DVA). Systemic vascular function was assessed in the same patients by means of flow mediated dilation (FMD). Mini-mental state examination (MMSE) and Addenbrooke's Cognitive Examination-Revised (ACE-R), as well as BP measurements and blood analyses for lipid metabolism markers were also performed in all cases.

Results: AD patients demonstrated differences in their vascular reaction times to three repeated cycles of flicker exposure at the retinal arterial level compared with healthy controls (p=0.038, p=0.049 and p=0.028 respectively), which correlated with their degree of cognitive impairment at the time of the test (R=-0.782, p=0.013).

Additionally AD patients demonstrated an increased venous baseline diameter fluctuation prior to the onset of flicker in comparison to controls (p=0.001). No significant differences in systemic endothelial function were found between groups (p>0.05).

Conclusions: Signs of microvascular dysfunction measured at the retinal level correlate with the degree of the cognitive decline at the time of diagnosis in mild AD patients. It is possible that this method of screening could offer valuable information about risk for future vascular complications as well as progressive cognitive decline these patients.

4.2 Introduction

In addition to other risk factors such as mechanical stress ^{463, 562}, genetics ⁵⁶⁸ and oxidative stress ⁵⁶⁵, vascular disease leading to chronic brain hypo-perfusion ^{566, 617, 618} and more recently cerebral vascular dysregulation have also been implicated in the development of AD ^{619, 630-632}. As such the assessment of cerebral vascular function is considered to be of increasing importance in regard to the early diagnosis and management of AD; however, due to the notorious difficulties associated with directly assessing and visualising the vasculature in this region the ability to do this non-invasively is currently limited.

One concept being increasingly explored is the possibility of using the eye as a "window to the brain"³. This is a particularly attractive concept as the retinal and brain vessels are known to share a large number of embryological and anatomical similarities ⁴ and alterations in retinal structure that replicate cortical neurodegeneration ^{3, 659, 663}, alterations in retinal blood flow ⁶⁶³ and deposits of amyloid-ß (Aß) ⁷³⁰ have all previously been demonstrated at the ocular level in AD patients. Additionally, the co-existence of either increased risk for, or clinical evident, cardiovascular disease (CVD) has previously been shown to result in a poor cognitive prognosis in AD ⁷³¹. All of these observations lead to the suggestion that through the assessment of retinal vasculature function information could potentially be gained not only on future systemic vascular risk in AD patients but also on cerebral vascular changes which may impact on cognitive prognosis. Indeed it has already previously been demonstrated that by assessing retinal vascular responses to flickering light early signs of vascular dysfunction in apparently healthy individuals with various degrees of risk for CVD can be identified ^{710, 732}. This study therefore aims to explore, through the use of DVA, if anomalies in vascular function are evident at the retinal level in mild newly diagnosed AD patients and whether

they can be linked to their degree of cognitive deficit and/or their degree of systemic vascular dysfunction, as measured by FMD.

4.3 Aims

The aim of this study was to identify if anomalies in vascular function are evident at the retinal level in mild newly diagnosed AD patients and whether any such anomalies can be linked to the degree of cognitive deficit and/or the presence of coexisting systemic vascular impairments in these patients.

4.4 Hypothesis

Disturbed vascular function will be present at the ocular level in mild newly diagnosed AD patients and will correlate with their degree of cognitive impairment and/or their degree of systemic vascular dysfunction.

4.5 Subjects and Methods

Mild newly diagnosed AD patients and healthy age matched controls were recruited for this study. The recruitment details, inclusion and exclusion criteria for these patients was detailed in section 3.1. The investigative procedures performed in this study are outlined below and were conducted in accordance with the protocols detailed in section 3.2.

- 1. Preliminary tests
- 2. Fasting venous blood sample obtained
- 3. Blood pressure measurement
- 4. Assessment of retinal vessel reactivity (DVA)
- 5. Assessment of systemic endothelial function (FMD)

4.6 Statistical analysis

The Kolmogorov-Smirnov test was used to determine the distribution of the data. As normality could not be confirmed in all cases the differences between groups were assessed using either the Student t-test for independent variables or the Mann-Whitney-U test as appropriate and presented as either mean ± SD or median (IQR) accordingly. Two factor repeated-measures ANOVA was used to compare the retinal reactivity responses across each flicker cycle and between groups, with log transformations being made where necessary. A Pearson's linear correlation analysis assessed the relationship between the level of cognitive impairment (MMSE score) and the vascular response. Multivariate analysis was performed to determine the influence of age, BMI, BP and the circulating markers on the measured variables; however no significant influences were found. P-values of less than 0.05 were considered significant, except in certain cases where a stricter p-value of less than 0.01 was adopted in order to correct for multiple comparisons. All analyses were performed using Statistica, version 6.0, Statsoft, Tusla, OK.

4.7 Power calculations

With regard to DVA, new measurement parameters and novel analysis methods were used in this study in patient groups which have not previously been examined with these techniques. Power calculation was therefore based on the results of previous studies which share the most similar protocols to that of the present research and was conducted using the computer based programme, GPower 3⁷³³. A retinal vessel reactivity response to flicker light of 6% with a standard deviation of 2.5% is considered normal on the basis of previous research and around a 50% alteration in this response has been shown to be clinically significant ⁶⁸⁶. Furthermore, with regard to FMD, a brachial artery dilation response of 8% with a standard deviation of 3% is considered normal on the basis of previous AD research and a 40% reduction in this response has

been shown to be clinically significant ⁶³¹. Analysis by t-tests for independent samples, as well as two factor repeated measures ANOVA was required in this study. Taking this into consideration, it was calculated that, in order to provide a statistical power of 80% at an α level of 0.05 a sample size of between 10-18 per group would be required (10 DVA t-test, 18 DVA within groups ANOVA, 18 DVA within/between ANOVA, 12 FMD t-test). The aim was therefore to recruit at least 18 patients in each study group.

4.8 Results

A total of 12 AD patients and 34 healthy controls were screened for inclusion in the present study however, in order to ensure all participants were matched on critical factors such as age and hypertensive status, all those under the age of 50, over the age of 75 or with a MBP of greater than 115 mmHg had to be excluded from analysis. Additionally, following the careful review of the obtained images, any patients who exhibited poor or incomplete results were also excluded meaning 10 mild newly diagnosed AD patients and 28 healthy controls were included in the final analysis. These numbers obviously fall below the intended target of 18 in the AD group, however statistical significance was still observed. The difficulties faced with recruitment of this category of patients is discussed in section 4.11

4.8.1 Baseline values

There were no significant differences in age, systemic BP, BMI, TGs, glucose, HDL cholesterol levels and total cholesterol levels between the two groups (all p>0.05). Furthermore there were no significant differences in IOP, MABP, OPP or Framingham risk score between groups (all p>0.05, table 4.1). The number of subjects with well controlled high BP was also proportionally similar in both groups (AD: n=3 and Controls: n=9, p>0.05; Chi-square test)

	AD	Controls	P-value
Ν	10	28	-
Gender	5F:5M	11F:17M	-
Age (years)	62.50±8.07	57.89±7.25	0.103
SBP (mmHg)	141.70±14.21	131.21±17.67	0.100
DBP (mmHg)	80.30±7.51	78.71±10.53	0.665
BMI	27.61±5.80	27.58±4.80	0.989
Glucose	4.40±1.44	4.99±0.96	0.163
Triglycerides	1.28±0.60	1.18±0.36	0.542
HDL-C (mmol/L)	1.33±0.25	1.16±0.29	0.121
Total-C (mmol/L)	4.77±0.64	4.66±0.82	0.708
IOP (mmHg)	16.50±2.12	18.00±2.62	0.470
MBP (mmHg)	100.77±7.67	96.21±11.92	0.269
OPP	84.96±9.46	80.26±12.48	0.299
Fram Risk Score	10.67±3.28	9.84±5.96	0.697
MMSE score	23.60 ± 3.57	-	-

Table 4.1: Summary of the baseline characteristics of the study groups. P<0.05 is considered a significant difference. SBP: systolic blood pressure; DBP: diastolic blood pressure; BMI: Body mass index; HDL-C: High density lipoprotein cholesterol; Total C: Total cholesterol; IOP: Intraocular pressure; MMSE: mini mental state examination. The presented SBP, DBP and IOP values are the baseline readings taken on the morning of the study and do not represent the 24 hour or diurnal averages. OPP was subsequently calculated using these baseline values.

4.8.2 Systemic endothelial dysfunction

No significant differences were found between groups with regard to brachial artery FMD or NMD (p>0.05, table 4.2). Furthermore no significant differences in circulating vWF levels were identified between groups

	AD	Controls	p-value
FMD (%)	10.97±11.28	11.04±8.53	0.985
NMD (%)	25.95 (13.66-39.74)	20.80 (18.85-26.38)	0.894
vWF	118.34±64.28	120.95±52.23	0.920

Table 4.2: Systemic endothelial function. Data presented as mean \pm SD or mean (IQR) depending on distribution. P<0.05 (*) is considered significant. FMD: flow mediated dilation; NMD: nitroglycerine mediated dilation; vWF: von Willebrand factor.

4.8.3 Dynamic retinal vessel analysis

For ease of interpretation the DVA response profile was considered in two parts, the first

being the dilation response (baseline to maximum dilation) and the second being the

constriction response (maximum dilation to maximum constriction). Each flicker cycle was analysed individually using traditional SDRA analysis with the artery and vein being considered separately. The dynamic nature or slope of the retinal vascular response profiles were fully explored using the polynomial fitted curves generated via our novel computational analysis (MatLab R2010a; MathWorks Inc., Natick, MA) (see section 3.3.3.11).

Arterial Response

With regard to the arterial dilation response no significant differences in baseline diameter fluctuation (BDF), maximum diameter (MD), reaction time (RT) or baseline corrected flicker response (BFR) were found between groups on average (p>0.01, table 4.3). On consideration of each flicker cycle individually however the arterial RT was found to be significantly longer in AD patients compared to controls on both the 1st and 3rd flicker cycles (p=0.039, p=0.028 respectively) and significantly shorter on the 2nd flicker cycle (p=0.049) (table 4.4). Furthermore although no within groups differences were found in RT on progressing from flicker 1 to flicker 3, the nature of this progression was however found to differ between groups, with AD patients showing a reduction in RT on going from flicker 1 to 2 and healthy controls showing a contrasting increase (p=0.007, table 4.4). With regard to the dynamic nature of the arterial dilation response profile, evaluated using our novel matlab analysis, no significant difference in arterial dilation slope (Slope_{AD}) was however found between study groups (p>0.05, table 4.5).

With regard to the second part of the dynamic response curve, no significant differences were found in the arterial constriction response (MC%) or the arterial constriction response time (tMC) between groups (p>0.01, table 4.3). Furthermore on consideration of the dynamic nature of the arterial constriction response profile no significant differences in constriction slope (Slope_{AC}) were found between groups (p>0.05, table 4.5).

ARTERY	AD	Controls	p-value
BDF	5.44±2.11	4.86±1.77	0.420
MD (%)	5.78±3.25	5.56±2.02	0.811
RT	21.83 (16.33-31.33)	17.83 (15.50-24.33)	0.273
BFR	4.88 (3.18-7.35)	4.66 (3.72-5.76)	0.529
MC (%)	-3.29±1.56	-2.59±1.62	0.267
tMC	32.63±10.24	28.07±8.51	0.101

Table 4.3: Arterial vascular function parameters determined using dynamic retinal vessel analysis (DVA, IMEDOS GmbH, Jena, Germany). Data presented as mean ± SD or mean (IQR) depending on distribution. P<0.01 (*) is considered as significant.BDF: baseline diameter fluctuation; MD(%): percentage change in diameter from baseline to maximum; RT: reaction time BFR: baseline corrected flicker response MC(%): percentage constriction below baseline; tMC: constriction time.

ARTERY	AD	Controls	p-value	Within/Between groups ANOVA
RT Flicker 1	29.30±16.61	18.59±12.21	0.039*	0.007*
Flicker 2	16.30±11.48	23.68±9.26	0.049*	
Flicker 3	27.89±17.62	17.07±10.43	0.028*	
Within group ANOVA	0.093	0.059		

Table 4.4: Arterial vascular function parameters by flicker cycle. P<0.05 (*) is considered as significant on two factor repeated measures ANOVA. RT: reaction time.

DYNAMIC RESPONSE	AD	Controls	p-value
Arteries Slope _{AD}	0.265±0.221	0.278±0.153	0.853
Slope _{AC}	-0.194±0.070	-0.187±0.113	0.848

Table 4.5: Dynamic characteristics of the retinal vascular response profiles determined using our novel computational model. P< 0.05 (*) is considered significant. Slope _{AD}: slope of arterial dilation; Slope _{AC}: slope of arterial constriction

Venous Response

With regard to the venous dilation response no significant differences in maximum diameter (MD), reaction time (RT) or baseline corrected flicker response (BFR) were found between groups (p>0.01, table 4.6). The venous baseline diameter fluctuation (BDF) however was found to be significantly greater in our AD patients compared to controls both on average (p=0.001) and across each individual flicker cycle (F1: p=0.013; F2: p=0.004; F3: p=0.014, table 4.7). On consideration of the dynamic nature of the venous dilation response profile, evaluated using our novel matlab analysis, no

significant difference in the venous dilation slope was found between study groups (p>0.05, table 4.8).

With regard to the second part of the dynamic response curve, no significant differences were found in the venous constriction response (MC%) or constriction time (tMC) between groups (p>0.01, table 4.6). Furthermore on analysis of the dynamic nature of the venous constriction response no significant differences in the venous constriction slope (Slope_{VC}) were found between groups on average (p>0.05, table 4.8).

VEIN	AD	Controls	p-value
BDF	5.83 (5.10-8.34)	3.24 (2.04-4.03)	0.001*
MD (%)	6.12±3.14	5.18±2.69	0.352
RT	20.33 (18.67-24.50)	20.00 (18.67-24.00)	0.688
BFR	1.84 (-1.01-6.34)	3.18 (1.54-4.66)	0.265
MC (%)	-2.61±2.13	-1.99±1.34	0.341
tMC	29.74±6.20	33.98±8.05	0.348

Table 4.6: Venous vascular function parameters determined using dynamic retinal vessel analysis (DVA, IMEDOS GmbH, Jena, Germany). Data presented as mean \pm SD or mean (IQR) depending on distribution. P<0.01 (*) is considered as significant.BDF: baseline diameter fluctuation; MD(%): percentage change in diameter from baseline to maximum; BFR: baseline corrected flicker response MC(%): percentage constriction below baseline.

VEIN	AD	Controls	p-value	Within/Between groups ANOVA
BDF				
Flicker 1	6.78±3.15	3.48±2.00	0.013*	0.961
Flicker 2	6.07±3.45	2.92±1.13	0.004*	
Flicker 3	7.08±4.21	3.59±2.52	0.014*	
Within groups ANOVA	0.820	0.295		

Table 4.7: Venous vascular function parameters by flicker cycle. P<0.05 (*) is considered as significant on two factor repeated measures ANOVA. BDF: baseline diameter fluctuation.

DYNAMIC RESPONSE	AD	Controls	t-test p-value
Veins			
Slope _{VD}	0.295±0.172	0.218±0.124	0.214
Slope _{VC}	-0.204±0.193	-0.164±0.122	0.451

Table 4.8: Dynamic characteristics of the retinal vascular response profiles determined using our novel computational model. P< 0.05 (*) is considered significant. Slope $_{VD}$: slope of venous dilation; Slope $_{VC}$: slope of venous constriction.

4.8.4 Correlations

No significant correlations were found between the retinal vascular reactivity parameters (BDF, MD%, RT, MC%, tMC, DA, BFR, Slope) and the systemic vascular parameters (FMD) in either group (P>0.05). Interestingly however a significant correlation was found between the degree of cognitive impairment in our AD patients, as indicated by MMSE score, and the arterial RT on the 1st flicker cycle (R=-0.782, p=0.013, figure 4.1)



Figure 4.1 Correlation between arterial reaction time (RT) on the 1st flicker cycle and MMSE score

4.9 Discussion

4.9.1 Main findings

This study has revealed differences in the response of the retinal vasculature to flicker light between mild newly diagnosed AD patients and healthy age matched controls. Newly diagnosed AD patients took significantly longer to reach the point of maximum arterial dilation on two out of the three occasions in which their retinal vessels were challenged by flickering light and this alteration in response was found to correlate with their degree of cognitive impairment. Furthermore AD patients demonstrated a consistently increased fluctuation in venous baseline diameter prior to the onset of flicker across all three cycles in comparison to healthy controls. No significant differences however were found in systemic endothelial dysfunction between groups.

4.9.2 Systemic endothelial function

No significant differences in systemic vascular endothelial function, as measured by FMD, were found between groups in this study. This is in contradiction to previous research by Dede et al (2007) who demonstrated impaired brachial artery FMD in a group of diagnosed AD patients compared to healthy controls ⁶³¹. Such contradiction in findings between previous research and our present study could partly be accounted for by differences in the measurement protocol and/or differences in the severity of the AD patients assessed. Our present study included only mild newly diagnosed AD patients and therefore it could be hypothesised that, as endothelial dysfunction is known to occur much earlier at the microvascular level than at the macrovascular level in a disease process ⁷³⁴, signs of systemic endothelial dysfunction, although not yet detectable in our newly diagnosed AD patients on FMD, could become more apparent as the disease progresses. In line with this Dede et al ⁶³¹ found that the degree of endothelial dysfunction as indicated by FMD correlated with the severity of cognitive impairment in their AD patients, with endothelial dysfunction getting worse as the severity of the

disease increased. Further research however would be required to confirm this hypothesis before any firm conclusions can be drawn. Alternatively, it is possible that the small sample size of AD patients included in this study did not provide sufficient power to detect a difference in FMD between groups. Indeed on power analysis it was identified that a sample size of 12 would be required to provide 80% power at an α level of 0.05, however only 10 AD participants were able to be recruited. Post-hoc analysis, conducted using G-power 3, reveals the sample sizes achieved in this study were in fact only able to provide a power of 69%. This could potentially explain these non-significant findings.

4.9.3 Retinal vessel reactivity

A large body of evidence exists linking AD to the presence of cerebral vascular dysfunction and highlighting the involvement of the ocular circulation in the AD disease process ^{3, 567, 617, 620, 621, 668, 669}. Furthermore, the easy access to the neural and vascular tissue at the retinal level, as well as the many anatomical and physiological similarities shared by the ocular and cerebral microcirculation, makes the retina an ideal screening target in cerebrovascular disease. The use of techniques such as the DVA method used in the present study, which are aimed at assessing retinal microvascular function, could therefore not only offer information regarding general microvascular function in AD patients but also offer an assessment of the potential risk for future decline and development of systemic vascular disease. In the present study, DVA analysis revealed how our newly diagnosed mild AD patients without manifest systemic vascular disease took significantly longer to reach the point of maximum dilation on two out of the three occasions in which their retinal vessels were challenged by flickering light. These results indicate that some form of microvascular dysfunction, detectable at the retinal level, does indeed appear to exist in AD patients; the cause of this vascular dysfunction, however, can only be hypothesised at this point.
The retinal vascular response to flicker light occurs due to an increase in retinal metabolic demand and is predominantly a neurovascular coupling driven response ^{226,} ^{685, 686, 691, 692}. It could therefore be hypothesised that the altered retinal vessel reactivity demonstrated in our AD patients is indicative of a disturbed neurovascular coupling mechanism, possibly related to endothelial dysfunction, a decreased bioavailability of the vasodilator NO or an alteration in the activity of astrocytes in these patients, all of which are known to be key mediators of the neurovascular response ⁶⁰⁰. Interestingly, both disturbed neurovascular coupling and dysfunction of the vascular endothelium have been previously linked to the aetiology of AD ^{631, 632, 735, 736}, as have alterations in astrocyte activity 737-739 and in NO production/release 565, 740. Such alterations in the production/release of NO could additionally be attributed to either cholinergic receptor degeneration in AD⁷⁴¹ and the subsequent reduction in acetylcholine mediated NO release, or to Aß deposition and the subsequent impairment of neuronal NO production. Indeed acetylcholine receptor stimulation and subsequent release of NO has been previously identified to play a role in the retinal vasodilation response to flicker light in rabbits ^{742, 743} and the accumulation of Aß, a key feature of AD, has been previously identified to reduce NO production in retinal neuronal cultures ⁷³⁰, adding plausibility to this theory.

Aside from disturbed neurovascular coupling mechanisms it could alternatively be hypothesised that factors such as relative arterial inertia due to either increased vascular stiffness or vasospasm, similar to that documented in AD and cognitively impaired patients at the cerebral level ^{636, 744, 745} may also contribute to the observed abnormalities in retinal vascular function. Such an alternative hypothesis could also partly explain the dissimilarities in reaction time observed in our AD patients on consecutive flicker cycles, in that the presence of limited arterial elasticity could theoretically lead to an incomplete baseline recovery after the initial stimulation cycle, reducing the time taken to then reach the point of maximum dilation on the subsequent flicker cycle. Exhaustion of vasoactive

mediator reserves, such as that of NO, which is a feature commonly associated with vascular endothelial dysfunction ²⁰¹, could then theoretically override this and prolong the reaction time again on the final cycle. Further research however would be required to validate these assumptions.

With regard to the venous dynamic retinal vessel response profile our AD patients demonstrated an increased fluctuation in baseline vessel diameter across all three cycles, prior to the onset of flicker, which was not replicated by our healthy controls. Consideration of baseline diameter fluctuation (BDF) was first recommended by Nagel et al ⁶⁸⁶ as a way of taking into account the effect of the spontaneous variations in vessel diameter that occur under normal resting conditions on the observed response of the vasculature to flicker light stimulation, however it is a parameter which is not commonly reported in the literature and which has, to date, mainly been considered in regard to the retinal arteries, where it has been tentatively linked to vascular disturbance in the form of instability or increased variation in vascular tone or rigidity ^{695, 746-748}. As such the cause and relevance of increased BDF in the venous circulation is currently unclear. Retinal veins are generally thought to play a more secondary role in retinal autoregulation, perhaps providing a fine tuning of the regulation response following the active reaction of the retinal arteries and instigating a regulatory contribution passively in response to increased blood flow ⁶⁸⁵. Interestingly increased retinal venous diameters have been previously linked to impaired cerebral blood flow and have been suggested as a marker of both retinal and cerebral ischemia and hypoxia ^{652, 749-751}. The finding here is somewhat different as fluctuations in diameter have been assessed dynamically as opposed to vessel diameter measurements being taken statically from photographs. Nevertheless, as both ischemia and hypoxia have been well linked to the development of AD, it could be hypothesised that the increased fluctuation in baseline diameter observed in these patients could be reflective of early hypoxic changes and perhaps an

increased risk of future damage. Further investigation however would be necessary to confirm these hypotheses.

4.9.4 Vascular function vs. Cognitive impairment

A significant positive correlation was found between the abnormal retinal arterial reaction to flickering light and the level of cognitive impairment in AD patients, suggesting a possible link may exist between the degree of vascular dysfunction and disease severity even in the early stages of the disease process. This is supported by previous research in which vascular dysregulation at the cerebral level has been shown to become more pronounced with increasing severity of AD ⁶²⁹ and by the finding of a correlation between cognition and the geometry of the retinal vessels in elderly people after correcting for age, visual acuity and apolipoprotein E status ⁷⁵². The possibility that the degree of vascular dysfunction at the retinal level could be a sensitive predictor of cognitive decline in mild AD patients emphasises the important role that vascular factors might play in the aetiology of the disease. In addition, it is tempting to propose that examining the function of retinal microvasculature could predict future cognitive decline in patients suffering from AD; nevertheless, more research is necessary to validate our presumption.

4.10 Conclusion

This study demonstrates for the first time that abnormalities in retinal vascular dysfunction, potentially indicative of a disturbed neurovascular coupling response, may be present in mild newly diagnosed AD patients and that the extent of these abnormalities may be a sensitive indicator of the degree of cognitive impairment. More research however would be necessary to validate our assumptions.

4.11 Limitations

One factor potentially limiting the conclusions which can be drawn from this study is the small sample size of AD patients. The strict inclusion/exclusion criteria were necessary to avoid any possible unwanted influences on the measured vascular parameters; however they also made patient recruitment a challenge. Regular contact was always maintained with the dementia team in order to try and maximise the recruitment of suitable participants and additionally all members of the team received a thorough briefing on the study protocol, inclusion and exclusion criteria. Although significant differences in DVA parameters were found, as mentioned previously, it is possible that the small sample size of AD patients and subsequent lack of statistical power may have limited the conclusions which can be drawn for other parameters, especially FMD. The results of this study do however go some way to providing a positive indicator that vascular dysfunction is a factor involved in the pathogenesis of AD and offer a relevant and adequate platform for further research in this area. Future research to include more patients with various degrees of cognitive impairment would be beneficial to validate these findings.

5. Ocular and Systemic Vascular Abnormalities in Newly-Diagnosed Normal Tension Glaucoma Patients

5.1 Abstract

Purpose: To investigate signs of ocular and systemic vascular abnormalities in newly diagnosed and previously untreated normal tension glaucoma patients (NTG). **Methods**: Retinal vascular reactivity to flickering light was assessed in 19 NTG and 28 healthy age-matched controls by means of dynamic retinal vessel analysis (DVA, IMEDOS GmbH, Jena, Germany). Using a newly developed computational model, the entire dynamic vascular response profile to flicker light was imaged and used for analysis. Assessments of systemic endothelial function (FMD), carotid intima-media thickness (IMT) and pulse wave analysis (PWA) were conducted on all participants, along with blood analyses for von Willebrand factor (vWf), glucose and lipid metabolism markers.

Results: Patients with NTG demonstrated an increased carotid IMT (p=0.015) and an elevated PWA augmentation index (p=0.017) in comparison to normal controls, as well as an altered retinal arterial vascular response profile and a significantly steeper retinal arterial constriction slope (slope _{AC}) following flicker light stimulation (p=0.031). No significant differences in systemic endothelial function were identified between groups (p>0.05)

Conclusions: Newly diagnosed and untreated NTG patients showed signs of subclinical vascular abnormalities at both ocular and systemic levels. Due to the importance of circulation-related pathologies in the aetiology and progression of this type of glaucoma early detection of such abnormalities and the introduction of disease modifying interventions could prove beneficial for disease prognosis.

5.2 Introduction

Although abnormal intraocular pressure (IOP) is still considered a crucial risk factor for the development of normal tension glaucoma (NTG) ^{36, 37}, other local culprits including ocular microvascular dysregulation ³²⁵ have been implicated in its aetiology. Moreover, systemic vascular pathologies, such as peripheral vasospasm ^{26, 353}, endothelial ^{42, 384} and autonomic nervous system (ANS) dysfunctions ^{413, 414} as well as high levels of oxidative stress ⁷⁵³ have also been reported in NTG patients, along with a high incidence of systemic cerebrovascular and/or cardiovascular disease (CVD) ^{24, 323, 444, 468, 477, 478, 754}. The impact and inter-relationships of such vascular alterations, which appear to affect both the macro- and micro-vascular beds, is however still poorly understood and many questions still remain around the exact pathogenesis of glaucomatous optic neuropathy (GON).

A common link between micro- and macro-vascular pathologies is endothelial dysfunction ²⁵⁵ and the assessment of this type of dysfunction across different vascular beds has been shown to be a good predictor of future vascular risk ^{263, 755, 756}. Interestingly, the presence of endothelial dysfunction has previously been reported in the systemic macro-circulation of NTG patients ⁴² however, in the course of vascular disease development, the microcirculation is thought to be affected much earlier than the macro-circulation ⁷³⁴. Dynamic retinal vessel analysis (DVA) allows an assessment of the retinal microvascular response to stress and using this technique, a diminished venous dilation response to flickering light has previously been reported in early primary open angle glaucoma (POAG) ⁴¹. No such studies have however yet been carried out in NTG patients. Additionally the majority of currently published DVA analyses focus almost solely on the dilatory response of the retinal vasculature to flickering light however, in order to completely understand the mechanisms behind vascular

entire profile of the retinal vascular response, including vessel recovery, should be considered. With this in mind the present study aimed, through the use of a new computational analysis of the complete DVA response profile, to study both the retinal dilation and constriction responses to flickering light in newly diagnosed and untreated NTG patients and to explore the coexistence of both micro- and macro-vascular abnormalities in these patients through the simultaneous evaluation of systemic macrovascular function markers. The assessment of vascular function in NTG patients is of particular interest as IOP is thought to play a less prominent role in the NTG disease process and hence diagnosis and treatment of the condition more commonly looks to IOP-independent causes, such as vascular alterations, whose roles are yet to be fully clarified.

5.3 Aims

The aim of this study was to assess ocular microvascular reactivity, in the form of both the retinal dilation and constriction responses to flickering light in otherwise healthy, newly diagnosed and untreated NTG patients and to fully explore the 'sick eye in a sick body' concept of glaucoma, through assessment of systemic endothelial function by means of FMD, systemic arterial stiffness and carotid artery IMT.

5.4 Hypothesis

Newly diagnosed and previously untreated NTG patients demonstrate altered retinal vessel reactivity to flicker light stimulation in conjunction with signs of vascular dysfunction at the systemic level.

5.5 Subjects and Methods

Newly diagnosed and previously untreated NTG patients and healthy age matched controls were recruited for this study. The recruitment details, inclusion and exclusion criteria for these patients was detailed in section 3.1. The investigative procedures performed in this study are outlined below and were conducted in accordance with the protocols outlined in section 3.2:

- 1. Preliminary tests
- 2. Fasting venous blood sample obtained
- 3. BP measurement
- 4. Assessment of retinal vessel reactivity (DVA)
- 5. Pulse wave analysis (PWA)
- 6. Intima-media thickness (IMT) measurement
- 7. Assessment of systemic endothelial function (FMD)

5.6 Statistical Analysis

The Kolmogorov-Smirnov test was used to determine the distribution of the data. As normality could not be confirmed in all cases the differences between groups were assessed using either the Student t-test for independent variables or Mann-Whitney-U test as appropriate and reported as either mean±SD or mean(IQR) accordingly. The only exception to this was with regard to IMT, where multiple regression analysis revealed systolic BP to be an influencing factor and hence a one-way ANCOVA, followed by Tukey's post-hoc analysis, was subsequently conducted. Correlations were determined according to Pearson's method or Spearman's rank method as appropriate. P-values of less than 0.05 were considered significant, except in certain cases where a stricter p-value of less than 0.01 was adopted in order to correct for multiple comparisons and minimise bias towards type I errors. All analyses were performed using Statistica, version 6.0, Statsoft, Tusla, OK.

5.7 Power calculations

With regard to DVA, new measurement parameters and novel analysis methods were used in this study in patient groups which have not previously been examined with these techniques. Power calculation was therefore based on the results of previous studies which share the most similar protocols to that of the present research and was conducted using the computer based programme, GPower 3⁷³³. A retinal vessel reactivity response of 6% with a standard deviation of 2.5% is considered normal on the basis of previous research and around a 50% alteration in this response has been shown to be clinically significant ⁶⁸⁶. With regard to FMD, a brachial artery dilation response of 7.5% with a standard deviation of 2.3% is considered normal on the basis of previous glaucoma research and around a 40% alteration in this response has been shown to be clinically significant ⁴². With regard to PWA AIx and IMT similar deductions to that above, regarding clinical significance, were made on the back of previous research ^{482, 715, 726}. Analysis by t-tests for independent samples was required in this study. Taking this into consideration, it was calculated that, in order to provide a statistical power of 80% at an α level of 0.05, a sample size of between 7-22 per group would be required (10 DVA, 7 FMD, 22 PWA, 9 IMT). The aim was to therefore recruit at least 22 participants in each study group.

5.8 Results

A total of 25 NTG patients and 34 healthy controls were screened for inclusion in the present study however, in order to ensure all participants were matched on critical factors such as age and hypertensive status, all those under the age of 45, over the age of 75 or with a MBP of greater than 115 mmHg had to be excluded from analysis. Additionally, following the careful review of the obtained images, any patients who exhibited poor or incomplete results were also excluded and 19 NTG patients (11 female, 8 male) and 28 healthy age matched controls (12 female, 16 male) were

included in the final analysis. The number of NTG patients ultimately fell slightly below the optimum sample size suggested by power calculation; however post-hoc analysis using GPower 3 revealed the obtained sample sizes were still able to provide a statistical power of 82%.

5.8.1 Baseline values

There were no significant differences in age, systemic BP, BMI, triglycerides, glucose, HDL cholesterol levels and total cholesterol levels between the two groups (all p>0.05). Furthermore there were no significant differences in IOP, MABP, OPP or Framingham risk score between groups (all p>0.05) (Table 5.1). The number of subjects with well controlled high BP was also similar in both groups (NTG: n=5 and Controls: n=9, p>0.05; Chi-square test).

	NTG	Controls	P-value
Ν	19	28	-
Gender	11F:8M	12F:16M	-
Age (years)	60.16±12.13	56.82±7.54	0.251
SBP (mmHg)	130.06±15.37	127.00±18.52	0.563
DBP (mmHg)	78.39±11.14	76.39±10.91	0.551
BMI	27.92±4.20	26.81±4.22	0.407
Glucose	5.08±0.86	4.93±0.94	0.594
Triglycerides	1.04±0.31	1.12±0.36	0.403
HDL-C (mmol/L)	1.12±0.29	1.18±0.32	0.556
Total-C (mmol/L)	4.58±1.04	4.75±0.86	0.560
IOP (mmHg)	17.40±1.80	16.71±2.36	0.460
MBP (mmHg)	95.61±10.93	93.26±13.00	0.529
OPP	76.09±12.66	80.85±12.87	0.223
Fram Risk Score	9.67±7.50	8.31±5,24	0.499

Table 5.1: Summary of the baseline characteristics of the study groups. P<0.05 is considered a significant difference. SBP: systolic blood pressure; DBP: diastolic blood pressure; BMI: Body mass index; HDL-C: High density lipoprotein cholesterol; Total C: Total cholesterol; IOP: Intraocular pressure. The presented SBP, DBP and IOP values are the baseline readings taken on the morning of the study and do not represent the 24 hour or diurnal averages. OPP was subsequently calculated using these baseline values.

5.8.2 Systemic vascular parameters

5.8.2.1 Pulse wave analysis

Pulse wave analysis AIx was found to be significantly greater in newly diagnosed NTG patients compared to healthy age matched controls (p=0.017) (Table 5.2).

5.8.2.2 Intima-media thickness measurement

Carotid artery IMT was found to be significantly higher in NTG patients compared to controls (p=0.015) (Table 5.2). Multiple regression analysis revealed significant positive correlations between IMT and SBP (p=0.035); therefore, the given p-value was calculated using ANCOVA correcting for the effect of SBP.

	NTG	NTG Controls	
PWA: Alx (%)	26.06 ± 11.25	17.18 ± 9.32	0.017*
IMT (cm)	0.064 ± 0.015	0.042 ± 0.223	0.015*

Table 5.2: Systemic macro-vascular parameters. P<0.05 (*) is considered significant. PWA:AIx: pulse wave analysis: augmentation index; IMT: intima-media thickness.

5.8.2.3 Systemic endothelial function

No significant differences were found between groups with regard to brachial artery

FMD or NMD. (p>0.05, table 5.3). Furthermore no significant differences in circulation

vWF levels were identified between groups (P>0.05).

	NTG	Controls	p-value
FMD (%)	11.01±6.97	12.55 ± 9.21	0.669
NMD (%)	25.89 (15.12-52.96)	22.66 (19.69-29.63)	0.665
vWF	118.17±63.46	126.58±55.35	0.696

Table 5.3: Systemic endothelial function. P<0.05 (*) is considered significant. Data presented as mean \pm SD or mean (IQR) depending on distribution. FMD: flow mediated dilation; NMD: nitroglycerine mediated dilation; vWF: von Willebrand factor.

5.8.3 Dynamic retinal vessel analysis

For ease of interpretation, the dynamic retinal vessel profile curve was considered in two parts, the first part being the dilation response (baseline to maximum dilation) and the second part being the constriction response (maximum dilation to maximum constriction). The principle results are given based on the average of the 3 flicker cycles with the artery and vein being considered separately. Although each flicker cycle was also analysed individually this analysis did not yield any additional information and has therefore not been presented. The dynamic nature of the retinal vascular response profiles were fully explored using the polynomial fitted curves generated via our novel computational analysis (MatLab R2010a; MathWorks Inc., Natick, MA). These curves are illustrated in figures 5.1 and 5.2 for the artery and vein respectively.



Figure 5.1: Averaged arterial response profile for NTG patients and healthy controls generated through Matlab. Demonstrates the significantly steeper arterial constriction slope found in NTG patients and the apparently greater percentage constriction response below baseline



Figure 5.2: Averaged venous response profile for NTG patients and healthy controls generated through Matlab.

Arterial response

Following the adoption of a stricter p value of less than 0.01 to account for the effects of multiple comparisons, no significant differences in the maximum arterial dilation (MD%), reaction time (RT), arterial baseline corrected flicker response (BFR) or the baseline diameter fluctuation (BDF) was found between study groups (all p>0.01, table 5.4). Furthermore with regard to the second part of the arterial dynamic profile curve, which looks at the activity of the retinal vessels following dilation and on cessation of flicker, again following the adoption of a stricter p-value of 0.01, no significant differences in the arterial constriction response (MC%) or constriction time (tMC) was also found between groups (p=0.028, p=0.462 respectively, table 5.4)

On consideration of the dynamic nature of the arterial dilation response, evaluated using our novel matlab analysis however, despite no significant differences being found between groups with regard to slope (p>0.05, table 5.5), the arterial dilation profile did appear to follow a different path in NTG patients compared to controls, with healthy controls demonstrating a 'two humped' arterial dilation response which was absent in our NTG patients (figure 5.2). Furthermore the arterial constriction slope (Slope_{AC}) was found to be significantly steeper in NTG patients compared to healthy controls on consideration of the dynamic nature of the arterial constriction profile, (p=0.031, table 5.5).

ARTERY	NTG	Controls	p-value
BDF	7.21 (4.50-9.08)	4.66 (3.51-5.45)	0.018
MD (%)	4.28 (2.80-6.31)	3.70 (2.29-4.44)	0.138
RT	21.09 ± 8.67	18.59 ± 5.79	0.256
BFR	-0.04±2.41	0.50±1.95	0.423
MC (%)	-2.64 ± 1.61	-1.56 ± 1.46	0.028
tMC	34.99 ± 8.30	32.99 ± 8.92	0.462

Table 5.4: Arterial vascular function parameters determined using dynamic retinal vessel analysis (DVA, IMEDOS GmbH, Jena, Germany). Data presented as mean \pm SD or mean (IQR) depending on distribution. P<0.01 is considered as significant (*).BDF: baseline diameter fluctuation; MD(%): percentage change in diameter from baseline to maximum; BFR: baseline corrected flicker response MC(%): percentage constriction below baseline.

DYNAMIC RESPONSE	NTG	Controls	p-value
Arteries			
Slope _{AD}	0.300 ± 0.159	0.250 ± 0.149	0.294
Slope _{AC}	-0.236 (-0.449—0.158)	-0.153 (-0.207—0.110)	0.031*

Table 5.5: Dynamic characteristics of the retinal vascular response profiles determined using our novel computational model. Data presented as mean \pm SD or mean (IQR) depending on distribution. P< 0.05 (*) is considered significant. Slope _{AD}: slope of arterial dilation; Slope _{AC}: slope of arterial constriction

Venous response

Following the adoption of a stricter p value of less than 0.01 to account for the effects of

multiple comparisons, no significant differences were found between groups with regard

to venous MD%, RT, BFR or BDF (p>0.01, table 5.6). Furthermore, with regard to the

second part of the dynamic response curve, no significant differences were found in the

venous constriction response (MC%) or the constriction response time between groups

(p>0.01, table 5.6)

On consideration of the dynamic nature of the venous response profile, evaluated using our novel matlab analysis, no significant difference in the venous dilation slope or the venous constriction slope (Slope_{VD}) was found between groups (p>0.05, table 5.7).

VEIN	NTG	Controls	t-test p-value
BDF	4.55 ± 1.72	3.36 ± 1.38	0.019
MD (%)	4.38 ± 2.16	4.15 ± 2.33	0.753
RT	22.63 ± 7.12	19.99 ± 4.08	0.128
BFR	-0.04 ± 2.56	1.72 ± 2.36	0.026
MC (%)	-0.61 ± 0.86	-0.93 ± 1.04	0.326
tMC	39.31 ± 6.93	37.73 ± 9.57	0.575

Table 5.6: Venous vascular function parameters determined using dynamic retinal vessel analysis (DVA, IMEDOS GmbH, Jena, Germany). P<0.01 (*) is considered as significant.BDF: baseline diameter fluctuation; MD(%): percentage change in diameter from baseline to maximum; BFR: baseline corrected flicker response MC(%): percentage constriction below baseline.

DYNAMIC RESPONSE	NTG	Controls	t-test p-value
Veins			
Slope _{VD}	0.235 ± 0.132	0.225 ± 0.133	0.816
Slope _{VC}	-0.147 ± 0.090	-0.164 ± 0.124	0.630

Table 5.7: Dynamic characteristics of the retinal vascular response profiles determined using our novel computational model. P< 0.05 (*) is considered significant. Slope $_{VD}$: slope of venous dilation; Slope $_{VC}$: slope of venous constriction.

5.8.3.1 Correlations

No significant correlations were found between the retinal vascular reactivity parameters

(BDF, MD, MC, DA and BFR, Slope) and the systemic vascular parameters (AIx, IMT

and FMD) in either of our groups (all p>0.05).

5.9 Discussion

5.9.1 Main findings

Using a novel computational model, our results reveal that newly diagnosed, previously untreated NTG patients with no overt clinical signs of systemic vascular disease show an altered retinal arterial response profile to flicker light along with a significantly steeper retinal arterial constriction response following cessation of flicker. Moreover, subclinical signs of systemic arterial stiffness along with increased carotid artery IMT are consistently exhibited by our NTG patients but not by our age-matched controls.

5.9.2 Systemic macro-vascular alterations in NTG

Both the presence of systemic vascular disease and alterations in systemic vascular function have been well linked with the development of glaucoma, particularly NTG ^{42, 311,} ^{446, 757, 758}. Nevertheless, previous research into the role of systemic measures, such as arterial stiffness in the aetiology of glaucoma have given inconsistent results with some showing a strong association ⁴⁸¹⁻⁴⁸³ and others showing no association at all ^{318, 485}. Such variability in results could be partly accounted for by differences in the inclusion/exclusion of patients suffering from already diagnosed systemic vascular disease in these studies, especially as arterial stiffness is a measure of vascular function ⁷⁵⁹. In this study we have demonstrated increased arterial stiffness and IMT in our NTG patients. Patients with well controlled hypertension were included in this study, however a similar number of subjects with such status were included in both the NTG and control group. The present report therefore simply shows that newly diagnosed NTG patients present with stronger signs of systemic vascular pathology, which could have possibly contributed to the onset of the ocular disease. In light of this it could be suggested that screening for such abnormalities in at risk or newly diagnosed individuals and addressing them through appropriate interventions could be beneficial with regard to disease prognosis.

With regard to macrovascular function, no significant differences in systemic endothelial function were found between NTG patients and controls in this study on brachial artery FMD or on the evaluation of the circulating endothelial marker, vWf. These findings are in contrary to the plentiful evidence in the literature indicating the involvement of systemic endothelial dysfunction in the glaucomatous disease process and to the findings of Su et al ⁵² who demonstrated impaired FMD in NTG patients. Such contradiction in findings between previous research and our present study could partly be accounted for by differences in both the measurement protocol and the patient inclusion criteria used, particularly as it is unclear whether the NTG patients included by Su et al were newly diagnosed and untreated or at a more advanced stage of the disease process. With this is mind it could be hypothesised that, as endothelial dysfunction is known to occur much earlier at the microvascular level than at the macrovascular level in a disease process ⁷³⁴, signs of systemic endothelial dysfunction, although not yet detectable in our newly diagnosed NTG patients on FMD, could become more apparent as the disease progresses. Further research however would be required to confirm this hypothesis and it is worth considering that there are studies which have previously indicated the presence of systemic endothelial dysfunction even in newly diagnosed NTG patients using alternative and now less favoured methods of assessment such as venous occlusion plethysmography ^{382, 384}, however, these studies whilst conclusive, did involve very small sample sizes and it is unclear to what extent coexisting systemic disease was considered.

The states of the macro- and microcirculation are known to be closely related ⁷⁶⁰⁻⁷⁶² and in addition to systemic vascular abnormalities, NTG has also been associated with ocular microcirculatory changes. Indeed, Oettli et al ²⁹⁵ recently demonstrated an increased retinal vessel rigidity in untreated NTG patients that correlated with the level of glaucomatous damage. Our present results also demonstrate that in addition to

systemic vascular abnormalities, newly diagnosed and untreated NTG patients show signs of abnormal vascular function at the retinal vessel level.

5.9.3 Retinal microvascular reactivity in NTG

Considering the arterial dynamic retinal vessel profile first, no differences in dilation response were found between groups in this study; however the dilation profile did appear to follow a different path in NTG patients compared to controls. As illustrated in figure 5.2, our healthy controls demonstrated a 'two humped' arterial dilation response that is consistent with that previously demonstrated by Lanzl et al in a sample of young healthy volunteers ⁷⁶³ and to a lesser extent in a sample of older healthy volunteers ⁶⁹⁶. Our patients fall between the age ranges included in the above research and whilst the 'two humped' aspect was present in our healthy control patients it was absent in our age-matched NTG patients. Lanzl et al 696 propose that such a 'two humped' profile is, amongst other factors, reflective of two separate systems that contribute to arterial dilation, the first being a fast onset, short duration system, mediated by endothelial NOsynthase and free NO and the second being a slow onset, long duration system which represents the summation of dilation and constriction factors. As no differences were found in age, maximum dilation or the time taken to reach maximum dilation between our groups it is difficult to determine which of these systems, if any, may be affected in our NTG patients and lead to the loss of this 'two humped' pattern. It could be hypothesised that whilst there is sufficient NO production/availability to induce a dilatory response in the retinal arteries in our NTG patients initially, this NO is either of short supply or is rapidly deactivated after release, leading to a loss of the second dilatory phase and a predominance of vasoconstrictive factors such as ET-1. In line with this assumption, our newly diagnosed NTG patients demonstrated a steeper arterial constriction slope compared to our healthy age matched controls on cessation of flicker. Slope calculation considers the interaction between the percentage change in vessel diameter and the rate at which this change occurs. The relevance of a steeper arterial

constriction slope, as demonstrated in our NTG patients, is currently unclear as very few previous studies looking at vascular reaction to flickering light have investigated the constriction responses in detail. One factor which could contribute towards such a steeper arterial constriction slope is an enhanced percentage constriction response below baseline following cessation of flicker. Re-establishment of baseline diameter following stimulation is part of a complex inherent biological control process and a certain degree of overshoot in vascular diameter below baseline is expected even in healthy individuals following flicker light exposure ^{685, 693}. Interestingly, in a similar way to our newly diagnosed NTG patients, a greater overshoot in arterial vessel diameter below baseline has previously been noted by Gugleta et al ⁷⁴⁶ in individuals with PVD, a condition known to be associated with both vascular dysfunction and NTG ^{26, 300, 334, 353,} ^{360, 383}. With this in mind we could therefore propose that the steeper constriction response demonstrated in our NTG patients could be indicative of some form of vascular dysfunction. The cause of this vascular dysfunction, however, can only be hypothesised at this point. One potential contributing factor, as mentioned previously, could be a predominance of vasoconstrictive factors such as ET-1, the levels of which, although not measured here, have previously been shown to be increased in NTG patients and vasospastic patients ^{340, 355, 356 391}. Alternatively an abnormal proliferation of astrocytes, which are known to be key mediators of the neurovascular coupling response ⁶⁰⁰ could also contribute to these findings. Indeed abnormal astrocyte proliferation has been previously linked to both the development of glaucoma¹⁴⁴ and to the re-establishment of vasomotor tone following neuronal stimulation ^{764, 765}. Further investigations however would be needed to validate these hypotheses.

With regard to the venous dynamic retinal vessel profile to flicker light stimulation, prior to correction for multiple comparisons a significantly reduced venous baseline corrected flicker response (BFR) was demonstrated in our newly diagnosed previously untreated NTG patients, which reinforces previous findings by Garhofer et al ⁴¹ who identified an

impaired retinal venous dilation response to flicker in early POAG patients. Similarly, baseline diameter fluctuation (BDF) was found to be significantly greater in both arteries and veins of those with NTG, however again this significance was lost on correction for multiple comparisons. Despite this there is evidence to suggest that increased BDF could be a relevant indicator of vascular disturbance ^{695, 746}, perhaps indicating instability or increased variation in vascular tone or rigidity ^{747, 748} and therefore it may be a parameter worth exploring further in future research, especially as increased arterial flow pulsations, associated with aging and arterial stiffening have been linked with the

5.9.4 Ocular microvasculature vs. Systemic macrovasculature

No correlations were found between the subclinical signs of systemic vascular disease and the retinal vascular response to flickering light in NTG patients; however, due to the complexity of comparing parameters from two different vascular beds using different measuring techniques this may not be surprising. Interestingly, this finding is in concordance with that of a number of recent studies which have similarly demonstrated no direct correlation between anomalies identified at the macro- and micro-vascular levels in various disease states ^{710, 768, 769}. This supports the view that, whilst the states of these two systems may be closely related, they may in fact act independently of each other with regard to the rate and pathophysiological mechanism of their vascular dysfunction development. The possibility that each may drive or influence the other however cannot be excluded ⁷⁷⁰ and this requires further investigation.

5.10 Conclusion

In conclusion this study demonstrates the coexistence of static macro-vascular abnormalities and functional retinal microvascular abnormalities in newly diagnosed and

previously untreated NTG patients, highlighting the importance of considering multi-level circulation-related pathologies in the development of this type of glaucoma.

5.11 Limitations

Only a moderately sized cohort of NTG patients could be recruited for this study which could potentially limit the statistical power of the analysis and the conclusions which can be drawn from the presented data. The overall limitations of the presented research are discussed further in section 8.3.

6. Ocular Vascular Dysregulation in AD compares to both POAG and NTG

6.1 Abstract

Purpose: To investigate and compare vascular function at both the ocular and systemic level in newly diagnosed mild AD patients, newly diagnosed normal tension glaucoma (NTG) patients and newly diagnosed primary open angle glaucoma (POAG) patients.

Methods: Retinal vessel reactivity to flickering light was assessed in 10 AD, 19 POAG, 19 NTG and 20 healthy age matched control patients by means of dynamic retinal vessel analysis (DVA, IMEDOS, GmbH, Jena, Germany). Systemic vascular endothelial function was additionally assessed in all patients by means of brachial artery flow mediated dilation and all patients underwent BP measurements and blood analysis for glucose and lipid metabolism markers.

Results: AD patients demonstrated altered arterial retinal vessel reactivity to flicker light stimulation which was comparable to that of POAG patients (p=0.013) and altered baseline venous reactivity which was comparable to that of NTG patients (p=0.001). Neither were replicated in healthy controls. No significant differences in systemic endothelial function were identified between groups (p>0.05).

Conclusion: AD patients demonstrate similar signs of retinal vascular dysfunction to both POAG and NTG patients at the early stages of their disease process, providing support for the concept of a common underlying vascular aetiology in both conditions and highlighting the need to consider ocular health in AD and cerebral health in glaucoma.

6.2 Introduction

The possibility that AD and glaucoma may share a common underlying vascular aetiology has been increasingly realised over recent years. This stems not only from their obvious similarities as chronic neurodegenerative diseases associated with aging but also from the results of numerous studies which have identified alterations in cerebral perfusion and dynamic cerebral autoregulation, more commonly associated with AD, in both POAG ³²³ and NTG patients ^{313, 323, 468}. Moreover reduced retinal perfusion and increased ONH cupping, more characteristic of glaucoma; have also been demonstrated in AD patients ^{663, 665}. Despite these apparent associations however there is currently no research which directly explores and compares the nature and coexistence of vascular function abnormalities in both AD and glaucoma patients simultaneously. Interestingly, in Chapters 4 and 5 of this thesis, evidence of vascular alteration was identified in both newly diagnosed AD patients and newly diagnosed NTG patients, individually, in comparison to healthy controls. In order to expand on these findings, this study aims, through the simple, non-invasive evaluation of both dynamic retinal vessel function and systemic vascular endothelial function, to investigate whether such ocular and systemic vascular alterations coexist in newly diagnosed glaucoma and AD patients and to explore the similarities and differences between them. Furthermore, through consideration of two separate categories of glaucoma patients, namely POAG and NTG, this study additionally aims to determine whether any vascular alterations identified in AD patients are more closely related to those of patients diagnosed with POAG or to those of patients diagnosed with NTG. Increasing our understanding of the nature of vascular dysfunction in all of these conditions, if present, could provide an important insight into their disease aetiology and lead to better awareness and understanding of their potential coexistence.

6.3 Aims

The aim of this study is to investigate whether ocular and systemic vascular alterations coexist in newly diagnosed POAG, NTG and AD patients and to explore the similarities and differences between them. Furthermore, this study aims to determine whether vascular alterations in AD, if identified, related to both POAG and NTG.

6.4 Hypothesis

Newly diagnosed mild AD patients will demonstrate signs of vascular dysfunction at the ocular and systemic level which are comparable to that also demonstrated by newly diagnosed POAG patients and/or NTG patients.

6.5 Subjects and Methods

Newly diagnosed and previously untreated NTG and POAG patients, mild newly diagnosed AD patients and healthy age matched controls were recruited for this study. The recruitment details, inclusion and exclusion criteria for these patients was detailed in section 3.1. The investigative procedures performed in this study are outlined below and were conducted in accordance with the protocols outlined in section 3.2:

- 1. Preliminary tests
- 2. Fasting venous blood sample obtained
- 3. Blood pressure measurement
- 4. Assessment of retinal vessel reactivity (DVA)
- 5. Assessment of systemic endothelial function (FMD)

6.6 Statistical Analysis

All data were reported as mean \pm standard deviation. The Kolmogorov-Smirnov test was used to determine the distribution of the data. Multivariate analysis was performed to

determine the influence of age, BMI, BP and circulating markers on the measured variables. Differences between groups were subsequently assessed using one-way ANOVA or ANCOVA, as appropriate, followed by Tukey's post hoc analysis. Two factor repeated-measures ANOVA was used to compare the retinal reactivity responses across each flicker cycle. In cases where the normality of the data could not be confirmed log transformations were made. Correlations between the ocular and systemic parameters were explored using either Pearson's linear correlation or Spearman's rank method as appropriate. P-values of less than 0.05 were considered significant, except in certain cases where a stricter p-value of less than 0.01 was adopted in order to correct for multiple comparisons and minimise bias towards type I errors. All analyses were performed using Satistica, version 6.0, Statsoft, Tusla, OK.

6.7 Power calculations

With regard to DVA, new measurement parameters and novel analysis methods were used in this study in patient groups which have not previously been examined with these techniques. Power calculation would normally be based on the results of previous studies which share the most similar protocols to that of the present research, however due to the nature of the statistical analysis required in this study, namely one-way ANOVA (or ANCOVA) and two factor repeated measures ANOVA and the uniqueness of the comparisons being made, appropriate previous research was not available and the power calculations were therefore ultimately made based on a number of assumptions. Firstly a large effect size of 0.40 was selected from Cohen's standardised effect sizes ⁷⁷¹, along with a Pearson's correlation coefficient among repeated measures of 0.5 and a non-sphericity correction of 1. The power calculation was made using the computer based programme, GPower 3 ⁷³³ and it was calculated that, in order to provide a statistical power of 80% at an α level of 0.05 a sample size of between 4-19 per group would be required (19 one-way ANOVA, 4 within groups ANOVA, 13 between groups

ANOVA, 5 within/between groups ANOVA). The aim was therefore to recruit at least 19 patients per group in this study.

6.8 Results

A total of 12 AD patients, 20 POAG patients, 25 NTG patients and 34 healthy controls were screened for inclusion in the present study however, in order to ensure all participants were matched on critical factors such as age and hypertensive status, all those under the age of 48, over the age of 75 or with a MBP of greater than 110 mmHg had to be excluded from analysis. Additionally, following the careful review of the obtained images, any patients who exhibited poor or incomplete results were also excluded meaning a total of 10 mild newly diagnosed AD patients, 19 POAG patients, 19 NTG patients and 20 healthy controls were recruited for this study. These numbers obviously fall below the target of 19 per group with regard to AD patients however on analysis of the results statistical differences between AD, glaucoma and control groups were obtained inferring sufficient power was still achieved. The difficulties associated with recruitment of this group of participants was discussed in section 4.11

6.8.1 Baseline values

There were no significant differences in age, systemic BP, BMI, triglycerides, glucose, HDL cholesterol levels and total cholesterol levels between the four groups (all p>0.05, table 6.1). Furthermore there were no significant differences in MBP or Framingham risk score (P>0.05, table 6.1) and the number of subjects with well controlled high BP was proportionally similar between groups (AD: n=3; POAG: n=3; NTG: n=5; Controls: n=6; Chi-square test). As expected IOP was found to be significantly greater in our POAG patients in comparison to all other groups (p<0.001, table 6.1) and consequently OPP was also found to be lower in our POAG patients but only significantly so with regard to AD patients and controls (p=0.017, table 6.1).

	AD (1)	POAG (2)	NTG (3)	Controls (4)	ANOVA	Post-
					p-value	hoc
N	10	19	19	20	-	-
Gender	5F:5M	9F:10M	11F:8M	8F:12M	-	-
Age (years)	62.50±8.07	65.26±9.52	60.16±12.13	58.00±4.32	0.143	-
SBP (mmHg)	141.70±14.21	136.32±15.42	130.06±15.37	131.70±17.90	0.278	-
DBP (mmHg)	80.30±7.51	79.11±9.80	78.39±11.14	79.70±9.39	0.952	-
ВМІ	27.61±5.80	27.71±5.00	27.92±4.20	27.56±4.67	0.997	-
Glucose	4.40±1.44	4.53±0.97	5.08±0.86	4.87±1.02	0.307	-
Triglycerides	1.28±0.60	1.22±0.50	1.04±0.31	1.17±0.40	0.488	-
HDL-C	1.33±0.25	1.17±0.26	1.12±0.29	1.14±0.32	0.315	-
(mmol/L)						
Total-C	4.77±0.64	4.29±0.78	4.58±1.04	4.73±0.67	0.376	-
(mmol/L)						
IOP (mmHg)	16.50±2.12	23.94±3.00	17.40±1.80	17.20±2.68	0.000*	2>1,3,4
MBP (mmHg)	100.77±7.67	99.29±11.35	95.61±10.93	97.03±11.67	0.637	-
OPP	84.96±9.46	47.12±16.93	76.09±12.66	82.65±12.06	0.002*	2<1,4
Fram Risk	10.67±3.28	11.18±5.56	9.67±7.50	10.11±5.20	0.921	-
Score						

Table 6.1: Summary of the baseline characteristics of the study groups. P<0.05 is considered a significant difference. SBP: systolic blood pressure; DBP: diastolic blood pressure; BMI: Body mass index; HDL-C: High density lipoprotein cholesterol; Total C: Total cholesterol; IOP: Intraocular pressure. The presented SBP, DBP and IOP values are the baseline readings taken on the morning of the study and do not represent the 24 hour or diurnal averages. OPP was subsequently calculated using these baseline values.

6.8.2 Systemic endothelial function

No significant differences were found between groups with regard to brachial artery

FMD or NMD (p>0.05, table 6.2). Furthermore no significant differences in circulating

vWF levels were identified between groups (p>0.05)

	POAG	NTG	AD	Controls	ANOVA p-	Post
					value	-hoc
FMD (%)	5.82±5.64	11.01±6.97	8.54±9.19	10.27±7.64	0.235	-
NMD (%)	15.30±12.44	29.09±17.83	24.74±16.42	23.76±7.63	0.258	-
vWF	149.09±50.93	96.39±39.05	118.34±64.28	108.60±49.23	0.496	

Table 6.2: Systemic endothelial function. P<0.05 (*) is considered significant. FMD: flow mediated dilation; NMD: nitroglycerine mediated dilation; vWF: von Willebrand factor.

6.8.3 Dynamic retinal vessel analysis

For ease of interpretation, the dynamic retinal vessel profile curve was considered in two parts, the first part being the dilation response (baseline to maximum dilation) and the second part being the constriction response (maximum dilation to maximum constriction). Each flicker cycle was analysed individually using traditional SDRA analysis and the artery and vein were considered separately. The dynamic nature or slope of the retinal vascular response profiles were fully explored using the polynomial fitted curves generated via our novel computational analysis (MatLab R2010a; MathWorks Inc., Natick, MA).

6.8.3.1 Arterial response

Dilation

With regard to the first part of dynamic profile curve, following the adoption of a stricter p value of 0.01 to correct for the effects of multiple comparisons no significant differences were found in the average arterial baseline diameter, maximum diameter (MD%), reaction time (RT) or baseline corrected flicker response (BFR) between all four study groups (all p>0.01, table 6.3). When considering each flicker cycle individually however using two factor repeated measures ANOVA, the arterial RT was found to be significantly longer on the final flicker cycle (F3) in both AD and POAG patients in comparison to healthy controls (p=0.016, table 6.4). Furthermore the sequential changes in the RT of the retinal arteries on progressing from flicker 1 to flicker 3 was found to vary significantly between groups (p=0.007, table 6.4), with healthy controls showing a significant decrease in RT on going from F2 to F3 (p=0.011, table 6.4) which was not replicated by any of the other groups, where in fact an increase was observed. On consideration of the dynamic nature of the arterial dilation response, evaluated using our novel matlab analysis, no significant differences were found between groups with regard to slope (p>0.05, table 6.5).

Constriction

With regard to the second part of the arterial dynamic profile curve, which looks at the activity of the retinal vessels following dilation and on cessation of flicker, again following the adoption of a stricter p value of 0.01 to account for the effects of multiple comparisons, no significant differences in maximum arterial constriction (MC%) or the time taken to reach maximum constriction (tMC) were found between groups (p>0.01, table 6.3). Furthermore, on consideration of the dynamic nature of the arterial constriction response, evaluated using our novel matlab analysis, no significant differences were found between groups with regard to slope (p>0.05, table 6.5).

ARTERY Average data	AD (1)	POAG (2)	NTG (3)	Controls (4)	ANOVA P-value	Post-hoc
BDF	5.44±2.11	7.67±4.17	7.85±4.18	4.74±1.94	0.012	-
MD (%)	5.78±3.25	5.90±3.65	7.27±3.02	5.19±2.19	0.188	-
RT	24.04±11.74	21.59±5.96	23.20±8.23	20.48±6.77	0.568	-
BFR	3.46±3.79	2.60±3.91	2.84±2.97	2.80±1.89	0.943	-
MC (%)	-3.29±1.56	-3.85±3.72	-3.85±1.93	-2.55±1.85	0.210	-
tMC (secs)	32.63±10.24	30.31±10.05	28.09±6.96	27.08±7.68	0.752	-

Table 6.3: Arterial vascular function parameters determined using dynamic retinal vessel analysis (DVA, IMEDOS GmbH, Jena, Germany). P<0.01 is considered as significant (*).BDF: baseline diameter fluctuation; MD(%): percentage change in diameter from baseline to maximum; RT: reaction time BFR: baseline corrected flicker response MC(%): percentage constriction below baseline; tMC: time taken to reach maximum constriction.

ARTERY	AD (1)	POAG (2)	NTG (3)	Controls (4)	ANOVA P-value	Post -hoc	Interaction p-value
RT							
Flicker 1	29.30±16.61	19.42±14.17	24.94±14.46	20.05±12.07	0.236		
Flicker 2	16.30±11.48	21.72±12.11	21.82±15.80	25.85±9.00	0.260		
Flicker 3	27.89±17.62	26.40±10.82	23.94±8.74	15.55±11.07	0.016*	1,2> 4	
Within groups ANOVA	0.093	0.314	0.787	0.011* (F2-F3)			0.007*

Table 6.4: Arterial vascular function parameters by flicker cycle. P<0.05 (*) is considered as significant on repeated measures ANOVA. RT: reaction time.

DYNAMIC RESPONSE	AD (1)	POAG (2)	NTG (3)	Controls (4)	ANVOA p-value	Post- hoc
Arteries Slope _{AD}	0.265±0.221	0.271±0.450	0.300±0.159	0.273±0.169	0.985	-
Slope _{AC}	-0.194±0.070	-0.269±0.154	-0.282±0.158	-0.170±0.095	0.055	-

Table 6.5: Dynamic characteristics of the retinal vascular response profiles determined using our novel computational model. P< 0.05 (*) is considered significant. Slope _{AD}: slope of arterial dilation; Slope _{AC}: slope of arterial constriction;

6.8.3.2 Venous response

Dilation

With regard to the first part of the venous dynamic profile curve, no significant differences were found in the average venous baseline diameter, maximum diameter (MD%), reaction time (RT) or baseline corrected flicker response (BFR) between all four study groups (all p>0.01, table 6.6). Venous baseline diameter fluctuation (BDF) however was found to be significantly greater on average in both our AD and NTG patients in comparison to our healthy controls (p=0.001, table 6.6). On consideration of the dynamic nature of the venous dilation response, evaluated using our novel matlab analysis, no significant differences were found between groups with regard to slope (p>0.05, table 6.7).

Constriction

With regard to the second part of the venous dynamic profile curve, which looks at the activity of the retinal vessels following dilation and on cessation of flicker, no significant differences in maximum constriction (MC%) or the time taken to reach maximum venous constriction (tMC) were found between groups (all p>0.01, table 6.6). Furthermore, on consideration of the dynamic nature of the venous constriction response, evaluated using our novel matlab analysis, no significant differences were found between groups with regard to slope (p>0.05, table 6.7).

VEIN	AD (1)	POAG (2)	NTG (3)	Controls (4)	ANOVA p-value	Post- hoc
BDF	6.44±2.67	4.98±2.44	5.29±1.99	3.27±1.56	0.002*	1,3>4
MD (%)	6.12±3.14	5.35±1.99	5.94±3.03	5.13±2.86	0.720	-
RT	22.67±9.39	19.94±4.39	21.69±8.37	20.48±3.68	0.756	-
BFR	2.16±4.03	2.81±2.56	2.72±2.92	3.30±2.28	0.765	-
MC (%)	-2.61±2.13	-2.60±2.46	-2.09±2.09	-1.77±1.40	0.572	-
tMC (secs)	29.74±6.20	34.25±8.65	31.63±12.99	34.25±9.40	0.664	-

Table 6.6: Venous vascular function parameters determined using dynamic retinal vessel analysis (DVA, IMEDOS GmbH, Jena, Germany). P<0.01 is considered as significant (*).BDF: baseline diameter fluctuation; MD(%): percentage change in diameter from baseline to maximum; RT: reaction time BFR: baseline corrected flicker response MC(%): percentage constriction below baseline; tMC: time taken to reach maximum constriction.

DYNAMIC RESPONSE	AD (1)	POAG (2)	NTG (3)	Controls (4)	ANVOA p-value	Post- hoc
Veins						
Slope _{VD}	0.295±0.172	0.229±0.094	0.219±0.141	0.222±0.132	0.518	-
Slope $_{VC}$	-0.204±0.193	-0.154±0.103	-0.160±0.10	-0.164±0.137	0.281	-

Table 6.7: Dynamic characteristics of the retinal vascular response profiles determined using our novel computational model. P< 0.05 (*) is considered significant; Slope $_{VD}$: slope of venous dilation; Slope $_{VC}$: slope of venous constriction.

6.8.4 Correlations

No significant correlations were found between the retinal vessel reactivity parameters

and the systemic parameters (FMD, blood analyses) between groups (all P>0.05).

6.9 Discussion

6.9.1 Main findings

This study has revealed for the first time evidence of altered reactivity to flicker light in the retinal arteries of both newly diagnosed, previously untreated POAG patients and newly diagnosed mild AD patients which is of a similar nature in both conditions and is not replicated by healthy control patients. The time taken for the retinal arteries to reach the point of maximum dilation following the onset of flicker light stimulation was found to be significantly greater in both AD and POAG patients in comparison to healthy controls on the final flicker cycle. Furthermore the sequential changes in the reaction time of the retinal arteries on progressing from flicker 1 to flicker 3 was found to vary significantly between groups with healthy controls showing a significant reduction in reaction time on heading into the final flicker cycle which was not replicated by any of the other groups. This study has also revealed evidence of altered baseline retinal venous activity which is of a similar nature in both our newly diagnosed and previously untreated NTG patients and our newly diagnosed mild AD patients and is again not replicated by healthy controls. No significant differences were identified between groups with regard to the systemic vascular parameters.

6.9.2 Systemic endothelial function

No significant differences in systemic vascular endothelial function, as measured by FMD, were found between all four groups in this study. This result indicates that the functioning of the systemic vascular endothelium is comparable between newly diagnosed glaucoma patients, mild AD patients and healthy controls. A comparison of this kind between all of these groups simultaneously has not previously been made however on an individual basis disturbed systemic endothelial function has previously been identified in both POAG ^{42, 385} and NTG patients ⁵² compared to healthy controls and also in AD patients compared to healthy controls ⁶³¹. The possibility therefore that

vascular dysfunction at the systemic level may play a role in their disease aetiologies individually cannot be ruled out. All of our participants were newly diagnosed and at the very earliest stages of their disease process. It could be hypothesised therefore that measureable signs of systemic macrovascular dysfunction were simply not yet detectable in these patients at the point of investigation. Indeed it has previously been demonstrated that endothelial dysfunction affects the microvasculature at an earlier stage in a disease process in comparison to the macrovasculature ⁷³⁴. Further research and follow-up would be required however in order to confirm this hypothesis.

6.9.3 Retinal vessel reactivity

With regard to retinal artery reactivity our mild AD patients and our newly diagnosed POAG patients took comparably and significantly longer to reach the point of maximum arterial dilation following the onset of flicker light compared to our healthy controls and NTG patients on the final stimulation cycle. A prolonged reaction time to flicker light is an established parameter which has been shown to indicate the presence of some form of vascular dysfunction at the retinal level ^{695, 710, 732}. The present results suggest that this vascular dysfunction appears to be shared by both our AD and POAG patients however its exact aetiology can only be hypothesised at this point. As the retinal vascular response to flickering light occurs due to an increase in retinal metabolic demand and is predominantly a neurovascular coupling driven response ^{226, 685, 686, 691, 692}, it could be hypothesised that the altered retinal vessel reactivity demonstrated here may be indicative of a similar disturbance of neurovascular coupling in AD and POAG patients which is exacerbated by repeated flicker stimulation. The variability in the progression of the arterial reaction time between groups over successive flicker cycles and the significant involvement of the final flicker cycle could further suggest that exhaustive factors, such as a progressive depletion of NO reserves, may play a role in the altered vascular reactivity observed here. Indeed altered NO activity and bioavailability has been previously described in both AD ^{772, 773} and POAG patients ^{368, 369, 377} and has also

previously been linked to alterations in retinal vessel reactivity to flicker light ^{41, 226}. The presence of a common endothelial dysfunction in AD and POAG, oxidative stress or a common alteration in astrocyte activity, all of which have been previously demonstrated to varying degrees in these disease processes individually ^{42, 144, 198, 631, 632, 739} and are known to be key mediators of the neurovascular response ⁶⁰⁰, could potentially explain these exhaustive alterations. Due to the complexity of the neurovascular coupling response however, before any firm conclusions can be made regarding the exact nature of the observed vascular dysfunction in our AD and POAG patients', further investigation would be required to validate all of these hypotheses.

On consideration of the retinal venous reactivity, our newly diagnosed mild AD patients demonstrated increased fluctuations in baseline vessel diameter on average, prior to the onset of flicker, which were comparable to that of our newly diagnosed NTG patients and were not replicated by healthy controls or POAG patients. Consideration of baseline diameter fluctuation (BDF) was first recommended by Nagel et al ⁶⁸⁶ as a way of taking into account the effect of the spontaneous variations in vessel diameter that occur under normal resting conditions on the observed response of the vasculature to flicker light stimulation, however it is a parameter which is not commonly reported in the literature and which has, to date, mainly been considered in regard to the retinal arteries, where it has been tentatively linked to vascular disturbance in the form of instability or increased variation in vascular tone or rigidity ^{695, 746-748}. As such the cause and relevance of increased BDF in the venous circulation is currently unclear. Retinal veins are generally thought to play a more secondary role in retinal autoregulation, perhaps providing a fine tuning of the regulation response following the active reaction of the retinal arteries and instigating a regulatory contribution passively in response to increased blood flow ⁶⁸⁵. Interestingly increased retinal venous diameters have been previously linked to impaired cerebral blood flow and have been suggested as a marker of both retinal and cerebral ischemia and hypoxia ^{652, 749-751}. The finding here is somewhat different as fluctuations in

diameter have been assessed dynamically as opposed to vessel diameter measurements being taken statically from photographs. Nevertheless as both ischemia and hypoxia have been well linked to the development of AD and NTG, it could be hypothesised that the increased fluctuation in baseline diameter observed in these patients could be reflective of early hypoxic changes and a common increased risk of future damage in these patients.

6.9.4 Retinal vascular dysfunction in AD compares to both POAG and NTG

The concept that AD and glaucoma may share a common underlying vascular aetiology has been increasingly realised over recent years and evidence of vascular dysfunction, related to either disturbed autoregulation or disturbed neurovascular coupling mechanisms, has been previously demonstrated at the cerebral level in both conditions ^{323, 567, 620, 626} and at the ocular level in both POAG and NTG patients ^{39, 41, 320}. Until now however the presence of dynamic vascular dysfunction at the ocular level has not been assessed in AD patients and directly compared to that of glaucoma patients and healthy controls. The present study addresses this and provides support for not only a shared vascular dysfunction in AD and glaucoma patients at the earliest stages of their disease process but also for the presence of vascular dysfunction in both POAG and NTG. Interestingly the elements of ocular vascular dysfunction which we have demonstrated in our AD patients appear to relate to that of both POAG and NTG patients in a slightly different manner. Indeed with regard to retinal artery reactivity AD patients demonstrated a dysfunction that was similar in nature to that displayed by our POAG patients and not replicated by our healthy controls or NTG patients, whereas with regard to the retinal venous reactivity, AD patients demonstrated baseline disturbances that were of a similar nature to that displayed in our NTG patients and were again not replicated by the other two groups. The key feature of both POAG and NTG is RGC loss and excavation of the

ONH ^{109, 113}. Vascular factors have been implicated in the aetiology of both conditions ^{141,} ^{320, 325} and both have been previously linked to AD ^{676, 677}. It has been recently suggested that rather than being divided into two separate entities, glaucoma should instead be considered as a disease continuum, extending from pure POAG at one end, in which raised IOP plays a predominate role, to pure NTG at the other end, in which IOPindependent or vascular factors play a more predominate role ³²⁻³⁴. This allows for a certain degree of overlap between the two extremes where a 'mixed' aetiology may exist. In a similar manner, due to the increasing evidence of vascular involvement in the AD process, AD dementia has also been described as a disease continuum extending from pure AD at one end to VaD at the other end, again with a mixed component separating the two extremes ^{645, 646}. VaD is a condition which bares similarities to AD dementia but has an established primary vascular cause such as stroke ⁶⁴³. Increasing evidence suggests that the majority of patients may in fact lie somewhere between the two extremes in both the glaucoma and AD continuums and exhibit a mixed aetiology with important vascular elements. This may in part explain the findings here of vascular dysfunction in AD patients which relates to that of both our POAG and NTG patients. Further research investigating vascular dysfunction in AD and glaucoma patients however is still required to elicit the exact nature of the relationship between the two conditions. The need for consideration of glaucoma and ocular health in AD and the need for consideration of AD in those diagnosed with all forms of glaucoma has nevertheless been highlighted by this study.

6.10 Conclusion

In conclusion this study demonstrates for the first time that retinal vascular reactivity to flickering light is altered in a similar manner in both AD and glaucoma patients providing support for a common underlying vascular aetiology in both conditions. Furthermore this study has identified differences in the nature of these retinal vascular reactivity
disturbances between AD and POAG patients and AD and NTG patients, however further investigation is required in order to determine the significance of these findings.

6.11 Limitations

Only a small sample size of AD patients and a moderate cohort of glaucoma patients could be recruited for this study. This could potentially limit the statistical power and conclusions which can be drawn from the presented data. The overall limitations of the presented research are discussed further in section 8.3.

7. Primary open angle glaucoma and Normal tension glaucoma: two separate diseases or one continuous entity? The vascular perspective

7.1 Abstract

Purpose: To compare and contrast the presence of ocular and systemic vascular function across multiple levels in newly diagnosed and previously untreated primary open angle glaucoma (POAG) and normal tension glaucoma (NTG) patients and to evaluate the validity of these conditions as distinct clinical entities.

Methods: Systemic vascular function was assessed in 19 POAG patients, 19 NTG patients and 20 healthy controls by means of 24 hour ambulatory blood pressure (ABPM), 24 hour heart rate variability (HRV) assessment, pulse wave analysis (PWA), carotid intima-media thickness (IMT) and flow mediated dilation (FMD). Ocular vascular function was assessed by means of retinal vascular reactivity to flicker light using dynamic retinal vessel analysis (DVA, IMEDOS, GmbH, Jena, Germany). All patients additionally underwent blood analysis for oxidative stress.

Results: When compared to normal controls both POAG and NTG patients exhibited increased nocturnal systolic blood pressure (SBP) variability (p=0.011); systemic arterial stiffness (p=0.015), carotid IMT (p=0.040) and a comparably reduced OPP (p=0.001). Furthermore, both groups of glaucoma patients also exhibited a significantly steeper retinal arterial constriction slope following cessation of flicker (p=0.007) and an increased fluctuation in arterial and venous baseline diameter (p=0.008; p=0.010) when compared to healthy controls. No significant difference in FMD, HRV or oxidative stress parameters were identified between groups.

Conclusion: POAG and NTG patients demonstrate multiple comparable alterations in both ocular and systemic vascular function at the early stages of their disease process. This highlights not only the importance of considering vascular factors in both conditions, but also the need to perhaps become less rigid in our separation of the two conditions into distinct clinical entities when considering vascular risk.

7.2 Introduction

The separation of OAG into two distinct clinical entities on the basis of IOP has been common practice for many years, with GON in the presence of IOP over 21 mmHg having long been described as POAG⁷⁷⁴ and all other cases having been described as either low tension glaucoma (LTG) or NTG ⁷⁷⁵. Over recent years however, additional risk factors for both of these forms of OAG have been identified and the concept of POAG and NTG as distinct clinical entities has been questioned. Indeed features more traditionally linked to NTG development, such as ocular and systemic vascular dysfunctions, have been similarly demonstrated in POAG patients ³⁹⁻⁴³ and the effects of IOP reduction on disease progression has been similarly demonstrated in NTG patients ³⁶⁻³⁸. Consequently recent research has recommended the abolishment of the terms POAG and NTG, along with the 'arbitrary' 21mmHg cut off value ³⁴. Furthermore it has been suggested instead that OAG should be considered a disease continuum in which the aetiology of the disease extends from being predominantly IOP dependent at one end (pure POAG), to predominantly IOP-independent at the other end (pure NTG), with the involvement of vascular factors increasing as the predominance of IOP decreases and a large area of overlap existing between the two extremes ³⁵. Nevertheless, a number of subtle but important differences have however previously been described between POAG and NTG patients in regard to both their structural and functional ONH changes ^{31, 44-46} as well as to their vascular risk ⁴⁷. Indeed NTG patients show stronger vasospastic tendencies ²⁶, greater systemic endothelial dysfunction ^{42, 52} and stronger links to the presence of hypotension ^{53, 54} in comparison to POAG patients. The possibility that these two conditions do in fact represent different clinical entities with different mechanisms of ONH damage can therefore not be ruled out and such a division is currently considered important with regard to both clinical diagnosis and management of the disease. Further investigation is therefore required into, not only the similarities and differences between POAG and NTG, but also into the aetiological role

and interactions of IOP and vascular factors in each condition. In Chapter 5 of this thesis abnormalities in both retinal vascular function and systemic vascular parameters were identified in our newly diagnosed NTG patients, highlighting not only vascular involvement in NTG but also the fact that a glaucomatous eye may represent a 'sick eye in a sick body'. Furthermore in Chapter 6 of this thesis, NTG patients were found to exhibit comparable alterations in retinal vascular function to that of AD patients, which differed in nature to that identified in POAG and AD patients. Expanding on these findings and exploring whether POAG patients exhibit similar or differing vascular abnormalities in comparison to NTG patients across an even broader range of ocular and systemic vascular parameters than used in previous chapters could not only allow the concept of a mixed aetiology and a glaucomatous disease continuum to be explored but could also allow further exploration of the 'sick eye in a sick body concept' in glaucoma. As such this study aims to explore, compare and contrast vascular function at the ocular level and at numerous systemic levels in NTG and POAG patients in comparison to healthy controls. Determining which vascular features, if any, are shared by both POAG and NTG patients and which features, if any, could potentially distinguish the two conditions could prove beneficial for enhancing disease prognosis, understanding and management in the future.

7.3 Aims

To compare and contrast the presence of vascular dysfunction at the ocular level in the form of retinal vascular dysfunction and at the systemic level in the form of systemic endothelial dysfunction, arterial stiffness, IMT, HRV, ambulatory blood pressure and the presence of oxidative stress, in POAG and NTG patients in comparison to healthy controls. Furthermore to evaluate the concept of a glaucomatous disease continuum and the concept that glaucoma represents a sick eye in a sick body.

7.4 Hypothesis

Evidence of ocular and systemic vascular dysfunction will be evident in both POAG and NTG patients but not healthy controls. The extent of vascular involvement will be more pronounced in NTG patients.

7.5 Subjects and Methods

Newly diagnosed and previously untreated NTG patients and POAG patients, along with healthy age matched controls were recruited for this study. The recruitment details, inclusion and exclusion criteria for these patients was detailed in section 3.1. The investigative procedures performed in this study are outlined below and were conducted in accordance with the protocols outlined in section 3.2:

- 1. Preliminary tests
- 2. Fasting venous blood sample obtained
- 3. 24 hour blood pressure and heart rate variability monitor fitted
- 4. Assessment of retinal vessel reactivity (DVA)
- 5. Pulse wave analysis
- 6. Intima-media thickness measurement
- 7. Assessment of systemic endothelial function (FMD)

7.6 Statistical Analysis

All data were reported as mean ± standard deviation. The Kolmogorov-Smirnov test was used to determine the distribution of the data. Multivariate analysis was performed to determine the influence of age, BMI, BP and circulating markers on the measured variables. Differences between groups were subsequently assessed using one-way ANOVA or ANCOVA, as appropriate, followed by Tukey's post hoc analysis. In cases where the normality of the data could not be confirmed log transformations were made. Correlations between the ocular and systemic parameters were explored using either Pearson's linear correlation or Spearman's rank method as appropriate. P-values of less than 0.05 were considered significant, except in certain cases where a stricter p-value of less than 0.01 was adopted in order to correct for multiple comparisons and minimise bias towards type II errors. All analyses were performed using Satistica, version 6.0, Statsoft, Tusla, OK

7.7 Power calculations

With regard to DVA, new measurement parameters and novel analysis methods were used in this study in patient groups which have not previously been examined with these techniques. Power calculation would normally be based on the results of previous studies which share the most similar protocols to that of the present research, however due to the nature of the statistical analysis required in this study, namely one-way ANOVA (or ANCOVA) and the uniqueness of the comparisons being made between the three groups, appropriate previous research was not available. The power calculation for this study was therefore made on the basis of Cohen's standardised effect sizes ⁷⁷¹, whereby a large effect size of 0.40 was selected. The calculation was conducted using the computer based programme, GPower 3 ⁷³³ and it was revealed that in order to provide a statistical power of 80% at a α level of 0.05 a sample size of 22 per group would be required. The aim was therefore to recruit at least 22 patients per group in this study.

7.8 Results

A total of 20 POAG patients, 25 NTG patients and 34 healthy controls were screened for inclusion in the present study however, in order to ensure all participants were matched on critical factors such as age and hypertensive status, all those under the age of 45, over the age of 75 or with a MBP of greater than 115 mmHg had to be excluded from analysis. Additionally, following the careful review of the obtained images, any patients

who exhibited poor or incomplete results were also excluded meaning 19 POAG patients (9 female, 10 male), 19 NTG patients (12 female, 7 male) and 20 healthy age matched controls (9 female, 11 male) were included in the final analysis. These sample sizes fall slightly short of the target of 22 recommended by power calculation however statistical significance was still achieved at a power level of 76%.

7.8.1 Baseline values

There were no significant differences in age, systemic BP, BMI, triglycerides, glucose, HDL cholesterol levels, total cholesterol levels and Framingham risk score between the three groups (all p>0.05, table 7.1). Furthermore the number of subjects with well controlled high BP was proportionally similar between groups (POAG: n=4; NTG: n=5; Controls: n=6, p>0.05). As expected IOP was found to be significantly greater in our POAG patients in comparison to NTG and controls (p<0.001, table 7.1). OPP on the other hand was found to be significantly lower in both our POAG and NTG patients in comparison to healthy controls (p=0.001, table 7.1).

	POAG(1)	NTG(2)	Controls(3)	ANOVA P-	Post-hoc
				value	
Ν	19	19	20	-	-
Gender	9F:10M	11F:8M	9F:11M	-	-
Age (years)	65.26±9.52	60.16±12.13	60.65±4.20	0.174	-
SBP (mmHg)	136.32±15.42	130.06±15.37	130.00±19.13	0.418	-
DBP (mmHg)	79.11±9.80	78.39±11.14	78.15±10.03	0.956	-
BMI	27.71±5.00	27.92±4.20	26.86±3.79	0.757	-
Glucose	4.53±0.97	5.08±0.86	4.84±1.03	0.270	-
Triglycerides	1.22±0.50	1.04±0.31	1.06±0.31	0.320	-
HDL-C	1.17±0.26	1.12±0.29	1.18±0.35	0.840	-
(mmol/L)					
Total-C	4.29±0.78	4.58±1.04	4.76±0.74	0.293	-
(mmol/L)					
IOP (mmHg)	23.94±3.00	17.40±1.80	17.20±2.68	<0.001*	1>2,3
MBP	99.29±11.35	95.61±10.93	96.14±13.09	0.594	-
(mmHg)					
OPP	47.12±16.93	46.24±7.84	72.77±18.01	0.001*	1,2<3
Fram Risk	11.18±5.56	9.67±7.50	10.22±5.14	0.824	-
Score					
MD	-1.58±0.54	-2.18±3.22	-	0.545	-

Table 7.1: Summary of the baseline characteristics of the study groups. P<0.05 is considered a significant difference. SBP: systolic blood pressure; DBP: diastolic blood pressure; BMI: Body mass index; HDL-C: High density lipoprotein cholesterol; Total C: Total cholesterol; IOP: Intraocular pressure; MD: mean deviation. The presented SBP, DBP and IOP values are the baseline readings taken on the morning of the study and do not represent the 24 hour or diurnal averages. OPP was subsequently calculated using these baseline values.

7.8.2 Ambulatory blood pressure

No significant difference in SBP, DBP or MBP was found between groups across the

diurnal, nocturnal or 24 hour measurement periods of this study (p>0.05, table 7.2).

Furthermore no significant difference in the percentage nocturnal dip in BP (p>0.05,

table 7.2) or in the number of non-dippers (less than 10% nocturnal dip in MBP),

physiological dippers (between 10-20% nocturnal dip in MBP) and over dippers (greater

than 20% nocturnal dip in MBP) between groups, were found in this study. A

significantly higher variability in the nocturnal SBP measurements however was

identified in both our POAG and NTG patients in comparison to healthy controls

	POAG(1)	NTG(2)	Controls(3)	ANOVA P-	Tukeys
				value	
24hr SBP	128.00±20.29	120.60±11.11	119.23±9.38	0.241	-
24 hr DBP	71.64±11.75	67.00±9.49	69.23±7.95	0.458	-
24 hr MBP	90.43±13.63	84.87±8.57	85.90±8.16	0.330	-
DSBP	133.20±18.86	127.19±12.86	125.31±11.87	0.343	-
DDBP	77.93±12.49	71.69±10.91	73.69±9.91	0.300	-
DMBP	96.36±13.66	90.19±10.11	90.90±10.32	0.286	-
NSBP	115.00±17.34	111.71±9.62	107.69±7.42	0.326	-
NDBP	62.09±11.31	60.07±7.57	61.31±6.12	0.832	-
NMBP	79.73±12.74	77.29±6.99	76.77±6.31	0.691	-
Dip (%)	13.43±9.89	13.40±7.10	15.04±6.87	0.835	-

(p=0.011, table 7.3).

Table 7.2: Ambulatory blood pressure parameters: SBP: systolic blood pressure; DBP: diastolic blood pressure; MBP: mean blood pressure; D- : diurnal; N- : nocturnal. P values <0.05 are considered significant

Coefficients of variation	POAG(1)	NTG(2)	Controls(3)	ANOVA p- value	Tukeys
24 hr SBP (%)	12.44±2.95	12.82±2.31	11.99±3.15	0.746	-
DSBP (%)	10.96±2.27	11.27±3.10	10.02±2.59	0.464	-
NSBP (%)	13.08±2.68	12.54±3.77	9.12±3.40	0.011*	1,2>3

Table 7.3: Coefficient of variation for systolic blood pressure across 24 hours, diurnally and nocturnally. SBP: systolic blood pressure; D- : diurnal-; N- : nocturnal-. P-values <0.05 (*) are considered significant

7.8.3 Heart rate variability

No significant differences in LF, HF or LF:HF were found between groups during the

diurnal, nocturnal or 24 hour measurement period (all p>0.05, table 7.4)

	POAG(1)	NTG(2)	Controls(3)	ANOVA P-	Post-hoc
				value	
24hr LF	57.15±18.77	63.81±10.01	63.30±11.27	0.714	-
24hr HF	37.46±16.55	31.56±7.12	32.80±11.82	0.640	-
24hr LF:HF	2.12±1.63	2.18±0.78	2.23±0.99	0.709	-
DLF	56.92±18.67	63.00±11.25	65.20±12.57	0.359	-
DHF	36.67±15.89	31.82±8.21	28.20±8.07	0.360	-
DLF:HF	2.20±1.97	2.17±0.81	2.53±1.00	0.773	-
NLF	54.42±19.92	65.31±10.06	60.67±15.54	0.366	-
NHF	41.08±18.21	31.13±7.99	36.78±15.17	0.298	-
NLF:HF	1.93±1.61	2.34±1.03	2.16±1.43	0.717	-

Table 7.4: Heart rate variability parameters: LF: low frequency; HF: high frequency; LF/HF: low to high frequency ratio; D- : diurnal; N- : nocturnal. P<0.05 is considered a significant difference

7.8.4 **Pulse wave analysis**

Following correction for DBP, identified as an influence factor on multivariate analysis,

pulse wave analysis augmentation index (AIx) was found to be significantly greater in

both our POAG and NTG patients compared to healthy age matched controls (p=0.015)

(Table 7.5).

	POAG(1)	NTG(2)	Controls(3)	ANCOVA	Post-hoc
				P-value	
PWA: Alx	27.88±8.20	26.06±11.25	16.00±10.66	0.015*	1,2>3

Table 7.5: Systemic arterial stiffness. P<0.05 (*) is considered significant. PWA:AIx: pulse wave analysis: augmentation index

7.8.5 Intima-media thickness measurement

Following correction for age and BMI, IMT was found to be increased in both our NTG

and POAG patients compared to controls (p=0.040, table 7.6).

	POAG(1)	NTG(2)	Controls(3)	ANCOVA	Tukeys post-
				P-value	hoc test
IMT (cm)	0.063±0.014	0.064±0.015	0.042±0.028	0.040*	1,2>3

Table 7.6: Carotid artery intima-media thickness. P<0.05 (*) is considered significant; IMT: intimamedia thickness

7.8.6 Systemic endothelial function

No significant differences were found between groups with regard to brachial artery FMD or nitroglycerine-mediated dilation (NMD) (p>0.05, table 7.7). Furthermore no significant differences in circulating vWF levels were identified between groups (p>0.05)

	POAG(1)	NTG(2)	Controls(3)	ANOVA P-	Tukeys
				value	
FMD (%)	6.38±8.66	11.01±6.97	13.35±8.71	0.190	-
NMD (%)	14.80±7.58	25.04±15.61	25.07±7.59	0.093	-
vWF	141.03±46.30	118.17±63.46	107.85±51.20	0.414	-

Table 7.7: Systemic endothelial function. P<0.05 (*) is considered significant. FMD: flow mediated dilation; NMD: nitroglycerine mediated dilation; vWF: von Willebrand factor

7.8.7 Dynamic retinal vessel analysis

For ease of interpretation, the dynamic retinal vessel profile curve was considered in two parts, the first part being the dilation response (baseline to maximum dilation) and the second part being the constriction response (maximum dilation to maximum constriction). The principle results are given based on the average of the 3 flicker cycles with the artery and vein being considered separately. Although each flicker cycle was also analysed individually this analysis did not yield any additional information and has therefore not been presented. The dynamic nature of the retinal vascular response profiles were fully explored using the polynomial fitted curves generated via our novel computational analysis (MatLab R2010a; MathWorks Inc., Natick, MA).

7.8.7.1 Arterial Response

Dilation

With regard to the first part of dynamic profile curve, no significant differences were found in the average arterial baseline diameter, maximum diameter (MD%), reaction time (RT) or baseline corrected flicker response (BFR) between all four study groups (all

p>0.01, table 7.8). The arterial baseline diameter fluctuation (BDF) however was found to be significantly greater in both POAG and NTG patients compared to controls (p=0.008, table 7.8). On consideration of the dynamic nature of the arterial dilation response, evaluated using our novel matlab analysis, no significant differences were found between groups with regard to slope (p>0.05, table 7.9).

Constriction

With regard to the second part of the arterial profile curve, which looks at the activity of the retinal vessels following dilation and on cessation of flicker, no significant differences in maximum constriction (MC%) or the time taken to reach maximum arterial constriction after dilation (tMC) were found between groups (all p>0.01, table 7.8). On consideration of the dynamic nature of the arterial constriction response however, a significantly steeper arterial constriction was demonstrated in both our POAG and NTG patients in comparison to our healthy controls (p=0.007, table 7.9).

ARTERY	POAG(1)	NTG(2)	Controls(3)	ANOVA P- value	Tukey's
BDF	7.28±3.18	7.47±3.84	4.55±1.90	0.008*	1,2>3
MD (%)	4.07±3.24	4.80±2.40	3.30±1.41	0.196	-
RT	24.93±9.16	21.09±8.67	18.47±6.35	0.061	-
BFR	-0.02±4.61	-0.04±2.41	0.05±2.01	0.997	-
MC (%)	-2.72±3.18	-2.64±1.61	-1.30±1.43	0.101	-
tMC (secs)	36.55±5.24	34.99±8.30	33.46±8.52	0.463	

Table 7.8: Arterial vascular function parameters determined using dynamic retinal vessel analysis (DVA, IMEDOS GmbH, Jena, Germany). P<0.01 is considered as significant (*).BDF: baseline diameter fluctuation; MD(%): percentage change in diameter from baseline to maximum; RT: reaction time BFR: baseline corrected flicker response MC(%): percentage constriction below baseline; tMC: time taken to reach maximum constriction.

DYNAMIC RESPONSE	POAG (1)	NTG (2)	Controls (3)	ANOVA P- value	Tukey's
Arteries Slope _{AD}	0.249±0.402	0.300±0.159	0.261±0.173	0.844	-
Slope _{AC}	-0.257±0.144	-0.282±0.158	-0.150±0.075	0.007*	1,2>3

Table 7.9: Dynamic characteristics of the arterial vascular response profiles determined using our novel computational model. P< 0.05 (*) is considered significant. Slope $_{AD}$: slope of arterial dilation; Slope $_{AC}$: slope of arterial constriction;

7.8.7.2 Venous response

Dilation

With regard to the first part of the venous dynamic profile curve, no significant differences were found in the average venous baseline diameter, MD%, RT, or BFR between all three study groups (all p>0.01, table 7.10). The venous BDF however was found to be significantly greater in both our POAG and NTG patients in comparison to our healthy controls (p=0.015, table 7.10). On consideration of the dynamic nature of the venous dilation response however, evaluated using our novel matlab analysis, no significant differences were found between groups with regard to slope (p>0.05, table 7.11).

Constriction

With regard to the second part of the venous dynamic profile curve, which looks at the activity of the retinal vessels following dilation and on cessation of flicker, no significant differences in MC% or the time taken to reach maximum venous constriction after dilation (tMC) were found between groups (all p>0.01, table 7.10). Furthermore, on consideration of the dynamic nature of the venous constriction response, evaluated using our novel matlab analysis, no significant differences were found between groups with regard to slope (p>0.05, table 7.11).

VEIN	POAG (1)	NTG (2)	Controls (3)	ANOVA P- value	Tukey's
BDF	4.84±2.28	5.07±2.15	3.29±1.41	0.010*	1,2>3
MD (%)	4.04±1.43	4.12±2.32	3.99±2.47	0.982	-
RT	20.89±7.98	23.47±7.65	18.55±2.22	0.081	-
BFR	0.59±1.44	-0.10±2.62	1.79±2.25	0.033	-
MC (%)	-1.38±1.57	-0.85±1.57	-1.09±1.18	0.543	-
tMC (secs)	39.08±7.93	39.02±6.80	36.29±9.34	0.489	-

Table 7.10: Venous vascular function parameters determined using dynamic retinal vessel analysis (DVA, IMEDOS GmbH, Jena, Germany). P<0.01 is considered as significant (*).BDF: baseline diameter fluctuation; MD(%): percentage change in diameter from baseline to maximum; RT: reaction time BFR: baseline corrected flicker response MC(%): percentage constriction below baseline; tMC: time taken to reach maximum constriction

DYNAMIC RESPONSE	POAG (1)	NTG (2)	Controls (3)	ANOVA P- value	Tukey's
Veins					
Slope _{VD}	0.261±0.200	0.219±0.141	0.225±0.140	0.705	-
Slope _{VC}	-0.161±0.094	-0.160±0.10	-0.167±0.133	0.982	-

Table 7.11: Dynamic characteristics of the arterial vascular response profiles determined using our novel computational model. P< 0.05 (*) is considered significant. Slope $_{VD}$: slope of venous dilation; Slope $_{VC}$: slope of venous constriction;

7.8.8 Oxidative stress markers

No significant differences in GSH, GSSG or total-GSH levels were found between

groups in this study. Furthermore the redox balance (ratio of GSH:GSSG) was also

found to be comparable between groups (all p>0.05, table 7.12)

	POAG(1)	NTG(2)	Controls(3)	ANCOVA P-	Tukey's
				value	
GSSG	38.98±21.99	37.59±23.88	61.89±46.34	0.169	-
GSH	574.89±217.27	623.09±269.65	687.67±393.90	0.644	-
tGSH	652.85±247.44	698.26±280.42	836.41±448.38	0.361	-
GSH:GSSG	17.34±7.12	20.43±10.81	13.93±8.53	0.159	-

Table 7.12: Oxidative stress analysis. GSSG: oxidised glutathione, GSH: reduced glutathione; tGSH: total glutathione; GSH:GSSG: redox balance. Calculated using ANCOVA, correcting for SBP, DBP and BMI.

7.8.9 Correlations

A significant positive correlation was identified between nocturnal SBP variability and

right IMT in POAG patients only (R= 0.74, p<0.001, figure 7.1) and between nocturnal

SBP variability and retinal arterial BDF in NTG patients only (R=0.71, p=0.001, figure

7.2). No other correlations between our systemic and ocular parameters could be found.



Figure 7.1 Nocturnal systolic blood pressure variability vs. intima-media thickness in POAG patients



Figure 7.2. Nocturnal systolic blood pressure variability vs. retinal arterial baseline diameter fluctuation in NTG patients

7.9 Discussion

7.9.1 Main findings

This study reveals evidence of systemic and ocular vascular abnormalities which are comparable in both newly diagnosed and previously untreated POAG and NTG patients and are not replicated by healthy age matched controls. At the systemic level ambulatory BP abnormalities, in the form of increased variability in nocturnal SBP, along with comparably reduced OPP, increased systemic arterial stiffness and increased IMT were identified in both groups of glaucoma patients. Furthermore, at the ocular level, signs of retinal vascular dysfunction, in the form of a significantly steeper retinal arterial constriction slope following cessation of flicker and an increased fluctuation in arterial and venous baseline diameter, was identified in both our POAG and NTG patients and was again not replicated in the healthy control group. No significant differences in HRV, systemic endothelial dysfunction or glutathione levels were found between groups.

7.9.2 Systemic vascular parameters

7.9.2.1 Ambulatory blood pressure assessment

Systemic BP follows a normal circadian rhythm and is constantly regulated by the ANS through modifications in cardiac output and total peripheral resistance ⁴⁰⁸. At the ocular level a close relationship exists between BP, IOP and OPP. As such the evaluation of ambulatory BP in patients with or at risk of ocular diseases such as glaucoma can not only give an insight into the presence of modifiable cardiovascular risk factors such as abnormal BP, but can also give an indication of the presence of local risk for ischemia. In the present study no significant differences in the average diurnal and nocturnal blood pressure parameters (SBP, DBP, MBP), or the nocturnal dipping status, were found between POAG, NTG and healthy control subjects following 24 hour BP assessment. This finding is consistent with that of a number of other studies that have assessed 24

hour BP in both newly diagnosed and more established glaucoma patients and have found no significant differences between groups ^{43, 414, 709, 776}. Indeed, in retrospect, such a comparable finding may not be surprising with regard to this study as very strict inclusion criteria was imposed on participants and there was a necessity for the groups to be not only age matched but also disease matched. It cannot be denied however that many other studies have been able to identify associations between abnormal systemic BP and the presence of GON. Indeed, whilst the evidence surrounding hypertension has a tendency to be somewhat variable and inconsistent ^{307, 441-443, 449}, the presence of hypotension, in particular nocturnal hypotension, has been more widely linked to the occurrence and progression of glaucoma²⁴, especially NTG^{54, 130, 459-462}, as has the presence of large nocturnal dips in BP (>20%)^{313, 433, 450, 463-466}. An increased variability/fluctuation in nocturnal SBP, as well as a reduced OPP at the time of testing, was however identified in both our POAG and NTG patients in comparison to healthy controls. Increased variability in BP is a parameter widely recognised in cardiovascular and hypertension research as a signal of increased risk for end organ damage 777 and over recent years its occurrence has also been increasingly considered in regard to other vascular diseases, including glaucoma ^{53, 467, 468}. Indeed, in line with our findings here, previous studies have also identified increased nocturnal variability of BP in both NTG patients ⁵³ and those with focal ischemic glaucoma ⁴⁶⁰, as well as in those with progressive NTG ⁴⁵¹ following 24 hour ABPM and in those with POAG during the daytime using a slightly different technique ⁷⁷⁸. Whilst an association does therefore appear to exist between increased BP variability and GON, the exact mechanisms by which such increased variability, over the nocturnal period in particular, may relate to the development of GON is currently unclear. It could be hypothesised that, due to the close relationship between BP and OPP, an increased variability in BP may subsequently lead to an increased variability or fluctuation in OPP, which may have particular effect nocturnally when OPP is already physiologically reduced and the ONH is more vulnerable. Indeed strong links have previously been made between the development of

GON, nocturnal hypotension ^{24, 433, 450} and fluctuating or reduced OPP ^{15, 779-781}, in that the risk of ischemia/reperfusion injury at the ONH and the subsequent risk of GON development is thought to be increased in the presence of these features, particularly if occurring in combination with autoregulatory dysfunction ¹⁴¹. In order to validate this hypothesis calculation of 24 hour OPP would ideally be needed; however in this study this information is not available as 24 hour IOP measurements were not taken. At the time of assessment however, OPP was found to be significantly reduced in both our POAG and NTG patients in comparison to healthy controls suggesting perfusion related vascular alterations are likely to be playing a part in the pathogenesis of both conditions. The relative contribution of IOP to this reduced OPP however is likely to vary between our glaucoma groups and as such consideration of the 24 hour IOP and its fluctuations, which themselves have been previously linked to GON development ^{129, 130}, would also be beneficial.

7.9.2.2 Arterial stiffness, IMT and systemic endothelial function

In conjunction with increased nocturnal SBP variability, a comparably increased systemic arterial stiffness and carotid artery IMT was identified in both our POAG and NTG patients in comparison to our healthy controls. Arterial stiffness is considered an independent predictor of cardiovascular disease ⁴⁸⁸ and increased IMT has been suggested as an indirect measure of generalised atherosclerosis and cardiovascular risk ⁷⁸². The presence of cardiovascular disease and structural vascular wall changes has been variably linked to the presence of GON over recent years, with a number of studies having identified strong associations in both POAG ^{307, 481, 482} and NTG patients ^{480, 483} individually and a number of further studies having revealed no such associations ^{318, 484, 485, 487}. The variability in results between these previous studies could in part be accounted for by differences in the inclusion/exclusion of patients suffering from already diagnosed systemic vascular disease, especially as arterial stiffness is a known

measure of vascular function ⁷⁵⁹. As most glaucoma patients seen in day-to-day practice suffer from a large variety of vascular pathologies, a careful selection of only those free from such diseases could bias results towards very rare occurrences. Consequently, the present study included POAG and NTG patients with well controlled hypertension, along with a similar number of subjects with such status in the control group. All groups were additionally matched on age and functional loss. In light of this and on the basis of our findings here, it could be hypothesised that systemic vascular wall changes may in some way contribute to, or signal a higher risk for, the development of GON in the early stages of the disease, regardless of the level of IOP. The mechanisms surrounding these associations however are currently unclear but could perhaps relate to the development of endothelial dysfunction or increased MABP, both of which have been widely linked to the presence of arterial stiffness and atherosclerosis ⁴⁸⁸ as well as to the presence of vascular dysfunction and the development of GON ^{198, 488}. On exploration however, no direct correlations could be found in this study between arterial stiffness, IMT and systemic endothelial function, however as systemic endothelial dysfunction, measured by FMD, was actually found to be comparable between groups here this finding may not be surprising. Interestingly however a significant positive correlation between nocturnal SBP variability and IMT was identified in POAG patients in this study. Such a relationship between these two parameters has also previously been reported in cardiovascular research 783, 784 and is thought to relate to the increased stress on the vascular wall imposed by increased fluctuations in BP which can result in structural alterations in the medial layer of the carotid artery ⁷⁸³. This finding provides support for the concept of a generalised systemic vascular dysfunction in glaucoma patients, however no such correlations were found in our NTG patients perhaps suggesting different mechanisms may be occurring here. Further investigation would be required to clarify this issue.

7.9.2.3 Heart rate variability (HRV) and the Autonomic nervous system (ANS)

The ANS supervises and influences the hemodynamic situation of the body through its constant regulation of HR and BP and plays a vital role in blood flow physiology ³⁹⁹. Through the assessment of HRV an indirect measure of the autonomic control of the heart and hence the state of the ANS can be gained. In the presence of ANS dysfunction blood perfusion can be either altered or impaired in multiple vascular beds throughout the body, including in the eye. Its assessment and detection in those with ocular diseases with known vascular associations, such as glaucoma, is therefore of significant interest. No significant differences in ANS parameters (LF, HF, LF:HF) where identified between POAG, NTG and healthy controls in this study following 24 hour HRV assessment. Previous studies, using similar techniques, have however been able to identify HRV alterations suggestive of a high sympathetic tone in both newly diagnosed POAG patients ⁴³ and NTG patients ⁴¹³, but those findings could not be replicated here. Stricter inclusion criteria with regard to cardiovascular disease were imposed on this study compared to that of previous studies, which could perhaps explain these findings, however before any firm conclusions can be drawn further investigation into the role of ANS in the development and progression of GON is required. Intriguingly there is evidence to suggest that the extent of any ANS dysfunction increases with glaucoma severity and progression, particularly in NTG patients ^{413, 414}, the possibility that the ANS may become involved later in the disease process of our newly diagnosed glaucoma patients can therefore not be ruled out, however again further follow up would be required to confirm this.

Interesting links have previously been made in cardiovascular research between the presence of ANS dysfunction and the presence of endothelial dysfunction in regard to the abnormal regulation of vascular tone ^{785, 786}, whereby a dysfunction in one system has been suggested to drive a dysfunction in the other. No such research however has

previously been conducted in glaucoma patients, although both parameters have been individually linked to the disease. In the present study, across all three study groups, no correlations were found between our ANS parameters (LF, HF and LF:HF) and our systemic endothelial function parameters (FMD), which themselves were all found to be comparable between groups. A larger scale study, perhaps looking at patients with varying degrees of, or progressive, POAG and NTG may help to explore the potential relationship between the ANS and the endothelium further and determine whether those associations identified in cardiovascular disease are evident in glaucoma patients at any stage of the disease process. Such a discovery would contribute towards an enhanced understanding of the pathological process involved in the development of GON.

7.9.3 Retinal vascular function

With regard to retinal vascular function both our POAG and NTG patients demonstrated a comparably steeper retinal arterial constriction slope following the cessation of flicker, as well as an increased arterial and venous baseline diameter fluctuation (BDF) in comparison to healthy controls. Slope is an important parameter which allows an evaluation of the dynamic nature of the vascular constriction profile and is influenced not only by the percentage constriction in vessel diameter below baseline following cessation of flicker but also by the time scale across which this happens. As discussed in chapter 5 the relevance of such a steeper arterial constriction slope in glaucoma patients is currently unclear as very few previous studies looking at vascular reaction to flickering light have investigated the constriction responses in detail. Nevertheless, due to the links which have previously been hypothesised between an increased overshoot in vessel diameter below baseline following to the previous studies constriction slope in glaucoma patienter below baseline following cessation of flicker, a parameter known to influence constriction slope, and the presence of PVD syndrome ⁷⁴⁶, as well as between altered astrocyte activity and re-establishment of vasomotor tone ^{764, 765}, it is not unreasonable to hypothesise that the steeper retinal arterial constriction slope identified

here signals the presence of retinal vascular dysfunction in both our POAG and NTG patients. It is interesting that a steeper arterial constriction slope was identified in both groups of glaucoma patients in comparison to controls, especially as some of the conditions to which we have theoretically linked this finding, namely PVD and raised ET-1 levels, have been more frequently related to NTG patients in the literature ^{26, 356}. As such it could be hypothesised that similar conditions to those more commonly exhibited in our NTG patients are also evident in the early stages of POAG, however in order to determine whether this is in fact the case additional information, for example on the presence of PVD symptoms, such as cold hands and feet, exploration of nail fold capillary perfusion and determination of ET-1 levels in both groups of patients would be required.

In addition to the steeper retinal arterial constriction slope, comparably increased fluctuations in baseline arterial and venous diameter, prior to the onset of flicker were also identified in both our POAG and NTG patients in comparison to healthy controls. BDF, as discussed in previous chapters, is a parameter, which although recognised is not commonly reported in the literature. Its consideration was first recommended by Nagel et al ⁶⁸⁶ as a way of taking into account the effect of spontaneous variations in vessel diameter, which occur under normal resting conditions, on the observed response of the vasculature to flicker light, however it is a parameter which is not commonly reported in the literature. As such the relevance of these findings with regard to GON development is currently unclear. Tentative links have been made in previous studies between the occurrence of increased arterial diameter fluctuations and the presence of vascular disturbance in both smokers and vasospastic subjects 695, 746-748 and between increased retinal venous diameters and the presence of retinal ischemia and hypoxia ^{749, 751}. On this basis it could be hypothesised that the increased BDF identified in both our POAG and NTG patients in this study is a further indicator that common alterations in retinal vascular function may exist in these individuals, perhaps

related to an increased variation in vascular tone or rigidity or the presence of ischemic and hypoxic changes. Interestingly a significantly positive correlation was found between arterial BDF and nocturnal BP variability in our NTG patients only. Variations in BP are known to influence baseline retinal vessel diameter even under normal resting conditions and this correlation confirms that in NTG patients in particular this influence is significant and may require consideration. Indeed on correcting for the effect of nocturnal BP variability on arterial BDF using ANCOVA the significant difference between NTG patients and controls is lost however the significance still remains between POAG patients and healthy controls. This not only highlights the importance of BP variability in NTG but also perhaps suggests that other factors separate to BP fluctuation may be affecting the stability of the arterial baseline diameter in POAG patients.

7.9.4 Oxidative stress

Oxidative stress has previously been implicated in the pathophysiology of both POAG and NTG. In this study however no significant differences in the circulating levels of the antioxidant glutathione could be found between groups (GSH, GSSG, GSH:GSSG). Such a direct comparison of oxidative stress status between POAG, NTG and controls has not previously been made, however there a number of studies which have been able to demonstrate evidence of oxidative stress in these glaucoma groups individually, in the form of increased DNA breaks ⁵²⁴, altered ET-1 and proteosome activity ^{386, 526} and decreased antioxidant activity ^{527, 528}. Of particular relevance is a study by Gherghel et al ⁷⁸⁷ which was able to identify low plasma levels of GSH and t-GSH in newly diagnosed POAG patients, suggestive of an increased oxidative burden. The patients included in this above study were slightly older than that of the present study which, due to the strong links between oxidative stress and age, could partly explain the difference in results; however it is difficult to draw any firm conclusions. Indeed, although no

significant differences in circulating glutathione levels were identified between groups in this study, the possibility that evidence of oxidative stress may be present in other forms in our POAG and NTG patients or be present in amounts which are not yet detectable systemically, cannot be ruled out. Further investigation may therefore be required in order to determine the exact associations between GON and oxidative stress.

7.9.5 POAG vs. NTG

The aim of this study was to compare and contrast the presence of vascular dysfunction at the ocular level and multiple systemic levels in POAG and NTG patients and to evaluate the validity of the concept that glaucoma may exist as a disease continuum and represent a sick eye in a sick body. This is the first time that direct comparisons such as these have been made between POAG and NTG patients across so many ocular and systemic parameters simultaneously, making it difficult to directly compare our findings to that of previous studies. Nevertheless we have identified evidence of altered retinal vascular reactivity, altered ocular perfusion, altered nocturnal BP variability and signs of systemic vascular pathology in both POAG and NTG patients, which are comparable and not replicated by healthy controls. Such findings suggest that a considerable overlap may in fact exist in the aetiology of both POAG and NTG, especially in the early stages of the disease. The finding of so many comparable parameters between our two groups of newly diagnosed glaucoma patients is perhaps somewhat surprising, especially as, on the basis of previous research, it was hypothesised that the evidence of vascular dysfunction would be more pronounced in our NTG patients. The relative influence of IOP on the results obtained in this study is difficult to determine and the possibility that the mechanisms by which these multiple vascular parameters become similarly altered could still vary between glaucoma groups cannot be ruled out. Regardless of this it does appear from our findings that the consideration of OAG as a disease continuum as opposed to as two separate clinical entities may well be a useful

and valid concept. Indeed we have clearly highlighted the need for consideration of vascular parameters in both POAG and NTG and have blurred the margins between these two forms of OAG.

An additional consideration of this study was to determine whether any further support for the concept that glaucoma should be considered a sick eye in a sick body could be found. Interestingly, in addition to the finding of altered ocular and systemic vascular parameters in our POAG and NTG patients, correlations were also found between increased nocturnal BP variability and both IMT at the systemic level and retinal artery baseline fluctuation at the ocular level. These findings highlight the possibility that a generalised vascular function, in which abnormalities at one level are influencing or being influenced by abnormalities at another level, could be present in glaucoma patients and that glaucoma, may indeed represent a sick eye in a sick body. With this in mind it could be broadly hypothesised that the increased nocturnal BP variability detected in this study, could precipitate, through the imposition of increased stress on the arterial wall, an increase in carotid artery IMT and an alteration in the functioning and reactivity of the retinal vessels in our glaucoma patients, contributing to both an increased cardiovascular risk and the presence of unstable ocular perfusion. Additional alterations in retinal microvascular regulation and astrocyte activity in these patients could then further contribute to these disruptions in retinal vascular reactivity and to the instability of ocular perfusion. The origins of the increased BP variability itself could theoretically be linked to increased arterial stiffness, the presence of which has previously been suggested to interfere with the buffering of BP alterations through reducing the compliance of the vascular wall ⁷⁸⁸. Indeed increased systemic arterial stiffness was also identified in our glaucoma patients and although not measured here, increased arterial stiffness at the ocular level has also previously been reported in NTG patients by other studies ²⁹⁵. If these factors were to combine as hypothesised and create an unstable ocular perfusion, the mechanisms by which they may then go on to

produce GON is currently uncertain, however it would not be unreasonable to hypothesise the involvement of recurrent ischemia/reperfusion injury. Interestingly however no differences in oxidative stress, a key component of ischemia/reperfusion injury or in ANS activity and systemic endothelial function, both of which are commonly linked to such vascular alterations, were found between groups in this study. Further investigation may therefore be required to elicit the exact role of these parameters in the aetiology of GON and to expand on the above mentioned hypothesis of generalised vascular dysfunction in glaucoma. Indeed it is possible that these parameters may become altered at a later stage of the disease process or may be already altered at a degree not yet detectable at the investigated level, for example in the microvasculature rather than the macrovasculature with regard to endothelial dysfunction in particular. As such their involvement although unclear cannot be ruled out completely.

7.10 Conclusion

In conclusion this study has demonstrated multiple comparable alterations in both ocular and systemic vascular function between POAG and NTG patients, which not only highlights the importance of considering vascular factors in the aetiology and treatment of both conditions, but also highlights the need to become less rigid in our separation of the two conditions into distinct clinical entities when considering vascular risk. Furthermore additional support for the concept that a generalised vascular dysfunction exists in glaucoma patients and that glaucoma should be considered a sick eye in a sick body has been provided.

7.11 Limitations

One factor not explored in this study was the precise nature of the structural ONH changes between our glaucoma groups, so it is unclear whether differences were present in this regard. Furthermore, considering the presented mean IOP values for

each group (table 7.1), it is possible that a certain degree of overlap may exist between the reported glaucoma categories, which could perhaps account for the large number of comparable findings between the POAG and NTG patients in this study. The categorising of patients into groups in this study based on clinical IOP levels above or below 21mmHg is however reflective of current practice. It would be interesting to determine through further research whether the similarities between POAG and NTG identified here remain consistent as the disease develops or whether they become less comparable with progression and also whether they remain in POAG patients with higher IOP levels (26 mmHg or over). This would be important not only from an aetiological point of view but also clinically as different treatment methods may be required at later stages of the disease process.

8. Summary and Conclusions

8.1 Summary

Evaluating the importance of ocular and systemic vascular risk factors in the aetiology of both glaucoma and AD has been a research area of interest for some time and the scientific literature relating to this has been reviewed in Chapter 1 of this thesis. It is still unclear however how hemodynamic disturbances, affecting multiple vascular beds, may interrelate to cause neuronal degeneration at the ocular and/or cerebral levels in glaucoma and AD and how valid the assessment of vascular function at the ocular level may be as an indicator of dysfunction at the cerebral level. Elucidating or validating these relationships could not only enhance our aetiological understanding of ocular and cerebral neurodegenerative disease but could also potentially open up new diagnostic or therapeutic avenues for both glaucoma and AD. As such this thesis has been concerned with investigating the presence, impact and interactions of ocular and systemic vascular alterations in POAG, NTG and AD patients and how they may relate to the pathogenesis of these neurodegenerative diseases. Furthermore this thesis has explored the concept of using the eye as a window to the brain in AD patients as well the validity of POAG and NTG as distinct clinical entities and the validity of considering glaucoma as a 'sick eye in a sick body'.

In summary the findings of this work were:

8.1.1 Is the eye a window to the brain? Ocular and systemic vascular dysfunction in Alzheimer's disease

Cerebral vascular dysregulation has previously been suggested to play a role in the development and progression of AD however the direct assessment of the cerebral vasculature is notoriously difficult. Furthermore questions still remain around exactly

how alterations in vascular function may lead to neuronal degeneration in AD. The aim of this study was therefore to determine for the first time whether vascular dysregulation could be detected at the ocular level in mild newly diagnosed AD patients using novel analysis methods and whether the degree of any dysregulation discovered could relate to either the degree of cognitive impairment or the presence of systemic vascular dysfunction in these patients. Following dynamic retinal vascular analysis, alterations in the retinal arterial response to flicker light, in the form of an increased arterial RT on two out of the three occasions in which the vessels were stimulated, along with alterations in baseline venous diameter, were detected in our mild AD patients. Furthermore this alteration in retinal arterial RT was found to correlate positively with degree of cognitive impairment in these patients. No significant differences in systemic endothelial function were however identified between groups. The exact cause of the retinal vascular alterations identified in this study can only be hypothesised at this point but nevertheless they provide support for a vascular aetiology in AD and, through the identification of retinal vascular dysfunction in mild AD patients for the first time, highlight the involvement of the ocular circulation in the disease process. Furthermore this study firmly introduces the idea that the functioning of the ocular circulation, as assessed by DVA, may be a useful marker for determining cognitive prognosis in AD patients and has provided a good basis for further exploration into vascular dysfunction in AD.

8.1.2 Ocular and systemic vascular abnormalities in newly

diagnosed normal tension glaucoma patients

NTG has been widely linked to the presence of ocular and systemic vascular alterations however very few studies have simultaneously explored and directly compared the presence of ocular and systemic vascular alterations in these patients and many questions still remain around exactly how these alterations may relate to RGC loss and GON development. As such the aim of this study was to investigate the presence of

cardiovascular risk factors and alterations in vascular function at both the ocular and systemic level in NTG patients, using novel investigative techniques, and to evaluate the concept that glaucoma may be considered a 'sick eye in a sick body'. It was hypothesised that, on the back of recent evidence highlighting the involvement of both the ocular and systemic circulations in NTG, our newly diagnosed and previously untreated NTG patients would demonstrate altered retinal vascular reactivity to flicker light in conjunction with signs of systemic vascular dysfunction. On investigation our NTG patients were indeed found to exhibit increased systemic arterial stiffness and carotid artery IMT in conjunction with alterations in retinal arterial reactivity to flicker light stimulation, in the form of an altered dynamic response profile and a steeper constriction slope following cessation of flicker. No direct correlations however could be found between the systemic and ocular vascular parameters explored in this study, and furthermore no significant differences in systemic endothelial function could be identified between groups. These findings indicate that subclinical signs of systemic vascular disease and disturbances in ocular vascular reactivity appear to be present in NTG patients at the earliest stages of the disease process and could potentially be contributing to the disease aetiology. The importance of considering systemic vascular factors in the diagnosis and management of NTG has therefore been highlighted and, furthermore, through the utilisation of novel analysis methods in conjunction with DVA assessment, a contribution to the aetiological understanding of vascular dysfunction at the ocular level in NTG has also been made. Additionally, although not conclusive, this research goes someway to reaffirming the concept that glaucoma may be considered 'a sick eye in a sick body'. Further exploration into this concept, using a wider range of systemic investigation techniques, was subsequently conducted in chapter 7 of this thesis.

8.1.3 Ocular Vascular Dysregulation in AD compares to both POAG and NTG

Having established the presence of retinal vascular dysfunction in both AD and NTG patients individually, an investigation into whether the nature of this vascular dysfunction is comparable between these groups of patients was the next logical step. A number of associations have previously been made between the presence and aetiology of these two neurodegenerative diseases; however the exact nature of their relationship is still uncertain. As such the aim of this study was to investigate for the first time, using novel techniques, the coexistence of vascular alterations at both the ocular and systemic level, in AD and both POAG and NTG patients, with the hypothesis that, on the back of recent evidence highlighting the potential aetiological similarities between these conditions, comparable alterations in vascular function would be found. On investigation our mild newly diagnosed AD patients were found to demonstrate an altered retinal arterial vessel reactivity to flicker light which was comparable to that of our POAG patients and an alteration in baseline venous reactivity which was comparable to that of our NTG patients; however no differences in systemic endothelial function were identified between groups. These findings go some way to provide support for the concept of a common underlying microvascular aetiology in AD and glaucoma and highlight the need to consider ocular health in AD and cognitive health in glaucoma. Interestingly, the similarities in the nature of the retinal vascular alterations identified in our AD and glaucoma patients appear to differ depending on whether POAG or NTG is being considered, this is a novel finding which would benefit from further investigation in order to clarify its significance. It also provokes questions about the similarities and differences which may exist in ocular and systemic vascular dysfunction between POAG and NTG patients themselves, which was subsequently explored in the following chapter.

8.1.4 Primary open angle glaucoma and Normal tension glaucoma: two separate diseases or one continuous entity? The vascular perspective

Although more traditionally associated with NTG, vascular alterations at both the ocular and systemic level have also previously been identified in POAG patients and in a similar manner the benefits of IOP reduction, more traditionally linked with POAG, have also been demonstrated in NTG patients. Overlaps such as these has lead to the questioning of whether these two forms of glaucoma should still be considered as distinct clinical entities or whether glaucoma may in fact represent a disease continuum with no distinct borders of separation. As such the aim of this study was to compare and contrast the presence of ocular and systemic vascular alterations at multiple levels in both newly diagnosed and previously untreated POAG and NTG patients and to not only evaluate the validity of these two conditions as distinct clinical entities but to also further explore the concept that glaucoma may represent a 'sick eye in a sick body'. It was hypothesised that, on the back of recent evidence, ocular and systemic vascular dysfunction would be present in both POAG and NTG patients but the extent of this vascular involvement would be greater in NTG patients. On investigation however comparative alterations in retinal vessel reactivity, OPP, nocturnal SBP variability and both systemic arterial stiffness and carotid IMT were found in our POAG and NTG patients. Furthermore significant correlations between systemic BP variability, IMT and retinal arterial baseline diameter fluctuation were found across the glaucoma groups. No significant differences in HRV, systemic endothelial function or oxidative stress parameters were however found between groups. This is the first time that direct comparisons between so many ocular and systemic parameters have been made in POAG and NTG patients. The inter-relationships identified between the investigated ocular and systemic parameters certainly provide support for the concept of a generalised vascular dysfunction in glaucoma and for the concept that glaucoma may

represent a 'sick eye in a sick body'. Furthermore, although the finding of so many comparable vascular alterations between both POAG and NTG patients was perhaps surprising, it highlights the importance of considering the role of vascular factors in both forms of glaucoma and goes some to way to supporting the concept that glaucoma should be considered as a disease continuum rather than a disease with distinct clinical forms.

8.2 Conclusions

The aims of this work were:

8.2.1 To investigate the presence and impact of ocular and systemic vascular alterations in AD and to explore the concept of using the 'eye as a window to the brain'

The findings of this work were:

- An altered retinal vascular response to flicker light stimulation, indicative of ocular vascular dysfunction, was identified on DVA analysis in mild newly diagnosed AD patients
- The degree of this retinal vascular dysfunction was found to correlate with the degree of cognitive impairment in these AD patients
- No significant differences in systemic endothelial dysfunction were identified between groups
- Support for the concept of using the 'eye as a window to the brain' for the diagnosis and screening of cognitive impairment was provided

8.2.2 To investigate the presence and impact of ocular and systemic vascular alterations in NTG

The findings of this work were:

- An altered retinal arterial vascular constriction response following flicker light stimulation, indicative of ocular vascular dysfunction, was identified on DVA analysis in newly diagnosed NTG patients
- Signs of subclinical vascular pathology in the form of increased systemic arterial stiffness and carotid artery IMT were also identified in these NTG patients
- No significant differences in systemic endothelial function were identified between groups

8.2.3 To investigate the possibility of a shared vascular aetiology, involving both the ocular and systemic circulations, in AD and both POAG and NTG.

The findings of this work were:

- AD and POAG patients demonstrated a common alteration in retinal arterial function in the form of a prolonged reaction time to flickering light
- AD and NTG patients demonstrated a common alteration in retinal venous function in the form of an increased baseline fluctuation in venous diameter
- No significant differences in systemic endothelial function were identified between groups
- The possibility that AD and glaucoma may share a common underlying microvascular aetiology was highlighted, along with possibility that the nature of this shared aetiology may differ between POAG and NTG patients

8.2.4 To compare and contrast vascular alterations at both the ocular and systemic level in POAG and NTG and to explore their validity as distinct clinical entities

The findings of this work were:

- POAG and NTG patients demonstrated comparable alterations in nocturnal SBP variability, OPP, retinal vascular reactivity, systemic arterial stiffness and carotid artery IMT
- Correlations were identified between nocturnal SBP variability and both retinal artery baseline diameter fluctuation and carotid IMT, highlighting the possibility that a generalised vascular dysfunction, affecting multiple vascular beds may be present in OAG patients and providing support for the concept that glaucoma may be considered a 'sick eye in a sick body'
- On the back of these finding the validity of POAG and NTG as distinct clinical entities can be questioned and the concept that glaucoma may represent a 'sick eye in a sick body' can be somewhat reaffirmed.

8.3 Overall Limitations

The studies outlined in this thesis are subject to a number of potential limitations. Firstly, only a small sample of AD patients and a moderate sample of glaucoma patients could be recruited. This, along with the cross-sectional design, potentially limits the conclusions which can be drawn from the presented results as well as the statistical power of the analysis. The reason for this limited sample of patients was the very strict patient inclusion/exclusion criteria, which were necessary to avoid any possible unwanted influences on the measured vascular parameters, but unavoidably limited the number of suitable study patients that could be recruited.

Secondly, with regard to the glaucoma patients, objective data such as the precise nature of the structural ONH changes and VF defects was not recorded and used in analysis. Such parameters were however considered by the clinicians in the diagnosis of POAG and NTG and were used indirectly to group the POAG and NTG patients. Inclusion of such objective data could have potentially allowed clearer distinctions between the glaucoma groups involved in this study to be made and enhanced the interpretation of results.

Finally multiple parameters were assessed and analysed simultaneously in AD, glaucoma and healthy control patients in this thesis. There is a risk when comparing multiple parameters that the rate of type 1 errors (false positives) will be increased. On statistical advice, a correction for multiple comparisons was therefore made in this thesis when comparing multiple sequential parameters on DVA analysis, whereby the p value for significance was reduced from p=0.05 to p=0.01. As the majority of other investigative techniques included in this thesis had only single output parameters a correction for multiple comparisons was not deemed necessary in those cases. It is however still possible that an increased family-wise error rate could have influenced the significant findings of this thesis, especially when reviewing the results of all of the investigative techniques as a whole. Significant parameters where the p-value is close to the 0.05 boundary should therefore be considered with caution.
8.4 Clinical implications

8.4.1 Implications with regard to AD

The need to consider the impact of vascular dysregulation and the role of vascular risk factors in the aetiology, diagnosis and management of AD has been highlighted. Furthermore the potential for utilising the easily accessible retinal vasculature to gain an insight into cerebral vascular alterations and/or cognitive risk, via DVA analysis, has been demonstrated. Finally the need to consider ocular health in AD patients, in the form of both POAG and NTG, has also been emphasised. On the basis of these findings and despite the need for further research and expansion of the study, it is recommended that the evaluation and assessment of ocular vascular function, ocular health and the presence of vascular risk factors should be considered in both the diagnosis and monitoring of AD, particularly in the early and pre-clinical stages of the disease.

8.4.2 Implications with regard to glaucoma

The potential role of both ocular and systemic vascular factors in the aetiology of POAG and NTG has been highlighted. Furthermore the coexistence of vascular alterations at multiple sites throughout the body and the impact of these alterations on the ocular circulation have been demonstrated in glaucoma patients. As such the extensive screening and assessment of both ocular vascular function and systemic vascular risk factors, to include 24 hour ABPM and arterial stiffness assessment, is recommended in the routine diagnosis and management of both of these groups of glaucoma patients and may be particularly relevant in cases where the disease is progressing.

8.4.3 Implications with regard to assessment of ocular vascular function

A number of different ocular and systemic vascular techniques were conducted in the thesis. With regard to ocular vascular function and DVA in particular, through the utilisation of SDRA and the introduction of our novel Matlab imaging analysis, the dynamic nature of the retinal vascular response profile to flicker light was able to been explored fully for the first time. The importance of considering the full vascular profile, including constriction response, slope and reaction times was highlighted by the significant differences found in these parameters in this thesis, which would otherwise have been missed. It is therefore recommended that such analysis should be conducted in all future studies utilising DVA.

8.5 Future Directions

A number of new questions have arisen from the data presented in the major chapters of this thesis from which the following avenues of future research are worth highlighting in particular:

8.5.1 Expansion of preliminary data

Despite statistically significant results being demonstrated in this thesis with the sample sizes obtained, expansion of the four main presented studies to include a larger cohort of patients would enhance the validity and application of our findings. Furthermore expansion of our blood analyses to include the endothelial marker ET-1 could offer additional information on the state of the vascular endothelium and vessel tonus, which could in turn offer additional information of both aetiological and clinical relevance with regard to neurodegenerative disease.

8.5.2 Assessment of retinal vascular dysfunction in mild cognitive impairment (MCI)

As well as expanding the presented research to include a larger sample of mild AD patients, it would be interesting to determine at what stage in the disease process such vascular alterations become apparent and whether their detection and management can impact on disease prognosis. Of particular interest in this regard would be the assessment of retinal vascular function in MCI patients. These patients are recognised as having memory impairment greater than that expected through aging alone and are considered to be at an increased risk for AD development. Indeed around 10-15% of MCI patients are thought to convert to AD annually, in comparison to only 1-2% of the normal elderly population . Assessment of retinal vascular function in MCI patients could not only provide an insight into whether vascular alterations are present at this mild level of cognitive impairment but could also, if conducted as part of a longitudinal study, provide an insight into whether the presence of retinal vascular dysfunction may predict or increase their risk of progression to AD. Furthermore a comparison between the nature of retinal vascular dysfunction in MCI patients and in those with mild and/or more advanced AD could provide an insight into how such vascular dysfunctions may contribute to neurodegenerative disease development and/or progression.

8.5.3 Impact of ocular and systemic vascular dysfunction on the progression of POAG and NTG

The presence of comparable alterations in ocular and systemic vascular dysfunction has been identified in mild newly diagnosed POAG and NTG patients in this thesis. As well as expanding on the aetiological relevance of these findings, future work assessing, through means of longitudinal study, whether the presence of such vascular alterations and risk factors at the earliest stages of the disease process affect the speed or severity

of disease progression and/or whether modification of these parameters, where possible, aids disease management would be beneficial in determining their clinical relevance. Furthermore the assessment of both ocular and systemic vascular function in glaucoma patients who are progressing despite maximum IOP therapy would also be of interest as it is widely thought that vascular or IOP-independent factors play a greater role in disease aetiology in such individuals. Indeed the assessment of ocular and systemic vascular function in progressive glaucoma patients not only has the potential to provide an insight into the vascular factors linked to this progression but could also open up new treatment and management options for these individuals without having to resort to surgery.

8.5.4 Development of dynamic retinal vessel analysis as a tool for the assessment of neurodegenerative disease

A database of normative values for retinal artery and venous function as assessed by DVA is not currently available. As such, although it has strong research implications, the viability of DVA as a diagnostic tool in neurodegenerative disease is currently limited. Future work to establish both a normative DVA database, across a range of ages and ethnic groups, using the parameters presented in this thesis and a disease related DVA database, across a variety of disease stages, is therefore needed.

8.5.5 Expansion of flow mediated dilation analysis

Current analysis of the systemic endothelial response to FMD is limited, with the percentage dilation response being the only parameter considered. Expansion of this analysis, to include parameters such as response and recovery times, dilation and constriction slopes and baseline diameter fluctuation, which are currently more commonly associated with DVA, could enhance the evaluation of systemic endothelial function and make comparisons between our ocular (DVA) and systemic vascular

parameters (FMD) more reliable. The development of new software programs and algorithms, capable of coping with large amounts of data, would be necessary for this to be conducted successfully.

9. References

- 1. Kawamata H, Manfredi G. Introduction to neurodegenerative diseases and related techniques. *Methods Mol Biol* 2011;793:3-8.
- 2. trust Asdr. Dementia statistics. . *wwwalzheimers-researchorguk/info/statistics*; 2010.
- 3. Guo L, Duggan J, Cordeiro MF. Alzheimer's Disease and Retinal Neurodegeneration. *Current Alzheimer research* 2010;7:3-14.
- 4. Patton N, Aslam T, Macgillivray T, Pattie A, Deary I, Dhillon B. Retinal vascular image analysis as a potential screening tool for cerebrovascular disease: a rationale based on homology between cerebral and retinal microvasculatures. *Journal of anatomy* 2005;206:319-348.
- 5. Van Buskirk E, Cioffi G. Glaucomatous optic neuropathy. *Am J Ophthalmol* 1992;113:447-452.
- 6. IGA IGA. About Glaucoma. Kent: PureBlue Creative; 2009:Webpage.
- 7. Mackenzie P, Cioffi G. How Does Lowering of Intraocular Pressure Protect the Optic Nerve? *Survey of Ophthalmology* 2008;53:S39-S43.
- 8. Gaasterland D, Kupfer C. Reports: Experimental Glaucoma in the Rhesus Monkey. *Invest Ophthalmol Vis Sci* 1974;13:455-457.
- 9. Nathan J. Hippocrates to Duke-Elder: an overview of the history of glaucoma. *Clinical and Experimental Optometry* 2000;83:116-118.
- 10. Jaeger E. Ueber Glaukom und seine Heilung durch Iridektomie. *Z GesAerzteWien* 1858;14:465-491.
- 11. Smith P. A Case of Chronic Glaucoma of Unusually Long Duration. *The American Journal of the Medical Sciences* 1886;181:292.
- 12. Rudnicka AR, Mt-Isa S, Owen CG, Cook DG, Ashby D. Variations in Primary Open-Angle Glaucoma Prevalence by Age, Gender, and Race: A Bayesian Meta-Analysis. *Investigative ophthalmology & visual science* 2006;47:4254-4261.
- 13. Nemesure B, Honkanen R, Hennis A, Wu SY, Leske MC. Incident Open-angle Glaucoma and Intraocular Pressure. *Ophthalmology* 2007;114:1810-1815.
- 14. Wolfs RCW, Klaver CCW, Ramrattan RS, van Duijn CM, Hofman A, de Jong PTVM. Genetic Risk of Primary Open-angle Glaucoma: Population-Based Familial Aggregation Study. *Arch Ophthalmol* 1998;116:1640-1645.
- 15. Leske MC, Wu S-Y, Hennis A, Honkanen R, Nemesure B. Risk Factors for Incident Open-angle Glaucoma - The Barbados Eye Studies. *Ophthalmology* 2008;115:85-93.
- 16. Hyman L, Wu S-Y, Connell AMS, et al. Prevalence and causes of visual impairment in the Barbados eye study1 1 The authors have no proprietary interest in the articles or devices mentioned herein. *Ophthalmology* 2001;108:1751-1756.
- 17. Mitchell P, Hourihan F, Sandbach J, Jin Wang J. The relationship between glaucoma and myopia: The blue mountains eye study. *Ophthalmology* 1999;106:2010-2015.
- 18. Xu L, Wang Y, Wang S, Wang Y, Jonas JB. High Myopia and Glaucoma Susceptibility: The Beijing Eye Study. *Ophthalmology* 2007;114:216-220.
- 19. Burk ROW, Rohrschneidern K, Noack H, Völcker HE. Are large optic nerve heads susceptible to glaucomatous damage at normal intraocular pressure? *Graefe's Archive for Clinical and Experimental Ophthalmology* 1992;230:552-560.
- 20. Herndon LW, Weizer JS, Stinnett SS. Central Corneal Thickness as a Risk Factor for Advanced Glaucoma Damage. *Arch Ophthalmol* 2004;122:17-21.
- 21. Doyle A, Bensaid A, Lachkar Y. Central corneal thickness and vascular risk factors in normal tension glaucoma. *Acta Ophthalmologica Scandinavica* 2005;83:191-195.

- 22. Bonomi L, Marchini G, Marraffa M, Bernardi P, Morbio R, Varotto A. Vascular Risk Factors for Primary Open Angle Glaucoma - The Egna-Neumarkt Study. *Ophthalmology* 2000;107:1287-1293.
- 23. Leske MC, Wu S-Y, Nemesure B, Hennis A. Incident Open-Angle Glaucoma and Blood Pressure *Arch Ophthalmol* 2002;120:954-959.
- 24. Leske MC, Heijl A, Hyman L, Bengtsson B, Dong L, Yang Z. Predictors of Longterm Progression in the Early Manifest Glaucoma Trial. *Ophthalmology* 2007;114:1965-1972.
- 25. Logan JFJ, Rankin SJA, Jackson AJ. Retinal blood flow measurements and neuroretinal rim damage in glaucoma. *British Journal of Ophthalmology* 2004;88:1049-1054.
- 26. Gasser P, Flammer J. Blood-cell velocity in the nail fold capillaries of patients with normal-tension and high-tension glaucoma. *Am J Ophthalmol* 1991;111:585-588.
- 27. Gherghel D, Griffiths H, Hilton E, Cunliffe I, Hosking S. Systemic reduction in glutathione levels occurs in patients with primary open-angle glaucoma. *Investigative ophthalmology & visual science* 2005;46:877-883.
- 28. Izzotti A, Bagnis A, Saccà SC. The role of oxidative stress in glaucoma. *Mutation Research/Reviews in Mutation Research* 2006;612:105-114.
- 29. Orgul S, Flammer J, Gasser P. Female preponderance in normal-tension glaucoma. *Annals of ophthalmology-glaucoma* 1995;27:355-359.
- 30. Pekmezci M, Vo B, Lim AK, et al. The Characteristics of Glaucoma in Japanese Americans. *Archives of ophthalmology* 2009;127:167-171.
- 31. Kitazawa Y, Shirato S, Yamamoto T. Optic disc hemorrhage in low-tension glaucoma. *Ophthalmology* 1986;93:853-857.
- 32. Shields M. Normal-tension glaucoma: is it different from primary open-angle glaucoma? *Curr Opin Ophthalmol* 2008;19:85-88.
- 33. EGS. European Glaucoma Society: Terminology and Guidelines for Glaucoma, 3rd Edition. Savona, Italy: Dogma; 2008.
- 34. Sommer A. Ocular Hypertension and Normal-Tension Glaucoma: Time for Banishment and Burial. *Arch Ophthalmol* 2011;129:785-787.
- 35. Shields MB, Spaeth GL. The Glaucomatous Process and the Evolving Definition of Glaucoma. *Journal of glaucoma* 2011;21:141-143.
- 36. CNTGS. Comparison of glaucomatous progression between untreated patients with normal-tension glaucoma and patients with therapeutically reduced intraocular pressures. *American Journal of Ophthalmology* 1998;126:487-497.
- 37. CNTGS. The effectiveness of intraocular pressure reduction in the treatment of normal-tension glaucoma. *American Journal of Ophthalmology* 1998;126:498-505.
- 38. Heijl A, Leske MC, Bengtsson B. Reduction of intraocular pressure and glaucoma progression: results from the Early Manifest Glaucoma Trial. *Arch Ophthalmol* 2002;120:1268-1279.
- 39. Feke GT, Pasquale LR. Retinal Blood Flow Response to Posture Change in Glaucoma Patients Compared with Healthy Subjects. *Ophthalmology* 2008;115:246-252.
- 40. Pemp B, Georgopoulos M, Vass C, et al. Diurnal fluctuation of ocular blood flow parameters in patients with primary open-angle glaucoma and healthy subjects. *British Journal of Ophthalmology* 2009;93:486-491.
- 41. Garhofer G, Zawinka C, Resch H, Huemer K, Schmetterer L, Dorner GT. Response of Retinal Vessel Diameters to Flicker Stimulation in Patients with Early Open Angle Glaucoma. *J Glaucoma* 2004;13:340-344.
- 42. Su W-W, Cheng S-T, Ho W-J, Tsay P-K, Wu S-C, Chang SHL. Glaucoma Is Associated with Peripheral Vascular Endothelial Dysfunction. *Ophthalmology* 2008;115:1173-1178.e1171.

- 43. Gherghel D, Hosking S, Armstrong R, Cunliffe I. Autonomic dysfunction in unselected and untreated primary open angle glaucoma patients: a pilot study. *Ophthalmic & physiological optics* 2007;27:336-341.
- 44. Thonginnetra O, Greenstein VC, Chu D, Liebmann JM, Ritch R, Hood DC. Normal Versus High Tension Glaucoma: A Comparison of Functional and Structural Defects. *Journal of Glaucoma* 2010;19:151-157 110.1097/IJG.1090b1013e318193c318145c.
- 45. Asrani S, Samuels B, Thakur M, Santiago C, Kuchibhatla M. Clinical Profiles of Primary Open Angle Glaucoma versus Normal Tension Glaucoma Patients: A Pilot Study. *Current Eye Research* 2011;36:429-435.
- 46. Sowka J. New thoughts on normal tension glaucoma. *Optometry* 2005;76:600-608.
- 47. Yamamoto T, Kitazawa Y. Vascular pathogenesis of normal-tension glaucoma: a possible pathogenetic factor, other than intraocular pressure, of glaucomatous optic neuropathy. *Progress in Retinal and Eye Research* 1998;17:127-143.
- 48. Caprioli J, Spaeth GL. Comparison of the Optic Nerve Head in High- and Low-Tension Glaucoma. *Arch Ophthalmol* 1985;103:1145-1149.
- 49. Fazio P, Krupin T, Feitl ME, Werner EB, Carre DA. Optic Disc Topography in Patients With Low-Tension and Primary Open Angle Glaucoma. *Arch Ophthalmol* 1990;108:705-708.
- 50. Caprioli J, Spaeth GL. Comparison of visual field defects in the low-tension glaucomas with those in the high-tension glaucomas. *Am J Ophthalmol* 1984;97:730-737.
- 51. Hitchings RA, Anderton SA. A comparative study of visual field defects seen in patients with low-tension glaucoma and chronic simple glaucoma. *British Journal of Ophthalmology* 1983;67:818-821.
- 52. Su W-W, Cheng S-T, Hsu T-S, Ho W-J. Abnormal Flow-Mediated Vasodilation in Normal-Tension Glaucoma Using a Noninvasive Determination for Peripheral Endothelial Dysfunction. *Invest Ophthalmol Vis Sci* 2006;47:3390-3394.
- 53. Plange N, Kaup M, Daneljan L, Predel HG, Remky A, Arend O. 24-h blood pressure monitoring in normal tension glaucoma: night-time blood pressure variability. *Journal of Human Hypertension* 2006;20:137-142.
- 54. Hayreh SS, Zimmerman MB, Podhajsky P, Alward WL. Nocturnal arterial hypotension and its role in optic nerve head and ocular ischemic disorders. *American Journal of Ophthalmology* 1994;117:603-624.
- 55. Phelps CD, Corbett JJ. MIGRAINE AND LOW-TENSION GLAUCOMA A CASE-CONTROL STUDY. *Investigative ophthalmology & visual science* 1985;26:1105-1108.
- 56. Dowling J. Organisation of vertebrate retinas. The Jonas M Friedenwald Memorial Lecture. *Investigative Ophthalmology* 1970;9:655-680.
- 57. Kolb H. Morphology and Circuitry of Ganglion Cells. In: Kolb H, Fernandez E, Nelson R (eds), *Webvision: The Organization of the Retina and Visual System* [*Internet*] Salt Lake City (UT): University of Utah Health Sciences Center; 1995-. 2001.
- 58. Kolb H, Fernandez E, Nelson R (eds). *Webvision: The Organization of the Retina and Visual System [Internet]*. Salt Lake City (UT): University of Utah Health Sciences Center; 1995-.
- 59. Kolb H. Photoreceptors. In: Kolb H, Fernandez E, Nelson R (eds), *Webvision: The Organization of the Retina and Visual System [Internet].* Salt Lake City (UT): University of Utah Health Sciences Center; 2005.
- 60. Kolb H. Outer plexiform layer. In: Kolb H, Fernandez E, Nelson R (eds), Webvision: The Organization of the Retina and Visual System [Internet]. Salt Lake City (UT): University of Utah Health Sciences Center; 2005.
- 61. Wichmann W, Müller-Forell W. Anatomy of the visual system. *European Journal* of *Radiology* 2004;49:8-30.

- 62. Hayreh SS. ANATOMY AND PHYSIOLOGY OF OPTIC NERVE HEAD. *Transactions - American Academy of Ophthalmology and Otolaryngology* 1974;78:O240-O254.
- 63. Wolter JR. The human optic papilla: A demonstration of new anatomic and pathologic findings. *Am J Ophthalmol* 1957;44:48-65.
- 64. Hayreh SS. Blood supply of the Optic Nerve Head. *Ophtalmologica* 1996;210:285-295.
- 65. Levitzky M, Henkind P. Angioarchitecture of the optic nerve II. Lamina Cribrosa. *Am J Ophthalmol* 1969;68:986-996.
- 66. Jonas JB, Gusek GC, Naumann GO. Optic disc, cup and neuroretinal rim size, configuration and correlations in normal eyes. *Investigative ophthalmology & visual science* 1988;29:1151-1158.
- 67. Jonas JB, Budde WM, Panda-Jonas S. Ophthalmoscopic Evaluation of the Optic Nerve Head. *Survey of Ophthalmology* 1999;43:293-320.
- 68. Britton R, Drance SM, Schulzer M. The area of the neuroretinal rim of the optic nerve in normal eyes. *Am J Ophthalmol* 1987;103:497-504.
- 69. Bengtsson B. The variation and covariation of cup and disk diameters. *Acta Ophthalmologica* 1976;54:804-818.
- 70. Hayreh SS. Orbital vascular anatomy. *Eye* 2006;20:1130-1144.
- 71. Hayreh SS. The Ophthalmic Artery: III. Branches. *British Journal of Ophthalmology* 1962;46:212-247.
- 72. Tsutsumi S, Rhoton ALJ. Microsurgical Anatomy of the Central Retinal Artery. *Neurosurgery* 2006;59:870-879 810.1227/1201.NEU.0000232654.0000215306.0000232654A.
- Hayreh SS. Posterior Ciliary Artery Circulation in Health and Disease The Weisenfeld Lecture. *Investigative ophthalmology & visual science* 2004;45:749-757.
- 74. Nickla DL, Wallman J. The multifunctional choroid. *Progress in Retinal and Eye Research* 2010;29:144-168.
- 75. Zhang H. Scanning Electron-Microscopic Study of Corrosion Casts on Retinal and Choroidal Angioarchitecture in Man and Animals. *Progress in Retinal and Eye Research* 1994;13:243-270.
- 76. Alm A, Bill A. Ocular and optic nerve blood flow at normal and increased intraocular pressures in monkeys (Macaca irus): a study with radioactively labelled microspheres including flow determinations in brain and some other tissues. *Experimental Eye Research* 1973;15:15-29.
- 77. Alm A, Bill A. Blood Flow and Oxygen Extraction in the Cat Uvea at Normal and High Intraocular Pressures. *Acta Physiologica Scandinavica* 1970;80:19-28.
- 78. Delaey C, van de Voorde J. Regulatory Mechanisms in the Retinal and Choroidal Circulation. *Ophthalmic Research* 2000;32:249-256.
- 79. Chou PI, Lu DW, Chen JT. Bilateral Superior Cervical Ganglionectomy Increases Choroidal Blood Flow in the Rabbit. *Ophthalmologica* 2000;214:421-425.
- 80. Steinle JJ, Krizsan-Agbas D, Smith PG. Regional regulation of choroidal blood flow by autonomic innervation in the rat. *Am J Physiol Regul Integr Comp Physiol* 2000;279:R202-209.
- 81. Polska E, Simader C, Weigert G, et al. Regulation of Choroidal Blood Flow during Combined Changes in Intraocular Pressure and Arterial Blood Pressure. *Invest Ophthalmol Vis Sci* 2007;48:3768-3774.
- 82. Riva CE. Effect of acute decreases of perfusion pressure on choroidal blood flow in humans. *Investigative ophthalmology & visual science* 1997;38:1752.
- 83. Hayreh SS. In vivo Choroidal Circulation and its Watershed Zones. *Eye* 1990;4:273-289.
- 84. Hickam JB. A study of retinal venous blood oxygen saturation in human subjects by photographic means. *Circulation* 1963;27:375.

- 85. Russell RW. Evidence for autoregulation in human retinal circulation. *The Lancet* 1973;302:1048-1050.
- 86. Dumskyj MJ, Eriksen JE, Doré CJ, Kohner EM. Autoregulation in the Human Retinal Circulation: Assessment Using Isometric Exercise, Laser Doppler Velocimetry, and Computer-Assisted Image Analysis. *Microvascular Research* 1996;51:378-392.
- 87. Riva CE, Sinclair SH, Grunwald JE. Autoregulation of retinal circulation in response to decrease of perfusion pressure. *Investigative ophthalmology & visual science* 1981;21:34-38.
- 88. Hayreh SS. Blood supply of the optic nerve head and its role in optic atrophy, glaucoma, and oedema of the optic disc. *British Journal of Ophthalmology* 1969;53:721.
- 89. Hofman P, Hoyng P, vanderWerf F, Vrensen GFJM, Schlingemann RO. Lack of Blood–Brain Barrier Properties in Microvessels of the Prelaminar Optic Nerve Head. *Investigative ophthalmology & visual science* 2001;42:895-901.
- 90. Rubin LL, Staddon JM. THE CELL BIOLOGY OF THE BLOOD-BRAIN BARRIER. Annual Review of Neuroscience 1999;22:11-28.
- 91. Funk R. Blood supply of the Retina. *Ophthalmic Research* 1997;29:320-325.
- 92. Rechtman E, Harris A. Choroidal blood flow regulation and possible implications to glaucoma. *Clinical & experimental ophthalmology* 2008;36:111-112.
- 93. Pillunat LE, Anderson DR, Knighton RW, Joos KM, Feuer WJ. Autoregulation of Human Optic Nerve Head Circulation in Response to Increased Intraocular Pressure. *Experimental Eye Research* 1997;64:737-744.
- 94. Movaffaghy Á, Chamot SŔ, Petrig BL, Riva ČE. Blood Flow in the Human Optic Nerve Head during Isometric Exercise. *Experimental Eye Research* 1998;67:561-568.
- 95. Flammer J, Orgül S, Costa VP, et al. The impact of ocular blood flow in glaucoma. *Progress in Retinal and Eye Research* 2002;21:359-393.
- 96. Cheung N, McNab AA. Venous Anatomy of the Orbit. *Investigative* ophthalmology & visual science 2003;44:988-995.
- 97. Dinn R, Kagemann L, Rechtman E, Harris A. A review of methods for human retinal oximetry. *Ophthalmic surgery, lasers & imaging* 2003;34:152-164.
- 98. Harris A, Weizman Y, Ehrlich R, Rechtman E, Ben Zion I. [An updated review of methods for human retinal oximetry measurements and current applications]. *Harefuah* 2008;147:812-817,836.
- 99. Weber AJ, Kaufman PL, Hubbard WC. Morphology of single ganglion cells in the glaucomatous primate retina. *Investigative ophthalmology & visual science* 1998;39:2304-2320.
- 100. Garcia-Valenzuela E, Shareef S, Walsh J, Sharma SC. Programmed cell death of retinal ganglion cells during experimental glaucoma. *Experimental Eye Research* 1995;61:33-44.
- 101. Nork TM. Swelling and loss of photoreceptors in chronic human and experimental glaucomas. *Archives of ophthalmology* 2000;118:235.
- 102. Janssen P, Naskar R, Moore S, Thanos S, Thiel H-J. Evidence for glaucomainduced horizontal cell alterations in the human retina. *Ger J Ophthalmol* 1996;5:378-385.
- 103. Yucel YH, Zhang Q, Gupta N, Kaufman PL, Weinreb RN. Loss of neurons in magnocellular and parvocellular layers of the lateral geniculate nucleus in glaucoma. *Archives of ophthalmology* 2000;118:378-384.
- 104. Yucel YH. Effects of retinal ganglion cell loss on magno-, parvo-, koniocellular pathways in the lateral geniculate nucleus and visual cortex in glaucoma. *Progress in Retinal and Eye Research* 2003;22:465.
- 105. Quigley HA, Addicks EM, Green WR, Maumenee AE. Optic Nerve Damage in Human Glaucoma: II. The Site of Injury and Susceptibility to Damage. *Arch Ophthalmol* 1981;99:635-649.

- 106. Quigley H. The dynamics and location of axonal transport blockade by acute intraocular pressure elevation in primate optic nerve. *Investigative ophthalmology & visual science* 1976;15:606.
- 107. Anderson DR, Hendrickson A. Effect of Intraocular Pressure on Rapid Axoplasmic Transport in Monkey Optic Nerve. *Investigative ophthalmology & visual science* 1974;13:771-783.
- 108. Bellezza AJ, Rintalan CJ, Thompson HW, Downs JC, Hart RT, Burgoyne CF. Deformation of the Lamina Cribrosa and Anterior Scleral Canal Wall in Early Experimental Glaucoma. *Investigative ophthalmology & visual science* 2003;44:623-637.
- 109. Yang H, Downs JC, Bellezza A, Thompson H, Burgoyne CF. 3-D Histomorphometry of the Normal and Early Glaucomatous Monkey Optic Nerve Head: Prelaminar Neural Tissues and Cupping. *Investigative ophthalmology & visual science* 2007;48:5068-5084.
- 110. Yang H. 3-D histomorphometry of the normal and early glaucomatous monkey optic nerve head: lamina cribrosa and peripapillary scleral position and thickness. *Investigative ophthalmology & visual science* 2007;48:4597.
- 111. Burgoyne CF, Crawford Downs J, Bellezza AJ, Francis Suh JK, Hart RT. The optic nerve head as a biomechanical structure: a new paradigm for understanding the role of IOP-related stress and strain in the pathophysiology of glaucomatous optic nerve head damage. *Progress in Retinal and Eye Research* 2005;24:39-73.
- 112. Jonas J, Gusek G, Naumann G. Optic disc morphometry in chronic primary open-angle glaucoma. *Graefe's Archive for Clinical and Experimental Ophthalmology* 1988;226:522-530.
- 113. Quigley HA, Hohman RM, Addicks EM, Massof RW, Green WR. Morphologic changes in the lamina cribrosa correlated with neural loss in open-angle glaucoma. *American Journal of Ophthalmology* 1983;95:673-691.
- 114. Jonas JB. Patterns of glaucomatous neuroretinal rim loss. *Ophthalmology* 1993;100:63.
- 115. Betz P, Camps F, Collignon-Brach J, Lavergne G, Weekers R. Biometric study of the disc cup in open-angle glaucoma. *Graefe's Archive for Clinical and Experimental Ophthalmology* 1982;218:70-74.
- 116. Gloster J. Quantitative relationship between cupping of the optic disc and visual field loss in chronic simple glaucoma. *British Journal of Ophthalmology* 1978;62:665-669.
- 117. Hoyt WF, Frisen L, Newman NM. Fundoscopy of Nerve Fiber Layer Defects in Glaucoma. *Investigative ophthalmology & visual science* 1973;12:814-829.
- 118. NICE-CG85. Glaucoma: Diagnosis and management of chronic open angle glaucoma and ocular hypertension. *NICE Clinical Guidelines 85*: National Institute for Health and Clinical Excellence (NICE); 2009.
- 119. Sigal IA, Ethier CR. Biomechanics of the optic nerve head. *Experimental Eye Research* 2009;88:799-807.
- 120. Pointer JS. The diurnal variation of intraocular pressure in non-glaucomatous subjects: relevance in a clinical context. *Ophthalmic & physiological optics* 1997;17:456-465.
- 121. Liu JH, Kripke DF, Twa MD, et al. Twenty-four-hour pattern of intraocular pressure in the aging population. *Investigative ophthalmology & visual science* 1999;40:2912-2917.
- 122. Liu JHK, Kripke DF, Twa MD, et al. Twenty-four-hour pattern of intraocular pressure in young adults with moderate to severe myopia. *Invest Ophthalmol Vis Sci* 2002;43:2351-2355.
- 123. Liu JHK, Gokhale PA, Loving RT, Kripke DF, Weinreb RN. Laboratory assessment of diurnal and nocturnal ocular perfusion pressures in humans. *Journal of ocular pharmacology and therapeutics* 2003;19:291-297.

- 124. Liu JHK, Kripke DF, Hoffman RE, et al. Nocturnal elevation of intraocular pressure in young adults. *Investigative ophthalmology & visual science* 1998;39:2707-2712.
- 125. Brown B, Burton P, Mann S, Parisi A. Fluctuations in intra-ocular pressure with sleep: II. Time course of IOP decrease after waking from sleep. *Ophthalmic & physiological optics* 1988;8:249-252.
- 126. Frampton P, Da Rin D, Brown B. Diurnal variation of intraocular pressure and the overriding effects of sleep. *Am J Optom Physiol Opt* 1987;64:54-61.
- 127. Liu JHK, Bouligny RP, Kripke DF, Weinreb RN. Nocturnal elevation of intraocular pressure is detectable in the sitting position. *Investigative ophthalmology & visual science* 2003;44:4439-4442.
- 128. Wang NL, Friedman DS, Zhou Q, et al. A Population-Based Assessment of 24-Hour Intraocular Pressure among Subjects with Primary Open-Angle Glaucoma: The Handan Eye Study. *Investigative ophthalmology & visual science* 2011;52:7817-7821.
- 129. Asrani S, Zeimer R, Wilensky J, Gieser D, Vitale S, Lindenmuth K. Large diurnal fluctuations in intraocular pressure are an independent risk factor in patients with glaucoma. *Journal of Glaucoma* 2000;9:134-142.
- 130. Liu JHK, Zhang X, Kripke D, Weinreb R. Twenty-four-hour intraocular pressure pattern associated with early glaucomatous changes. *Investigative ophthalmology & visual science* 2003;44:1586-1590.
- 131. Hughes E, Spry P, Diamond J. 24-hour monitoring of intraocular pressure in glaucoma management: a retrospective review. *Journal of Glaucoma* 2003;12:232-236.
- 132. Bellezza AJ, Hart RT, Burgoyne CF. The Optic Nerve Head as a Biomechanical Structure: Initial Finite Element Modeling. *Investigative ophthalmology & visual science* 2000;41:2991-3000.
- 133. Yan DB, Coloma FM, Metheetrairut A, Trope GE, Heathcote JG, Ethier CR. Deformation of the lamina cribrosa by elevated intraocular pressure. *British Journal of Ophthalmology* 1994;78:643-648.
- 134. Yang H, Thompson H, Roberts MD, Sigal IA, Downs JC, Burgoyne CF. Deformation of the Early Glaucomatous Monkey Optic Nerve Head Connective Tissue after Acute IOP Elevation in 3-D Histomorphometric Reconstructions. *Investigative ophthalmology & visual science* 2011;52:345-363.
- 135. Cartwright MJ, Anderson DR. Correlation of Asymmetric Damage With Asymmetric Intraocular Pressure in Normal-Tension Glaucoma (Low-Tension Glaucoma). *Arch Ophthalmol* 1988;106:898-900.
- 136. Magitot A. The circulation in glaucoma. *Ann d'Ocul* 1925;162:729-763.
- 137. Prunte C, Orgul S, Flammer J. Abnormalities of microcirculation in glaucoma: facts and hints. *Curr Opin Ophthalmol* 1998;9:50-55.
- 138. Flammer J. Psychophysics in glaucoma. A modified concept of the disease. In: Greve E, Leydhecker W, Raitta C (eds), *Proceedings of the European Glaucoma Society.* Dondrecht: Dr W Junk Publishers; 1985:11-17.
- 139. Anderson DR. Introductory comments on blood flow autoregulation in the optic nerve head and vascular risk factors in glaucoma. *Surv Ophthalmol* 1999;43:S5-S9.
- 140. Osborne NN, Melena J, Chidlow G, Wood JPM. A hypothesis to explain ganglion cell death caused by vascular insults at the optic nerve head: possible implication for the treatment of glaucoma. *British Journal of Ophthalmology* 2001;85:1252-1259.
- 141. Flammer J, Mozaffarieh M. What Is the Present Pathogenetic Concept of Glaucomatous Optic Neuropathy? *Survey of Ophthalmology* 2007;52:S162-S173.
- 142. Hayreh SS. The Role of Age and Cardiovascular Disease in Glaucomatous Optic Neuropathy. *Survey of Ophthalmology* 1999;43:S27-S42.

- 143. Ostadal P. What is reperfusion injury? *European Heart Journal* 2005;26:99.
- 144. Prasanna G, Krishnamoorthy R, Yorio T. Endothelin, astrocytes and glaucoma. *Experimental Eye Research* 2010;In Press, Corrected Proof.
- 145. Sherwood L. *Human Physiology: from cells to systems*. Belmont: Yolanda Cossio; 2006.
- 146. Strauss AL, Kedra AW. EXPERIENCES WITH A NEW PROCEDURE FOR THE MEASUREMENT OF THE OPHTHALMIC ARTERY PRESSURE: OPHTHALMOMANOMETRY-DOPPLER. *Medical Instrumentation* 1987;21:255-261.
- 147. Henkind P. Retinal arteriolar annuli. *Investigative ophthalmology & visual science* 1968;7:584.
- 148. Luxenber.Mn, Mausolf FA. RETINAL CIRCULATION IN HYPERVISCOSITY SYNDROME. *American Journal of Ophthalmology* 1970;70:588.
- 149. Perry MC, Hoagland HC. The Hyperviscosity Syndrome. *JAMA: The Journal of the American Medical* 393. *Association* 1976;236:392-
- 150. Henrion D. Pressure and flow-dependent tone in resistance arteries. *Archives des Maladies du Coeur et des Vajsseaux* 2005;98:913-921.
- 151. Johnson PC. Autoregulation of blood flow. *Circulation research* 1986;59:483.
- 152. Weigert G, Findl O, Luksch A, et al. Effects of Moderate Changes in Intraocular Pressure on Ocular Hemodynamics in Patients with Primary Open-Angle Glaucoma and Healthy Controls. *Ophthalmology* 2005;112:1337-1342.
- 153. Pournaras C, Rungger-Brändle E, Riva C, Hardarson S, Stefansson E. Regulation of retinal blood flow in health and disease. *Progress in Retinal and Eye Research* 2008;27:284.
- 154. Harris A, Anderson DR, Pillunat L, et al. Laser Doppler Flowmetry Measurement of Changes in Human Optic Nerve Head Blood Flow in Response to Blood Gas Perturbations. *Journal of glaucoma* 1996;5:258-265.
- 155. Luksch A, Garhöfer G, Imhof A, et al. Effect of inhalation of different mixtures of O2 and CO2 on retinal blood flow. *British Journal of Ophthalmology* 2002;86:1143-1147.
- 156. Orgul S, Gugleta K, Flammer J. Physiology of Perfusion as It Relates to the Optic Nerve Head. *Survey of Ophthalmology* 1999;43:S17-S26.
- 157. Polska E, Ehrlich P, Luksch A, FuchsjĤger-Mayrl G, Schmetterer L. Effects of Adenosine on Intraocular Pressure, Optic Nerve Head Blood Flow, and Choroidal Blood Flow in Healthy Humans. *Investigative ophthalmology & visual science* 2003;44:3110-3114.
- 158. Sullivan SM, Johnson PC. Effect of oxygen on blood flow autoregulation in cat sartorius muscle. *American Journal of Physiology Heart and Circulatory Physiology* 1981;241:H807-H815.
- 159. Kontos HA, Wei EP, Raper AJ, Rosenblum WI, Navari RM, Patterson JL. Role of tissue hypoxia in local regulation of cerebral microcirculation. *American Journal of Physiology Heart and Circulatory Physiology* 1978;234:H582-H591.
- 160. Gilmore ED, Hudson C, Preiss D, Fisher J. Retinal arteriolar diameter, blood velocity, and blood flow response to an isocapnic hyperoxic provocation. *American Journal of Physiology - Heart and Circulatory Physiology* 2005;288:H2912-H2917.
- 161. Eperon G, Johnson M, David NJ. The effect of arterial PO2 on relative retinal blood flow in monkeys. *Investigative ophthalmology & visual science* 1975;14:342-352.
- 162. Takagi C, King GL, Takagi H, Lin YW, Clermont AC, Bursell SE. Endothelin-1 action via endothelin receptors is a primary mechanism modulating retinal circulatory response to hyperoxia. *Investigative ophthalmology & visual science* 1996;37:2099-2109.

- 163. Busse R, Förstermann U, Matsuda H, Pohl U. The role of prostaglandins in the endothelium-mediated vasodilatory response to hypoxia. *Pflügers Archiv European Journal of Physiology* 1984;401:77-83.
- 164. Busse R, Pohl U, Kellner C, Klemm U. Endothelial cells are involved in the vasodilatory response to hypoxia. *Pflügers Archiv European Journal of Physiology* 1983;397:78-80.
- 165. Roth S, Park SS, Sikorski CW, Osinski J, Chan R, Loomis K. Concentrations of adenosine and its metabolites in the rat retina/choroid during reperfusion after ischemia. *Current Eye Research* 1997;16:875-885.
- 166. Crosson CE, DeBenedetto R, Gidday JM. Functional Evidence for Retinal Adenosine Receptors. *Journal of Ocular Pharmacology and Therapeutics* 1994;10:499-507.
- 167. Geiser MH. Response of choroidal blood flow in the foveal region to hyperoxia and hyperoxia-hypercapnia. *Current Eye Research* 2000;21:669.
- 168. Kergoat H, Marinier J-A, Lovasik JV. Effects of Transient Mild Systemic Hypoxia on the Pulsatile Choroidal Blood Flow in Healthy Young Human Adults. *Current Eye Research* 2005;30:465-470.
- 169. Dorner GT, Garhoefer G, Zawinka C, Kiss B, Schmetterer L. Response of retinal blood flow to CO2-breathing in humans. *European journal of ophthalmology* 2002;12:459-466.
- 170. Venkataraman ST, Hudson C, Fisher JA, Rodrigues L, Mardimae A, Flanagan JG. Retinal arteriolar and capillary vascular reactivity in response to isoxic hypercapnia. *Experimental Eye Research* 2008;87:535-542.
- 171. Petropoulos IK, Munoz J-L, Pournaras CJ. Metabolic Regulation of the Hypercapnia-Associated Vasodilation of the Optic Nerve Head Vessels. *Invest Ophthalmol Vis Sci* 2005;46:3908-.
- 172. Bayliss WM. On the local reactions of the arterial wall to changes of internal pressure. *The Journal of Physiology* 1902;28:220-231.
- 173. Schmidl D, Garhofer G, Schmetterer L. The complex interaction between ocular perfusion pressure and ocular blood flow Relevance for glaucoma. *Experimental Eye Research* 2010;doi: 10.1016/j.exer.2010.09.002:1-15.
- 174. Schubert R, Mulvany MJ. The myogenic response: established facts and attractive hypotheses. *Clin Sci (Lond)* 1999;96:313-326.
- 175. Harder D. Pressure-induced myogenic activation of cat cerebral arteries is dependent on intact endothelium. *Circ Res* 1987;60:102-107.
- 176. Weinstein JM. Regional optic nerve blood flow and its autoregulation. Investigative ophthalmology & visual science 1983;24:1559.
- 177. Alm A. The effect of sympathetic stimulation on blood flow through the uvea, retina and optic nerve in monkeys (Macaca irus). *Experimental Eye Research* 1977;25:19-24.
- 178. Ehinger B. Adrenergic Nerves to the Eye and to Related Structures in Man and in the Cynomolgus Monkey (Macaca Irus). *Investigative ophthalmology & visual science* 1966;5:42-52.
- 179. Laties AM. Central Retinal Artery Innervation: Absence of Adrenergic Innervation to the Intraocular Branches. *Arch Ophthalmol* 1967;77:405-409.
- 180. Ye XD, Laties AM, Stone RA. Peptidergic innervation of the retinal vasculature and optic nerve head. *Investigative ophthalmology & visual science* 1990;31:1731-1737.
- 181. Lütjen-Drecoll E. Choroidal innervation in primate eyes. *Experimental Eye Research* 2006;82:357-361.
- 182. Bill A. Autonomic Nervous Control of Uveal Blood Flow. *Acta Physiologica Scandinavica* 1962;56:70-81.
- 183. Ferrari-Dileo G. Beta 1 and beta 2 adrenergic binding sites in bovine retina and retinal blood vessels. *Investigative ophthalmology & visual science* 1988;29:695-699.

- 184. Forster BA, Ferrari-Dileo G, Anderson DR. Adrenergic alpha 1 and alpha 2 binding sites are present in bovine retinal blood vessels. *Investigative ophthalmology & visual science* 1987;28:1741-1746.
- 185. Bill A, Linder J. Sympathetic control of cerebral blood flow in acute arterial hypertension. *Acta Physiologica Scandinavica* 1976;96:114.
- 186. Lovasik JV, Kergoat H, Riva CE, Petrig BL, Geiser M. Choroidal Blood Flow during Exercise-Induced Changes in the Ocular Perfusion Pressure. *Investigative ophthalmology & visual science* 2003;44:2126-2132.
- 187. Riva CE, Cranstoun SD, Mann RM, Barnes GE. Local choroidal blood flow in the cat by laser Doppler flowmetry. *Investigative ophthalmology & visual science* 1994;35:608-618.
- 188. Nilsson SFE, Bill A. Vasoactive intestinal polypeptide (VIP): Effects in the eye and on regional blood flows. *Acta Physiologica Scandinavica* 1984;121:385-392.
- 189. Stjernschantz J, Geijer C, Bill A. Electrical stimulation of the fifth cranial nerve in rabbits: Effects on ocular blood flow, extravascular albumin content and intraocular pressure. *Experimental Eye Research* 1979;28:229-238.
- 190. Wiencke AK, Nilsson H, Nielsen PJ, Nyborg NC. Nonadrenergic noncholinergic vasodilation in bovine ciliary artery involves CGRP and neurogenic nitric oxide. *Investigative ophthalmology & visual science* 1994;35:3268-3277.
- 191. Nilsson SFE, Linder J, Bill A. Characteristics of uveal vasodilation produced by facial nerve stimulation in monkeys, cats and rabbits. *Experimental Eye Research* 1985;40:841-852.
- 192. Kurihara H, Yazaki Y. Regulation of vascular tone. In: Haber E (ed), *Molecular Cardiovascular Medicine*. New York: Scientific American; 1995:275-288.
- 193. Ferrari-Dileo G, Davis EB, Anderson DR. Angiotensin binding sites in bovine and human retinal blood vessels. *Investigative ophthalmology & visual science* 1987;28:1747-1751.
- 194. Sossi N, Anderson DR. Blockage of Axonal Transport in Optic Nerve Induced by Elevation of Intraocular Pressure: Effect of Arterial Hypertension Induced by Angiotensin I. *Arch Ophthalmol* 1983;101:94-97.
- 195. Krejcy K. Characterization of angiotensin-II effects on cerebral and ocular circulation by noninvasive methods. *British journal of clinical pharmacology* 1997;43:501.
- 196. Morgan TR, Mirate DJ, Bowman K, Green K. Topical Epinephrine and Regional Ocular Blood Flow in Aphakic Eyes of Rabbits. *Arch Ophthalmol* 1983;101:112-116.
- 197. Malik AB, van Heuven WA, Satler LF. Effects of isoproterenol and norepinephrine on regional ocular blood flows. *Investigative ophthalmology & visual science* 1976;15:492-495.
- 198. Resch H, Garhofer G, Fuchsjager-Mayrl G, Hommer A, Schmetterer L. Endothelial dysfunction in glaucoma. *Acta Ophthalmologica* 2009;87:4-12.
- 199. Deanfield JE, Halcox JP, Rabelink TJ. Endothelial function and dysfunction -Testing and clinical relevance. *Circulation* 2007;115:1285-1295.
- 200. Burton AC. Relation of structure to function of the tissues of the wall of blood vessels. *Physiological Reviews* 1954;34:619.
- 201. Endemann DH, Schiffrin EL. Endothelial Dysfunction. *Journal of the American Society of Nephrology* 2004;15:1983-1992.
- 202. Furchgott RF, Zawadzki JV. The obligatory role of endothelial cells in the relaxation of arterial smooth muscle by acetylcholine. *Nature* 1980;288:373-376.
- 203. Furchgott RF. Role of endothelium in responses of vascular smooth muscle. *Circulation research* 1983;53:557.
- 204. Vane JR, Änggård EE, Botting RM. Regulatory Functions of the Vascular Endothelium. *New England Journal of Medicine* 1990;323:27-36.
- 205. Bassenge E. Endothelial modulation of coronary tone*1. *Progress in cardiovascular diseases* 1988;30:349-380.

- 206. Landmesser U, Hornig B, Drexler H. Endothelial Function: A Critical Determinant in Atherosclerosis? *Circulation* 2004;109:II-27-33.
- 207. Haefliger IO, Flammer J, Lüscher TF. Nitric oxide and endothelin-1 are important regulators of human ophthalmic artery. *Investigative ophthalmology & visual science* 1992;33:2340-2343.
- 208. Palmer RMJ, Ferrige AG, Moncada S. Nitric oxide release accounts for the biological activity of endothelium-derived relaxing factor. *Nature* 1987;327:524-526.
- 209. Luscher T, Yang Z, Tschudi M, et al. Interaction between endothelin-1 and endothelium-derived relaxing factor in human arteries and veins. *Circ Res* 1990;66:1088-1094.
- 210. Moncada S, Palmer RM, Higgs EA. Nitric oxide: physiology, pathophysiology, and pharmacology. *Pharmacological Reviews* 1991;43:109-142.
- 211. Bredt DS, Snyder SH. Nitric Oxide: A Physiologic Messenger Molecule. *Annual Review of Biochemistry* 1994;63:175-195.
- 212. Liversidge J, Grabowski P, Ralston S, Benjamin N, Forrester JV. Rat retinal pigment epithelial cells express an inducible form of nitric oxide synthase and produce nitric oxide in response to inflammatory cytokines and activated T cells. 1993.
- 213. Osborne NN, Barnett NL, Herrera AJ. NADPH diaphorase localization and nitric oxide synthetase activity in the retina and anterior uvea of the rabbit eye. *Brain Research* 1993;610:194-198.
- 214. Yamamoto R, Bredt DS, Snyder SH, Stone RA. The localization of nitric oxide synthase in the rat eye and related cranial ganglia. *Neuroscience* 1993;54:189-200.
- 215. Chakravarthy U, Stitt AW, Mcnally J, Bailie JR, Hoey EM, Duprex P. Nitric oxide synthase activity and expression in retinal capillary endothelial cells and pericytes. *Current Eye Research* 1995;14:285-294.
- 216. Yao K, Tschudi M, Flammer J, Lüscher TF. Endothelium-dependent regulation of vascular tone of the porcine ophthalmic artery. *Investigative ophthalmology & visual science* 1991;32:1791-1798.
- 217. Rubanyi GM, Romero JC, Vanhoutte PM. Flow-induced release of endotheliumderived relaxing factor. *American Journal of Physiology - Heart and Circulatory Physiology* 1986;250:H1145-H1149.
- 218. Cocks TM, Angus JA. Endothelium-dependent relaxation of coronary arteries by noradrenaline and serotonin. *Nature* 1983;305:627-630.
- 219. Katusic Z, Shepherd J, Vanhoutte P. Vasopressin causes endotheliumdependent relaxations of the canine basilar artery. *Circ Res* 1984;55:575-579.
- 220. Haefliger IO, Zschauer A, Anderson DR. Relaxation of retinal pericyte contractile tone through the nitric oxide-cyclic guanosine monophosphate pathway. *Investigative ophthalmology & visual science* 1994;35:991-997.
- 221. Haefliger IO. Endothelium-dependent vasoactive modulation in the ophthalmic circulation. *Progress in Retinal and Eye Research* 2001;20:209.
- 222. Toda N, Nakanishi-Toda M. Nitric oxide: Ocular blood flow, glaucoma, and diabetic retinopathy. *Progress in Retinal and Eye Research* 2007;26:205-238.
- 223. Mann RM, Riva CE, Stone RA, Barnes GE, Cranstoun SD. Nitric oxide and choroidal blood flow regulation. *Investigative ophthalmology & visual science* 1995;36:925-930.
- 224. Becquet F, Courtois Y, Goureau O. Nitric oxide in the eye: Multifaceted roles and diverse outcomes. *Survey of Ophthalmology* 1997;42:71-82.
- 225. Huemer K-H, Garhofer G, Aggermann T, Kolodjaschna J, Schmetterer L, FuchsjĤger-Mayrl G. Role of Nitric Oxide in Choroidal Blood Flow Regulation during Light/Dark Transitions. *Investigative ophthalmology & visual science* 2007;48:4215-4219.

- 226. Dorner GT, Garhofer G, Kiss B, et al. Nitric oxide regulates retinal vascular tone in humans. *Am J Physiol Heart Circ Physiol* 2003;285:H631-636.
- 227. Luksch A, Polak K, Beier C, et al. Effects of Systemic NO Synthase Inhibition on Choroidal and Optic Nerve Head Blood Flow in Healthy Subjects. *Investigative ophthalmology & visual science* 2000;41:3080-3084.
- 228. Schmetterer L, Findl O, Fasching P, et al. Nitric oxide and ocular blood flow in patients with IDDM. *Diabetes* 1997;46:653-658.
- 229. Yanagisawa M, Kurihara H, Kimura S, et al. A novel potent vasoconstrictor peptide produced by vascular endothelial cells. *Nature* 1988;332:411-415.
- 230. Levin E. Endothelins. *N Engl J Med* 1995;333:356-363.
- 231. Siekmeier R, Grammer T, März W. Roles of oxidants, nitric oxide, and asymmetric dimethylarginine in endothelial function. *Journal of cardiovascular pharmacology and therapeutics* 2008;13:279-297.
- 232. Wagner OF, Christ G, Wojta J, et al. Polar secretion of endothelin-1 by cultured endothelial cells. *Journal of Biological Chemistry* 1992;267:16066-16068.
- 233. Grieshaber MC, Mozaffarieh M, Flammer J. What is the link between vascular dysregulation and glaucoma. *Surv Ophthalmol* 2007;52:S144-S154.
- 234. Arai H, Hori S, Aramori I, Ohkubo H, Nakanishi S. Cloning and expression of a cDNA encoding an endothelin receptor. *Nature* 1990;348:730-732.
- 235. Pollock D, Keith T, Highsmith R. Endothelin receptors and calcium signaling. *The FASEB Journal* 1995;9:1196-1204.
- 236. Haynes WG, Strachan FE, Webb DJ. Endothelin ETA and ETB Receptors Cause Vasoconstriction of Human Resistance and Capacitance Vessels In Vivo. *Circulation* 1995;92:357-363.
- 237. Kiowski W, Luscher T, Linder L, Buhler F. Endothelin-1 induced vasoconstriction in humans. Reversal by calcium channel blockade but not by nitrovasodilators or endothelium derived relaxing factors. *Circulation* 1991;83:469-475.
- 238. Lippton HL, Hauth TA, Summer WR, Hyman AL. Endothelin produces pulmonary vasoconstriction and systemic vasodilation. *Journal of Applied Physiology* 1989;66:1008-1012.
- 239. Verhaar MC, Strachan FE, Newby DE, et al. Endothelin-A Receptor Antagonist– Mediated Vasodilatation Is Attenuated by Inhibition of Nitric Oxide Synthesis and by Endothelin-B Receptor Blockade. *Circulation* 1998;97:752-756.
- 240. Abraham D, Distler O. How does endothelial cell injury start? The role of endothelin in systemic sclerosis. *Arthritis Research & Therapy* 2007;9:S2.
- 241. Haynes WG, Webb DJ. Contribution of endogenous generation of endothelin-1 to basal vascular tone. *The Lancet* 1994;344:852-854.
- 242. Stokely ME, Brady ST, Yorio T. Effects of Endothelin-1 on Components of Anterograde Axonal Transport in Optic Nerve. *Investigative ophthalmology & visual science* 2002;43:3223-3230.
- 243. Miyauchi T, Masaki T. PATHOPHYSIOLOGY OF ENDOTHELIN IN THE CARDIOVASCULAR SYSTEM. *Annual Review of Physiology* 1999;61:391-415.
- 244. Haynes W, Webb DJ. Endothelin as a regulator of cardiovascular function in health and disease. *J Hypertens* 1998;16:1081-1098.
- 245. Shah R. Endothelins in health and disease. *European journal of internal medicine* 2007;18:272-282.
- 246. MacCumber MW, D'Anna SA. Endothelin Receptor-Binding Subtypes in the Human Retina and Choroid. *Arch Ophthalmol* 1994;112:1231-1235.
- 247. Ripodas A, De Juan JA, Moya FJ, Fernandez-Cruz A, Fernandez-Durango R. Identification of Endothelin Receptor Subtypes in Rat Ciliary Body Using Subtype-Selective Ligands. *Experimental Eye Research* 1998;66:69-79.
- 248. Ripodas A, de Juan JA, Roldán-Pallarés M, et al. Localisation of endothelin-1 mRNA expression and immunoreactivity in the retina and optic nerve from human and porcine eye. Evidence for endothelin-1 expression in astrocytes. *Brain Research* 2001;912:137-143.

- 249. Strenn K, Matulla B, Wolzt M, et al. Reversal of endothelin-1 induced ocular hemodynamic effects by low-dose nifedipine in humans. *Clin Pharmacol Ther* 1998;63:54-63.
- 250. Polak K, Petternel V, Luksch A, et al. Effect of Endothelin and BQ123 on Ocular Blood Flow Parameters in Healthy Subjects. *Investigative ophthalmology & visual science* 2001;42:2949-2956.
- 251. Schmetterer L, Findl O, Strenn K, et al. Effects of endothelin-1 (ET-1) on ocular hemodynamics. *Curr Eye Res* 1997;16:687-692.
- 252. Polak K, Luksch A, Frank B, Jandrasits K, Polska E, Schmetterer L. Regulation of human retinal blood flow by endothelin-1. *Experimental Eye Research* 2003;76:633-640.
- 253. Fuchsjager-Mayrl G, Luksch A, Malec M, Polska E, Wolzt M, Schmetterer L. Role of Endothelin-1 in Choroidal Blood Flow Regulation during Isometric Exercise in Healthy Humans. *Investigative ophthalmology & visual science* 2003;44:728-733.
- 254. Chauhan BC. Endothelin and its potential role in glaucoma. *Can J Ophthalmol* 2008;43:356-360.
- 255. Feletou M, Vanhoutte PM. Endothelial dysfunction: a multifaceted disorder (The Wiggers Award Lecture). *American Journal of Physiology Heart and Circulatory Physiology* 2006;291:H985-H1002.
- 256. Vita JA, Keaney JF. Endothelial function A barometer for cardiovascular risk? *Circulation* 2002;106:640-642.
- 257. Griendling KK, FitzGerald GA. Oxidative Stress and Cardiovascular Injury: Part I: Basic Mechanisms and In Vivo Monitoring of ROS. *Circulation* 2003;108:1912-1916.
- 258. Vapaatola H, Mervaala E. Clinically important factors influencing endothelial function. *Medical science monitor international medical journal of experimental and clinical research* 2001;7:1075-1085.
- 259. Ehrlich R, Kheradiya N, Winston D, Moore D, Wirostko B, Harris A. Age-related ocular vascular changes. *Graefe's Archive for Clinical and Experimental Ophthalmology* 2009;247:583-591.
- 260. Taddei S, Virdis A, Mattei P, Ghiadoni L, Sudano I, Salvetti A. Defective L-Arginine–Nitric Oxide Pathway in Offspring of Essential Hypertensive Patients. *Circulation* 1996;94:1298-1303.
- 261. Celermajer DS, Sorensen KE, Gooch VM, et al. Non-invasive detection of endothelial dysfunction in children and adults at risk of atherosclerosis. *The Lancet* 1992;340:1111-1115.
- 262. Halcox JPJ, Schenke WH, Zalos G, et al. Prognostic Value of Coronary Vascular Endothelial Dysfunction. *Circulation* 2002;106:653-658.
- 263. Schachinger V, Britten MB, Zeiher AM. Prognostic Impact of Coronary Vasodilator Dysfunction on Adverse Long-Term Outcome of Coronary Heart Disease. *Circulation* 2000;101:1899-1906.
- 264. Griendling KK, FitzGerald GA. Oxidative Stress and Cardiovascular Injury: Part II: Animal and Human Studies. *Circulation* 2003;108:2034-2040.
- 265. Cai H, Harrison DG. Endothelial Dysfunction in Cardiovascular Diseases: The Role of Oxidant Stress. *Circ Res* 2000;87:840-844.
- 266. Jin N, Packer C, Rhoades R. Reactive oxygen-mediated contraction in pulmonary arterial smooth muscle: cellular mechanisms. *Canadian Journal of Physiology and Pharmacology* 1991;69:383-388.
- 267. Duvall W. Endothelial dysfunction and antioxidants *Mt Sinai J Med* 2005;72:71-80.
- 268. Brown AA, Hu FB. Dietary modulation of endothelial function: implications for cardiovascular disease. *The American Journal of Clinical Nutrition* 2001;73:673-686.

- 269. Kelm M. Flow-mediated dilatation in human circulation: diagnostic and therapeutic aspects. *American Journal of Physiology Heart and Circulatory Physiology* 2002;282:H1-H5.
- 270. Constans J, Conri C. Circulating markers of endothelial function in cardiovascular disease. *Clinica Chimica Acta* 2006;368:33-47.
- 271. Ruggeri ZM. Structure of von Willebrand factor and its function in platelet adhesion and thrombus formation. *Best practice & research Clinical haematology* 2001;14:257-279.
- 272. Blann AD, McCollum CN. von Willebrand factor, endothelial cell damage and atherosclerosis. *European journal of vascular surgery* 1994;8:10-15.
- 273. Lip GYH, Blann A. von Willebrand factor: a marker of endothelial dysfunction in vascular disorders? *Cardiovascular research* 1997;34:255-265.
- 274. Stehouwer CA, Zeldenrust GC, den Ottolander GH, Hackeng WHL, Donker AJM, Nauta JJP. Urinary albumin excretion, cardiovascular disease, and endothelial dysfunction in non-insulin-dependent diabetes mellitus. *The Lancet* 1992;340:319-323.
- 275. Blann A, Nagvi T, Waite M, McCollum CN. von Willebrand factor and endothelial damage in essential hypertension. *J Hum Hypertens* 1993;7:107-111.
- 276. Liu L, Lin Z, Shen Z. Changes of von Willebrand factor and antithrombin III levels in acute stroke: Difference between thrombotic and haemorrhagic stroke. *Thrombosis Research* 1993;72:353-358.
- 277. Jansson JH, Nilsson TK, Johnson O. von Willebrand factor in plasma: a novel risk factor for recurrent myocardial infarction and death. *British heart journal* 1991;66:351-355.
- 278. Chong AY, Blann AD, Patel J, Freestone B, Hughes E, Lip GYH. Endothelial Dysfunction and Damage in Congestive Heart Failure. *Circulation* 2004;110:1794-1798.
- 279. Rajagopalan S, Somers EC, Brook RD, et al. Endothelial cell apoptosis in systemic lupus erythematosus: a common pathway for abnormal vascular function and thrombosis propensity. *Blood* 2004;103:3677-3683.
- 280. Mannucci PM. von Willebrand Factor : A Marker of Endothelial Damage? Arteriosclerosis, Thrombosis, and Vascular Biology 1998;18:1359-1362.
- 281. Pottinger BE, Read RC, Paleolog EM, Higgins PG, Pearson JD. von Willebrand factor is an acute phase reactant in man. *Thrombosis Research* 1989;53:387-394.
- 282. Blann A. von Willebrand factor antigen as an acute phase reactant and marker of endothelial cell injury in connective tissue diseases: a comparison with CRP, rheumatoid factor, and erythrocyte sedimentation rate. *Z Rheumatol* 1991;50:320-322.
- 283. Vanhoutte PM. How to assess endothelial function in human blood vessels. *Journal of Hypertension* 1999;17:1047-1058.
- 284. Raitakari OT, Celermajer DS. Testing for endothelial dysfunction. *Annals of Medicine* 2000;32:293-304.
- 285. Michelson G, MJ L, MJ G. Perfusion of the juxtapapillary retina and the neuroretinal rim area in primary open angle glaucoma. *J Glaucoma* 1996;5:91-98.
- 286. Piltz-Seymour J, Grunwald J, Hariprasad S, Dupont J. Optic nerve blood flow is diminished in eyes of primary open angle glaucoma suspects. *Am J Ophthalmol* 2001;132:63-69.
- 287. Grunwald J, J P, Hariprasad S, Dupont J. Optic nerve and choroidal circulation in glaucoma. *Invest Ophthalmol Vis Sci* 1998;39:2329-2336.
- 288. Findl O, Rainer G, Dallinger S, et al. Assessment of optic disk blood flow in patients with open-angle glaucoma. *American Journal of Ophthalmology* 2000;130:589-596.

- 289. Sato E, Ohtake Y, Shinoda K, Mashima Y, Kimura I. Decreased blood flow at neuroretinal rim of optic nerve head corresponds with visual field deficit in eyes with normal tension glaucoma. *Graefe's Archive for Clinical and Experimental Ophthalmology* 2006;244:795-801.
- 290. Mitchell P, Leung H, Wang J, et al. Retinal vessel diameter and open-angle glaucoma: the Blue Mountains Eye Study *Ophthalmology* 2005;112:245-250.
- 291. Rader J, Feuer W, Anderson D. Peripapillary vasoconstriction in the glaucomas and the anterior ischemic optic neuropathies. *Am J Ophthalmol* 1994;117:72-80.
- 292. Arend O, Remky A, Plange N, Martin B, Harris A. Capillary density and retinal diameter measurements and their impact on altered retinal circulation in glaucoma: a digital fluorescein angiographic study. *Br J Ophthalmol* 2002;86:429-433.
- 293. Wolf S, Arend O, W.E. S, Schulte K, Cantor LB, Reim M. Retinal hemodynamics using scanning laser ophthalmoscopy and hemorheology in chronic open angle glaucoma. *Ophthalmology* 1993;100:1561-1566.
- 294. Duijm H, van den Berg T, Greve E. A comparison of retinal and choroidal hemodynamics in patients with primary open angle glaucoma and normal-pressure glaucoma *Am J Ophthalmol* 1997;123:644-656.
- 295. Oettli A, Gugleta K, Kochkorov A, Katamay R, Flammer J, Orgul S. Rigidity of Retinal Vessel in Untreated Eyes of Normal Tension Primary Open-angle Glaucoma Patients. *Journal of glaucoma* 2010;20:303-306.
- 296. Cellini M, Possati G, Caramazza N, Caramazza R. Colour Doppler analysis of the choroidal circulation in chronic open angle glaucoma. *Ophthalmologica* 1996;210:200-202.
- 297. Fuchsjager-Mayrl G, Wally B, Georgopoulos M, et al. Ocular Blood Flow and Systemic Blood Pressure in Patients with Primary Open-Angle Glaucoma and Ocular Hypertension. *Invest Ophthalmol Vis Sci* 2004;45:834-839.
- 298. Kerr J, Nelson P, O'Brien C. A comparison of ocular blood flow in untreated primary open-angle glaucoma and ocular hypertension. *Am J Ophthalmol* 1998;126:42-51.
- 299. Plange N, Remky A, Arend O. Colour Doppler imaging and fluorescein filling defects of the optic disc in normal tension glaucoma. *Br J Ophthalmol* 2003;87:731-736.
- 300. Butt Z, McKillop G, O'Brien C, Allan P, Aspinal P. Measurement of ocular blood flow velocity using colour Doppler imaging in low tension glaucoma. *Eye* 1995;9:29-33.
- 301. Rankin SJA. Color Doppler imaging of the retrobulbar circulation in glaucoma *Surv Ophthalmol* 1995;43 (suppl 1):S176-182.
- 302. Zeitz O, Galambos P, Wagenfeld L, et al. Glaucoma progression is associated with decreased blood flow velocities in the short posterior ciliary artery. *Br J Ophthalmol* 2006;90:1245-1248.
- 303. Yamazaki Y, Drance S. The relationship between progression of visual field defects and retrobulbar circulation in patients with glaucoma. *Am J Ophthalmol* 1997;124:287-295.
- 304. Galassi F, Sodi A, Ucci F, Renieri G, Pieri B, Baccini M. Ocular hemodynamics and glaucoma prognosis A color Doppler imaging study. *Arch Ophthalmol* 2003;121:1711-1715.
- 305. Lam A, Piltz-Seymour JR, DuPont J, Grunwald JE. Laser Doppler Flowmetry in Asymmetric Glaucoma. *Current Eye Research* 2005;30:221-227.
- 306. Tielsch JM, Katz J, Sommer A, Quigley HA, Javitt JC. Hypertension, Perfusion Pressure, and Primary Open-angle Glaucoma: A Population-Based Assessment. *Arch Ophthalmol* 1995;113:216-221.
- 307. Hulsman C, Vingerling J, Hofman A, Witterman J, De Jong P. Blood Pressure, Arterial Stiffness and Open-angle Glaucoma. *Arch Ophthalmol* 2007;125:805-812.

- 308. Ramdas W, Wolfs R, Hofman A, de Jong P, Vingerling J, Jansonius NM. Ocular perfusion pressure and the incidence of glaucoma: real effect or artifact? The Rotterdam Study. *Invest Ophthalmol Vis Sci* 2011;[Epub ahead of print].
- 309. Nicolela M, Drance S, Rankin SJA, Buckley CH, Walman B. Color Doppler imaging in patients with asymmetric glaucoma and unilateral visual field loss. *Am J Ophthalmol* 1996;121:502-510.
- 310. Susanna R, Basseto FL. Hemorrhage of the optic disc and neurosensorial dysacousia. *J Glaucoma* 1992;1:248-253.
- 311. Waldmann E, Gasser P, Dubler B, Huber C, Flammer J. Silent myocardial ischemia in glaucoma and cataract patients. *Graefe's Archive for Clinical and Experimental Ophthalmology* 1996;234:595-598.
- 312. Ong K, Farinelli A, Billson F, Houang M, Stern M. COMPARATIVE-STUDY OF BRAIN MAGNETIC-RESONANCE-IMAGING FINDINGS IN PATIENTS WITH LOW-TENSION GLAUCOMA AND CONTROL SUBJECTS. *Ophthalmology* 1995;102:1632-1638.
- 313. Stroman G, Stewart W, Golnik K, Cure J, Olinger R. Magnetic resonance imaging in patients with low-tension glaucoma. *Arch Ophthalmol* 1995;133:168-172.
- 314. O'Brien C, Saxton V, Crick RP, Meire H. Doppler carotid artery studies in asymmetric glaucoma. *Eye (London, England)* 1992;6 (Pt 3):273-276.
- 315. O'Brien C, Butt Z. Blood flow velocity in the peripheral circulation in glaucoma patients. *Ophthalmologica* 1999;213:150-153.
- 316. Pache M, Dubler B, Flammer J. Peripheral vasospasm and nocturnal blood pressure dipping--two distinct risk factors for glaucomatous damage? *European journal of ophthalmology* 2003;13:260-265.
- 317. Grieshaber MC, Flammer J. Blood flow in glaucoma. *Curr Opin Ophthalmol* 2005;16:79-83.
- 318. Chiba T, Chiba N, Kashiwagi K. Systemic arterial stiffness in glaucoma patients. *Journal of Glaucoma* 2008;17:15-18.
- 319. Hickman SJ, Brierley CM, Brex PA, et al. Continuing optic nerve atrophy following optic neuritis: a serial MRI study. *Multiple Sclerosis* 2002;8:339-342.
- 320. Evans D, Harris A, Garrett M, Chung H, Kagemann L. Glaucoma patients demonstrate faulty autoregulation of ocular blood flow during posture change. *Br J Ophthalmol* 1999;83:809-813.
- 321. Nagel E, Vilser W, Lanzl I. Retinal vessel reaction to short term IOP elevation in ocular hypertensive and glaucoma patients. *Eur J Ophthalmol* 2001;11:338-344.
- 322. Galambos P, Vafiadis J, Vilchez SE, et al. Compromised Autoregulatory Control of Ocular Hemodynamics in Glaucoma Patients after Postural Change. *Ophthalmology* 2006;113:1832-1836.
- 323. Tutaj M, Brown CM, Brys M, et al. Dynamic cerebral autoregulation is impaired in glaucoma. *Journal of the Neurological Sciences* 2004;220:49-54.
- 324. Emre M, Orgul S, Gugleta K, Flammer J. Ocular blood flow alteration in glaucoma is related to systemic vascular dysregulation. *Br J Ophthalmol* 2004;88:662-666.
- 325. Pournaras C, Riva C, Bresson-Dumont H, De Gottrau P, Bechetoille A. Regulation of optic nerve head blood flow in normal tension glaucoma patients. *Eur J Ophthalmol* 2004;14:226-235.
- 326. Plange N, Kaup M, Remky A, Arend K. Prolonged retinal arteriovenous passage time is correlated to ocular perfusion pressure in normal tension glaucoma. *Graefe's Archive for Clinical and Experimental Ophthalmology* 2008;246:1147-1152.
- 327. Gherghel D, Orgul S, Gugleta K, Gekkieva M, Flammer J. Relationship between ocular perfusion pressure and retrobulbar blood flow in patients with glaucoma with progressive damage. *Am J Ophthalmol* 2000;130:597 605.

- 328. Gugleta K, Orgül S, Hasler P, Picornell T, Gherghel D, Flammer J. Choroidal Vascular Reaction to Hand-Grip Stress in Subjects with Vasospasm and its Relevance in Glaucoma. *Invest Ophthalmol Vis Sci* 2003;44:1573-1580.
- 329. Riva CE, Grunwald JE, Petrig BL. AUTOREGULATION OF HUMAN RETINAL BLOOD-FLOW - AN INVESTIGATION WITH LASER DOPPLER VELOCIMETRY. *Investigative ophthalmology & visual science* 1986;27:1706-1712.
- 330. Anderson DR. Glaucoma, capillaries and pericytes. 1. Blood flow regulation. *Ophtalmologica* 1996;210:257-262.
- 331. Delaney Y, Walshe T, O'Brien C. Vasospasm in glaucoma: clinical and laboratory aspects. *Optom Vis Sci* 2006;83:406-414.
- 332. Flammer J, Mozaffarieh M. Autoregulation, a balancing act between supply and demand. *Can J Ophthalmol* 2008;43:317-321.
- 333. Flammer J, Pache M, Resink T. Vasospasm, its role in the pathogenesis of diseases with particular reference to the eye. *Progress in Retinal and Eye Research* 2001;20:319-349.
- 334. Gherghel D, Orgül S, Dubler B, Lubeck P, Gugleta K, Flammer J. Is vascular regulation in the central retinal artery altered in persons with vasospasm? *Arch Ophthalmol* 1999;117:1359-1362.
- 335. Prunte-Glowazki A, Flammer J. Ocular Vasospasm 4: Clinical Examples. *Klin Monatsbl Augenheilkd* 1991;198:415-418.
- 336. Beltrame JF, Sasayama S, Maseri A. Racial heterogeneity in coronary artery vasomotor reactivity: differences between Japanese and caucasian patients. *Journal of the American College of Cardiology* 1999;33:1442-1452.
- 337. Mocco J, Ransom E, Komotar R, et al. Racial differences in cerebral vasospasm: a systematic review of the literature. *Neurosurgery* 2006;58:305-314.
- 338. Pache M, Kräuchi K, Cajochen C, et al. Cold feet and prolonged sleep-onset latency in vasospastic syndrome. *The Lancet* 2001;358:125-126.
- 339. Orgül S, Kaiser HJ, Flammer J, Gasser P. Systemic blood pressure and capillary blood-cell velocity in glaucoma patients: a preliminary study. *Eur J Ophthalmol* 1995;5:88-91.
- 340. Teuchner B, Orgül S, Ulmer H, Haufschild T, Flammer J. Reduced thirst in patients with a vasospastic syndrome. *Acta Ophthalmologica Scandinavica* 2004;82:738-740.
- 341. Wunderlich K, Zimmerman J, H G. Vasospastic persons exhibit differential expression of ABC-transport proteins. *Mol Vis* 2003;9:756-761.
- 342. Gasser P, Flammer J. Influence of vasospasm on visual function. *Doc Ophthalmol* 1987;66:3-18.
- 343. Guthauser U, Flammer J, Mahler F. The relationship between digital and ocular vasospasm. *Graefe's Arch Clin Exp Ophthalmol* 1988;226:224-226.
- 344. Flammer J, Gasser P, Prunte C, Yao K. The probable involvement of factors other than ocular pressure in the pathogenesis of glaucoma. In: Drance S, Van Buskirk E, Neufeld A (eds), *Pharmacology of Glaucoma*. Baltimore: Williams and Wilkins; 1992:273-283.
- 345. Prunte C, Flammer J. Choroidal capillary and venous congestion in central serous chorioretinopathy. *Am J Ophthalmol* 1996;121:26-34.
- 346. Messerli J, Flammer J. [Central vein thrombosis in younger patients]. *Klin Monatsbl Augenheilkd* 1996;208:303-305.
- 347. Kaiser HJ, Flammer J, Messerli J. Vasospasm-a risk factor for nonarteritic anterior ischaemic optic neuropathy? *Neuro Ophthalmol* 1996;16:5-10.
- 348. Flammer J, Kaiser HJ, Haufschild T. Susac syndrome: a vasospastic disorder? *Eur J Ophthalmol* 2001;11:175-179.
- 349. Gasser P, Meienberg O. Finger microcirculation in classical migraine. A videomicroscopic study of nailfold capillaries. *European Neurology* 1991;31:168-171.

- 350. Kaiser HJ, Flammer J, Burckhardt D. SILENT-MYOCARDIAL-ISCHEMIA IN GLAUCOMA PATIENTS. *Ophthalmologica* 1993;207:6-8.
- 351. Lip GYH, Ray KK, Shiu MF. Coronary spasm in acute myocardial infarction. *Heart* 1998;80:197-199.
- 352. Grieshaber MC, Terhorst T, Flammer J. The pathogenesis of optic disc splinter haemorrhages: a new hypothesis. *Acta Ophthalmologica Scandinavica* 2006;84:62-68.
- 353. Drance S, Douglas G, Wijsman K, Schulzer M, Britton R. Response of blood flow to warm and cold in normal and low-tension glaucoma patients. *Am J Ophthalmol* 1988;105:35-39.
- 354. Gasser P, Flammer J, Guthauser U, Mahler F. Do vasospasms provoke ocular diseases? *Angiology* 1990;41:213-220.
- 355. Cellini M, Possati GL, Profazio V, Sbrocca M, Caramazza N, Caramazza R. Color doppler imaging and plasma levels of endothelin-1 in low-tension glaucoma. *Acta Ophthalmologica Scandinavica* 1997;75:11-13.
- 356. Šugiyama T, Moriya S, Oku H, Azuma I. Association of endothelin-1 with normal tension glaucoma: Clinical and fundamental studies. *Survey of Ophthalmology* 1995;39:S49-S56.
- 357. Ghanem A, Elewa A, Arafa L. Endothelin-1 and Nitric Oxide Levels in Patients with Glaucoma. *Ophthalmic Research* 2011;46:98.
- 358. Pillunat LE, Lang GK, Harris A. *The visual response to increased ocular blood flow in normal pressure glaucoma*; 1994.
- 359. Harris A, Sergott RC, Spaeth GL, Katz JL, Shoemaker JA, Martin BJ. Color Doppler analysis of ocular vessel blood velocity in normal-tension glaucoma. *American Journal of Ophthalmology* 1994;118:642-649.
- 360. Goldberg I, Chai E, Chia A, Benscher C, Bauman A, Chen J. The Hettinger Hand Vibration Test, Vasospasm, and Glaucoma. *Survey of Ophthalmology* 1999;43:S66-S78.
- 361. Butt Z, O'Brien C, McKillop G, Aspinall P, Allan P. Color Doppler imaging in untreated high- and normal-pressure open-angle glaucoma. *Investigative ophthalmology & visual science* 1997;38:690-696.
- 362. Hosking SL, Harris A, Chung HS, et al. *Ocular haemodynamic responses to induced hypercapnia and hyperoxia in glaucoma*; 2004.
- 363. Schulzer M, Drance SM, Carter CJ, Brooks DE, Douglas GR, Lau W. Biostatistical evidence for two distinct chronic open angle glaucoma populations. *British Journal of Ophthalmology* 1990;74:196-200.
- 364. Hafez A, Bizzarro R, Descovich D, Lesk M. Correlation between finger blood flow and changes in optic nerve head blood flow following therapeutic intraocular pressure reduction. *J Glaucoma* 2005;14:448-454.
- 365. Creutzig A, Hiller S, Appiah R, Thum J, Caspary L. Nailfold capillaroscopy and laser Doppler fluxmetry for evaluation of Raynaud's phenomenon: how valid is the local cooling test? *Vasa* 1997;26:205-209.
- 366. Mozaffarieh M, Grieshaber, M C., Flammer, J. Oxygen and blood flow: players in the pathogenesis of glaucoma. *Molecular Vision* 2008;14:224-233.
- 367. Harris KF, Matthews KA. Interactions between autonomic nervous system activity and endothelial function: A model for the development of cardiovascular disease. *Psychosomatic medicine* 2004;66:153-164.
- 368. Doganay S, Evereklioglu C, Turkoz Y, Er H. Decreased nitric oxide production in primary open-angle glaucoma. *European journal of ophthalmology* 2002;12:44-48.
- 369. Nathanson JA, McKee M. Alterations of ocular nitric oxide synthase in human glaucoma. *Investigative ophthalmology & visual science* 1995;36:1774-1784.
- 370. Tezel G, Kass MA, Kolker AE, Becker B, Wax MB. Plasma and aqueous humor endothelin levels in primary open-angle glaucoma. *Journal of glaucoma* 1997;6:83.

- 371. Noske W, Hensen J, Wiederholt M. Endothelin-like immunoreactivity in aqueous humor of patients with primary open-angle glaucoma and cataract. *Graefe's Archive for Clinical and Experimental Ophthalmology* 1997;235:551-552.
- 372. Polak K. Altered nitric oxide system in patients with open-angle glaucoma. *Archives of ophthalmology* 2007;125:494.
- 373. Riva CE, Salgarello T, Logean E, Colotto A, Galan EM, Falsini B. Flicker-Evoked Response Measured at the Optic Disc Rim Is Reduced in Ocular Hypertension and Early Glaucoma. *Investigative ophthalmology & visual science* 2004;45:3662-3668.
- 374. Oku H, Sugiyama T, Kojima S, Watanabe T, Azuma I. Experimental Optic Cup Enlargement Caused by Endothelin-1-Induced Chronic Optic Nerve Head Ischemia. *Survey of Ophthalmology* 1999;44:S74-S84.
- 375. Orgül S, Cioffi GA, Wilson DJ, Bacon DR, Van Buskirk EM. An endothelin-1 induced model of optic nerve ischemia in the rabbit. *Investigative ophthalmology* & visual science 1996;37:1860-1869.
- 376. Chauhan BC, LeVatte TL, Jollimore CA, et al. Model of Endothelin-1–Induced Chronic Optic Neuropathy in Rat. *Investigative ophthalmology & visual science* 2004;45:144-152.
- 377. Galassi F, Renieri G, Sodi A, Ucci F, Vannozzi L, Masini E. Nitric oxide proxies and ocular perfusion pressure in primary open angle glaucoma. *British Journal of Ophthalmology* 2004;88:757-760.
- 378. Galassi F, Sodi A, Ucci F, Renieri G, Pieri B, Masini E. Ocular haemodynamics and nitric oxide in normal pressure glaucoma. *Acta Ophthalmologica Scandinavica* 2000;78:37-38.
- 379. Venugopal SK, Devaraj S, Yuhanna I, Shaul P, Jialal I. Demonstration That C-Reactive Protein Decreases eNOS Expression and Bioactivity in Human Aortic Endothelial Cells. *Circulation* 2002;106:1439-1441.
- Leibovitch I, Kurtz S, Kesler A, Feithliher N, Shemesh G, Sela B-A. C-Reactive Protein Levels in Normal Tension Glaucoma. *Journal of glaucoma* 2005;14:384-386.
- 381. Su W-W, Ho W-J, Cheng S-T, Chang SHL, Wu S-C. Systemic High-sensitivity Creactive Protein Levels in Normal-tension Glaucoma and Primary Open-angle Glaucoma. *Journal of glaucoma* 2007;16:320-323 310.1097/IJG.1090b1013e3180391a3180383.
- 382. Henry E, Newby DE, Webb DJ, O'Brien C. Peripheral endothelial dysfunction in normal pressure glaucoma. *Investigative ophthalmology & visual science* 1999;40:1710-1714.
- 383. Henry E, Newby DE, Webb DJ, Hadoke PWF, O'Brien CJ. Altered Endothelin-1 Vasoreactivity in Patients with Untreated Normal-Pressure Glaucoma. *Invest Ophthalmol Vis Sci* 2006;47:2528-2532.
- 384. Buckley C. Systemic vascular endothelial cell dysfunction in normal pressure glaucoma. *British Journal of Ophthalmology* 2002;86:227.
- 385. Fadini G, Pagano C, Baesso I, et al. Reduced endothelial progenitor cells and brachial artery flow-mediated dilation as evidence of endothelial dysfunction in ocular hypertension and primary open-angle glaucoma. *Acta Ophthalmologica* 2010;88:135.
- 386. Emre M, Orgul S, Haufschild T, Shaw SG, Flammer J. Increased plasma endothelin-1 levels in patients with progressive open angle glaucoma. *The British journal of ophthalmology* 2005;89:60-63.
- 387. Kunimatsu S, Mayama C, Tomidokoro A, Araie M. Plasma Endothelin-1 Level in Japanese Normal Tension Glaucoma Patients. *Current Eye Research* 2006;31:727 - 731.
- Holló G, Lakatos P, Farkas K. Cold pressor test and plasma endothelin-1 concentration in primary open-angle and capsular glaucoma. *Journal of glaucoma* 1998;7:105.

- 389. Chen H, Chang Y, Chen W, Lane H. Association Between Plasma Endothelin-1 and Severity of Different Types of Glaucoma. *J Glaucoma* 2011;Epub ahead of print.
- 390. Kaiser HJ, Flammer J, Wenk M, Lüscher T. Endothelin-1 plasma levels in normal-tension glaucoma: abnormal response to postural changes. *Graefe's Archive for Clinical and Experimental Ophthalmology* 1995;233:484-488.
- 391. Nicolela MT, Ferrier SN, Morrison CA, et al. Effects of Cold-Induced Vasospasm in Glaucoma: The Role of Endothelin-1. *Investigative ophthalmology & visual science* 2003;44:2565-2572.
- 392. Lip PL, Felmeden DC, Blann AD, et al. Plasma vascular endothelial growth factor, soluble VEGF receptor FLT-1, and von Willebrand factor in glaucoma. *Br J Ophthalmol* 2002;86:1299-1302.
- 393. Logan JFJ, Chakravarthy U, Hughes AE, Patterson CC, Jackson JA, Rankin SJA. Evidence for Association of Endothelial Nitric Oxide Synthase Gene in Subjects with Glaucoma and a History of Migraine. *Invest Ophthalmol Vis Sci* 2005;46:3221-3226.
- 394. Tunny TJ, Richardson KA, Clark CV. Association study of the 5' flanking regions of endothelial-nitric oxide synthase and endothelin-1 genes in familial primary open-angle glaucoma. *Clinical and experimental pharmacology & physiology* 1998;25:26-29.
- 395. Ayub H, Khan MI, Micheal S, et al. Association of eNOS and HSP70 gene polymorphisms with glaucoma in Pakistani cohorts. *Molecular Vision* 2010;16:18-25.
- 396. Kim S, Kim J, Kim D, et al. Investigations on the association between normal tension glaucoma and single nucleotide polymorphisms of the endothelin-1 and endothelin receptor genes. *Molecular Vision* 2006;12:1016-1021.
- 397. Ishikawa K, Funayama T, Ohtake Y, et al. Association between glaucoma and gene polymorphism of endothelin type A receptor. *Molecular Vision* 2005;11:431-437.
- 398. Pumprla J, Howorka K, Groves D, Chester M, Nolan J. Functional assessment of heart rate variability: physiological basis and practical applications. *Int J Cardiol* 2002;84:1-14.
- 399. Jaradeh SS, Prieto TE. Evaluation of the autonomic nervous system. *Phys Med Rehabil Clin N Am* 2003;14:287-305.
- 400. Tamura H, Kawakami H, Kanamoto T, et al. High frequency of open-angle glaucoma in Japanese patients with Alzheimer's disease. *Journal of the Neurological Sciences* 2006;246:79.
- 401. Gherghel D, Hosking S, Orgul S. Autonomic nervous system, circadian rhythms, and primary open-angle glaucoma. *Surv Ophthalmol* 2004;49:491 508.
- 402. Wolf MM, Varigos GA, Hunt D, Sloman JG. Sinus arrhythmia in acute myocardial infarction. *The Medical journal of Australia* 1978;2:52-53.
- 403. Kitney RI, Rompleman O. *The Study of Heart Rate Variability*: Clarendon, Oxford; 1980.
- 404. Pagani M, Lombardi F, Guzzetti S, et al. Power spectral analysis of heart rate and arterial pressure variabilities as a marker of sympatho-vagal interaction in man and conscious dog. *Circulation research* 1986;59:178-193.
- 405. Sleight P, La Rovere MT, Mortara A, et al. Physiology and pathophysiology of heart rate and blood pressure variability in humans: is power spectral analysis largely an index of baroreflex gain? *Clinical science* 1995;88:103-109.
- 406. Scholz UJ, Bianchi AM, Cerutti S, Kubicki S. Vegetative Background of Sleep: Spectral Analysis of the Heart Rate Variability. *Physiology & Behavior* 1997;62:1037-1043.
- 407. Goldberger JJ. Sympathovagal balance: how should we measure it? . *Am J Physiol Heart Circ Physiol* 1999;276:H1273-H1280.

- 408. Portaluppi F, Smolensky MH. Circadian rhythm and environmental determinants of blood pressure regulation in normal and hypertensive conditions. In: White WB (ed), *Blood pressure monitoring in cardiovascular medicine and therapeutics*: Human Press, Totowa NJ; 2001:79-138.
- 409. Schilder P. Ocular blood flow responses to pathology of the carotid and cerebral circulations. *Survey of Ophthalmology* 1994;38:S52-S58.
- 410. Brown CM, Dutsch M, Michelson G, Neundorfer B, Hilz MJ. Impaired cardiovascular responses to baroreflex stimulation in open-angle and normal-pressure glaucoma. *Clinical science* 2002;102:623-630.
- 411. Kumar R, Ahuja VM. A study of changes in the status of autonomic nervous system in primary open angle glaucoma cases. *Indian Journal of Medical Sciences* 1999;53:529-534.
- 412. Clark C, Mapstone R. Systemic autonomic neuropathy in open-angle glaucoma. *Doc Ophthalmol* 1986;64:179-185.
- 413. Kashiwagi K, Tsumura T, Ishii H, Ijiri H, Tamura K, Tsukahara S. Circadian rhythm of autonomic nervous function in patients with normal-tension glaucoma compared with normal subjects using ambulatory electrocardiography. *Journal of Glaucoma* 2000;9:239-246.
- 414. Riccadonna M, Covi G, Pancera P, et al. Autonomic system activity and 24-hour blood pressure variations in subjects with normal- and high-tension glaucoma. *Journal of Glaucoma* 2003;12:156-163.
- 415. Na KS, Lee NY, Park SH, Park CK. Autonomic Dysfunction in Normal Tension Glaucoma: The Short-term Heart Rate Variability Analysis. *Journal of glaucoma* 2010;19:377-381.
- 416. Gherghel D, Hosking S, Cunliffe IA. Abnormal systemic and ocular vascular response to temperature provocation in primary open angle glaucoma patients: a case for autonomic failure? . *Invest Ophthalmol Vis Sci* 2004;45:3546 3554.
- 417. Bullock R, Hammond G. Realistic expectations: the management of severe Alzheimer's disease. *Alzheimer Dis Assoc Disord* 2003;17 (Suppl 3):S80-85.
- 418. Vickers JC. The cellular mechanism underlying neuronal degeneration in glaucoma Parallels with Alzheimer's disease. *Clinical & experimental ophthalmology* 1997;25:105.
- 419. Guo L, Salt TE, Luong V, et al. Targeting amyloid-beta in glaucoma treatment. *Proceedings of the National Academy of Sciences of the United States of America* 2007;104:13444-13449.
- 420. Shiose Y, Kitazawa Y, Tsukahara S, et al. EPIDEMIOLOGY OF GLAUCOMA IN JAPAN A NATIONWIDE GLAUCOMA SURVEY. *Japanese Journal of Ophthalmology* 1991;35:133-155.
- 421. Henderson VW. The epidemiology of estrogen replacement therapy and Alzheimer's disease. *Neurology* 1997;48:S27-S35.
- 422. Engelborghs S, De Vreese K, Van de Casteele T, et al. Diagnostic performance of a CSF-biomarker panel in autopsy-confirmed dementia. *Neurobiology of Aging* 2008;29:1143.
- 423. Yoneda S, Hara H, Hirata A, Fukushima M, Inomata Y, Tanihara H. Vitreous fluid levels of beta-amyloid((1-42)) and tau in patients with retinal diseases. *Japanese Journal of Ophthalmology* 2005;49:106-108.
- 424. McKinnon SJ, Lehman DM, Kerrigan-Baumrind LA, et al. Caspase activation and amyloid precursor protein cleavage in rat ocular hypertension. *Investigative ophthalmology & visual science* 2002;43:1077-1087.
- 425. Blanks JC, Hinton DR, Sadun AA, Carol AM. Retinal ganglion cell degeneration in Alzheimer's disease. *Brain Research* 1989;501:364.
- 426. Kessing LV, Lopez AG, Andersen PK, Kessing SV. No increased risk of developing Alzheimer disease in patients with glaucoma. *Journal of Glaucoma* 2007;16:47-51.

- 427. Parisi V, Restuccia R, Fattapposta F, Mina C, Bucci MG, Pierelli F. Morphological and functional retinal impairment in Alzheimer's disease patients. *Clinical neurophysiology* 2001;112:1860-1867.
- 428. Costa VP, Arcieri ES, Harris A. Blood pressure and glaucoma. *British Journal of Ophthalmology* 2009;93:1276-1282.
- 429. O'Brien E, Murphy J, Tyndall A, et al. Twenty-four-hour ambulatory blood pressure in men and women aged 17 to 80 years: the Allied Irish Bank Study. *Journal of hypertension* 1991;9:355-360.
- 430. Blanks JC, Torigoe Y, Hinton DR, Blanks RHI. Retinal pathology in Alzheimer's disease .1. Ganglion cell loss in foveal/parafoveal retina. *Neurobiology of Aging* 1996;17:377-384.
- 431. Iseri PK, Altinas O, Tokay T, Yuksel N. Relationship between cognitive impairment and retinal morphological and visual functional abnormalities in Alzheimer disease. *Journal of neuro-ophthalmology* 2006;26:18-24.
- 432. O'Brien E, Sheridan J, O'Malley K. Dippers and non-dippers. *Lancet* 1988;13:397.
- 433. Tokunaga T, Kashiwagi A, Tsumura T, Taguchi K, Tsukahara S. Association between nocturnal blood pressure reduction and progression of visual field defect in patients with primary open-angle glaucoma or normal-tension glaucoma. *Jpn J Ophthalmol* 2004;48.
- 434. Verdecchia P, Schillaci G, Porcellati C. Dippers versus non-dippers. *Journal of hypertension Supplement* 1991;9:S42-44.
- 435. Chandra V, Bharucha NE, Schoenberg BS. Conditions associated with Alzheimer's disease at death: case-control study. *Neurology* 1986;36:209-211.
- 436. Bayer AU, Ferrari G, Erb C. High occurence rate of glaucoma among patients with Alzheimer's disease. *Eur Neurol* 2002;47:165-168.
- 437. Hansen TW, Kikuya M, Thijs L, et al. Prognostic superiority of daytime ambulatory over conventional blood pressure in four populations: a metaanalysis of 7030 individuals. *Journal of Hypertension* 2007;25:1554-1564 1510.1097/HJH.1550b1013e3281c1549da1555.
- 438. Dolan E, Stanton A, Thijs L, et al. Superiority of Ambulatory Over Clinic Blood Pressure Measurement in Predicting Mortality: The Dublin Outcome Study. *Hypertension* 2005;46:156-161.
- 439. White WB. Relating Cardiovascular Risk to Out-of-Office Blood Pressure and the Importance of Controlling Blood Pressure 24 Hours a Day. *The American journal of medicine* 2008;121:S2-S7.
- 440. O'Brien E, Asmar R, Beilin L, et al. European Society of Hypertension recommendations for conventional, ambulatory and home blood pressure measurement. *Journal of Hypertension* 2003;21:821-848.
- 441. Wang N, Peng Z, Fan B, et al. [Case control study on the risk factors of primary open angle glaucoma in China]. *zhong hua liu xing bing xue za zhi* 2002;23:293-296.
- 442. Wilson MR, Hertzmark E, Walker AM, Childsshaw K, Epstein DL. A CASE-CONTROL STUDY OF RISK-FACTORS IN OPEN ANGLE GLAUCOMA. *Archives of ophthalmology* 1987;105:1066-1071.
- 443. Mitchell P, Lee AJ, Rochtchina E, Wang JJ. Open-Angle Glaucoma and Systemic Hypertension: The Blue Mountains Eye Study. *Journal of glaucoma* 2004;13:319-326.
- 444. Goldberg I, Hollows FC, Kass MA, Becker B. Systemic factors in patients with low-tension glaucoma. *British Journal of Ophthalmology* 1981;65:56-62.
- 445. Kashiwagi K, Hosaka O, Kashiwagi F, et al. Systemic Circulatory Parameters: Comparison Between Patients with Normal Tension Glaucoma and Normal Subjects Using Ambulatory Monitoring. *Japanese Journal of Ophthalmology* 45:388-396.

- 446. Orzalesi N, Rossetti L, Omboni S. Vascular risk factors in glaucoma: the results of a national survey. *Graefe's Archive for Clinical and Experimental Ophthalmology* 2007;245:795-802.
- 447. Leighton DA, Phillips CI. SYSTEMIC BLOOD-PRESSURE IN OPEN-ANGLE GLAUCOMA, LOW TENSION GLAUCOMA, AND NORMAL EYE. British Journal of Ophthalmology 1972;56:447.
- 448. Dielemans I, Vingerling JR, Algra D, Hofman A, Grobbee DE, de Jong PT. Primary open-angle glaucoma, intraocular pressure, and systemic blood pressure in the general elderly population. The Rotterdam Study. *Ophthalmology* 1995;102:54-60.
- 449. Bayer AU, Keller ON, Ferrari G, Maag K-P. Association of glaucoma with neurodegenerative diseases with apoptotic cell death: Alzheimer's disease and Parkinson's disease. *Am J Ophthalmol* 2002;133:135-137.
- 450. Collignon N, Dewe W, Guillaume S, Collignon-Brach J. Ambulatory blood pressure monitoring in glaucoma patients. The nocturnal systolic dip and its relationship with disease progression. *International Ophthalmology* 1998;22:19-25.
- 451. Kashiwagi K, Hosaka O, Kashiwagi F, et al. Systemic Circulatory Parameters: Comparison Between Patients with Normal Tension Glaucoma and Normal Subjects Using Ambulatory Monitoring. *Japanese Journal of Ophthalmology* 2001;45:388-396.
- 452. Detry M, Boschi A, Ellinghaus G, De Plaen JF. Simultaneous 24-hour monitoring of intraocular pressure and arterial blood pressure in patients with progressive and non-progressive primary open-angle glaucoma. *European journal of ophthalmology* 1996;6:273-278.
- 453. Shields MBMD. INTRAOCULAR PRESSURE AND SYSTEMIC BLOOD PRESSURE: LONGITUDINAL PERSPECTIVE: THE BEAVER DAM EYE STUDY. *Evidence-based ophthalmology* 2005;6:133-134.
- 454. Hennis A, Wu S-Y, Nemesure B, Leske MC. Hypertension, diabetes, and longitudinal changes in intraocular pressure. *Ophthalmology* 2003;110:908-914.
- 455. Klein BEK, Klein R, Knudtson MD. Intraocular pressure and systemic blood pressure: longitudinal perspective: the Beaver Dam Eye Study. *British Journal of Ophthalmology* 2005;89:284-287.
- 456. Constad WH, Fiore P, Samson C, Cinotti AA. Use of an angiotensin converting enzyme inhibitor in ocular hypertension and primary open-angle glaucoma. *American Journal of Ophthalmology* 1988;105:674-677.
- 457. Wang R-F, Podos SM, Mittag TW, Yokoyoma T. Effect of CS-088, an angiotensin AT1 receptor antagonist, on intraocular pressure in glaucomatous monkey eyes. *Experimental Eye Research* 2005;80:629-632.
- 458. Hvidberg A, Kessing SV, Fernandes A. EFFECT OF CHANGES IN PC02 AND BODY POSITIONS ON INTRAOCULAR PRESSURE DURING GENERAL ANAESTHESIA. Acta Ophthalmologica 1981;59:465-475.
- 459. Kaiser HJ, Flammer J, Graf T, Stumpfig D. Systemic blood pressure in glaucoma patients. *Graefe's Archive for Clinical and Experimental Ophthalmology* 1993;231:677-680.
- 460. Bechetoille A, Bresson-Dumont H. DIURNAL AND NOCTURNAL BLOOD-PRESSURE DROPS IN PATIENTS WITH FOCAL ISCHEMIC GLAUCOMA. *Graefe's Archive for Clinical and Experimental Ophthalmology* 1994;232:675-679.
- 461. Hayreh SS, Podhajsky P, Zimmerman MB. Beta-blocker eyedrops and nocturnal arterial hypotension. *American Journal of Ophthalmology* 1999;128:301-309.
- 462. Graham SL, Drance SM, Wijsman K, Douglas GR, Mikelberg FS. Ambulatory blood pressure monitoring in glaucoma. The nocturnal dip. *Ophthalmology* 1995;102:61-69.

- 463. Wostyn P, Audenaert K, De Deyn PP. The Valsalva Maneuver and Alzheimer's Disease: Is there a link? *Current Alzheimer research* 2009;6:59-68.
- 464. Kaiser HJ, Flammer J. Systemic hypotension: a risk factor for glaucomatous damage? *Ophthalmologica* 1991;203:105-108.
- 465. Bresson-Dumont H, Bechetoille A. [Role of arterial blood pressure in the development of glaucomatous lesions]. *Journal français d'ophtalmologie* 1996;19:435-442.
- 466. Meyer JH, BrandiDohrn J, Funk J. Twenty four hour blood pressure monitoring in normal tension glaucoma. *British Journal of Ophthalmology* 1996;80:864-867.
- 467. Joe SG, Choi J, Sung KR, Park SB, Kook MS. Twenty-Four Hour Blood Pressure Pattern in Patients With Normal Tension Glaucoma in the Habitual Position. *Korean J Ophthalmol* 2009;23:32-39.
- 468. Sugiyama T, Utsunomiya K, Ota H, Ogura Y, Narabayashi I, Ikeda T. Comparative study of cerebral blood flow in patients with normal-tension glaucoma and control subjects. *American Journal of Ophthalmology* 2006;141:394-396.
- 469. Harris A, Zarfati D, Zalish M. Reduced cerebrovascular blood flow velocities and vasoreactivity in open-angle glaucoma. *Am J Ophthalmol* 2003;135:144-147.
- 470. Topouzis F, Coleman AL, Harris A, et al. Association of Blood Pressure Status With the Optic Disk Structure in Non-glaucoma Subjects: The Thessaloniki Eye Study. *American Journal of Ophthalmology* 2006;142:60-67.e61.
- Muskens R, de Voogd S, Wolfs RCW, et al. Systemic antihypertensive medication and incident open-angle glaucoma. *Ophthalmology* 2007;114:2221-2226.
- 472. Cioffi GA. Three Common Assumptions About Ocular Blood Flow and Glaucoma. *Survey of Ophthalmology* 2001;45:S325-S331.
- 473. Seeley R, Stephens T, Tate P. *Essentials of Anatomy and Physiology, Sixth Edition*. New York: McGraw-Hill Companies; 2007.
- 474. Silverthorn D. *Human Physiology: An integrated approach*. San Francisco: Pearson Education Inc; 2010.
- 475. Fruchart J-C, Nierman MC, Stroes ESG, Kastelein JJP, Duriez P. New Risk Factors for Atherosclerosis and Patient Risk Assessment. *Circulation* 2004;109:III-15-III-19.
- 476. Peräsalo R, Raitta C, Peräsalo J. Optic nerve fiber loss in relation to atrial fibrillation and blood pressure. *International Ophthalmology* 1992;16:259-263.
- 477. Demailly P, Cambien F, Plouin PF, Baron P, Chevallier B. Do Patients with Low Tension Glaucoma Have Particular Cardiovascular Characteristics? *Ophthalmologica* 1984;188:65-75.
- 478. Drance SM, Sweeney VP, Morgan RW, Feldman F. Studies of Factors Involved in the Production of Low Tension Glaucoma. *Arch Ophthalmol* 1973;89:457-465.
- 479. Lee AJ, Wang JJ, Kifley A, Mitchell P. Open-Angle Glaucoma and Cardiovascular Mortality: The Blue Mountains Eye Study. *Ophthalmology* 2006;113:1069-1076.
- 480. Gungor IU, Gungor L, Ozarslan Y, et al. Is Symptomatic Atherosclerotic Cerebrovascular Disease a Risk Factor for Normal-Tension Glaucoma? *Medical Principles and Practice* 2011;20:220-224.
- 481. Visontai Z, Mersich B, Holló G. Carotid artery elasticity and baroreflex sensitivity in patients with glaucoma. *Journal of Glaucoma* 2005;14:30-35.
- 482. Siasou G. RELATIONSHIPS BETWEEN GLAUCOMA, ARTERIAL STIFFNESS AND INFLAMMATORY PROCESS. *Journal of the American College of Cardiology* 2010;55:A159.
- 483. Ohguro I, Ishikawa F, Ohguro H, Shimokawa R, Nakazawa M. Arterial Sclerosis Grade in Normal-Tension Glaucoma Patients. *J Eye* 2004;21:397-400.
- 484. de Voogd S, Wolfs RCW, Jansonius NM, Witteman JCM, Hofman A, de Jong PTVM. Atherosclerosis, C-Reactive Protein, and Risk for Open-Angle Glaucoma:

The Rotterdam Study. *Investigative ophthalmology & visual science* 2006;47:3772-3776.

- 485. Stewart WC, Sine C, Sutherland S, Stewart JA. Total cholesterol and highdensity lipoprotein levels as risk factors for increased intraocular pressure. *Am J Ophthalmol* 1996;122:575-577.
- 486. Kang J, Pasquale LR, Rosner B, et al. Prospective study of cigarette smoking and the risk of primary open-angle glaucoma. *Arch Ophthalmol* 2003;121:1762-1768.
- 487. Ramdas WD, Wolfs RCW, Hofman A, de Jong PTVM, Vingerling JR, Jansonius NM. Lifestyle and Risk of Developing Open-Angle Glaucoma: The Rotterdam Study. *Arch Ophthalmol* 2011;129:767-772.
- 488. Cohn JN. Arterial Stiffness, Vascular Disease, and Risk of Cardiovascular Events. *Circulation* 2006;113:601-603.
- 489. Mackenzie IS, Wilkinson IB, Cockcroft JR. Assessment of arterial stiffness in clinical practice. *QJM* 2002;95:67-74.
- 490. O'Rourke MF, Staessen J, Vlachopoulos C, Duprez D, Plante G. Clinical applications of arterial stiffness; definitions and reference values. *American journal of hypertension* 2002;15:426-444.
- 491. Sies H. Oxidative stress: oxidants and antioxidants. *Experimental Physiology* 1997;82:291-295.
- 492. Halliwell B. Reactive Species and Antioxidants. Redox Biology Is a Fundamental Theme of Aerobic Life. *Plant Physiology* 2006;141:312-322.
- 493. Dröge W. Free Radicals in the Physiological Control of Cell Function. *Physiological Reviews* 2002;82:47-95.
- 494. Cabrerizo FM, Laura Dántola M, Petroselli G, et al. Reactivity of Conjugated and Unconjugated Pterins with Singlet Oxygen (O2(1Δg)): Physical Quenching and Chemical Reaction[†]. *Photochemistry and Photobiology* 2007;83:526-534.
- 495. Donzelli S, Switzer CH, Thomas DD, et al. The Activation of Metabolites of Nitric Oxide Synthase by Metals Is Both Redox and Oxygen Dependent: A New Feature of Nitrogen Oxide Signaling. *Antioxidants & Redox Signaling* 2006;8:1363-1371.
- 496. Hodgson EK, Fridovich I. Reversal of the superoxide dismutase reaction. Biochemical and Biophysical Research Communications 1973;54:270-274.
- 497. Qi X, Hauswirth W, Guy J. Dual gene therapy with extracellular superoxide dismutase and catalase attenuates experimental optic neuritis. *Mol Vis* 2007;13:1-11.
- 498. Ursini F, Maiorino M, Brigelius-Flohe R, et al. Diversity of glutathione peroxidases. *Methods in enzymology* 1995;252:38-53.
- 499. Meyer A, Hell R. Glutathione homeostasis and redox-regulation by sulfhydryl groups. *Photosynthesis Research* 2005;86:435-457.
- 500. Kosower NS, Kosower EM, G.H. Bourne JFD, Jeon KW. The Glutathione Status of Cells. *International Review of Cytology*: Academic Press; 1978:109-160.
- 501. Diller KR. STRESS PROTEIN EXPRESSION KINETICS. Annual Review of Biomedical Engineering 2006;8:403-424.
- 502. Gutteridge JM, Quinlan GJ, Evans TW. Transient iron overload with bleomycin detectable iron in the plasma of patients with adult respiratory distress syndrome. *Thorax* 1994;49:707-710.
- 503. Cho SY, Lee E-Y, Lee JS, et al. Efficient prefractionation of low-abundance proteins in human plasma and construction of a two-dimensional map. *PROTEOMICS* 2005;5:3386-3396.
- 504. Jarrett SG, Albon J, Boulton M. The contribution of DNA repair and antioxidants in determining cell type-specific resistance to oxidative stress. *Free Radical Research* 2006;40:1155-1165.
- 505. Baynes JW. Role of oxidative stress in development of complications in diabetes. *Diabetes* 1991;40:405-412.

- 506. Halliwell B. Free radicals, reactive oxygen species and human disease: a critical evaluation with special reference to atherosclerosis. *Br J Exp Pathol* 1989;70:737-757.
- 507. Mapp PI, Grootveld MC, Blake DR. Hypoxia, oxidative stress and rheumatoid arthritis. *British Medical Bulletin* 1995;51:419-436.
- 508. Hack V, Schmid D, Breitkreutz R, et al. Cystine levels, cystine flux, and protein catabolism in cancer cachexia, HIV/SIV infection, and senescence. *The FASEB Journal* 1997;11:84-92.
- 509. Shen J, Dong A, Hackett S, Bell W, Green W, Campochiaro P. Oxidative damage in age-related macular degeneration. *Histol Histopathol* 2007;22:1301-1308.
- 510. Finkel T, Holbrook NJ. Oxidants, oxidative stress and the biology of ageing. *Nature* 2000;408:239-247.
- 511. Meister A, Anderson M. Glutathione. Ann Rev Biochem 1983;52:711-760.
- 512. Sies H. Glutathione and its role in cellular functions. *Free Radical Biology and Medicine* 1999;27:916-921.
- 513. Pompella A, Visvikis A, Paolicchi A, Tata VD, Casini AF. The changing faces of glutathione, a cellular protagonist. *Biochemical Pharmacology* 2003;66:1499-1503.
- 514. Deneke S. Thiol-based antioxidants. *Current Topics in Cellular Regulation* 2000;36:151-180.
- 515. Townsend DM, Tew KD, Tapiero H. The importance of glutathione in human disease. *Biomedicine & Pharmacotherapy* 2003;57:145-155.
- 516. Owen JB, Butterfield DA, Bross P, Gregersen N. Measurement of Oxidized/Reduced Glutathione Ratio
- Protein Misfolding and Cellular Stress in Disease and Aging. In: Walker JM (ed): Humana Press; 2010:269-277.
- 517. Ames BN, Shigenaga MK, Hagen TM. Oxidants, antioxidants, and the degenerative diseases of aging. *Proceedings of the National Academy of Sciences* 1993;90:7915-7922.
- 518. Pennathur S, Heinecke JW. Mechanisms for Oxidative Stress in Diabetic Cardiovascular Disease. *Antioxidants & Redox Signaling* 2007;9:955-969.
- 519. Pennathur S, Heinecke J. Oxidative stress and endothelial dysfunction in vascular disease. *Current Diabetes Reports* 2007;7:257-264.
- 520. Guzik TJ, Harrison DG. Vascular NADPH oxidases as drug targets for novel antioxidant strategies. *Drug Discovery Today* 2006;11:524-533.
- 521. Taddei S, Virdis A, Ghiadoni L, Magagna A, Salvetti A. Vitamin C Improves Endothelium-Dependent Vasodilation by Restoring Nitric Oxide Activity in Essential Hypertension. *Circulation* 1998;97:2222-2229.
- 522. Ohara Y, Peterson TE, Sayegh HS, Subramanian RR, Wilcox JN, Harrison DG. Dietary Correction of Hypercholesterolemia in the Rabbit Normalizes Endothelial Superoxide Anion Production. *Circulation* 1995;92:898-903.
- 523. Ting HH, Timimi FK, Haley EA, Roddy M-A, Ganz P, Creager MA. Vitamin C Improves Endothelium-Dependent Vasodilation in Forearm Resistance Vessels of Humans With Hypercholesterolemia. *Circulation* 1997;95:2617-2622.
- 524. Moenkemann H, Flammer J, Wunderlich K, Breipohl W, Schild HH, Golubnitschaja O. Increased DNA breaks and up-regulation of both G1 and G2 checkpoint genes p21^{WAF1/CIP1} and 14-3-3 σ in circulating leukocytes of glaucoma patients and vasospastic individuals. *Amino Acids* 2005;28:199-205.
- 525. Golubnitschaja O, Yeghiazaryan K, Liu R, et al. Increased Expression of Matrix Metalloproteinases in Mononuclear Blood Cells of Normal-Tension Glaucoma Patients. *Journal of glaucoma* 2004;13:66-72.
- 526. Wunderlich K, Golubnitschaja O, Pache M, Eberle A, Flammer J. Increased plasma levels of 20S proteasome alpha-subunit in glaucoma patients: an observational pilot study. *Mol Vis* 2002;8:431-435.

- 527. Sorkhabi R, Ghorbanihaghjo A, Javadzadeh A, Rashtchizadeh N, Moharrery M. Oxidative DNA damage and total antioxidant status in glaucoma patients. *Molecular Vision* 2011;17:41-46.
- 528. Majsterek I, Malinowska K, Stanczyk M, et al. Evaluation of oxidative stress markers in pathogenesis of primary open-angle glaucoma. *Experimental and Molecular Pathology* 2011;90:231-237.
- 529. Russell P, Johnson D. Enzymes protective of oxidative damage present in all decades of life in the trabecular meshwork, as detected by two-dimensional gel electrophoresis protein maps. *J Glaucoma* 1996;5:317-324.
- 530. Saccà SC, Izzotti A, Rossi P, Traverso C. Glaucomatous outflow pathway and oxidative stress. *Experimental Eye Research* 2007;84:389-399.
- 531. Ferreira SM, Lerner SFn, Brunzini R, Evelson PA, Llesuy SF. Oxidative stress markers in aqueous humor of glaucoma patients. *American Journal of Ophthalmology* 2004;137:62-69.
- 532. Bunin A, Filina A, Erichev V. [A glutathione deficiency in open-angle glaucoma and the approaches to its correction]. *Vestnik Oftalmologii* 1992;112:3-5.
- 533. Wang N, Chintala SK, Fini ME, Schuman JS. Activation of a tissue-specific stress response in the aqueous outflow pathway of the eye defines the glaucoma disease phenotype. *Nat Med* 2001;7:304-309.
- 534. Ghanem AA, Arafa LF, El-Baz A. Oxidative Stress Markers in Patients with Primary Open-Angle Glaucoma. *Current Eye Research* 2010;35:295-301.
- 535. Izzotti A, Saccà SC, Cartiglia C, De Flora S. Oxidative deoxyribonucleic acid damage in the eyes of glaucoma patients. *The American journal of medicine* 2003;114:638-646.
- 536. Kahn MG, Giblin FJ, Epstein DL. Glutathione in calf trabecular meshwork and its relation to aqueous humor outflow facility. *Investigative ophthalmology & visual science* 1983;24:1283-1287.
- 537. Grace P. Ischaemia-reperfusion injury. Br J Surg 1994;81:637-647.
- 538. McKinnon SJ. Glaucoma, apoptosis, and neuroprotection. *Current Opinion in Ophthalmology* 1997;8:28-37.
- 539. Quigley HA, Nickells RW, Kerrigan LA, Pease ME, Thibault DJ, Zack DJ. RETINAL GANGLION-CELL DEATH IN EXPERIMENTAL GLAUCOMA AND AFTER AXOTOMY OCCURS BY APOPTOSIS. *Investigative ophthalmology & visual science* 1995;36:774-786.
- 540. Murakami A, Okisaka S. [Neuronal cell death mechanism in glaucomatous optic neuropathy]. ニホン ガンカ ガッカイ ザッシ 1998;102:645-653.
- 541. Hernandez MR, Miao H, Lukas T. Astrocytes in glaucomatous optic neuropathy. In: Carlo Nucci LCNNO, Giacinto B (eds), *Progress in Brain Research*: Elsevier; 2008:353-373.
- 542. Neufeld AH, Hernandez MR, Gonzalez M. Nitric Oxide Synthase in the Human Glaucomatous Optic Nerve Head. *Arch Ophthalmol* 1997;115:497-503.
- 543. Luthra A, Gupta N, Kaufman PL, Weinreb RN, Yücel YH. Oxidative injury by peroxynitrite in neural and vascular tissue of the lateral geniculate nucleus in experimental glaucoma. *Experimental Eye Research* 2005;80:43-49.
- 544. Feilchenfeld Z, Yücel YH, Gupta N. Oxidative injury to blood vessels and glia of the pre-laminar optic nerve head in human glaucoma. *Experimental Eye Research* 2008;87:409-414.
- 545. Zeitz O, Wagenfeld L, Wirtz N, et al. Influence of oxygen free radicals on the tone of ciliary arteries: a model of vasospasms of ocular vasculature. *Graefe's Archive for Clinical and Experimental Ophthalmology* 2007;245:1327-1333.
- 546. Zhang X, Neufeld AH. Activation of the Epidermal Growth Factor Receptor in Optic Nerve Astrocytes Leads to Early and Transient Induction of Cyclooxygenase-2. *Investigative ophthalmology & visual science* 2005;46:2035-2041.

- 547. Liu B, Chen H, Johns TG, Neufeld AH. Epidermal Growth Factor Receptor Activation: An Upstream Signal for Transition of Quiescent Astrocytes into Reactive Astrocytes after Neural Injury. *The Journal of Neuroscience* 2006;26:7532-7540.
- 548. Mojon DS, Hess CW, Goldblum D, Böhnke M, Körner F, Mathis J. Primary openangle glaucoma is associated with sleep apnea syndrome. *Ophthalmologica* 2000;214:115.
- 549. Drance SM. The Visual Field of Low Tension Glaucoma and Shock-Induced Optic Neuropathy. *Arch Ophthalmol* 1977;95:1359-1361.
- 550. Ritchie K, Lovestone S. The dementias. The Lancet 2002;360:1759-1766.
- 551. Gottfries C. Dementia: Classification and aspects of treatment. *Psychopharmacol Ser* 1988;5:187-195.
- 552. Morris J. Classification of dementia and Alzheimer's disease. *Acta Neurol Scand Suppl* 1996;165:41-50.
- 553. Gustafson L. Clinical classification of dementia conditions. *Acta Neurologica Scandinavica* 1992;85:16-20.
- 554. Alzheimer A. Uber eine eigenartige Erkrankung der Hirnrinde. *Allg Zeits Psychiatry PsychischYGerichtlich Med* 1907;64:146-148.
- 555. Hebert LE, Scherr PA, Beckett LA, et al. Age-specific incidence of Alzheimer's disease in a community population. *Journal of the American Medical Association* 1995;273:1354-1359.
- 556. Castellani RJ, Rolston RK, Smith MA. Alzheimer Disease. *Disease-a-Month* 2011;56:484-546.
- 557. Galimberti D, Scarpini E. Progress in Alzheimer's disease. *Journal of Neurology* 2011;1-11.
- 558. Glenner G. Amyloid beta protein and the basis for Alzheimer's disease. *Prog Clin Biol Res* 1989;317:857-868.
- 559. Sorrentino G, Bonavita V. Neurodegeneration and Alzheimer's disease: the lesson from tauopathies. *Neurological Sciences* 2007;28:63-71.
- 560. Rowan MJ, Klyubin I, Wang Q, Hu NW, Anwyl R. Synaptic memory mechanisms: Alzheimer's disease amyloid beta-peptide-induced dysfunction. *Biochemical society transactions* 2007;35:1219.
- 561. Medeiros R, Baglietto-Vargus D, Laferla F. The Role of Tau in Alzheimer's Disease and Related Disorders. *CNS Neurosci Ther* 2010.
- 562. Wostyn P, Audenaert K, De Deyn P. Alzheimer's disease-related changes in diseases characterized by elevation of intracranial or intraocular pressure. *Clinical Neurology and Neurosurgery* 2008;110:101-109.
- 563. Akiyama H, Barger S, Barnum S, et al. Inflammation and Alzheimer's disease. *Neurobiology of aging* 2000;21:383-421.
- 564. Maruszak A, Zekanowski C. Mitochondrial dysfunction and Alzheimer's disease. *Progress in Neuro-Psychopharmacology and Biological Psychiatry* 2010;35:320-330.
- 565. Zhu X, Smith MA, Honda K, et al. Vascular oxidative stress in Alzheimer disease. *Journal of the Neurological Sciences* 2007;257:240-246.
- 566. de la Torre JC, Mussivand T. Can disturbed brain microcirculation cause Alzheimer's disease? *Neurological Research* 1993;15:146.
- 567. Humpel C. Chronic mild cerebrovascular dysfunction as a cause for Alzheimer's disease? *Experimental Gerontology* 2010;46:225-232.
- 568. Corder E, Saunders A, Strittmatter W, et al. Gene dose of apolipoprotein E type 4 allele and the risk of Alzheimer's disease in late onset families. *Science* 1993;261:921-923.
- 569. Dubois B, Feldman HH, Jacova C, et al. Research criteria for the diagnosis of Alzheimer's disease: revising the NINCDS-ADRDA criteria. *The Lancet Neurology* 2007;6:734-746.

- 570. McKhann G, Drachman D, Folstein M, Katzman R, Price D, Stadlan EM. Clinical diagnosis of Alzheimer's disease: report of the NINCDS-ADRDA Work Group under the auspices of Department of Health and Human Services Task Force on Alzheimer's Disease. *Neurology* 1984;34:939-944.
- 571. American Psychiatric A. *Diagnostic and statistical manual of mental disorders: DSM-IV-TR*; 2000.
- 572. Varma AR, Snowden JS, Lloyd JJ, Talbot PR, Mann DMA, Neary D. Evaluation of the NINCDS-ADRDA criteria in the differentiation of Alzheimer's disease and frontotemporal dementia. *Journal of Neurology, Neurosurgery & Psychiatry* 1999;66:184-188.
- 573. Blacker D, Albert MS, Bassett SS, Go RCP, Harrell LE, Folstein MF. Reliability and Validity of NINCDS-ADRDA Criteria for Alzheimer's Disease: The National Institute of Mental Health Genetics Initiative. *Arch Neurol* 1994;51:1198-1204.
- 574. Waldemar G, Dubois B, Emre M, et al. Recommendations for the diagnosis and management of Alzheimer's disease and other disorders associated with dementia: EFNS guideline. *European Journal of Neurology* 2007;14:e1-e26.
- 575. Bäckman L, Jones S, Berger AK, Laukka EJ, Small BJ. Multiple cognitive deficits during the transition to Alzheimer's disease. *Journal of Internal Medicine* 2004;256:195-204.
- 576. Folstein MF, Folstein SE, McHugh PR. "Mini-mental state" : A practical method for grading the cognitive state of patients for the clinician. *Journal of Psychiatric Research* 1975;12:189-198.
- 577. NICE-CG42. Dementia. *NICE Clinical Guideline 42*: National Institute for Health and Clinical Excellence (NICE); 2011.
- 578. Crum RM, Anthony JC, Bassett SS, Folstein MF. Population-Based Norms for the Mini-Mental State Examination by Age and Educational Level. *JAMA: The Journal of the American Medical Association* 1993;269:2386-2391.
- 579. Escobar J, Burnam A, Karno M, Forsythe A, Landsverk J, Golding J. Use of the Mini-Mental State Examination (MMSE) in a community population of mixed ethnicity. Cultural and linguistic artifacts. *J Nerv Ment Dis* 1986;174:607-614.
- 580. Farkas E, Luiten PGM. Cerebral microvascular pathology in aging and Alzheimer's disease. *Progress in Neurobiology* 2001;64:575-611.
- 581. Alpers BJ, Berry RG, Paddison RM. Anatomical Studies of the Circle of Willis in Normal Brain. *AMA Arch Neurol Psychiatry* 1959;81:409-418.
- 582. Klein B, Kuschinsky W, Schrock H, Vetterlein F. Interdependency of local capillary density, blood flow, and metabolism in rat brains. *American Journal of Physiology Heart and Circulatory Physiology* 1986;251:H1333-H1340.
- 583. Gjedde A, Diemer NH. Double-tracer study of the fine regional blood-brain glucose transfer in the rat by computer-assisted autoradiography. *Journal of cerebral blood flow and metabolism* 1985;5:282.
- 584. Perlmutter LS, Helena Chang C. Microangiopathy, the vascular basement membrane and Alzheimer's disease: a review. *Brain Research Bulletin* 1990;24:677-686.
- 585. Janzer RC. The blood-brain barrier: Cellular basis. *Journal of Inherited Metabolic Disease* 1993;16:639-647.
- 586. Koehler RC, Gebremedhin D, Harder DR. Role of astrocytes in cerebrovascular regulation. *Journal of Applied Physiology* 2006;100:307-317.
- 587. Cohen Z, Molinatti G, Hamel E. Astroglial and Vascular Interactions of Noradrenaline Terminals in the Rat Cerebral Cortex. *J Cereb Blood Flow Metab* 1997;17:894-904.
- 588. Kulik T. Regulation of cerebral vasculature in normal and ischemic brain. *Neuropharmacology* 2008;55:281.
- 589. Fog M. CEREBRAL CIRCULATION: THE REACTION OF THE PIAL ARTERIES TO A FALL IN BLOOD PRESSURE. Arch Neurol Psychiatry 1937;37:351-364.

- 590. Fog M. CEREBRAL CIRCULATION: II. REACTION OF PIAL ARTERIES TO INCREASE IN BLOOD PRESSURE. *Arch Neurol Psychiatry* 1939;41:260-268.
- 591. Lassen NA. Cerebral Blood Flow and Oxygen Consumption in Man. *Physiological Reviews* 1959;39:183-238.
- 592. Faraci FM, Heistad DD. Regulation of the Cerebral Circulation: Role of Endothelium and Potassium Channels. *Physiological Reviews* 1998;78:53-97.
- 593. Strandgaard S. Cerebral autoregulation. *Stroke* 1984;15:413.
- 594. Baumbach GL. Effects of sympathetic stimulation and changes in arterial pressure on segmental resistance of cerebral vessels in rabbits and cats. *Circulation research* 1983;52:527.
- 595. Heistad DD. Effects of sympathetic nerves on cerebral vessels in dog, cat, and monkey. *American journal of physiology Heart and circulatory physiology* 1978;235:H544.
- 596. Branston NM, Branston. The physiology of the cerebrovascular parasympathetic innervation. *British journal of neurosurgery* 1995;9:319.
- 597. Mosso A. Sulla circolazione del cervello dell'uomo. *Atti R Accad Lincei* 1880;5:237-358.
- 598. Roy CS, Sherrington CS. On the Regulation of the Blood-supply of the Brain. *The Journal of Physiology* 1890;11:85-158.
- 599. Filosa JA, Blanco VM. Neurovascular coupling in the mammalian brain. *Experimental Physiology* 2007;92:641-646.
- 600. Lecrux C, Hamel E. The neurovascular unit in brain function and disease. *Acta Physiologica* 2011;'Accepted Article'; doi: 10.1111/j.1748-1716.2011.02256.x.
- 601. Carmignoto G. The contribution of astrocyte signalling to neurovascular coupling. *Brain Research Reviews* 2010;63:138.
- 602. Kitaura H, Uozumi N, Tohmi M, et al. Roles of nitric oxide as a vasodilator in neurovascular coupling of mouse somatosensory cortex. *Neuroscience Research* 2007;59:160-171.
- 603. Lindauer U, Megow D, Matsuda H, Dirnagl U. Nitric oxide: a modulator, but not a mediator, of neurovascular coupling in rat somatosensory cortex. *American Journal of Physiology Heart and Circulatory Physiology* 1999;277:H799-H811.
- 604. Iadecola C. The overlap between neurodegenerative and vascular factors in the pathogenesis of dementia. *Acta Neuropathologica* 2010;120:287-296.
- 605. Metea MR, Newman EA. Signalling within the neurovascular unit in the mammalian retina. *Experimental Physiology* 2007;92:635-640.
- 606. Buerk DG, Riva CE, Cranstoun SD. Frequency and luminance-dependent blood flow and K+ ion changes during flicker stimuli in cat optic nerve head. *Investigative ophthalmology & visual science* 1995;36:2216-2227.
- 607. Formaz F. Diffuse luminance flicker increases retinal vessel diameter in humans. *Current Eye Research* 1997;16:1252.
- 608. Beydoun MA, Beydoun HA, Wang Y. Obesity and central obesity as risk factors for incident dementia and its subtypes: a systematic review and meta-analysis. *Obesity reviews* 2008;9:204-218.
- 609. Lee Y, Back JH, Kim J, et al. Systematic review of health behavioral risks and cognitive health in older adults. *International Psychogeriatrics* 2010;22:174-187.
- 610. Qiu C, Winblad B, Fratiglioni L. The age-dependent relation of blood pressure to cognitive function and dementia. *The Lancet Neurology* 2005;4:487-499.
- 611. Cedazo-Mínguez A, Ismail M, Mateos L. Cholesterol and Risk for Late-Onset Alzheimer's Disease: Expert Commentary & Five-year View. *Expert Review of Neurotherapeutics* 2011;11:495-498.
- 612. Carlsson CM. Type 2 Diabetes Mellitus, Dyslipidemia, and Alzheimer's Disease. *Journal of Alzheimer's Disease* 2010;20:711-722.
- 613. Akter K, Lanza EA, Martin SA, Myronyuk N, Rua M, Raffa RB. Diabetes mellitus and Alzheimer's disease: shared pathology and treatment? *British journal of clinical pharmacology* 2011;71:365-376.

- 614. Yoshitake T, Kiyohara Y, Kato I, et al. Incidence and risk factors of vascular dementia and Alzheimer's disease in a defined elderly Japanese population. *Neurology* 1995;45:1161-1168.
- 615. De la Torre JC. How do heart disease and stroke become risk factors for Alzheimer's disease? *Neurological Research* 2006;28:637-644.
- 616. Sparks DL, Martin TA, Gross DR, Hunsaker JC. Link between heart disease, cholesterol, and Alzheimer's disease: A review. *Microscopy Research and Technique* 2000;50:287-290.
- 617. de la Torre JC. The Vascular Hypothesis of Alzheimer's Disease: Bench to Bedside and Beyond. *Neurodegenerative disease* 2010;7:116-121.
- 618. Helzner E, J L, Scarmeas N, et al. Contribution of vascular risk factors to the progression in Alzheimer's disease. *Arch Neurol* 2009;66:343-348.
- 619. Jeynes B, Provias J. The case for blood–brain barrier dysfunction in the pathogenesis of Alzheimer's disease. *Journal of Neuroscience Research* 2011;89:22-28.
- 620. Marchesi VT. Alzheimer's dementia begins as a disease of small blood vessels, damaged by oxidative-induced inflammation and dysregulated amyloid metabolism: implications for early detection and therapy. *The FASEB Journal* 2011;25:5-13.
- 621. Altman R, Rutledge JC. The vascular contribution to Alzheimer's disease. *Clinical science* 2010;119:407-421.
- 622. O'Brien JT, Eagger S, Syed GM, Sahakian BJ, Levy R. A study of regional cerebral blood flow and cognitive performance in Alzheimer's disease. *Journal of Neurology, Neurosurgery & Psychiatry* 1992;55:1182-1187.
- 623. Ruitenberg A, Heijer Td, Bakker SLM, et al. Cerebral hypoperfusion and clinical onset of dementia: The Rotterdam study. *Annals of Neurology* 2005;57:789-794.
- 624. Johnson NA, Jahng G-H, Weiner MW, et al. Pattern of Cerebral Hypoperfusion in Alzheimer Disease and Mild Cognitive Impairment Measured with Arterial Spinlabeling MR Imaging: Initial Experience1. *Radiology* 2005;234:851-859.
- 625. Pakrasi S, O'Brien JT. Emission tomography in dementia. *Nuclear Medicine Communications* 2005;26:189-196.
- 626. Bateman GA, Levi CR, Schofield P, Wang Y, Lovett EC. Quantitative measurement of cerebral haemodynamics in early vascular dementia and Alzheimer's disease. *Journal of Clinical Neuroscience* 2006;13:563-568.
- 627. Prohovnik I, Mayeux R, Sackeim HA, Smith G, Stern Y, Alderson PO. Cerebral perfusion as a diagnostic marker of early Alzheimer's disease. *Neurology* 1988;38:931.
- 628. Bell R, Zlokovic B. Neurovascular mechanisms and blood–brain barrier disorder in Alzheimer's disease. *Acta Neuropathologica* 2009;118:103-113.
- 629. Mentis MJ, Alexander GE, Krasuski J, et al. Increasing Required Neural Response to Expose Abnormal Brain Function in Mild Versus Moderate or Severe Alzheimer's Disease: PET Study Using Parametric Visual Stimulation. *Am J Psychiatry* 1998;155:785-794.
- 630. de la Torre JC. Is Alzheimer's disease a neurodegenerative or a vascular disorder? Data, dogma, and dialectics. *The Lancet Neurology* 2004;3:184-190.
- 631. Dede DS, Yavuz B, Yavuz BB, et al. Assessment of Endothelial Function in Alzheimer's Disease: Is Alzheimer's Disease a Vascular Disease? *Journal of the American Geriatrics Society* 2007;55:1613-1617.
- 632. Bomboi G, Castello L, Cosentino F, Giubilei F, Orzi F, Volpe M. Alzheimer's disease and endothelial dysfunction. *Neurological Sciences* 2010;31:1-8.
- 633. Fischer VW, Siddiqi A, Yusufaly Y. Altered angioarchitecture in selected areas of brains with Alzheimer's disease. *Acta Neuropathologica* 1990;79:672-679.
- 634. Selnes OA, Vinters HV. Vascular cognitive impairment. *Nat Clin Pract Neuro* 2006;2:538-547.
- 635. Beach T, Wilson J, Sue L, et al. Circle of Willis atherosclerosis: association with Alzheimer's disease, neuritic plaques and neurofibrillary tangles. *Acta Neuropathologica* 2007;113:13-21.
- 636. Honig LS, Kukull W, Mayeux R. Atherosclerosis and AD: Analysis of data from the US National Alzheimer's Coordinating Center. *Neurology* 2005;64:494-500.
- 637. Zhang X, Le W. Pathological role of hypoxia in Alzheimer's disease. *Experimental Neurology* 2010;223:299-303.
- 638. Aliev G, Smith M, Obrenovich M, de la Torre J, Perry G. Role of vascular hypoperfusion-induced oxidative stress and mitochondria failure in the pathogenesis of Alzheimer disease. *Neurotoxicity Research* 2003;5:491-504.
- 639. de la Torre JC. Is Alzheimer's disease preceded by neurodegeneration or cerebral hypoperfusion? *Annals of Neurology* 2005;57:783-784.
- 640. de la Torre JC. Vascular risk factor detection and control may prevent Alzheimer's disease. *Ageing Research Reviews* 2010;9:218-225.
- 641. Lobo A, Launer L, Fratiglioni L, et al. Prevalence of dementia and major subtypes in Europe : A collaborative study of population-based cohorts. *Neurology* 2000;54:S4-9.
- 642. Fratiglioni L, Launer L, Andersen K, et al. Incidence of dementia and major subtypes in Europe : A collaborative study of population-based cohorts. *Neurology* 2000;54:S10-15.
- 643. Iemolo F, Duro G, Rizzo C, Castiglia L, Hachinski V, Caruso C. Pathophysiology of vascular dementia. *Immunity & Ageing* 2009;6:1-9.
- 644. Duthie A, Chew D, Soiza RL. Non-psychiatric comorbidity associated with Alzheimerâ€[™]s disease. *QJM* 2011.
- 645. Raj K. Similarities between Alzheimer's disease and vascular dementia. *Journal* of the Neurological Sciences 2002;203-204:29-34.
- 646. Rockwood K, Bowler J, Erkinjuntti T, Hachinski V, Wallin A. Subtypes of Vascular Dementia. *Alzheimer Disease & Associated Disorders* 1999;13:S59-S65.
- 647. Hardy P, Varma DR, Chemtob S. CONTROL OF CEREBRAL AND OCULAR BLOOD FLOW AUTOREGULATION IN NEONATES. *Pediatric Clinics of North America* 1997;44:137-152.
- 648. Tornquist P, Alm A. Carrier-mediated transport of amino acids through the bloodretinal and blood-brain barriers. *Graefe's Archive for Clinical and Experimental Ophthalmology* 1986;224:21-25.
- 649. Gao H, Hollyfield JG. Aging of the human retina. Differential loss of neurons and retinal pigment epithelial cells. *Investigative ophthalmology & visual science* 1992;33:1-17.
- 650. Kwa VIH, van der Sande JJ, Stam J, Tijmes N, Vrooland JL, Group* ftAVM. Retinal arterial changes correlate with cerebral small-vessel disease. *Neurology* 2002;59:1536-1540.
- 651. Wong TY, Klein R, Sharrett AR, et al. Retinal Microvascular Abnormalities and Cognitive Impairment in Middle-Aged Persons. *Stroke* 2002;33:1487-1492.
- 652. Ikram MK, De Jong FJ, Van Dijk EJ, et al. Retinal vessel diameters and cerebral small vessel disease: the Rotterdam Scan Study. *Brain* 2006;129:182-188.
- 653. Katz B, Rimmer S. Ophthalmologic manifestations of Alzheimer's disease. *Survey of Ophthalmology* 1989;34:31-43.
- 654. Cronin-Golomb A, Corkin S, Rizzo JF, Cohen J, Growdon JH, Banks KS. Visual dysfunction in Alzheimer's disease: Relation to normal aging. *Annals of Neurology* 1991;29:41-52.
- 655. Jackson GR, Owsley C. Visual dysfunction, neurodegenerative diseases, and aging. *Neurologic clinics* 2003;21:709-728.
- 656. Gilmore GC, Whitehouse PJ. Contrast Sensitivity in Alzheimer's Disease: A 1-Year Longitudinal Analysis. *Optometry & Vision Science* 1995;72:83-91.

- 657. Cormack FK, Tovee M, Ballard C. Contrast sensitivity and visual acuity in patients with Alzheimer's disease. *International Journal of Geriatric Psychiatry* 2000;15:614-620.
- 658. Blanks JC, Hinton DR, Sadun AA, Miller CA. Retinal ganglion cell degeneration in Alzheimer's disease. *Brain Research* 1989;501:364-372.
- 659. Paquet C, Boissonnot M, Roger F, Dighiero P, Gil R, Hugon J. Abnormal retinal thickness in patients with mild cognitive impairment and Alzheimer's disease. *Neuroscience Letters* 2007;420:97-99.
- 660. Blanks JC, Torigoe Y, Hinton DR, Blanks RHI. Retinal pathology in Alzheimer's disease. I. Ganglion cell loss in foveal/parafoveal retina. *Neurobiology of aging* 1996;17:377-384.
- 661. Ning A, Cui J, To E, Ashe KH, Matsubara J. Amyloid-{beta} Deposits Lead to Retinal Degeneration in a Mouse Model of Alzheimer Disease. *Invest Ophthalmol Vis Sci* 2008;49:5136-5143.
- 662. Shimazawa M, Inokuchi Y, Okuno T, et al. Reduced retinal function in amyloid precursor protein-over-expressing transgenic mice via attenuating glutamate-N-methyl-d-aspartate receptor signaling. *Journal of Neurochemistry* 2008;107:279-290.
- 663. Berisha F, Feke GT, Trempe CL, McMeel JW, Schepens CL. Retinal Abnormalities in Early Alzheimer's Disease. *Invest Ophthalmol Vis Sci* 2007;48:2285-2289.
- 664. Hinton DR, Sadun AA, Blanks JC, Miller CA. Optic-Nerve Degeneration in Alzheimer's Disease. *New England Journal of Medicine* 1986;315:485-487.
- 665. Danesh-Meyer HV, Birch H, Ku JY-F, Carroll S, Gamble G. Reduction of optic nerve fibers in patients with Alzheimer disease identified by laser imaging. *Neurology* 2006;67:1852-1854.
- 666. Tsai CS, Ritch R, Schwartz B, et al. Optic Nerve Head and Nerve Fiber Layer in Alzheimer's Disease. *Arch Ophthalmol* 1991;109:199-204.
- 667. Blennow K, Hampel H. CSF markers for incipient Alzheimer's disease. *The Lancet Neurology* 2003;2:605-613.
- 668. Valenti DA. Alzheimer's Disease and Glaucoma: Imaging the Biomarkers of Neurodegenerative Disease. *International Journal of Alzheimer's Disease* 2010;2010.
- 669. Ding J-D, Lin J, Mace BE, Herrmann R, Sullivan P, Bowes Rickman C. Targeting age-related macular degeneration with Alzheimer's disease based immunotherapies: Anti-amyloid-[beta] antibody attenuates pathologies in an age-related macular degeneration mouse model. *Vision Research* 2008;48:339-345.
- 670. Johnson LV, Leitner WP, Rivest AJ, Staples MK, Radeke MJ, Anderson DH. The Alzheimer's A beta-peptide is deposited at sites of complement activation in pathologic deposits associated with aging and age-related macular degeneration. *Proceedings of the National Academy of Sciences of the United States of America* 2002;99:11830-11835.
- 671. Wostyn P, Audenaert K, De Deyn P. Alzheimer's disease: cerebral glaucoma? *Medical hypotheses* 2010;74:973.
- 672. Ohno-Matsui K. Parallel findings in age-related macular degeneration and Alzheimer's disease. *Progress in Retinal and Eye Research* 2011;30:217-238.
- 673. McKinnon SJ. Glaucoma: Ocular Alzheimer's disease? *Front Biosci* 2003;8:s1140-1156.
- 674. Bayer AU, Ferrari F. Severe progression of glaucomatous optic neuropathy in patients with Alzheimer/'s disease. *Eye* 2002;16:209-212.
- 675. Yochim B, Mueller A, Kane K, Kahook M. Prevalence of Cognitive Impairment, Depression, and Anxiety Symptoms Among Older Adults With Glaucoma. *J Glaucoma* 2011;[Epub ahead of print].

- 676. Bach-Holm D, Kessing SV, Mogensen U, Forman JL, Andersen PK, Kessing LV. Normal tension glaucoma and Alzheimer disease: comorbidity? *Acta Ophthalmologica* 2011;no-no.
- 677. Tsolaki F, Gogaki E, Tiganita S, et al. Alzheimer's disease and primary openangle glaucoma: is there a connection? *Clinical Ophthalmology* 2011;5:887-890.
- 678. Gupta N, Fong J, Ang L, Yücel Y. Retinal tau pathology in human glaucomas. *Canadian journal of ophthalmology* 2008;43:53-60.
- 679. Wostyn P, Audenaert K, De Deyn PP. Alzheimer's disease and glaucoma: is there a causal relationship? *British Journal of Ophthalmology* 2009;93:1557-1559.
- 680. Huang W, Qiu C, von Strauss E, Winblad B, Fratiglioni L. APOE Genotype, Family History of Dementia, and Alzheimer Disease Risk: A 6-Year Follow-up Study. *Arch Neurol* 2004;61:1930-1934.
- 681. Vickers JC, Craig J, Stankovich J, et al. The apolipoprotein epsilon4 gene is associated with elevated risk of normal tension glaucoma. *Molecular Vision* 2002;8:389-393.
- 682. McKhann G, Drachman D, Folstein M, Katzman R, Price D, Stadlan EM. Clinical diagnosis of Alzheimer's disease: Report of the NINCDS-ADRDA Work Group* under the auspices of Department of Health and Human Services Task Force on Alzheimer's Disease. *Neurology* 1984;34:939-.
- 683. Mills RP, Budenz DL, Lee PP, et al. Categorizing the Stage of Glaucoma From Pre-Diagnosis to End-Stage Disease. *American Journal of Ophthalmology* 2006;141:24-30.
- 684. Mioshi E. The Addenbrooke's Cognitive Examination Revised (ACE-R): a brief cognitive test battery for dementia screening. *International Journal of Geriatric Psychiatry* 2006;21:1078.
- 685. Kotliar KE, Vilser W, Nagel E, Lanzl IM. Retinal vessel reaction in response to chromatic flickering light. *Graefe's Archive for Clinical and Experimental Ophthalmology* 2004;242:377-392.
- 686. Nagel E, Vilser W, Lanzl I. Age, blood pressure, and vessel diameter as factors influencing the arterial retinal flicker response. *Investigative ophthalmology & visual science* 2004;45:1486-1492.
- 687. Seifert BU, Vilser W. RETINAL VESSEL ANALYZER (RVA) DESIGN AND FUNCTION. *Biomedizinische Technik/Biomedical Engineering* 2002;47:678-681.
- 688. Garhofer G, Bek T, Boehm AG, et al. Use of the retinal vessel analyzer in ocular blood flow research. *Acta Ophthalmologica* 2010;88:717-722.
- 689. Pache M, Nagel E, Flammer J. Reproduzierbarkeit der Messungen mit dem Retinal Vessel Analyzer unter Optimalbedingungen. *Klin Monatsbl Augenheilkd* 2002;219:523,527.
- 690. Nagel E, Vilser W, Lanzl IM. [Comparison of diameter response of retinal arteries and veins to flicker. A Clinical Study on the healthy]. *Der Ophthalmologe* 2005;102:787-793.
- 691. Nagel E, Vilser W. Flicker observation light induces diameter response in retinal arterioles: a clinical methodological study. *British Journal of Ophthalmology* 2004;88:54-56.
- 692. Riva CE, Logean E, Falsini B. Visually evoked hemodynamical response and assessment of neurovascular coupling in the optic nerve and retina. *Progress in Retinal and Eye Research* 2005;24:183-215.
- 693. Polak K, Schmetterer L, Riva C. Influence of flicker frequency on flicker-induced changes of retinal vessel diameter. *Investigative ophthalmology & visual science* 2002;43:2721-2726.
- 694. Nagel E, Vilser W, Fink A, Riemer T. Varianz der Netzhautgefäßreaktion auf Flickerlicht. *Der Ophthalmologe* 2006;103:114-119.

- 695. Heitmar R, Blann A, Cubbidge R, Lip G, Gherghel D. Continuous retinal vessel diameter measurements the future of retinal vessel assessment? *Invest Ophthalmol Vis Sci* 2010;51:5833-5839.
- 696. Lanzl IM, Seidova S-F, Maier M, et al. Dynamic retinal vessel response to flicker in age-related macular degeneration patients before and after vascular endothelial growth factor inhibitor injection. *Acta Ophthalmologica* 2010;89:472-479.
- 697. Nguyen TT, Kawasaki R, Wang JJ, et al. Flicker Light Induced Retinal Vasodilation in Diabetes and Diabetic Retinopathy. *Diabetes Care* 2009;32:2075-2080.
- 698. Bek T, Hajari J, Jeppesen P. Interaction between flicker-induced vasodilatation and pressure autoregulation in early retinopathy of Type 2 diabetes. *Graefe's Archive for Clinical and Experimental Ophthalmology* 2008;246:763-769.
- 699. Mandecka A, Dawczynski J, Blum M, et al. Influence of Flickering Light on the Retinal Vessels in Diabetic Patients. *Diabetes Care* 2007;30:3048-3052.
- 700. Anderson TJ, Elstein E, Haber H, Charbonneau F. Comparative study of ACEinhibition, angiotensin II antagonism, and calcium channel blockade on flowmediated vasodilation in patients with coronary disease (BANFF study). *Journal of the American College of Cardiology* 2000;35:60-66.
- 701. Laurent S, Lacolley P, Brunel P, Laloux B, Pannier B, Safar M. FLOW-DEPENDENT VASODILATION OF BRACHIAL-ARTERY IN ESSENTIAL-HYPERTENSION. *American journal of physiology* 1990;258:H1004-H1011.
- 702. Corretti MC, Anderson TJ, Benjamin EJ, et al. Guidelines for the ultrasound assessment of endothelial-dependent flow-mediated vasodilation of the brachial artery A report of the International Brachial Artery Reactivity Task Force. *Journal of the American College of Cardiology* 2002;39:257-265.
- 703. Vogel RA. Measurement of endothelial function by brachial artery flow-mediated vasodilation. *The American journal of cardiology* 2001;88:31E-34E.
- 704. Clarkson P, Celermajer DS, Donald AE, et al. Impaired vascular reactivity in insulin-dependent diabetes mellitus is related to disease duration and low density lipoprotein cholesterol levels. *Journal of the American College of Cardiology* 1996;28:573-579.
- 705. Favaloro EJ, Smith J, Petinos P, Hertzberg M, Koutts J. Laboratory testing for von Willebrand's disease: an assessment of current diagnostic practice and efficacy by means of a multi-laboratory survey. RCPA Quality Assurance Program (QAP) in Haematology Haemostasis Scientific Advisory Panel. *Thrombosis and haemostasis* 1999;82:1276-1282.
- 706. Favaloro EJ. Laboratory assessment as a critical component of the appropriate diagnosis and sub-classification of von Willebrand's disease. *Blood Reviews* 1999;13:185-204.
- 707. O'Brien E, Atkins N, Stergiou G, et al. European Society of Hypertension International Protocol revision 2010 for the validation of blood pressure measuring devices in adults. *Blood Pressure Monitoring* 2010;15:23-38 10.1097/MBP.1090b1013e3283360e3283398.
- 708. Donic V, Donicova V, Lesko R, Tomori R. [Blood pressure and heart rate monitoring as a method of detection of sleep apnea]. *Sb Lek* 2002;103:85-90.
- 709. Gherghel D, Hosking SL, Cunliffe IA, Heitmar R. Transient cardiac ischaemia and abnormal variations in systemic blood pressure in unselected primary openangle glaucoma patients. *Ophthalmic & physiological optics* 2010;in press.
- 710. Heitmar R, Cubbidge R, Lip G, Gherghel D, Blann A. Altered blood vessel responses in the eye and finger in coronary artery disease. *Invest Ophthalmol Vis Sci* 2011;52:6199-6205.
- 711. O'Rourke MF, Pauca A, Jiang XJ. Pulse wave analysis. *British journal of clinical pharmacology* 2001;51:507-522.

- 712. Wilkinson IB, Fuchs SA, Jansen IM, et al. Reproducibility of pulse wave velocity and augmentation index measured by pulse wave analysis. *Journal of Hypertension* 1998;16:2079-2084.
- 713. Crilly M, Coch C, Bruce M, Clark H, Williams D. Indices of cardiovascular function derived from peripheral pulse wave analysis using radial applanation tonometry: a measurement repeatability study. *Vascular Medicine* 2007;12:189-197.
- 714. Skinner S, Barin E, Gallery E, Beattie J, Kamen P. SphygmoCor manual: A clinical guide; Pulse wave analysis. Australia; 2011.
- 715. Janner JH, Godtfredsen NS, Ladelund S, Vestbo J, Prescott E. Aortic Augmentation Index: Reference Values in a Large Unselected Population by Means of the SphygmoCor Device. *Am J Hypertens* 2009;23:180-185.
- 716. Segers P, Kips J, Trachet B, et al. Limitations and pitfalls of non-invasive measurement of arterial pressure wave reflections and pulse wave velocity. *Artery Research* 2009;3:79-88.
- 717. Simon A, Gariepy J, Chironi G, Megnien JL, Levenson J. Intima-media thickness: a new tool for diagnosis and treatment of cardiovascular risk. *Journal of hypertension* 2002;20:159-169.
- 718. Bots ML, Hoes AW, Koudstaal PJ, Hofman A, Grobbee DE. Common Carotid Intima-Media Thickness and Risk of Stroke and Myocardial Infarction : The Rotterdam Study. *Circulation* 1997;96:1432-1437.
- 719. O'Leary DH, Polak JF, Kronmal RA, Manolio TA, Burke GL, Wolfson SK. Carotid-Artery Intima and Media Thickness as a Risk Factor for Myocardial Infarction and Stroke in Older Adults. *New England Journal of Medicine* 1999;340:14-22.
- 720. Elias-Smale SE, Kavousi M, Verwoert GC, et al. Common carotid intima-media thickness in cardiovascular risk stratification of older people: the Rotterdam Study. *European Journal of Cardiovascular Prevention & Rehabilitation* 2011.
- 721. Simon A, Megnien J-L, Chironi G. The Value of Carotid Intima-Media Thickness for Predicting Cardiovascular Risk. *Arteriosclerosis, Thrombosis, and Vascular Biology* 2010;30:182-185.
- 722. Persson J, Formgren J, Israelsson B, Berglund G. Ultrasound-determined intimamedia thickness and atherosclerosis. Direct and indirect validation. *Arteriosclerosis, Thrombosis, and Vascular Biology* 1994;14:261-264.
- 723. Salonen R, Haapanen A, Salonen JT. Measurement of intima-media thickness of common carotid arteries with high-resolution B-mode ultrasonography: Inter- and intra-observer variability. *Ultrasound in Medicine & Biology* 1991;17:225-230.
- 724. Schmidt, Wendelhag. How can the variability in ultrasound measurement of intima-media thickness be reduced? Studies of interobserver variability in carotid and femoral arteries. *Clinical Physiology* 1999;19:45-55.
- 725. Rembold KE, Ayers CR, Wills MB, Rembold CM. Usefulness of carotid intimal medial thickness and Flow-Mediated dilation in a preventive cardiovascular practice. *The American journal of cardiology* 2003;91:1475-1477.
- 726. Kocak H, Gumuslu S, Sahin E, et al. Relationship between carotid artery intimamedia thickness and brachial artery flow-mediated dilation in peritoneal dialysis patients. *International Urology and Nephrology* 2009;41:409-416.
- 727. Lloyd-Jones DM, Wilson PWF, Larson MG, et al. Framingham risk score and prediction of lifetime risk for coronary heart disease. *The American journal of cardiology* 2004;94:20-24.
- 728. D'Agostino RB, Grundy S, Sullivan LM, Wilson P, Group ftCRP. Validation of the Framingham Coronary Heart Disease Prediction Scores. *JAMA: The Journal of the American Medical Association* 2001;286:180-187.
- 729. Rossi R, Milzani A, Dalle-Donne I, et al. Blood Glutathione Disulfide: In Vivo Factor or in Vitro Artifact? *Clin Chem* 2002;48:742-753.

- 730. Oliveira L, Louzada P, de Mello F, Ferreira S. Amyloid-β Decreases Nitric Oxide Production in Cultured Retinal Neurons: A Possible Mechanism for Synaptic Dysfunction in Alzheimer's Disease? *Neurochemical Research* 2011;36:163-169.
- 731. Ettorre E, Cerra E, Marigliano B, et al. Role of cardiovascular risk factors (CRF) in the patients with mild cognitive impairment (MCI). *Archives of Gerontology and Geriatrics* 2011.
- 732. Patel SR, Bellary S, Qin L, et al. Abnormal retinal vascular function and lipid levels in a sample of healthy UK South Asians. *British Journal of Ophthalmology* 2011;95:1573-1576.
- 733. Faul F, Erdfelder E, Lang AG, Buchner A. G*Power 3: a flexible statistical power analysis program for the social, behavioral, and biomedical sciences. *Behav Res Methods* 2007;39:175-191.
- 734. Gates PE, Strain WD, Shore AC. Human endothelial function and microvascular ageing. *Experimental Physiology* 2009;94:311-316.
- 735. Liu R, Zhang T-t, Wu C-x, Lan X, Du G-h. Targeting the neurovascular unit: Development of a new model and consideration for novel strategy for Alzheimer's disease. *Brain Research Bulletin* 2011;86:13-21.
- 736. Rosengarten B, Paulsen S, Burr O, Kaps M. Neurovascular coupling in Alzheimer patients: Effect of acetylcholine-esterase inhibitors. *Neurobiology of aging* 2009;30:1918-1923.
- 737. Filosa JA, Bonev AD, Nelson MT. Calcium Dynamics in Cortical Astrocytes and Arterioles During Neurovascular Coupling. *Circ Res* 2004;95:e73-81.
- 738. Mulligan SJ, MacVicar BA. Calcium transients in astrocyte endfeet cause cerebrovascular constrictions. *Nature* 2004;431:195-199.
- 739. Rossi D, Volterra A. Astrocytic dysfunction: Insights on the role in neurodegeneration. *Brain Research Bulletin* 2009;80:224-232.
- 740. Selley ML. Increased concentrations of homocysteine and asymmetric dimethylarginine and decreased concentrations of nitric oxide in the plasma of patients with Alzheimerâ€[™]s disease. *Neurobiology of aging* 2003;24:903-907.
- 741. Kása P, Rakonczay Z, Gulya K. The cholinergic system in Alzheimer's disease. *Progress in Neurobiology* 1997;52:511-535.
- 742. Berra A, GANZINELLI S, SARAVIA M, BORDA E, STERIN-BORDA L. Inducible nitric oxide synthase subserves cholinergic vasodilation in retina. *Visual Neuroscience* 2005;22:371-377.
- 743. Neal M. Modulation by endogenous ATP of the light-evoked release of ACh from retinal cholinergic neurones. *British Journal of Pharmacology* 1994;113:1085.
- 744. Hanon O, Haulon S, Lenoir H, et al. Relationship Between Arterial Stiffness and Cognitive Function in Elderly Subjects With Complaints of Memory Loss. *Stroke* 2005;36:2193-2197.
- 745. Price J, Chi X, Hellermann G, Sutton ET. Physiological levels of beta-amyloid induce cerebral vessel dysfunction and reduce endothelial nitric oxide production. *Neurological Research* 2001;23:506-512.
- 746. Gugleta K, Zawinka C, Rickenbacher I, et al. Analysis of Retinal Vasodilation after Flicker Light Stimulation in Relation to Vasospastic Propensity. *Invest Ophthalmol Vis Sci* 2006;47:4034-4041.
- 747. Kochkorov A, Gugleta K, Zawinka C, Katamay R, Flammer J, Orgul S. Short-Term Retinal Vessel Diameter Variability in Relation to the History of Cold Extremities. *Investigative ophthalmology & visual science* 2006;47:4026-4033.
- 748. Kotliar KE, Nagel E, Vilser W, Lanzl IM. Functional in vivo assessment of retinal artery microirregularities in glaucoma. *Acta Ophthalmologica* 2008;86:424-433.
- 749. de Jong FJ, Vernooij MW, Ikram MK, et al. Arteriolar Oxygen Saturation, Cerebral Blood Flow, and Retinal Vessel Diameters: The Rotterdam Study. *Ophthalmology* 2008;115:887-892.
- 750. Ikram MK, de Jong FJ, Bos MJ, et al. Retinal vessel diameters and risk of stroke. *Neurology* 2006;66:1339-1343.

- 751. Klijn CJM, Kappelle LJ, van Schooneveld MJ, et al. Venous Stasis Retinopathy in Symptomatic Carotid Artery Occlusion. *Stroke* 2002;33:695-701.
- 752. Patton N, Pattie A, MacGillivray T, et al. The Association between Retinal Vascular Network Geometry and Cognitive Ability in an Elderly Population. *Investigative ophthalmology & visual science* 2007;48:1995-2000.
- 753. Kumar DM. Oxidative stress in glaucoma: a burden of evidence. *Journal of glaucoma* 2007;16:334.
- 754. Tsai J. Influencing ocular blood flow in glaucoma patients: the cardiovascular system and healthy lifestyle choices. *Canadian journal of ophthalmology* 2008;43:347-350.
- 755. Gokce N, Keaney JF, Jr, Hunter LM, et al. Predictive value of noninvasivelydetermined endothelial dysfunction for long-term cardiovascular events inpatients with peripheral vascular disease. *J Am Coll Cardiol* 2003;41:1769-1775.
- 756. Corrado E, Camarda P, Coppola G, et al. Prognostic role of endothelial dysfunction and carotid intima-media thickness in patients undergoing coronary stent implantation. *Int Angiol* 2009;28:12-19.
- 757. Gherghel D, Hosking S. Heart rate variability, silent myocardial ischaemia, and 24-h blood pressure in primary open-angle glaucoma patients. *Ophthalmic & physiological optics* 2004;24.
- 758. He Z, Vingrys AJ, Armitage JA, Bui BV. The role of blood pressure in glaucoma. *Clinical and Experimental Optometry* 2011;94:133-149.
- 759. Coutinho T. Arterial dysfunction and functional performance in patients with peripheral artery disease: A review. *Vasc Med* 2011;16:203-211.
- 760. Wong TY, Klein R, Klein BEK, Tielsch JM, Hubbard L, Nieto FJ. Retinal microvascular abnormalities and their relationship with hypertension, cardiovascular disease, and mortality. *Surv Ophthalmol* 2001;46:59-80.
- 761. O'Rourke M, Safar M. Relationship between aortic stiffening and microvascular disease in brain and kidney: cause and logic of therapy. *Hypertension* 2005;46:200.
- 762. Mitchell G, Vita J, Larson M, et al. Cross-sectional relations of peripheral microvascular function, cardiovascular disease risk factors, and aortic stiffness: the Framingham Heart Study. *Circulation* 2005;112:3722.
- 763. Lanzl IM, Kotliar KE, Bock A, Vilser W, Halle M, Schmidt-Trucksäss A. How are healthy retinal vessels getting old? *Dtsch Med Wochenschr* 2006;131:180-181.
- 764. Harder DR, Zhang C, Gebremedhin D. Astrocytes Function in Matching Blood Flow to Metabolic Activity. *Physiology* 2002;17:27-31.
- 765. Filosa J. Vascular tone and neurovascular coupling: considerations towards an improved in vitro model. *Frontiers in Neuroenergetics* 2010;2.
- 766. Hirata K, Yaginuma T, O'Rourke MF, Kawakami M. Age-Related Changes in Carotid Artery Flow and Pressure Pulses: Possible Implications for Cerebral Microvascular Disease. *Stroke* 2006;37:2552-2556.
- 767. O'Rourke MF. Arterial aging: pathophysiological principles. *Vascular Medicine* 2007;12:329-341.
- 768. Sandoo A, Carroll D, Metsios G, Kitas G, Veldhuijzen van Zanten J. The association between microvascular and macrovascular endothelial function in patients with rheumatoid arthritis: a cross-sectional study. *Arthritis Research & Therapy* 2011;13:R99.
- 769. Arosio E, De Marchi S, Rigoni A, Prior M, Delva P, Lechi A. Forearm haemodynamics, arterial stiffness and microcirculatory reactivity in rheumatoid arthritis. *J Hypertens* 2007;25:1273-1278.
- 770. Krentz AJ, Clough G, Byrne CD. Interactions between microvascular and macrovascular disease in diabetes: pathophysiology and therapeutic implications. *Diabetes, Obesity and Metabolism* 2007;9:781-791.

- 771. Cohen J. *Statistical Power Analysis for the Behavioral Sciences*. 2nd ed. New Jersey: Lawrence Erlbaum Associates; 1988.
- 772. de la Torre JC, Stefano GB. Evidence that Alzheimer's disease is a microvascular disorder: the role of constitutive nitric oxide. *Brain Research Reviews* 2000;34:119-136.
- 773. Bonkale WL, Winblad B, Ravid R, Cowburn RF. Reduced nitric oxide responsive soluble guanylyl cyclase activity in the superior temporal cortex of patients with Alzheimer's disease. *Neuroscience Letters* 1995;187:5-8.
- 774. Leydhecker W. *Glaucoma: Tutzing symposium : symposium in Tutzing Castle, August 5-10, 1966*: Karger; 1967.
- 775. Hollows FC, Graham PA. Intra-ocular pressure, glaucoma, and glaucoma suspects in a defined population. *Br J Ophthalmol* 1966;50:570-586.
- 776. Kim YK, Oh WH, Park KH, Kim JM, Kim DM. Circadian blood pressure and intraocular pressure patterns in normal tension glaucoma patients with undisturbed sleep. *Korean journal of ophthalmology : KJO* 2010;24:23-28.
- 777. Su D-F, Miao C-Y. Blood Pressure Variability And Organ Damage. *Clinical and Experimental Pharmacology and Physiology* 2001;28:709-715.
- 778. Kochkorov A, Gugleta K, Katamay R, Flammer J, Orgul S. Short-term variability of systemic blood pressure and submacular choroidal blood flow in eyes of patients with primary open-angle glaucoma. *Graefe's Arch Clin Exp Ophthalmol* 2010.
- 779. Choi J, Kim KH, Jeong J, Cho H-s, Lee CH, Kook MS. Circadian Fluctuation of Mean Ocular Perfusion Pressure Is a Consistent Risk Factor for Normal-Tension Glaucoma. *Invest Ophthalmol Vis Sci* 2007;48:104-111.
- 780. Sung KR, Lee S, Park SB, et al. Twenty-four Hour Ocular Perfusion Pressure Fluctuation and Risk of Normal-Tension Glaucoma Progression. *Invest Ophthalmol Vis Sci* 2009;50:5266-5274.
- 781. Choi J, Jeong J, Cho HS, Kook MS. Effect of nocturnal blood pressure reduction on circadian fluctuation of mean ocular perfusion pressure: A risk factor for normal tension glaucoma. *Investigative ophthalmology & visual science* 2006;47:831-836.
- 782. Simon A, Gariepy J, Chironi G, Megnien J-L, Levenson J. Intima-media thickness: a new tool for diagnosis and treatment of cardiovascular risk. *Journal of Hypertension* 2002;20:159-169.
- 783. Zakopoulos NA, Tsivgoulis G, Barlas G, et al. Time Rate of Blood Pressure Variation Is Associated With Increased Common Carotid Artery Intima-Media Thickness. *Hypertension* 2005;45:505-512.
- 784. Shintani Y, Kikuya M, Hara A, et al. Ambulatory blood pressure, blood pressure variability and the prevalence of carotid artery alteration: the Ohasama study. *Journal of Hypertension* 2007;25:1704-1710.
- 785. Aronson D, Mittleman MA, Burger AJ. Role of endothelin in modulation of heart rate variability in patients with decompensated heart failure. *Pacing and clinical electrophysiology* 2001;24:1607-1615.
- 786. Aso Y, Fujiwara Y, Tayama K, Inukai T, Takemura Y. Elevation of von Willebrand factor in plasma in diabetic patients with neuropathic foot ulceration. *Diabetic medicine* 2002;19:19-26.
- 787. Gherghel D, Griffiths HR, Hilton EJ, Cunliffe IA, Hosking SL. Systemic reduction in glutathione levels occurs in patients with primary open-angle glaucoma. *Investigative ophthalmology & visual science* 2005;46:877-883.
- 788. Glasser S. On arterial physiology, pathophysiology of vascular compliance and cardiovascular disease. *Heart Dis* 2000;2:375-379.

10. Appendices

- 1. Summary of Ocular Blood Flow Studies in Glaucoma
- 2. Addenbrooke's Cognitive Examination-Revised (ACE-R)
- 3. Framingham Risk Score Calculation Chart
- Conference Presentation: Is the eye a window to the mind? Retinal vascular reactivity as a marker for endothelial function in Alzheimer's disease. EVER 2010; Acta Ophthalmologica, Suppl 2010

Study	Participants	Site and technique of measurement	Findings
		Retrobulbar	
Plange et al (2003)	29 NTG / 29 Controls	CRA and SPCA	\downarrow blood flow parameters and \uparrow resistance in most
		Color Doppler Imaging and Fluorescein	vessels, along with larger ONH filling defects in NTG
-		angiograms of ONH	patients
Butt et al (1995)	34 LTG / 17 Controls	CRA and OA	↓ BF and ↑ resistance in CRA and OA of LTG
		Color Doppler Imaging	patients
Nasemann et al (1994)	8 NIG / 8 Controls	CRA and OA	↓ blood flow velocity in OA and CRA
		Fluorescence perfusion scintigraphy	
Yamazaki et al (1997)	16 NIG, prog VF	SPCA, CRA and OA	↓ blood flow velocity and ↑ resistance in CRA and
	15 NIG Stable VF	Color Doppler Imaging	SPCA of NIG patients with progressive visual fields
	14 HIG, prog VF		
Dealis et al (1005)	14 HTG, stable VF		
Rankin et al (1995)	52 POAG	CRA and SPCA	\downarrow blood flow velocity and \uparrow resistance in CRA and CRA and CRA at both NTC and DOAC patients
	24 NIG	Color Doppler Imaging	SPCA of both NTG and POAG patients
Boiopopopopup et el (1002)		00	blood flow valagity in QA of both DQAC and NTC
Rojanapongpun et al (1993)		Transpropial Doppler ultrasound (2MHz)	↓ blood now velocity in OA or both POAG and NTG
	42 NTG		patients
Gallassi et al (2003)		00	blood flow velocity and thresistance in patients with
		Color Doppler imaging and VF analysis	deteriorating visual fields
Satilmis et al (2003)	20 progressive POAG	CRA	blood flow velocity in CRA correlates with visual
	over approx 4 years	Color Doppler imaging and VF analysis	field progression rate
Zeitz et al (2006)	114 NTG / 40 Controls	CRA, SPCA and OA	↓ blood flow velocity in the CRA and SPCA of
		Color Doppler imaging	progressing glaucoma patients
Schumann et al (2000)	20 progressive POAG	CRA and OA	↓ blood flow velocity in OA and ↑ resistance in CRA.
		Color Doppler imaging and VF analysis	Interocular differences in VF progression correlate
			with interocular differences in blood flow parameters.
Januleviciene et al (2008)	30 POAG / 30 Controls	CRA, OA and SPCA	\downarrow blood flow velocity correlated with thinner RNFL in
		Color Doppler Imaging	POAG patients.
		RNFL thickness using scanning laser	
		polarimetry	
		Optic nerve head	
Michelson et al (1998)	91 POAG / 44 Controls	Juxtapapillary retina and NRR	\downarrow ONH and juxtapapillary retina blood flow in POAG
		Scanning laser Doppler flowmetry	with no field defect, borderline field defect and
		Analysis of visual fields	advanced disease

Plitz-Seymour et al (2001)	21 POAG suspects 22 POAG 15 Controls	ONH – 4 quadrants, cup and foveola Laser Doppler flowmetry	Similar levels of ↓ blood flow in the cup, superotemporal and inferotemporal NRR of suspect POAG and POAG patients
Lam et al (2005)	16 glauc with asymmetry between eyes 20 glauc with asymmetry superior to inferior	Superotemporal and inferotemporal NRR and cup Laser Doppler flowmetry	Reduction in blood flow greater in the eye with worse damage and the hemifield of the disc with greater damage
Fontana et al (1998)	95 NTG / Controls	Pulsatile ocular blood flow	POBF \downarrow in NTG patients with and without field loss and greater \downarrow in eye with field defect compared to that with no field defect.
Adam et al (1980)	171 POAG, OHT and Controls	Filling defects in the rim, wall and floor of cup. Optic disc fluorescein angiograms	Greater % of filling defects in the wall of the cup in glaucoma patients and ↑ with degree of field loss
Grunwald et al (1998)	19 POAG / 15 Controls	ONH – 4 quadrants and cup and foveola Laser Doppler flowmetry	↓ in blood flow in inferotemporal and superotemporal NRR in POAG. Lower blood flow in those with more advanced VF defects.
Findl et al (2000)	90 POAG / 61 Controls	Cup and NRR Scanning laser Doppler flowmetry Fundus pulsation amplitude – cup and macula	↓ blood flow and pulsation amplitude in cup, NRR and macula of POAG patients and correlation with degree of field defect
Michelson et al (1996)	43 POAG / 43 Controls	NRR and juxtapapillary retina Scanning laser Doppler flowmetry	↓ blood flow at both the NRR and juxtapapillary retina in POAG
Sato et al (2006)	54 NTG	Superior and inferior NRR Heidelberg retina flowmetry	Region NRR with greatest reduction in blood flow corresponded with region of visual field defect
		Retina	
Mitchell et al (2005)	59 POAG 163 OHT 3065 Controls	Stereo optic disc photography – analysis of retinal vessel diameter	Significantly narrower retinal arteriolar diameters in POAG compared to OHT and controls
Rader et al (1994)	226 POAG/NTG 206 Controls / OHT	Analysis of retinal vessel diameter	Vessels more constricted closer to the disc compared to downstream in POAG/NTG patients and correlation with degree and site of ON damage
Arend et al (2002)	36 NTG 31 Controls	Arteriovenous passage (AVP) time and peripapillary arterial and venous diameters. Digital scanning laser fluorescein	Prolonged AVP time in NTG patients but no differences in vessel diameters compared to controls

		angiograms	
Duijm et al	45 POAG 43 NTG 11 OHT 20 Controls	Retinal arteriovenous passage time. Video fluorescein angiograms	Retinal AVP prolonged in POAG compared to other groups
Berisha et al (2008)	12 Early POAG 8 Controls	Inferotemporal retinal artery blood flow parameters - Canon laser Doppler blood flow instrument Peripapillary RNFL thickness - OCT	↓ blood flow and speed and thinner RNFL in POAG patients. Negative correlation between blood flow and RNFL thickness in POAG
Logan et al (2004)	58 POAG 76 NTG 38 Controls	Retinal blood flow – Heidelberg retinal flowmetry Structural damage of ONH – Heidelberg retinal tomography	↓ retinal blood flow in both NTG and POAG. ONHs with abnormal segments had lower corresponding blood flow parameters. Glaucoma patients with normal ONH segments had ↓ blood flow compared to controls with normal ONH segments.
Wolf et al (1993)	51 OAG Controls	Retinal parameters including AVP time Video fluorescein angiograms	Prolonged AVP time, ↓ dye velocity, increased plasma viscosity in POAG
		Choroid	
Duijm et al (1997)	45 POAG 43 NTG 11 OHT 20 Controls	Choroidal blood refreshment time Video fluorescein angiograms	Slower choroidal circulation in NTG compared to other groups
Yin et al (1997)	25 POAG 5 Optic Atrophy 18 Controls	Choroidal filling time – fluorescein angiography Choroidal thickness – light microscopy	Thinner choroids and delayed choroidal perfusion in POAG patients
Cellini et al (1996)	15 POAG	Choroid, OA and SPCA Color Doppler Imaging	↓ blood flow velocity and ↑ resistance in the SPCA and choroid in POAG
Fuchsjager-Mayrl et al (2004)	140 POAG/OHT 102 Controls	Temporal NRR and cup – scanning laser Doppler flowmetry Choroidal blood flow – laser interferometry	↓ ONH and choroidal blood flow in POAG/OHT patients and an abnormal association between BP and ocular perfusion
Kerr et al (1998)	10 POAG 14 OHT	Temporal NRR and cup and peripapillary retina – scanning laser Doppler flowmetry and pulsatile ocular blood flow	↓ blood flow at NRR, lamina cribrosa and choroid in POAG

Appendix 2: Addenbrooke's Cognitive Examination-Revised (ACE-R)

ADDEN	IBROOKE	'S COGI Final Revise	NITIVE E	XAMINA A (2005)	TION - A	CE-R	
Name : Date of birth : Hospital no. :			Date of tes Tester's na Age at lea Occupatio	sting:/ ame: ving full-time e n:	education:		
		Addressograp	h Handedne	SS:			
ORIENTATION	۱ <u> </u>		•				
> Ask: What is the	Day	Date	Month	Year	Season	[Score 0-5]	z o
							-
> Ask: Which	Building	Floor	Town	County	Country	[Score 0-5]	A T
							z
REGISTRATIO	N	•		<u>*</u>	•		ш
Tell: 'I'm going to After subject repe the first trial (repe Register number of tr	give you three wo ats, say 'Try to re at 3 times if nece ials	ords and i'd like y member them b ssary).	you to repeat af ecause i'm goir	ter me: lemon, k ng to ask you lat	ey and ball'. er'. Score only	[Score 0-3]	N & O R I
ATTENTION 8	CONCENT	RATION					0
Ask the subject: ' to take away ano check the subsec Stop after five subtrace	could you take 7 ther 7 to a total of juent answer (i.e. ctions (93, 86, 79,	away from a 100 5 subtractions. 93, 84, 77, 70, 0 , 72, 65).	0? After the sub If subject make 33 -score 4)	ject responds, a a mistake, can	ask him or her ry on and	[Score 0-5]	TENTI
 Ask: could you p 	lease spell work	LD for me? The		spen it backwa			<
MEMORY - Rec	all						
Ask: "Which 3 wo	rds did I ask you t	to repeat and re	member?"			[Score 0-3]	7
			*******	••••••	•••••		
 MEMORY - Ante Tell: ' I'm going to doing that 3 times 	give you a name s, so you have a c	and address an chance to learn i	nd I'd like you to t. I'll be asking y	repeat after me you later'	e. We'll be	[Score 0-7]	œ
Score only the third to	ial						
Harry Barnes	1 ^{et} Trial	2 nd Tr	ial	3 ^{re} Trial			0
73 Orchard Close	4						
Kingsbridge							2
Devon			-				
MEMORY-Retro	grade Memory			1			ш
Name of current Name of the won Name of the USA Name of the USA	Prime Minister nan who was Prim president	e Minister	l in the 1960's			[Score 0 -4]	W
						dabt 2000 John D	Unders

ADDENR	ROOKE'S COGNITIVE EX	AMINATION - ACE-R	Fna	Revised Version (2005)	
VERBAL FLUE	N.C.Y Letter 'P' and anim	nals			
Letters Say: 'I'm going to giv as you can beginning got a minute and the	e you a letter of the alphabe) with that letter, but not nan letter is P'	et and I'd like you to gener: nes of people or places. Ar	ate as many words re you ready? You've	[Score 0 - 7]	Y
				s17 7 14-17 6 11-13 5	υ
				6-7 3 4-5 2 2-3 1	2
> Animals		<u>.</u>		<2 0 total conect	ш
Say: 'Now can you na	ame as many animals as po	ossible, beginning with any	letter?	[Score 0 - 7]	D
				17-21 C 14-16 5 11-13 4 9-10 S	Г
				.7-8 2 5-6 1 ⊲5 C total correct	L
LANGUAGE - Co	mprehension				
Show written instru	udion:			[Score 0-1]	ш
	Close	your e	yes		J
					٩
➢ 3 stage command: 'Take the paper in you	ur right hand. Fold the pa	per in half. Put the paper	on the floor'	[Score 0-3]	D
	iting	- /			U
> Ask the subject to Score 1 if sentence con	make up a sentence and wi ntains a subject and a verb	rite it in the space below: (see guide for examples)		[Score 0-1]	z
					٩
					L







 Ask the subject to ider 	ntify the letters			[Score 0-4]
	-			
-	4	1	_	
/	1 . M			
ī	•	(· _	
di n e				
		J		5
				· · ·
	L			
	- N			
	· · · ·			
	<u> </u>			
	a .		•	
RECALL	4		-	
RECALL > Ask "Now tell me what	t you remember of that name	e and address we were repe	ating at the beg	jinning'''
RECALL Ask "Now tell me what Harry Barnes	t you remember of that name	e and address we were repe	ating at the beg	jinning''' Score 0-7]
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close	t you remember of that name	e and address we were repe	ating at the beg	inning'''
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge	t you remember of that name	e and address we were repe	ating at the beg	inning'''
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon	t you remember of that name	e and address we were repe	ating at the beg	inning'''
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon RECOGNITION	t you remember of that name	e and address we were repe	ating at the beg	inning''' [Soore 0-7]
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon RECOGNITION This test should be donu	t you remember of that name t	e and address we were repe	ating at the beg	inning'''
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon RECOGNITION This test should be don test and score 5. If only right hand side. Then to 72" and score 1. Such score	t you remember of that name t you remember of that name e if subject failed to recall one part is recalled start by ticking est not recalled items one one one	e and address we were repe or more items. If all items we g items recalled in the shadow g "ok, I'll give you some hints: twoich it give you some hints:	eating at the beg re recalled, skip t was the name X, thingh by calling	inning''' [Score 0-7] [Score 0-7] [Score 0-5] [Score 0-5] [Score 0-5]
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon RECOGNITION , This test should be don test and score 5. If only right hand side. Then te Z?" and so on. Each rec erry Barnes	t you remember of that name 	e and address we were repe or more items, If all items we gitems recalled in the shadow g "ok, I'll give you some hints: t which is added to the point g Harry Bradford	eating at the beg re recalled, skip t ed column on th was the name X, jained by recallin	inning''' [Score 0-7] he [Score 0-5] e Yor ig.
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon RECOGNITION This test should be don test and score 5. If only right hand side. Then to Z?" and so on. Each rec	t you remember of that name t you remember of that name e if subject failed to recall one part is recalled start by ticking est not recalled items by tellin ognised item scores one point Harry Barnes 73	e and address we were repe or more items. If all items we g items recalled in the shadow g "ok, I'll give you some hints: t which is added to the point g Harry Bradford 76	re recalled, skip t red column on th was the name X, jained by recalled recalled	inning''' IScore 0-7]
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon RECOGNITION This test should be don test and score 5. If only right hand side. Then to Z?" and so on. Each rec terry Barnes 7 Orchard Place	t you remember of that name t you remember of that name e if subject failed to recall one part is recalled start by ticking est not recalled items by tellin ognised item scores one point Hany Barnes 73 Oak Close	e and address we were repe or more items. If all items we g items recalled in the shadow g "ok, I'll give you some hints: t which is added to the point o Harry Bradford 76 Orchard Close	re recalled, skip t red column on th was the name X, jained by recalled recalled recalled	inning''' [Soore 0-7] he [Soore 0-5] y or ig.
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon RECOGNITION This test should be don test and score 5. If only right hand side. Then to Z?" and so on. Each rec erry Barnes 7 Orchard Place	t you remember of that name t you remember of that name e if subject failed to recall one part is recalled start by ticking est not recalled items by tellin ognised item scores one point Hany Barnes 73 Oak Close Kingsbridge	e and address we were repe or more items. If all items we g items recalled in the shadow g "ok, I'll give you some hints: t which is added to the point o Harry Bradford 76 Orchard Close Dartington	re recalled, skip t red column on th was the name X, jained by recalled recalled recalled recalled	inning''' [Soore 0-7] he [Soore 0-5] y or ig.
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon RECOGNITION This test should be don test and score 5. If only right hand side. Then to Z?" and so on. Each rec erry Barnes 7 Orchard Place Devon	t you remember of that name t you remember of that name e if subject failed to recall one part is recalled start by ticking est not recalled items by tellin ognised item scores one point Harry Barnes 73 Oak Close Kingsbridge Dorset	e and address we were repe or more items, if all items we g items recalled in the shadow g "ok, I'll give you some hints: t which is added to the point g Harry Bradford 76 Orchard Close Dartington Somerset	re recalled, skip t re recalled, skip t red column on th was the name X, jained by recalled recalled recalled recalled recalled recalled	inning''' IScore 0-7] he Soore 0-6] e Yor g.
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon RECOGNITION This test should be don test and score 5. If only right hand side. Then to Z?" and so on. Each rec erry Barnes 7 Orchard Place Dakhampton Devon Seneral Scores	t you remember of that name t you remember of that name e if subject failed to recall one part is recalled start by ticking est not recalled items by tellin ognised item scores one point Harry Barnes 73 Oak Close Kingsbridge Dorset	e and address we were repe or more items. If all items we g items recalled in the shadow g "ok, I"ll give you some hints: t which is added to the point g Harry Bradford 76 Orchard Close Dartington Somerset	re recalled, skip t re recalled, skip t ved column on th was the name X, jained by recallin recalled recalled recalled recalled recalled	inning''' IScore 0-7] he Yor G. IScore 0-6]
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon RECOGNITION This test should be dom test and score 5. If only right hand side. Then te Z?" and so on. Each rec erry Barnes 7 Orchard Place Dakhampton Devon General Scores	t you remember of that name t you remember of that name e if subject failed to recall one part is recalled start by ticking est not recalled items by tellin ognised item scores one point Harry Bames 73 Oak Close Kingsbridge Dorset	e and address we were repe or more items. If all items we g items recalled in the shadow g "ok, I'll give you some hints: t which is added to the point g Harry Bradford 76 Orchard Close Dartington Somerset	re recalled, skip t re recalled, skip t ved column on th was the name X, jained by recalled recalled recalled recalled recalled recalled recalled	inning''' IScore 0-7] he (Score 0-6] y or SE //30
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon RECOGNITION This test should be don test and score 5. If only right hand side. Then to Z?" and so on. Each rec erry Barnes 7 Orchard Place Dakhampton Devon Seneral Scores	t you remember of that name t you remember of that name e if subject failed to recall one part is recalled start by ticking est not recalled items by tellin ognised item scores one point Harry Barnes 73 Oak Close Kingsbridge Dorset	e and address we were repe or more items. If all items we g items recalled in the shadow g "ok, I'll give you some hints: t which is added to the point of Harry Bradford 76 Orchard Close Dartington Somerset	eating at the beg	inning'''
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon RECOGNITION This test should be don test and score 5. If only right hand side. Then to Z?" and so on. Each rec erry Barnes 7 Orchard Place Dakhampton Devon Seneral Scores Subscores	t you remember of that name	e and address we were repe or more items. If all items we g items recalled in the shadow g "ok, I'll give you some hints: t which is added to the point of Harry Bradford 76 Orchard Close Dartington Somerset	eating at the beg	inning'''
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon RECOGNITION This test should be don test and score 5. If only right hand side. Then to Z?" and so on. Each rec lerry Barnes TO Drohard Place Dakhampton Devon Seneral Scores Subscores	t you remember of that name	e and address we were repe or more items. If all items we g items recalled in the shadow g "ok, I'll give you some hints: t which is added to the point of Harry Bradford 76 Orchard Close Dartington Somerset Attent	eating at the beg	inning'''
RECALL Ask "Now tell me what Harry Barnes 73 Orchard Close Kingsbridge Devon RECOGNITION This test should be don test and score 5. If only right hand side. Then to Z?" and so on. Each rec Jerry Barnes Jorohard Place Dakhampton Devon General Scores Subscores	t you remember of that name	e and address we were repe or more items. If all items we g items recalled in the shadow g "ok, I'll give you some hints: t which is added to the point g Harry Bradford 76 Orohard Close Dartington Somerset	eating at the beg	inning'''

Normative values based on 63 controls aged 52-75 and 142 dementia patients aged 46-86

Cut-off <88 gives 94% sensitivity and 89% specificity for dementia Cut-off <82 gives 84% sensitivity and 100% specificity for dementia



FRAMINGHAM RISK SCORE to predict 10 year ABSOLUTE RISK of CHD EVENT WEST HERTFORDSHIRE CARDIOLOGY

This risk assessment only applies to assessment for PRIMARY PREVENTION of CHD, in people who do not have evidence of established vascular disease. Patients who *already* have evidence of vascular disease usually have a >20% risk of further events of over 10 years, and require vigorous **SECONDARY PREVENTION**. People with a Family History of premature vascular disease and some Asians are at higher risk than predicted; Southern Europeans may have a lower risk in relation to standard risk factors.

A	ge		Total Cho	lester	rol	HDL Chole	ester	ol	Systolic BP	P Diastolic BP					Diabetes	М	F	Smoking	М	F
	Μ	F		М	F		М	F	Male	<80	80-84	85-89	90-99	≥100	No	0	0	No	0	0
30-34	- 1	- 9	< 4.1	- 3	- 2	< 0.9	2	5	<120	0	0	1	2	3	Yes	2	4	Yes	2	2
35-39	0	- 4	4.1 – 5.1	0	0	0.9 - 1.16	1	2	120-129	0	0	1	2	3						
40-44	1	0	5.2 - 6.2	1	1	1 .17 - 1.2 9	0	1	130-139	1	1	1	2	3						
45-49	2	3	6.3 – 7.1	2	1	1.30 - 1.55	0	0	140-159	2	2	2	2	3						
50-54	3	6	≥7.2	3	3	≥1.56	- 2	- 3	≥160	3	3	3	3	3						
55-59	4	7					•		Female	<80	80-84	85-89	90-99	≥100		Cat	egor	isaticn of 10 y	/ear	Risk
60-64	5	8							<120	- 3	0	0	2	3			-	of CHD Event		
65-69	6	8							120-129	0	0	0	2	3		Ve	ry Lo	w risk	< '	10%
70-74	7	8							130-139	0	0	0	2	3		Lo	w ris	k	< 1	15%
									140-159	2	2	2	2	3		Мо	dera	te risk	15	20%
									≥160	3	3	3	3	3		Hig	gh ri	sk	> 2	20%
									If Systolic and	Diastoli	c BP fall i	nto differ	ent cateo	ories.						

use score from higher category

STEP 1: Add scores by sex for Age, Total Cholesterol, HDL-Cholesterol, BP, Diabetes and Smoking. (If HDL unknown, assume 1.1 in Males, 1.4 in Females)

STEP 2: Use total score to determine Predicted 10 year Absolute Risk of CHD Event (Coronary Death, Myocardial Infarction, Angina) by sex

Total Score	≤-2	-1	0	1	2	3	4	5	6	7	8	9	10	11	12	3	14	15	16	≥17
10 year Risk: Male		<2%	3%	3%	4%	5%	7%	8%	10%	13%	16%	20%	25%	31%	37%	45%	≥53%	≥53%	≥53%	≥53%
10 year Risk: Female	<1%	2%	2%	2%	3%	3%	4%	4%	5%	6%	7%	8%	10%	11%	13%	15%	18%	20%	24%	≥27%

STEP 3: Compare Predicted 10 year Absolute Risk with "Average" and "Ideal" 10 year Risks, to give Relative Risks

Age	30 - 34	35 - 39	40 - 44	45 - 49	50 - 54	55 - 59	60 - 64	65 - 69	70 - 74	
"Average" Male	3%	5%	7%	11%	14%	16%	21%	25%	30%	
"Ideal" Male	2%	3%	4%	4%	6%	7%	9%	11%	14%	
"Average" Female	< 1%	< 1%	2%	5%	8%	12%	12%	13%	14%	
"Ideal" Female	<1%	1%	2%	3%	5%	7%	8%	8%	8%	

"Ideal" risk represents
Total Cholesterol = 4.1 - 5.1
HDL = 1.2 (Male), 1.4 (Female)
BP < 120/80
No Diabetes, Non Smoker

<u>People with an absolute risk of \geq 30% should be considered for treatment</u>: with a Statin to achieve a Total Cholesterol <5 and/or LDL cholesterol <3 <u>People with an absolute risk of \geq 15% should be considered for treatment</u> with anti-hypertensives to achieve a BP ideally \leq 140/90

Appendix 4: Conference Presentation

Is the eye a window to the mind? Retinal vascular reactivity as a marker for endothelial function in Alzheimer's disease.

Stephanie Mroczkowska, Alexandra Benavente-Perez, Sunni Patel, Lu Qin, Doina Gherghel

School of Life & Health Sciences, Aston University, Birmingham, B4 7ET, UK

Aim: To assess the retinal and systemic vascular function in patients diagnosed with mild Alzheimer's disease in comparison to healthy age matched controls.

Methods: Nine newly diagnosed mild AD patients (MMSE score 18- 24) and 23 healthy age-matched controls without any cognitive dysfunction (ACE-R score \geq 88) were recruited for the study. Retinal vessel reactivity was assessed using the retinal vessel analyser (RVA, IMEDOS, Germany). From these recordings the time taken to reach maximum dilation (RT), was determined for each individual flicker cycle. Systemic vascular function was assessed using flow mediated dilation (FMD) technique at the brachial artery level (Siemens; Acuson Sequoia, UK). Intraocular pressure (IOP) and systemic blood pressure (BP) were also recorded for each participant and OPP was then calculated.

Results: There were no significant differences in age, mean BP, IOP or OPP between the two study groups (p>0.05). The retinal arterial RT to flicker light stimulation was found to be significantly longer in AD patients as compared to healthy control for both the first (p= 0.01) and the third (p=0.049) flicker cycles. In addition, the RT measured at the chosen vein level was significantly longer in AD patients compared to controls for the first (p=0.046) and second (p=0.043) flicker cycles. No significant differences were found in the brachial arterial diameter between the two groups (p>0.05).

Conclusion: In patients suffering from AD, the prolonged retinal vessel RT to flicker provocation could represent an early sign of vascular dysfunction evident at the microvascular level.

Awarded 'Best paper in section' – 500 euro travel grant

